

# Enabling Health Professionals to Provide Effective Self-Management Support for Cancer-Related Fatigue

By

# Oluwaseyifunmi Andi Agbejule

**Bachelor of Radiation Therapy** 

Thesis submitted to Flinders University for the degree of

# **Doctor of Philosophy**

College of Nursing and Health Sciences 17/11/2023

# TABLE OF CONTENTS

TABLE OF CONTENTS	I
SUPERVISION TEAM	V
ABSTRACT	VI
DECLARATION	VIII
ACKNOWLEDGEMENTS	IX
LIST OF RELEVANT PUBLICATIONS COMPLETED DURING CANDIDATURE	Х
Relating to Doctoral Thesis:	x
Relating to Self-management Support	
CONFERENCE PRESENTATIONS, POSTERS, AND AWARDS	XI
Conference Presentations and Posters	xi
Awards	xi
LIST OF FIGURES	XII
LIST OF TABLES	XIII
1.0 INTRODUCTION	14
Background and Scope of the Problem	14
Cancer-related fatigue	14
Self-management, Cancer-Related Fatigue, and Self-management Support	16
Translation of CRF Management Strategies to Practice	18
Research problem	22
Research Plan	22
Aims	22
Objectives	23
Research Questions	24
Thesis Outline	24
2.0 THEORETICAL AND CONCEPTUAL FRAMEWORK	25
Chapter Introduction	25
History of Chronic Illness Self-management	25
Self-efficacy, self-management, and cancer-related fatigue	26
Conceptual Framework: Flinders Capabilities for Self- Management Support	29
Introduction	29
Background	30
19 Core Capabilities	
Conclusions	34
3.0 COMPONENTS, PRINCIPLES, AND PROCESSES OF SELF-MANAGEMENT SUI FOR CANCER-RELATED FATIGUE (STUDY 1)	
Chapter Introduction	35
Background	
Methods	
Identification of studies and inclusion criteria	

Study screening and data extraction	
Study quality assessment and data analysis	
Results	
Characteristics of Included Studies	
Risk of Bias	
Description of Self-management Support Programs	
Self-management Support Components, Strategies and Approaches	45
Effectiveness of Self-management Support Programs	51
Fatigue outcomes	51
Self-management Behavioural outcomes	52
Discussion	54
Cancer-related Fatigue Management Interventions	55
Self-management Support Delivery	55
Facilitation of Self-management Support	56
Theories Underpinning Self-Management Support Programs	56
Self-Management Support Components	57
Strengths and Limitations	
Conclusion	59
4.0 MODIFIED DELPHI STUDY (STUDY 2)	60
Chapter Introduction	60
Background	60
Methodology and Research Design	61
Methodology	61
Research Design	62
Defining a Practice Framework	62
Developing the Preliminary Practice Framework	63
Participants	67
Round 1 Modified Delphi Study	69
Round 2 Modified Delphi Study	69
Analysis	69
Ethical Considerations	70
Results	70
Demographics	70
Consensus Building	72
Discussion	74
Future Work	77
Strengths and Limitations	
Conclusion	78
5.0 CONVERSATION ANALYSIS STUDY (STUDY 3)	79
Chapter Introduction	79
Methodology	80

Research Design	82
Telehealth Cancer-related Fatigue Self-management Support Clinic	82
Context	83
Participants	83
Procedure	83
Data	
Analysis	87
Findings	87
Establishing a focus on fatigue self-management support from the outset of the cor	nsultation 88
Formulations	94
Using formulations to focus on matters relevant for fatigue self-management planni	ng 95
Using formulations to close down matters not relevant for fatigue self-management	planning98
Discussion	100
Limitations and Strengths	102
Conclusions	102
6.0 DISCUSSION AND CONCLUSIONS	104
Chapter Introduction	104
Overview, Key Research Findings, and Implications	104
Establishment of CRF SMS Best Practices	104
Exploration of how SMS Practice Could Be Delivered in Clinical Care	106
Limitations	107
Exclusion of Concurrent Symptoms and Psychological Outcomes	107
Specialised Settings	108
Focus on the Health Professional System of Support	108
Research for Contextual Evidence	110
Social Systems of Support	111
Conclusions	112
BIBLIOGRAPHY	
APPENDICES	132
Appendix 1: A Systematic Scoping Review of Implementation of Interventions for Car Related Fatigue Management	
Appendix 2: Population Intervention Comparison Outcome (PICO) Selection Criteria	• •
Appendix 3: Search Strategy SMS Systematic Review (Study 1) - Updated 20/06/202	:1 166
Appendix 4: Summary of Study Characteristics for Systematic Review (Study 1)	167
Appendix 5: Risk of Bias for each Included Study (Study 1)	179
Appendix 6: SMS Taxonomy Components in each study (Study 1)	
Appendix 7: Reported outcomes of included studies, post-intervention, and follow-up	
Annendiy 9, Auglitative Eventulate of Identified Themas (Study 9)	
Appendix 8: Qualitative Examples of Identified Themes (Study 2)	
Appendix 9: Changes from Round 1 of Modified Delphi Study (Study 2)	

Appendix 10: Cancer-Related Fatigue Self-Management Support Practice Framework (Study 2)	
	3

# SUPERVISION TEAM

### Principal Supervisor:

Professor Raymond Chan, Matthew Flinders Professor of Cancer Nursing, Director Caring Futures Institute and Dean (Research), College of Nursing and Health Sciences, Flinders University, South Australia, Australia

### Associate Supervisor:

Associate Professor Nicolas Hart School of Sport, Exercise and Rehabilitation, Faculty of Health, University of Technology Sydney (UTS), Sydney, New South Wales, Australia

### **External Supervisor:**

Associate Professor Stuart Ekberg School of Psychology and Counselling, Faculty of Health, Queensland University of Technology (QUT), Brisbane, Queensland, Australia

# ABSTRACT

**Background:** Cancer-related fatigue (CRF) is one of the most prevalent and distressing symptoms experienced by people diagnosed with cancer. Managing CRF requires individuals to adopt a range of self-management behaviours, thus quality self-management support (SMS) is critical. Clinicians report that a lack of clear guidance on SMS practices hinders their provision of supportive care for CRF. This doctoral thesis focuses on enabling health professionals to support cancer survivors to self-manage their CRF. Specifically, this study contributes to the literature by identifying the core practices required by health professionals to cancer survivors experiencing CRF.

**Aims:** The aims of this doctoral thesis were to: 1) examine and identify the core components, theories, and effectiveness of self-management interventions in relation to management of CRF for cancer survivors; 2) develop a framework of best practice guidance for SMS in managing CRF in cancer survivors; and 3) gain an understanding of clinical interactions as part of SMS practices in relation to CRF between health professionals and cancer survivors.

**Methods:** Three studies were conducted. <u>Study 1</u> was a systematic literature review of randomised controlled trials that examined CRF SMS programs. <u>Study 2</u> was a modified Delphi study (informed by Study 1) with cancer care clinicians, researchers, and cancer survivors to inform the development of a clinical practice guidance framework for CRF SMS. <u>Study 3</u> was a conversation analysis of video-/audio-recorded consultations between trained cancer nurse-counsellors and cancer survivors in CRF SMS clinic sessions, to understand how SMS tasks are accomplished in a real-world clinical setting.

**Results:** In Study 1, 51 papers representing 50 unique studies were identified following a systematic literature search. Synthesis of data illustrated CRF SMS delivered after cancer treatment, facilitated by health professionals, with at least one in-person contact appeared to produce the most favourable fatigue and behavioural outcomes in cancer survivors. Additionally, studies reporting the provision of additional training to intervention facilitators most frequently produced positive intervention effects for CRF and associated behavioural outcomes. Study 2 comprised two modified Delphi study rounds. Fifty-two panel participants in Round 1, and 32 panel participants in Round 2, produced consensus on a clinical practice framework with 44 items (13 Key Practices and 31 Practice Components). Lastly, the investigation of CRF SMS communication in Study 3 indicated that during supportive care sessions, clinicians should focus the conversation on CRF SMS early in the consultation, by clearly introducing the agenda of the consultation from the outset, followed by seeking client agreement. Additionally, Study 3 found that formulating or summarising patient's talk allows clinicians to maintain a focus on matters relevant for self-management fatigue planning; tie divergent conversation back to support for CRF; and potentially provide supportive care within limited time frames.

**Conclusions:** The clinical practice framework offers an evidence- and consensus-based model of best practice for health professionals providing SMS for CRF to cancer survivors. Future work is required to identify different stakeholders' needs in supporting the implementation of the framework in their local settings.

# DECLARATION

I, Oluwaseyifunmi Andi Agbejule, certify that this thesis does not incorporate without acknowledgment any material previously submitted for a degree or diploma in any university; and the research within will not be submitted for any other future degree or diploma without the permission of Flinders University; and that to the best of my knowledge and belief it does not contain any material previously published or written by another person except where due reference is made in the text.

Signed:

Date: 21/07/2023

# ACKNOWLEDGEMENTS

I would like to express my gratitude and appreciation to my Principal Supervisor and mentor, Professor Raymond Chan for introducing me to the world of cancer survivorship research. From undertaking a short vacation research program during my undergraduate degree, to submitting a PhD dissertation, I have accomplished so much under your guidance and mentorship.

I am deeply thankful to my entire supervisory team comprising Professor Raymond Chan, Associate Professor Stuart Ekberg, and Associate Professor Nicolas Hart, for their invaluable advice and support throughout this research journey. Their expertise, encouragement, and insightful feedback have been instrumental in shaping the direction of this thesis and enhancing its quality. I also extend my heartfelt appreciation to my peers and friends for their camaraderie, fruitful discussions, and moral support, which made this experience enriching and enjoyable. I want to give a special shoutout to Ria Joseph, my PhD sister, whose positivity, and enthusiasm uplifted me especially during the challenging moments. Thank you Ria, for being an incredible source of inspiration and laughs, and for making my PhD journey memorable!

I would also like to extend my appreciation to the participants of my research who generously shared their time, knowledge, and experiences. Their contribution has been invaluable, and without their cooperation, this research would not have been possible. I am also grateful for the funding support of the Australian Government Research Training Program Scholarship.

Lastly, I am immensely grateful to my family for their unconditional love and encouragement throughout my studies. I give special thanks to my beautiful parents. Their unwavering love and belief in my abilities has been a constant source of motivation, and I owe the successful completion of this thesis to their support. Mummy and Daddy E sé, mo dúpé!

# LIST OF RELEVANT PUBLICATIONS COMPLETED DURING CANDIDATURE

## **Relating to Doctoral Thesis:**

Agbejule, O. A., Hart, N. H., Ekberg, S., Koczwara, B., Ladwa, R., Simonsen, C., Pinkham, E. P., & Chan, R. J. (2021). Bridging the research to practice gap: a systematic scoping review of implementation of interventions for cancer-related fatigue management. BMC Cancer, 21(1), 809. https://doi.org/10.1186/s12885-021-08394-3

<u>Agbejule, O. A.</u>, Hart, N. H., Ekberg, S., Crichton, M., & Chan, R. J. (2022). Self-management support for cancer-related fatigue: A systematic review. International Journal of Nursing Studies, 129, e104206. <u>https://doi.org/10.1016/j.ijnurstu.2022.104206</u>

<u>Agbejule, O. A.</u>, Hart, N. H., Ekberg, S., & Chan, R. J. (2023). Development of a self-management support practice framework for addressing cancer-related fatigue: a modified Delphi study. Journal of Cancer Survivorship. Advance online publication. <u>https://doi.org/10.1007/s11764-023-01348-7</u>

**Agbejule, O.A.,** Chan, R. J., Ekberg, S., Ashbury F. D., Kleckner., A. S., Hart, N. H. (2023) Cancer-related fatigue self-management: a MASCC-endorsed practice framework for healthcare professionals to optimally support cancer survivors. Supportive Care in Cancer, 31(12), 666. <u>https://doi.org/10.1007/s00520-023-08130-6</u>

## **Relating to Self-management Support**

Chan, R. J., Mayer, D. K., Koczwara, B., Loerzel, V., Charalambous, A., <u>Agbejule, O. A</u>., & Howell, D. (2020). Building capacity in cancer nurses to deliver self-management support: A call for action paper. Cancer Nursing, 43(5), 341–342. <u>https://doi.org/10.1097/NCC.00000000000855</u>

Crichton, M., Yates, P. M., <u>Agbeiule, O. A.</u>, Spooner, A., Chan, R. J., & Hart, N. H. (2022). Non-pharmacological self-management strategies for chemotherapy-induced peripheral neuropathy in people with advanced cancer: A Systematic Review and Meta-Analysis. Nutrients, 14(12), e2403. <u>https://doi.org/10.3390/nu14122403</u>

Chan, R. J., <u>Agbejule, O. A.</u>, Crawford-Williams, F., Yates, P. M., Koczwara, B., Erickson, J., So, W. K. W., Howell, D., & ISNCC Knowledge Development and Dissemination Committee (2022). Development of an international competency framework for nurses in the provision of self-management support to cancer populations. [Advanced Online Publication]. Cancer Nursing.

Ladwa, R., Pinkham, E. P., Teleni, L., Hanley, B., Lock, G., Nixon, J., <u>Agbejule, O. A.,</u> Crawford-Williams, F., Jones, L., Pinkham, M. B., Turner, J., Yates, P., McPhail, S. M., Aitken, J. F., Escalante, C. P., Hart, N. H., & Chan, R. J. (2022). Telehealth cancer-related fatigue clinic model for cancer survivors: a pilot randomised controlled trial protocol (the T-CRF trial). BMJ Open, 12(5), e059952. <u>https://doi.org/10.1136/bmjopen-2021-059952</u>

# **CONFERENCE PRESENTATIONS, POSTERS, AND AWARDS**

## **Conference Presentations and Posters**

- PC4 Scientific Symposium 2021
  - Survivorship Education for Primary Care Providers (E-poster Presentation)
- <u>Clinical Oncology Society of Australia (COSA) Cancer Survivorship Conference, 18th-19th</u>
   <u>March 2021</u>
  - Cancer Survivorship Education for Primary Care Providers (*E-poster presentation*)
  - Bridging the Research to Practice Gap: A Systematic Scoping Review of Implementation of Cancer-Related Fatigue Management (*E-poster presentation*)
- <u>4th Victorian Cancer Survivorship Conference 25th 26th March 2022</u>
  - Self-management Support for Cancer-related fatigue (Presentation)
- <u>COSA Cancer Survivorship Conference, 9th-10th March 2023</u>
  - Developing of a Self-management Support Framework for Cancer-related Fatigue (Physical Poster)
- <u>Multinational Association for Supportive Care in Cancer, 22<sup>nd</sup> -24<sup>th</sup> June, Japan 2023</u>
  - A Clinician Practice Framework for Supporting The Management Of Cancer-Related Fatigue: A Modified Delphi Study (*E-poster presentation*)

### Awards

- Primary Care Collaborative Cancer Clinical Trials Group (PC4) Symposium Best Poster <u>Award 2021</u>
  - Survivorship Education for Primary Care Providers
- Margaret Fay Fuller PhD Top-Up Scholarship 2022
- <u>College of Health and Nursing Science Best HDR Student Publication Award 2022</u>
  - o Self-management Support for Cancer-related fatigue: A Systematic Review
- <u>College of Health and Nursing Science HDR Student Publication Award 2023</u>
  - Development of a self-management support practice framework for addressing cancer-related fatigue: a modified Delphi study

# LIST OF FIGURES

Figure 1: Graphical Representation of Research Plan	23
Figure 2a: Theoretical Model of Perceived Self Efficacy for Fatigue Self-Management	27
Figure 3: The Chronic Care Model (Wagner, 1998)	31
Figure 4: PRISMA Flow Diagram	41
Figure 5: Risk of Bias in Included Studies	43
Figure 6: Fatigue Outcomes	53
Figure 7: Behavioural Outcomes	54
Figure 8: Key Features of Effective Self-management Support for Cancer-related Fatigue	58
Figure 9: Preliminary Practice Framework Development Process	63
Figure 10: Modified Delphi Study Process	73
Figure 11: The Telehealth Cancer-Related Fatigue (T-CRF) Nurse Consultation Model	82
Figure 12: Jeffersonian's Conversation Analysis Transcription Convention	85
Figure 13: Systems of Support (Rogers et al., 2011)	110
Figure 14: Conceptualising Evidence	111

# LIST OF TABLES

Table 1: Nineteen Core Capabilities and Self-efficacy (Lawn & Battersby, 2009)	. 32
Table 2: SMS Core Capabilities and Self-efficacy	. 33
Table 3: Summary of self-management support components and strategies used across studies           with positive intervention effects	
Table 4: Preliminary Practice Framework	. 64
Table 5: Demographics of panel participants	. 70
Table 6: Domains and Key Practices of Self-management Support Practice Framework	74
Table 7: Use of the Practice Framework	. 76

# **1.0 INTRODUCTION**

## **Background and Scope of the Problem**

For the last 15 years, cancer has been the leading cause of health burden in Australia, currently accounting for 18% of the total disease burden in the nation (Australian Institute of Health and Welfare, 2022). Due to advancements in early detection, diagnostics, and treatment, it is estimated that by 2040, almost 1.9 million Australians will be living with, or beyond cancer, increasing the non-fatal burden of cancer on the Australian population and health system (Cancer Australia, 2017; Hunter et al., 2019). As survival continues to improve, the pertinent need to address the many symptoms, late effects, and resulting long-term health needs of individuals living with cancer also increases. The term 'cancer survivor' encompasses the entire spectrum of an individual's experience of cancer including diagnosis, treatment, remission, surveillance, after-cancer care, and end of life (Miller et al., 2019; Rodriguez & Foxhall, 2018). The provision of continual treatment support and long-term survivorship care is made complex as survivors must often manage multiple new and pre-existing comorbidities, highlighting the increasing role of the cancer survivor as well as their health care teams in self-managing follow-up care. This doctoral thesis examines supporting the self-management of one of the most prevalent symptoms experienced by cancer survivors.

### **Cancer-related fatigue**

Cancer-related fatigue (CRF) is one of the most common and debilitating symptoms reported in people diagnosed with cancer (Banipal et al., 2017; Bower et al., 2000; Horneber et al., 2012; Luthy et al., 2011; Weis & Horneber, 2014). Although there are several existing definitions, CRF is commonly defined as "a distressing, persistent, subjective sense of physical, emotional, and/or cognitive exhaustion related to cancer or cancer treatment that is not proportional to recent activity and interferes with usual functioning" (Berger, Mooney, et al., 2015). Moreover, unlike 'normal' fatigue, CRF is not alleviated with rest or sleep and leaves the individual with a prolonged and overwhelming sense of exhaustion (Narayanan & Koshy, 2009).

Several studies have demonstrated that the duration and intensity of CRF greatly affects the quality of life (QOL) of cancer survivors during and after treatment; reducing physical, mental, emotional and social wellbeing (Banipal et al., 2017; Bower et al., 2000; Charalambous & Kouta, 2016; Curt et al., 2000; Lis et al., 2009; Luthy et al., 2011; Rodríguez Antolín et al., 2019). Studies have also identified that the adverse effects associated with CRF often persist long after treatment cessation, with one longitudinal study (Bower et al., 2006) reporting that

34% (n=259) of participants experienced significant fatigue 5-10 years post treatment (Biering et al., 2020; Goedendorp et al., 2012; Servaes et al., 2002). Additionally, a recent retrospective study found that cancer survivors reported more general (p=0.04), and mental fatigue (p=0.02) compared with controls, up to 15 years post-cancer diagnosis (Gernier et al., 2020).

Numerous management strategies have been investigated for alleviating and reducing CRF severity. Pharmacological treatments (e.g., erythropoietin, methylphenidate, modafinil, steroids, anti-depressants, etc.) are largely ineffective for CRF, with those that do demonstrate effectiveness often resulting in treatment-related adverse effects (Bohlius et al., 2014; Finnegan-John et al., 2013; Tomlinson et al., 2018). For example, in a recent meta-analysis of 117 trials, Tomlinson and colleagues found that CRF was significantly reduced with pharmacologic agents including erythropoietin [standardised mean difference (SMD): –0.52; 95%CI: –0.89 to –0.14] and methylphenidate (SMD: –0.36; 95%CI: –0.56 to –0.15). However, the study authors highlighted that the limited sustainability of such improvements in fatigue, and the adverse effects relating to both pharmacologic agents (e.g., insomnia, decreased appetite, weight loss, nausea, rapid pulse rate, etc.), reduced their clinical usefulness. As such, non-pharmacological interventions (e.g., exercise, psychosocial therapy, integrative therapies, energy conservation, multimodal therapy) have been recommended as first-line therapies for managing CRF (NCCN, 2017).

A persuasive body of quality evidence (Corbett et al., 2019; Finnegan-John et al., 2013; Hilfiker et al., 2018b; Mustian et al., 2017) concludes that physical activity and psychological-based interventions (e.g., cognitive behaviour therapy; CBT) are both effective and superior in managing CRF. For example, in a meta-analysis by Mustian and colleagues (Mustian et al., 2017), physical activity (WES, 0.30; 95% CI, 0.25–0.36; P < .001) and psychological interventions (WES, 0.27; 95% CI, 0.21–0.33; P < .001) exhibited significant moderate effects on CRF improvement both during and after primary cancer treatment. Similarly, Hilfiker and colleagues explored the effects of specific exercises and psychological interventions in their indirect-comparisons meta-analysis (Hilfiker et al., 2018a), concluding that relaxation exercise ranked highest for managing CRF during treatment (SMD of -0.77; 95% CI: -1.22 to -0.31); whilst yoga had the greatest effect after treatment (SMD of -0.68; 95% CI-0.93 to -0.43); with combined physical activity and CBT; aerobic exercise, resistance training, and combined aerobic-resistance training; being beneficial both during and after treatment, despite their lower effect size level (Hilfiker et al., 2018a). The impact of integrative therapies (i.e., acupuncture, massage, mindfulness-based stress reduction, energy conservation) on alleviating CRF have also been explored. Although, several trials have demonstrated integrative therapies can result in reduced CRF, these studies often had limitations (i.e., small

sample sizes or non-randomised); with two systematic reviews concluding there was insufficient and conflicting data to facilitate clinical recommendations on the effectiveness of such therapies (Finnegan-John et al., 2013; Sood et al., 2007).

### Self-management, Cancer-Related Fatigue, and Self-management Support

With the increased side effects and co-morbidities that come with cancer, cancer treatment and living longer, cancer survivors are expected to assume a greater role in self-managing their follow-up care. Although conceptualisations vary, self-management can be broadly defined as the "day-to-day management of chronic conditions by individuals over the course of an illness" (Grady & Gough, 2014). The self-management model of care has been highlighted as a critical component to ensure patient wellbeing, particularly in chronic illness and the use of non-pharmacological approaches to symptom management (Grady & Gough, 2014). In fact, all evidence-based CRF management strategies presented in the section above aim to address behavioural risk factors (e.g., sleep disturbance, physical inactivity, poor nutrition, depression) and require cancer survivors to play a key role in guiding their care through self-management (e.g., increasing exercise, conserving energy, improving nutritional status) (Chan et al., 2016). Engagement in these fatigue self-management behaviours can be complex, and can often require cancer survivors to recognise, track, self-monitor, self-report, and apply problem solving skills to manage their fatigue along with their other comorbid conditions (Howell et al., 2021) – tasks that are not typically part of their regular behaviour. Additionally, factors such as fatigue symptom burden, socioeconomic status, mental health, cognitive status, age, performance status, or other situational factors also make it challenging for those diagnosed with cancer to self-manage their fatigue (Berger, 2019; Howell et al., 2021). To navigate the complexities associated with fatigue management, cancer survivors therefore require access to comprehensive self-management support (SMS) that assists and enables them to manage their fatigue while accounting for their capacity, capability, confidence levels, and priorities. Moreover, SMS strategies (e.g., motivational interviewing, goal setting, action planning, progress evaluations) which assist cancer survivors to address behavioural risk factors and incorporate behaviour change are increasingly recognised as essential elements of CRF management (Girgis, 2020).

SMS is defined by the Institute of Medicine as "the systematic provision of education and supportive interventions to increase patients' skills and confidence in managing their health problems, including regular assessment of progress and problems, goal setting, and problem-solving support" (Institute of Medicine Committee on Identifying Priority Areas for Quality Improvement, 2003). As the non-fatal burden of cancer in Australia continues to increase (Cancer Australia, 2017; Hunter et al., 2019), and cancer expenditure consistently remains in

the top three spending categories in the nation (including expenditure on hospital admissions, pharmaceuticals, medical specialists, and national screening programs) (Australian Institute of Health and Welfare, 2021), the need to strengthen the self-management model of care through SMS initiatives has become more apparent (Grady & Gough, 2014). A 2017 report by the Australian Government Department of Health included SMS as one of five priority areas for Australia in a national framework for improving the quality of life of individuals with chronic conditions (Australian Health Ministers' Advisory Council, 2017). More specifically, Australian policymakers called for increased efforts to provide tailored SMS interventions as well as appropriate SMS training and education of the health workforce. Systematic evidence has shown that clinicians supporting people to self-manage their chronic symptoms reduces health service utilisation without compromising patient health outcomes (de Silva, 2011; Panagioti et al., 2014). Further, the provision of clinician SMS moves beyond the paternalistic 'expert' view of health care to a more patient-centred approach that prioritises an individual's goals in treatment decisions (Gudgeon et al., 2022). This patient-centred approach to care has been shown to contribute to better patient outcomes, and patient satisfaction with the quality of care received (Kuipers et al., 2019).

Health professionals play a key role in promoting and empowering cancer survivors to selfmanage their CRF. For example, such support could involve oncologists or primary care providers overseeing treatment and offering guidance specific to an individual's cancer therapies; nurses providing symptom monitoring and education; or specialists like exercise physiologists, psychologists, and nutritionists contributing with tailored exercise, mental health support, and dietary guidance aimed at managing and alleviating CRF during and after cancer treatment. It is therefore integral that clinicians possess the appropriate CRF SMS knowledge and expertise. Moreover, effective SMS requires the specific education and training of health professionals (Jordan et al., 2008), with studies demonstrating that clinician knowledge and education of SMS is associated with the improved uptake and implementation of patient selfmanagement programs and interventions (Holman & Lorig, 2004; Jordan et al., 2008; Lawn & Battersby, 2009). Further, the significance of empowering and providing cancer survivors with the knowledge and skills needed to manage their symptoms and adjust health behaviours through self-management is well recognised, with several studies reporting on the efficacy of SMS programs on a range of cancer-related symptoms (Boland et al., 2018; Boogaard et al., 2016; Cuthbert et al., 2019; Goldberg et al., 2019; Hammer et al., 2015; Howell et al., 2017; Kim et al., 2017; Smith-Turchyn et al., 2016).

#### **Translation of CRF Management Strategies to Practice**

As noted at the beginning of this chapter, the effective management of CRF is well established, with multiple prominent organisations - National Comprehensive Cancer Network (USA), American Society of Clinical Oncology, European Society of Medical Oncology, Canadian Association of Psychosocial Oncology – producing and disseminating evidence-based recommendations and guidelines for cancer fatigue management (Berger et al., 2015; Bower et al., 2014; Fabi et al., 2020; Howell et al., 2013; NCCN, 2017). However, the effective translation of these guidelines to clinical practice is rare (Agbejule et al., 2021; Hilarius et al., 2011; Jones et al., 2021; Pearson et al., 2017), resulting in CRF being under-reported, underdiagnosed, and undertreated (Hilarius et al., 2011; Lisy et al., 2019; Stone et al., 2000; Vogelzang et al., 1997; Wang et al., 2018). A systematic scoping review undertaken to inform the research reported in this doctoral thesis synthesised knowledge around the implementation of CRF interventions in routine clinical practice and presented the outcomes of implementation strategies used (Agbejule et al., 2021; see manuscript in Appendix 1). The review examined original research articles (i.e., randomised controlled trials, observational studies, qualitative studies, mixed methods studies, study abstracts, and program evaluations) as well as other grey literature (e.g., evaluations of modules, online programs, and institutional/government interventions) to understand efforts to implement CRF interventions in clinical care, and the outcomes of these implementation efforts. Despite the high prevalence of CRF and evidence-based interventions for managing CRF, the review found that there was limited evidence informing the sustainable implementation of CRF interventions. Potential reasons for lack of implementation are discussed in detail below. This limited evidence highlights the disconnect between the established 'gold-standard' management strategies for CRF, what occurs presently in routine clinical care, and what cancer survivors experience (i.e., CRF being under-reported and under-treated) (Agbejule et al., 2021). This systematic scoping review was not included as a main study of this doctoral thesis as it did not focus on the implementation of SMS programs. However, the key gaps in knowledge, and priorities for research identified in the review, informed the aims and objectives of this doctoral thesis.

It is well recognised across evidenced-based literature that systemic (e.g., policy work, embedding supportive care monitoring into the electronic health records, reimbursement for supportive care services), organisational (e.g., difficulties reviewing past notes, difficulties for HCPs to follow-up on referrals, a reactive approach to symptom management) and cancer survivor-level changes (e.g., improving survivor attitudes and beliefs about CRF) are needed to accelerate the translation of evidence on CRF management to clinical practice (Abdalrahim et al., 2014; Agbejule et al., 2021; Berger et al., 2015; Berger & Mooney, 2016; Hilarius et al., 2011; Jones et al., 2021). However, Berger and colleagues (2015), note that although

improved processes and policies can enable the translation of CRF guidelines to clinical practice, the role of the health professional is one of the most important. This is because clinicians remain the access point for cancer survivor care. Further, Berger and colleagues emphasise the importance of collaboration between health professionals and cancer survivors, concluding that the adoption of CRF guidelines is reliant on health professionals supporting cancer survivors to create tailored CRF management strategies, and providing cancer survivors with adequate instruction to support the execution of those strategies (Berger, et al., 2015). For this reason, the current doctoral thesis focuses on the role of health professionals in supporting cancer survivors to self-manage their CRF.

# Lack of clinician knowledge of CRF management strategies and poor communication affecting clinician-cancer survivor partnerships

For guided self-management to be successful, a positive patient-physician relationship has been shown to be a key factor (Coulter, 1997; Clark and Gong, 2000; Holman and Lorig, 2000). Difficulties in communication between cancer survivors and health professionals represent a key barrier to the implementation of CRF management strategies to practice, and is a large contributor to cancer survivors feeling dissatisfied with the CRF SMS they currently receive from clinicians (Jones et al., 2021; Jones et al., 2020). Implementation of CRF selfmanagement strategies relies on cancer survivors' understanding of the strategies and their importance. Further, when communication between cancer survivors and their healthcare teams is poor or ineffective, cancer survivors are left to self-initiate strategies through trial and error, despite often having limited knowledge and misconceptions of fatigue management strategies (Fitch et al., 2008). This poor communication can lead cancer survivors to distrust their health providers, and can result in non-compliance with, or inadequate implementation of, fatigue management strategies (Fitch et al., 2008; Pertl et al., 2014). A recent mixedmethods study emphasised that a lack of health professional knowledge on CRF management strategies along with breakdown in cancer survivor-provider communication, are key factors hindering the translation of evidence-based CRF management to practice (Jones et al., 2021). The study found that cancer survivors rarely received support from their health care teams on how to effectively manage their fatigue and often felt that they could not rely on health professionals for CRF management support (Jones et al., 2021). Additionally, cancer survivors often felt dismissed and discouraged in managing their CRF due to inconsistencies in clinician attitudes; screening and assessment strategies used; and the normalization of CRF without specific management strategies (Jones et al., 2021). Similarly, a survey of 2508 cancer survivors found that almost 60% did not feel well informed about fatigue, and 87% of cancer survivors that perceived they had severe fatigue did not receive support from their health teams (Schmidt et al., 2021). Deficits in clinicians' knowledge of the causes and appropriate management strategies for CRF, negatively impact cancer survivors (Pertl et al., 2014; Schmidt et al., 2021), with clinicians reporting their lack of CRF knowledge contributes to their unwillingness to communicate and initiate support discussions with survivors (Borneman et al., 2011).

Although the importance of health professional knowledge in the management of CRF is well documented; clinicians often find existing management guidelines too long, complicated, or lacking relevant information for optimal clinical utility (Hilarius et al., 2011; Jones et al., 2021; Pearson et al., 2017). A mixed-methods study by Pearson and colleagues (2017) found that the translation of CRF management guidelines to clinical practice was often hindered by guideline complexity, with clinicians reporting barriers such as guideline content being too broad (i.e., assessing the literature) whilst also lacking specific steps or strategies to inform clinical guidance (Pearson et al., 2017). Moreover, their study demonstrated that there is a pertinent need for pragmatic, practical, clinician-friendly frameworks that detail CRF management strategies and can be applied in a variety of settings (e.g., acute cancer, primary care, maintenance, etc.).

#### Lack of clinician knowledge on self-management support strategies

Despite being emphasised by national (Australian Health Ministers' Advisory Council, 2017) and international policy agendas (Institute of Medicine National Research Council, 2006; National Health Service, 2019), SMS is not always integrated into clinical care (Duprez et al., 2018; Kantilal et al., 2022; Rimmer et al., 2021; Slev et al., 2017). Education on SMS competencies are often not integrated into medical, nursing, or allied health undergraduate curricula or professional development pathways (Duprez et al., 2017; Gudgeon et al., 2022). Additionally, evidence shows that clinicians rarely provide SMS (Jongerden et al., 2019; Norris & Kilbride, 2014); and lack competency and confidence in support activities such as collaborative goalsetting, shared decision making, and coordinating follow-up care (Duprez et al., 2018). A study of health professionals in an Australian health care setting, found that 65% of participants had received no formal training in self-management techniques (Lake & Staiger, 2010). Further, when SMS is provided, it primarily consists of information provision alone (e.g., presenting patients with leaflets) which coincidentally involves the lowest level of patient participation (Duprez et al., 2018; Elissen et al., 2013). Findings from qualitative studies highlight that health professionals frequently rely on their own expert opinions during patient interactions, and struggle to relinquish control, often disregarding the expertise of the patient or their support networks (Matthews & Trenoweth, 2015; Mudge et al., 2015; Norris & Kilbride, 2014). Further, a perceived lack of guidance and time are commonly cited barriers to SMS provision (Gudgeon et al., 2022). A recent review by Kantilal and colleagues (2022) sought to understand the influences involved in facilitating clinicians to provide routine SMS to cancer survivors. Study authors found that health practitioners are more likely to engage patients in discussions of self-management if they have the appropriate SMS knowledge, consultation skills, a clear understanding of their SMS role and responsibilities, and an organisation and health system configuration that enables integration of support provision into to routine care (Kantilal et al., 2022). SMS is integral in the management of CRF; however, several studies have demonstrated that health professionals often lack the confidence, knowledge, and ability to provide effective SMS to cancer survivors experiencing CRF which hinders the uptake of CRF management guidelines to practice (Hilarius et al., 2011; Jones et al., 2021). Evidence highlights that cancer survivors report they do not regularly receive support from their health care teams to effectively manage their CRF (Hilarius et al., 2011). Additionally, in instances where supportive counselling does occur, education on ineffective CRF management strategies (such as increased rest) are commonly provided (Fitch et al., 2008; Hilarius et al., 2011). Hilarius and colleagues (2011) attribute these deficiencies to complex guidance and posit that guidance on SMS and counselling for CRF need to be further clarified and reformulated into best practices, with the most essential support strategies being highlighted (Hilarius et al., 2011).

### Difficulties in conceptualising self-management support for CRF

The identification of effective SMS strategies for CRF, together with the development and optimisation of SMS best practices, is needed to ensure health professionals (in acute, primary, and community care) have the needed capabilities to support cancer survivors to manage their fatigue and improve CRF outcomes. Current reviews on SMS interventions for individuals experiencing CRF do not examine or identify the essential self-management elements and mechanisms needed to establish a successful CRF SMS program. Lack of awareness of effective SMS practices and CRF management strategies are barriers to the successful implementation of CRF management guidelines to practice, and as such, efforts must be directed to improving health professional understanding in this area through the identification of SMS best practices. Recent reviews on SMS programs for individuals experiencing CRF do not explore or identify key SMS components or principles; a gap that has been routinely highlighted across the survivorship literature. For example, in their systematic review on SMS interventions for cancer survivors, Boland and colleagues (2018) emphasised that standardised definitions of self-management and essential SMS elements are needed to determine the effectiveness of certain components to improve health outcomes. Additionally, Howell and colleagues' review of self-management education programs identified that consensus on the essential components of self-management programs is critical to facilitate consistent and effective delivery of SMS (Howell et al., 2017). Inconsistency of self-

management conceptualisations, terms, and definitions has been identified as a key issue in self-management literature (Grady & Gough, 2014), and is a significant barrier to intervention development, the translation of research to clinical practice and, self-management knowledge advancement (Ryan & Sawin, 2009). In fact, systematic reviews of self-management interventions regularly conclude that variability in self-management definitions and intervention reporting impedes study synthesis and increases difficulty in determining the distinction of outcomes gained as a result of self-management practices (Boland et al., 2018; Boogaard et al., 2016; Cuthbert et al., 2019; Goldberg et al., 2019; Hammer et al., 2015; Howell et al., 2017; Smith-Turchyn et al., 2016). Moreover, the lack of clarity, definition and identification of SMS strategies utilised in CRF management hinders clinical implementation efforts (Hoffmann et al., 2014).

## **Research problem**

An extensive body of evidence on CRF and its effective management exist. Despite this, management strategies are rarely translated to clinical practice, and CRF remains one of the most prevalent symptoms and unmet needs in cancer survivors (Lisy et al., 2019; Molassiotis et al., 2017; Wang et al., 2018). Lack of health professional knowledge on effective CRF management and CRF SMS strategies, as well as low cancer-provider communication, have been highlighted as key barriers to CRF intervention translation. There is urgent need to move beyond the investigation of intervention effectiveness to the identification, dissemination, and implementation of evidence based CRF self-management strategies into clinical care, through the establishment of effective SMS practices (Agbejule et al., 2021).

## **Research Plan**

This doctoral research develops an evidence-based framework of best clinical practice guidance for SMS in managing CRF in cancer survivors (see Figure 1). This best practice framework outlines the essential SMS practices needed for health professionals to successfully facilitate cancer survivor uptake of CRF management strategies. Additionally, this doctoral thesis examines how these practices have been applied by clinicians in current clinical practice at an operational level and provides insight how practices could be executed at an interactional level.

### Aims

The aims for this doctoral thesis are to:

i. examine and identify the core components, theories, and effectiveness of SMS interventions in relation to management of CRF for cancer survivors (Study 1).

- ii. develop a framework of best practice guidance for SMS in managing CRF in cancer survivors (Study 2).
- gain an understanding of clinical interactions as part of SMS practices to manage CRF in cancer survivors (Study 3)



Figure 1: Graphical Representation of Research Plan

### Objectives

The objectives of this doctoral thesis are to:

- i. Conduct a systematic review to identify the core components and effectiveness of SMS interventions for CRF in cancer survivors.
- ii. Conduct a modified Delphi study with cancer care clinicians (medical, nursing, and allied health), researchers, and cancer survivors to inform the development of a clinical practice guidance framework for SMS in managing CRF in cancer survivors.
- iii. Conduct a conversation analysis of video-/audio-recorded consultations between trained cancer counselling nurses and cancer survivors in their CRF clinic sessions to understand how SMS tasks are accomplished in clinical practice.

### **Research Questions**

**Study 1 –** Systematic review of core components and effectiveness of SMS interventions for CRF in cancer survivors.

- 1. What is the effectiveness of SMS interventions on improving CRF and other relevant behavioural outcomes (e.g., frequency of behaviours, self-efficacy, intention) for cancer survivors?
- 2. What self-management theories, strategies and intervention components are associated with higher acceptability, and improved behavioural and CRF outcomes in cancer survivors?

**Study 2 –** Modified Delphi study to develop consensus on a health professional practice framework for the SMS of CRF.

1. What are the key practices and practice components needed for health professionals to deliver comprehensive SMS for CRF?

**Study 3 –** Conversation analysis of interactions between cancer nurses and survivors with CRF to understand how SMS for CRF is accomplished in clinical practice.

- 1. What communication practices are used by cancer nurses during SMS intervention consult actions with cancer survivors experiencing CRF?
- 2. What are the interactional consequences of these practices?

### **Thesis Outline**

This chapter has presented a background to the research problem, aims, and objectives for this doctoral thesis. Chapter 2 explores relevant health behaviour and self-management theories and contextualises the theoretical framework that will underpin the doctoral research. Chapter 3 presents a systematic review on the effectiveness of SMS interventions for relieving CRF. This systematic review also identifies and synthesises the key SMS components of effective interventions. Chapter 4 details the research design and results of a modified Delphi study that will inform the development of clinical practice guidance for SMS in managing CRF in cancer survivors. Chapter 4 also presents the resulting best practice guidance framework for health professionals providing support for those experiencing CRF. Chapter 5 details the methodology, methods, research design, and findings of a conversation analytic study examining clinical interactions as part of CRF SMS practices used by cancer nurse counsellors. Chapter 6 presents a discussion of the key findings and implications of the doctoral work, as well as future directions. The Appendices include a systematic scoping review that was conducted and published during the early stages of the doctoral program, as well as other supporting information.

# 2.0 THEORETICAL AND CONCEPTUAL FRAMEWORK

## **Chapter Introduction**

This chapter explores the relevant health behaviour and self-management theories that underpin this doctoral research.

## **History of Chronic Illness Self-management**

One of the earliest theories of self-management was produced by Corbin and Strauss in 1985. Their gualitative study identified three core sets of tasks associated with managing chronic illness at home: (1) medical management (e.g., medication adherence); (2) meaningful behaviour change and maintenance (e.g., smoking cessation); and (3) emotional management – learning to manage the emotions that often accompany chronic conditions – (e.g., fear, depression, anxiety, frustration) (Corbin & Strauss, 1985). Over subsequent years, several researchers built on the model purported by Corbin and Strauss to enhance understanding of self-management in chronic illness (Dunbar et al., 2008; Grey et al., 2006; Lorig & Holman, 2003; Ryan & Sawin, 2009). For example, Lorig and Holman (2008) expanded on Corbin and Strauss' three tasks and proposed five core self-management skills required by individuals with chronic illness: problem solving, decision making, resource utilisation, the formation of patient-provider partnerships, and action-planning. Ryan and Sawin's (2009) Individual and Family Self-management Theory, delved into the context (e.g., condition specific factors, individual and family factors), process (e.g., knowledge and beliefs, skills and abilities); and proximal (e.g., uptake of symptom management strategies) and distal (e.g., overall health status) outcomes of self-management practices. The internationally recognised Chronic Care Model (CCM) (Wagner et al., 1996) also recommended selfmanagement as one of six essential elements needed to provide better care for individuals with chronic conditions and emphasises the importance of clinician-patient collaboration in improving care.

Although conceptualisations, terms and definitions of self-management differ across the literature, a central objective across many self-management models, theories and support programs is the achievement of sustainable positive outcomes through the preparation and strengthening of an individual's self-efficacy (a person's belief in their capabilities to exert control over their own functioning and over events that affect their lives) to adopt self-management strategies and adapt health behaviours (Bandura, 1977; Lorig & Holman, 2003; Ryan & Sawin, 2009). The enhancement of self-efficacy is an important mechanism for self-management programs (Chan et al., 2016; Peters et al., 2019; Tan et al., 2021), with several

systematic reviews (Gong et al., 2021; Thornton et al., 2021; Zhang et al., 2022), and experimental studies (O'Leary, 1992; Peters et al., 2019; Sarkar et al., 2007) demonstrating that self-efficacy are associated with improvements in health outcomes, and are one of the largest predictors of positive health practices (Yarcheski et al., 2004). For example, a systematic review by Náfrádi and colleagues found that high self-efficacy was routinely associated with improved medication adherence (Náfrádi et al., 2017). Additionally, the authors of the review concluded that promoting medication adherence requires health professionals engaging in support activities to foster clinician-patient relationships and promote patient empowerment. Similarly, high levels of self-efficacy in cancer survivors have consistently been found to promote physical and psychosocial symptom self-management, nutritional intake, physical activity and exercise, and other health self-management behaviours (Gong et al., 2021; Thornton et al., 2021; White et al., 2019). High self-efficacy has also been associated with lower symptom occurrence and distress, and improved general health and quality of life (White et al., 2019). Evidently, self-efficacy is a necessary component of selfmanagement and SMS programs, and a key concept in how and why SMS programs facilitate the adoption and adaptation of health behaviours. Thus, before considering the SMS model of care that frames this doctoral thesis, it is important to understand the specific factors that influence self-efficacy and ascertain the significance of self-efficacy in CRF management.

### Self-efficacy, self-management, and cancer-related fatigue

Self-efficacy is a fundamental construct in several self-management and behaviour change theories. Bandura (1977) conceptualised self-efficacy as a descriptive model of human behaviour, whereby self-efficacy has a causal influence with expected behavioural outcomes (i.e., the higher an individual's self-efficacy, the increased likelihood they will engage in a behaviour and succeed). Bandura suggests four primary sources that influence the development of self-efficacy: (1) mastery experience, (2) vicarious experience, (3) verbal persuasion, and (4) psychological arousal (Bandura, 1977). Mastery experience (or performance outcomes) refers to the experience an individual gains when they succeed or fail at a particular task. Bandura posits that past experiences provide evidence as to whether an individual perceives they can succeed, with past successes building perceived efficacy and past failures undermining self-efficacy (particularly before a sense of self-efficacy is established) (Bandura, 1977). Vicarious experiences (or social role modelling) refer to the positive impacts of observing how other people succeed, whereby an individual is more likely to believe they can complete a task if they observe other individuals (particularly in similar circumstances) successfully completing a task (Bin Hasan et al., 2014). Verbal (or social) persuasion involves the continuous provision of explanation and feedback to convince an

individual they have the capability to succeed at undertaking a task. Lastly, physiological arousal describes how the emotional, physical, and psychological well-being of an individual influences their judgement in their abilities to complete a task. For example, depression and anxiety are significantly associated with low total, physical and emotional self-efficacy (Tahmassian & Jalali Moghadam, 2011), which can result in the creation of unrealistic and unattainable goals during action planning activities (Kenioua & El-Kadder, 2016). Thus, strategies that minimise negative states (e.g., worry, anxiety, sadness) and promote positive affects (e.g., tranquillity, happiness) should be used to assist in enhancing perceived self-efficacy.

As mentioned in Chapter 1, the effective management of CRF involves the use of nonpharmacological strategies such as physical activity and exercise, cognitive behaviour therapy, and energy conservation through activity management. Further, all CRF management strategies (including those that are pharmacological) require individuals to selfmonitor, self-manage and undergo a level of lifestyle change (Chiba et al., 2019). Thus, the importance of cancer survivor self-management in managing CRF cannot be overstated. Chan and colleagues demonstrate perceived self-efficacy as a significant factor underpinning cancer survivor's perceived uptake and adoption of CRF self-management behaviours (Chan et al., 2016). Similarly, in a theoretical model designed to map the role of perceived selfefficacy for self-management on an individual's ability to manage CRF, Hoffman and colleagues identified that greater perceived self-efficacy predicted greater physical functional



Figure 2a: Theoretical Model of Perceived Self Efficacy for Fatigue Self-Management

(Hoffman et al., 2009). Reproduced with permission from Wolters Kluwer Health, Inc.

Solid line means a significant direct path (t > -/+ 2.0). • Dashed line means a nonsignificant direct path (t < 2.0). • The numerical values represent standardized path coefficients.

status, as illustrated in Figure 2a (Hoffman et al., 2009). In contrast, individuals with greater CRF severity had reduced self-efficacy to manage their fatigue, leading to worse treatmentrelated outcomes. Similar relationships between self-efficacy and CRF have been replicated in numerous studies across literature (Chen et al., 2018; Chin et al., 2021; Saito et al., 2022; Zhang et al., 2022). For example, in Chen and colleagues' (2018) prospective cohort study of 456 lung cancer patients, self-efficacy had a direct influence on CRF, and was found to be a high (r=0.69) predictor of increased quality of life, and a moderate predictor (r=-0.42) of reduced CRF (see Figure 2b).



**Figure 2b:** Self-efficacy, Cancer-related fatigue, and Quality of Life Model (Chen et al., 2018). *Reproduced with permission from John Wiley and Sons.* 

*PF=physical functioning, BP=bodily pain, GH=general health, VT=vitality, SF=social functioning, RP=role limitations due to physical problems, RE = role limitations due to emotional problem, MH=mental health* 

Accordingly, these findings emphasise that SMS strategies and programs that enhance perceived self-efficacy are crucial to facilitate sustained behavioural uptake of fatigue self-management activities, better CRF control, and improved health outcomes (e.g., improvements in physical functioning, bodily pain, general health, vitality, social functioning, role limitations due to physical and emotional problem, and mental health). The following section will present the SMS conceptual framework that informs this doctoral thesis.

## Conceptual Framework: Flinders Capabilities for Self- Management Support

### Introduction

This doctoral thesis focuses on enhancing the support provided to health professionals engaging in SMS activities for CRF. The Capabilities for Supporting Prevention and Chronic Condition Self-Management offers an ideal framework for this doctoral thesis as it emphasises the role health professionals play in building self-efficacy and conceptualises selfmanagement as an active process that requires an essential set of collaborative and partnership building behaviours, skills, knowledge, and attitudes. Other self-management models and frameworks such as the Chronic Care Model, Individual and Family Selfmanagement Theory, or the Chronic Disease Self-Management Program have been widely adopted by various health care systems in the USA, United Kingdom, Canada, Australia, and New Zealand (Grady & Gough, 2014). However, these frameworks focus on patient selfmanagement education programs, and generic systems-level changes that may impact on patient-related and provider-related factors, as opposed to specific strategies to enhance clinician SMS (Grady & Gough, 2014). A review by O'Connell and colleagues (2018) identified eight SMS frameworks for chronic disease that were published between 2008 and 2017. Although six out of the eight frameworks prioritised training and education for health care professionals as key courses of action to support self-management, none of the frameworks provided specific recommendations of the required competencies, skills or activities needed to undertake SMS (O'Connell et al., 2018), and instead presented a general view of the SMS training needed (e.g., priority statements, calls to action, general objectives and missions statements).

Clinicians frequently report they are unaware of fatigue support strategies, are often unwilling to initiate support discussions about fatigue with patients (Borneman et al., 2011; Mudge et al., 2015; Vogelzang et al., 1997), and have called for specific guidance on CRF SMS practices (Hilarius et al., 2011). Thus, a health professional capability framework was considered particularly appropriate, as this doctoral thesis seeks to improve the capacity of health professionals to provide SMS care for CRF through the development of a practice guidance framework. The Capabilities for Supporting Prevention and Chronic Condition Self-Management framework operationalises the self-management models such as the Chronic Care Model and other SMS frameworks by providing actionable, pragmatic capabilities for SMS that can be easily translated to clinical practice and applied to health professionals caring for cancer survivors. Additionally, the Capabilities for Supporting Prevention and Chronic Condition Self-Management framework was designed using input from prominent national organisations representing Australian patients, primary health professionals, and educators,

and therefore is relevant and suitable for use in the Australian context. Finally, the framework is particularly relevant to the doctoral research as it was used to inform the T-CRF intervention that was analysed in the conversation analysis study (Study 3).

### Background

The Capabilities for Supporting Prevention and Chronic Condition Self-Management (hereinafter referred to as The Flinders SMS Capabilities Framework, or 'FCF') was designed as a joint health initiative between Flinders University and the Australian Department of Health and Ageing (Lawn & Battersby, 2009). The framework provides evidence-based guidance on the core skills required by health care professionals for SMS to be successfully implemented into practice. Although primarily developed for use by the primary health care workforce, the FCF has been applied in a variety of health contexts and is considered appropriate for use in the context of cancer-related symptom management (Lawn & Lawton, 2011; Turner et al., 2014). Most recently, the framework was used to inform a self-management intervention for head and neck cancer survivors that was facilitated by oncology nurses (Turner et al., 2014). For the remainder of this document, the FCF will be described in the context of cancer and CRF.

The FCF proposes 19 Core Capabilities needed for health professionals to successfully support individuals and their carers to self-manage chronic conditions. These 19 capabilities were informed by an extensive review of SMS literature and curricula and a survey of key stakeholders. These stakeholders included the national primary health care workforce; training organisations delivering self-management education; tertiary education providers from the medical, nursing and allied health disciplines; professional accreditation bodies; and other clinical delivery sectors (Lawn & Battersby, 2009). Additionally, the framework was heavily underpinned by the Chronic Care Model introduced earlier in this chapter (Wagner et al., 1996).

The CCM was developed to improve health outcomes by enhancing and optimising interactions between health professionals and patients. Wagner suggests that improved health outcomes are reliant on the reconfiguration of clinical systems to address the needs and concerns of individuals with chronic conditions (Grover & Joshi, 2014). The CCM classifies six key elements that interact to enable quality chronic disease care: (1) a health system design that fosters a quality improvement culture of chronic disease service delivery; (2) clinical information systems that efficiently support chronic care and facilitate exchange of information between patients and providers (e.g., quality records, follow-up, recall); (3) decision support using evidence-based guidelines; (4) delivery system design that enables individuals to receive co-ordinated, tailored care by well-informed clinicians; (5) SMS; and (6)

the development of partnerships and alliances with the community to support the health care efforts of clinicians (Grover & Joshi, 2014; Savage, 2009) (see Figure 3).



## The Chronic Care Model

**Improved Outcomes** 

Figure 3: The Chronic Care Model (Wagner, 1998)Reproduced with permission from ACP-ASIM Journals and Book.

Although, placing particular emphasis on the SMS component, the FCF considers all elements of the CCM and suggests that to action and operationalise the six CCM features posited, clinicians require appropriate knowledge, skills and attitudes (i.e., skills to support behaviour change; knowledge on planning care and accessing external resources; an understanding of social determinants of health and health promotion; how to work collaboratively within teams and systems; skills for effective communication). Moreover, the importance of cancer survivor-centred communication is particularly emphasised in both the CCM and FCF, and is considered integral to empower survivors to self-manage, reduce risk, and maintain wellness (Lawn & Battersby, 2009).

The importance of clinician-survivor partnership in the CCM is demonstrated in Figure 3, which displays that improved health outcomes are dependent on productive interactions between patients and the health care team. Similarly, the FCF asserts that to successfully support cancer survivors to self-manage their symptoms (e.g., fatigue), clinicians must acquire patient-centred skills for effective engagement and communication between cancer survivors, health

professionals and health systems. The required knowledge and attitudes posited in the FCF are expanded further in the suggested 19 Core Capabilities needed for providing SMS.

### **19 Core Capabilities**

The 19 Core Capabilities proposed in the FCF encompass the essential knowledge, attitudes and skills needed by health professionals to support cancer survivors to self-manage. These core capabilities are displayed in Table 1 below.

General Patient-Centred	Behaviour Change	Organisational/Systems
Capabilities	Capabilities	Capabilities
1. Health promotion	9. Models of health	14. Working in
approaches	behaviour change	multidisciplinary
2. Assessment of health risk	10. Motivational interviewing	teams/interprofessional
factors	11. Collaborative problem	learning and practice
3. Communication skills	definition	15. Information, assessment,
4. Assessment of self-	12. Goal setting and goal	and communication
management capacity	achievement	management systems
(understanding strengths	13. Structured problem	16. Organisational change
and barriers)	solving and action	techniques
5. Collaborative care	planning	17. Evidence-based
planning		knowledge
6. Use of peer support		18. Conducting practice-
7. Cultural awareness		based research/quality
8. Psychosocial assessment		improvement framework
and support skills		19. Awareness of community
		resources

Table 1: Nineteen Core Capabilities and Self-efficacy (Lawn & Battersby, 2009)

The FCF emphasises the importance of operationalising the SMS Core Capabilities into a selfmanagement care plan. In the context of CRF, a best practice care plan should contain an assessment of self-management capacity and provide cancer survivors with specific skills and tasks to assist in managing CRF (e.g., goal setting, action planning, problem solving, emotional management, fatigue management, psychosocial skills, cognitive change skills, and relapse prevention skills). Further, these components will engage cancer survivors in managing their treatment and care, enhance the cancer survivor-clinician relationship, and enhance survivors' ability to maintain changes once fatigue self-management outcomes are achieved (Lawn & Battersby, 2009).

The FCF suggests that an additional key objective of the SMS Core Capabilities (and selfmanagement care plan) is to enhance the self-efficacy and confidence of cancer survivors to perform self-management tasks (e.g., 150 minutes of moderate intensity aerobic activity a week) and achieve improved health outcomes (e.g., reduced fatigue severity) (Lawn & Battersby, 2009). This is particularly evident as strategies addressing and/or incorporating the four factors influencing self-efficacy (mastery experience, vicarious experience, verbal persuasion, psychological arousal) can be found across many of the proposed SMS Core Capabilities. For example, goal setting, and motivational interviewing techniques use 'verbal persuasion' to help cancer survivors overcome resistance and build confidence in their ability to perform a task (e.g., exercise). Additionally, the incremental achievement of goals derived during action planning and goal setting (e.g., completing 4,000 steps a day) is an example of 'mastery experience', as direct performance accomplishments increase perceived capability and self-efficacy, which in turn, increases the likelihood of task progression and performance. The assessment of health risk factors and self-management capacity also link into mastery experience as they assist health professionals to create tailored, smaller, more manageable goals that are more likely to be accomplished. Likewise, psychosocial assessment and support and other assessments can also minimise the negative effects of 'psychological arousal' on perceived ability and self-efficacy. Additional examples of self-efficacy influencing factors included across the SMS Core Capabilities are displayed in Table 2.

Factors influencing Self-Efficacy	FCF Capability	Methods
Mastery Experience	<ul> <li>(1) Health promotion approaches</li> <li>(2) Assessment of health risk factors</li> <li>(4) Assessment of self- management capacity</li> <li>(12) Goal setting</li> <li>(13) Structured problem solving and action planning</li> </ul>	Successful perseverant efforts such as incremental achievement of goals (e.g., achieving 4000 steps per day) and other manageable tasks which in turn increase survivor willingness and confidence to engage and improve other CRF self-management behaviours.
Vicarious Experience	(6) Use of peer support	Observing similar role models complete a task (e.g., group classes, cancer survivor stories) increases perceived ability to undergo behaviour change (e.g., increase daily exercise).
Verbal Persuasion	<ul> <li>(3) Communication Skills</li> <li>(5) Collaborative Care</li> <li>Planning</li> <li>(10) Motivational Interviewing</li> <li>(11) Collaborative Problem</li> <li>definition</li> <li>(12) Goal setting</li> <li>(13) Structured Problem</li> <li>Solving and action planning</li> </ul>	The provision of encouragement, feedback and coaching through communication to foster cancer survivor confidence in their capability to achieve success and build a sense of mastery (e.g., verbal feedback from clinicians, app notifications).

Table 2: SMS Core	Capabilities an	d Self-efficacy
-------------------	-----------------	-----------------

Psychological Arousal	(2) Assessment of health risk	Assessment and collaboration with
	factors	cancer survivors to determine the
	(4) Assessment of self-	psychological, emotional, and social
	management capacity	barriers that may impede confidence in
	(5) Collaborative Care	ability; with the referral to relevant
	Planning	professionals as needed.
	(8) Psychosocial assessment	The provision of tailored support (coping
	and support skills	skills, relaxation exercises, other
	(11) Collaborative Problem	techniques) to minimise distress (e.g.,
	definition	anxiety) and cultivate eustress
		(beneficial stress) to improve perceived
		ability to perform tasks (e.g., exercise).

## Conclusions

This chapter presented a brief history of self-management, the influence of self-efficacy in SMS and fatigue management, and introduced the FCF as the conceptual framework for this research. The FCF is particularly relevant for this doctoral thesis as it was used to guide the development of the clinician SMS guidance framework for CRF (Study 2 – Chapter 4). The FCF also informed the Telehealth Cancer-Related Fatigue (T-CRF) intervention (Ladwa et al., 2022) described in the conversation analysis study (Study 3 – Chapter 5). Briefly, cancer nurses delivering the T-CRF intervention received training and utilised the SMS Core Capabilities proposed in the FCF (e.g., motivational interviewing, goal setting, collaborative care planning) to empower cancer survivors to manage their CRF. The FCF focuses on the role of health professionals in supporting individuals to self-manage chronic conditions – which aligns with the aims of the current doctoral thesis which focuses on the role of clinicians in supporting cancer survivors to manage their CRF. For this reason, the FCF was considered a suitable conceptual framework for this doctoral thesis.

# 3.0 COMPONENTS, PRINCIPLES, AND PROCESSES OF SELF-MANAGEMENT SUPPORT FOR CANCER-RELATED FATIGUE (STUDY 1)

## **Chapter Introduction**

As highlighted in Chapter 1, current reviews on SMS interventions for individuals experiencing CRF do not examine or identify the essential self-management elements and mechanisms needed to establish a successful CRF SMS program. Further a lack of identification of these important components has been highlighted as a barrier to the successful implementation of CRF management guidelines to practice, subsequently affecting the care cancer survivors receive (Hilarius et al., 2011; Jones et al., 2021). This chapter details the methodology and findings of a systematic review, that aimed to collate and examine knowledge on the effectiveness, elements, and processes of SMS interventions for individuals with CRF found in published literature. This review informs the development of a CRF SMS guidance framework for health professionals and explores the following two questions:

- What is the effectiveness of SMS interventions on improving cancer related fatigue and other relevant behavioural outcomes (e.g., frequency of behaviours, self-efficacy, intention) for cancer survivors?
- 2. What self-management theories, strategies and intervention components are associated with higher acceptability, and improved behavioural and CRF outcomes in cancer survivors?

This systematic review was accepted for publication in the International Journal of Nursing Studies on 10<sup>th</sup> February 2022 (DOI: <u>https://doi.org/10.1016/j.ijnurstu.2022.104206</u>) and has been included in this chapter.
# Background

As previously stated in Chapter 1, the effective management of CRF mainly involves the use of non-pharmacological strategies such as physical activity, exercise, cognitive behaviour therapy (CBT), and daily activity management (Corbett et al., 2019; Finnegan-John et al., 2013; Hilfiker et al., 2018b; Mustian et al., 2017). These management strategies aim to address behavioural risk factors for CRF (e.g., sleep disturbance, physical inactivity, poor nutrition, depression) and require individuals to play a key role in guiding their care through the adoption of fatigue self-management behaviours (e.g., increasing weekly physical activity, conserving energy, improving nutritional status) (Chan et al., 2016).

Such engagement is a complex patient-led endeavour that involves active self-monitoring, adherence to treatment, purposeful changes to multiple health and lifestyle behaviours, contact with healthcare providers, and engagement with peer and family support (Araújo-Soares et al., 2019; Schulman-Green et al., 2012). SMS for CRF aims to promote and facilitate cancer survivor behaviour change and can occur through a variety of approaches, including self-efficacy enhancement; motivational interviewing; behavioural counselling based on stages of change; counselling and behavioural modification; and other complex interventions (Minton et al., 2013). Further, effective SMS through the provision of problem solving and decision-making support, education, resources, regular assessment of progress, defined goals, and an established patient-health care provider partnership (Dineen-Griffin et al., 2019), have been widely identified as necessary and vital components of CRF management (Chiba et al., 2019; Girgis, 2020; Huang et al., 2019).

Although the effective management of CRF is well established, the effective translation of management strategies to practice is rare (Agbejule et al., 2021; Hilarius et al., 2011; Jones et al., 2021), resulting in CRF being under-reported, under-diagnosed and undertreated.

As highlighted earlier in Chapter 1, recent reviews on SMS programs in the cancer population do not explore or identify key SMS components or principles; a gap that has been routinely highlighted across cancer survivorship literature. In addition to the literature discussed in Chapter 1, three systematic reviews (Hernandez Silva et al., 2019; Kim et al., 2017; Xu et al., 2019) have investigated the impact of SMS programs on CRF demonstrating varying results of program efficacy and effectiveness. However, these reviews only focused on web-based interventions (e.g., mobile health applications, electronic health) and did not explore or identify the essential self-management components, theories, or behaviour change techniques of included interventions; a gap that has been routinely highlighted across SMS literature in

cancer survivors (Boland et al., 2018; Cuthbert et al., 2019; Howell et al., 2017; Kim et al., 2017).

Identifying SMS components is key to ascertaining whether a program can be successfully implemented, effectively evaluated, and improved over time; and is essential to determine the relevant educational and training requirements of clinicians and the impact, efficacy, and effectiveness of specific components on certain outcomes (Blase & Fixsen, 2013). This review seeks to address the identified gaps in literature by moving beyond the examination of program effectiveness, to gain a greater level of understanding of SMS for cancer survivors experiencing CRF.

To date, no systematic review has examined the theories, components, and effectiveness of SMS interventions for individuals experiencing CRF. To address this gap, the aim of this systematic review was to understand (1) the effectiveness of SMS interventions on improving CRF and other related behavioural outcomes (e.g., frequency of behaviours, self-efficacy, and intention) for cancer survivors; and (2) the self-management theories, strategies, and intervention components associated with improved behavioural and CRF outcomes in cancer survivors.

# **Methods**

This systematic review was reported according to the 2020 Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) Statement (Page et al., 2021) and was registered in the PROSPERO international prospective register of systematic reviews by the National Institute for Health Research (NIHR). Protocol registration ID: CRD42020207121.

## Identification of studies and inclusion criteria

Population, Intervention, Comparison and Outcome (PICO) framework information are presented in Appendix 2. The following databases were searched for articles published from inception to June 2021: Cumulative Index to Nursing and Allied Health Literature (CINAHL), US National Library of Medicine National Institutes of Health (PubMed), Cochrane CENTRAL, and the Excerpta Medica Database (EMBASE), as they provided collective coverage of medical, nursing, and allied health journals. Free text terms and relevant subject headings (i.e., MeSH, EMTREE) for "self-management support" and "cancer-related fatigue" were used. The full search strategy is presented in Appendix 3. Citation searching of relevant studies (e.g., self-management support systematic reviews, and included studies) was also conducted.

Included studies were required to meet the following criteria: (1) evaluate the effects of interventions designed for cancer survivors, where fatigue associated with cancer or cancer treatment is an outcome of interest; (2) aimed at supporting the self-management of cancer survivors; (3) published in English; and (4) a randomised controlled trial. To ensure that included studies reported on interventions where fatigue associated with cancer, or cancer-treatment was a prominent outcome of interest, studies which only reported tiredness using a question in a greater quality of life scale (e.g., European Organization for Research and Treatment of Cancer Quality of Life Questionnaire Core 30 - EORTC-QLQ-C30) were not included.

SMS interventions for CRF target cancer survivor behaviour change and support the reduction of fatigue severity through behavioural mechanisms (Hoffman et al., 2017). As such, fatigue severity was the primary outcome of this review, and behavioural outcomes were assessed as secondary outcomes (see Appendix 2). Behavioural outcomes can include fatigue management behaviours such as diet, physical activity, and exercise; and associated determinants such as fatigue self-efficacy (confidence to undertake fatigue self-management behaviours), other types of self-efficacy outcomes, and the intention to change behaviours.

While there is no 'gold standard' definition of a SMS intervention, this review employed the Taxonomy of Self-Management Support (or The Self-management Support Taxonomy) developed by Howell and colleagues (Howell et al., 2019) – an adaptation of the Practical Reviews in Self-Management Support Taxonomy (Pearce et al., 2016). The Taxonomy of Self-management Support comprises 14 components that can be used to support self-management when delivered to a cancer survivor or their carer (Pearce et al., 2016). Briefly, components comprise the provision of information about the condition, and its physical and psychosocial management; goal setting; regular clinical review; provision of equipment; monitoring of the condition with feedback, social support; lifestyle advice and support, and coaching for different coping strategies. Studies including an intervention with at least one of the core SMS components described in this taxonomy were considered for inclusion.

### Study screening and data extraction

Search titles and abstracts were screened by the thesis author and another researcher (Stuart Ekberg). The full text of eligible studies were then independently assessed for inclusion. Disagreements were resolved with another researcher (Raymond Chan) acting as arbiter where consensus could not be reached. All articles that met the specified inclusion criteria were included in the systematic review.

Data extraction was conducted by the thesis author and checked by another researcher (Megan Crichton). Data extraction included: study setting; participant and demographic characteristics; disease-specific factors (i.e. tumour type, tumour stage, treatment characteristics); time since completion of treatment; self-management or behavioural theories underpinning the intervention; intervention information for each arm of the study (e.g. method of delivery, intervention type, content, duration, number of sessions, duration of each session); descriptions of providers of the intervention and comparison intervention/s, timing, frequency and duration of follow-up for each outcome and key study findings. The Self-management Support Taxonomy (Howell et al., 2019) was used to extract SMS components. While primarily focussing on support delivered directly to cancer survivors or caregivers, the taxonomy also refers to indirect SMS provided to intervention facilitators (e.g., nurses, social worker, physical trainers, psychologists etc.). As such, information on any assistance (i.e., education and training; provision of equipment; prompts; feedback and review; financial incentives) that enabled facilitators to deliver SMS was also extracted. SMS strategies not included in the taxonomy were also recorded.

### Study quality assessment and data analysis

Quality assessment was independently conducted by the thesis author and another researcher (Megan Crichton) using the Revised Cochrane Risk-of-Bias tool for Randomised Trials (Sterne et al., 2019). This validated critical appraisal tool includes five domains: (1) bias arising from the randomisation process, (2) bias due to deviations from intended interventions, (3) bias due to missing outcome data, (4) bias in measurement of the outcome, (5) bias in selection of the reported result. Each domain was given a risk of bias judgement (low, some concerns, high). These decisions contribute to an overall risk of bias (low, some concerns, high) for each study. Any disagreements were resolved by another researcher (Raymond Chan) acting as arbiter. Due to study heterogeneity (i.e., diversity of populations studied; outcome measures; intervention duration, components, and aims; and duration of follow-up period), a meta-analysis could not be conducted. Rather, data were synthesised descriptively as having statistically significant, positive, negative, or null effects as reported by authors of the included studies.

# Results

Initial search results produced 772 articles. After removal of duplicates and title-abstract screening, 96 articles were assessed for full-text eligibility. Articles that reported findings from the same intervention were considered as one study, using the author's name associated with the earliest publication date. As such, 50 studies involving a total of 7,383 cancer survivors were included in the review (Abrahams et al., 2017; Armes et al., 2007; Bantum et al., 2014;

Barsevick et al., 2010; Barsevick et al., 2004; Bennett et al., 2007; Bruggeman-Everts et al., 2017; Chan et al., 2011; Dodd et al., 2010; Donnelly et al., 2011; Fillion et al., 2008; Foster et al., 2016; Freeman et al., 2015; Galiano-Castillo et al., 2016; Godino et al., 2006; Gokal et al., 2016; Golsteijn et al., 2018; Grégoire et al., 2020; Hoffman et al., 2017; Johns et al., 2016; Johns et al., 2015; Johnston et al., 2011; Lam et al., 2018; Lee et al., 2014; W. H. C. Li et al., 2018; Mooney et al., 2017; Naraphong, 2015; Purcell et al., 2011; Ream et al., 2015; Ream et al., 2006; Reif et al., 2013; Ritterband et al., 2012; Sandler et al., 2017; Savard et al., 2005; Schjolberg et al., 2014; Smith et al., 2019; Spahn et al., 2013; Stanton et al., 2005; Steel et al., 2016; Vallerand et al., 2018; van der Lee & Garssen, 2012; van Waart et al., 2015; van Weert et al., 2010; Vargas et al., 2014; Wang et al., 2011; Wangnum et al., 2013; Willems et al., 2017; Yates et al., 2005; Yun et al., 2012; Zhang et al., 2018). The selection of articles, including reasons for exclusion are outlined in Figure 4. All included studies used patient self-reported scales and questionnaires (e.g., Greater Fatigue Scale, Brief Fatigue Inventory, Schwartz Cancer Fatigue Scale, Multidimensional Fatigue Inventory) to measure all outcomes of interest.

## Figure 4: PRISMA Flow Diagram



41

### **Characteristics of Included Studies**

Study characteristics are summarised in Appendix 4. Studies and SMS programs were mostly developed in the United States of America (n=15/50, 30%), United Kingdom (n=6/50, 12%), and the Netherlands (n=7/50, 14%). Participant sample size across studies ranged from 13 to 588. Follow-up assessments were completed in 29 studies with follow-up periods ranging from 2 to 12 months. Fourteen studies (28%) used a waitlist control, 12 studies (24%) used an attention control group, one study (2%) did not specify control type, and the remaining studies (n=23/50, 46%) utilised usual care. Interventions were predominantly targeted towards cancer survivors with mixed cancer type (n=16/50; 32%), breast cancer survivors (n=16/50; 32%), and lung cancer survivors (n=6/50, 12%). The average intervention duration was 9.6 weeks (median = 8.5 weeks). Twenty-eight (55%) interventions were delivered post cancer treatment, 20 (39%) delivered during treatment, and three (6%) delivered during and following cancer treatment. One study (2%) compared post-cancer treatment intervention delivery alone with intervention delivery during and post-cancer treatment.

## **Risk of Bias**

A summary of the risk assessment across all studies is presented in Figure 5. Risk of bias for each included study is presented in Appendix 5. Twenty-two studies (n=22/50; 44%) had a high risk of bias, and 28 studies (n=28/50; 56%) had some concerns, with no studies presenting with an overall low risk of bias. Randomisation processes were sufficiently described in most studies and were a source of the lowest risk of bias (62% of studies reported low risk for this domain). The domains with the highest risk of bias were measurement of the outcome (84% some concerns, 4% high risk), missing outcome data (54% some concerns, 22% high risk), and deviations from intended interventions (56% some concerns, 20% high risk). The largest source of bias was the influence of self-reporting on blinding and outcome assessments, insufficient evidence that results were not biased by missing data (e.g., lack of sensitivity analyses), and a lack of detailed trial protocols.





## **Description of Self-management Support Programs**

### Cancer-related Fatigue Management Interventions

CRF management strategies described as interventions included: education (Bantum et al., 2014; Foster et al., 2016; Godino et al., 2006; Hoffman et al., 2017; Purcell et al., 2011; Reif et al., 2013; Schjolberg et al., 2014; Smith et al., 2019; Wangnum et al., 2013; Yun et al., 2012), physical activity (Bennett et al., 2007; Bruggeman-Everts et al., 2017; Donnelly et al., 2011; Gokal et al., 2016; Golsteijn et al., 2018; Lam et al., 2018; W. H. C. Li et al., 2018; van Waart et al., 2015; Wang et al., 2011), psychoeducation (Chan et al., 2011; Fillion et al., 2008; Stanton et al., 2005; Steel et al., 2016; Willems et al., 2017; Yates et al., 2005), exercise (Dodd et al., 2010; Galiano-Castillo et al., 2016; Naraphong, 2015), CBT (Abrahams et al., 2017; Ritterband et al., 2012; Savard et al., 2005), energy conservation (Barsevick et al., 2010; Barsevick et al., 2004), mindfulness-based cognitive therapy (Bruggeman-Everts et al., 2017; van der Lee & Garssen, 2012), mindfulness-based stress reduction (Johns et al., 2016; Johns et al., 2015), imagery-based behavioural therapy (Freeman et al., 2015), a symptom monitoring system (Mooney et al., 2017), cognitive behavioural stress management (Vargas et al., 2014); and combinations of self-care education and self-hypnosis (Grégoire et al., 2020), education and acupuncture (Johnston et al., 2011), exercise and diet education (Lee et al., 2014), exercise and mind-body therapy (Spahn et al., 2013), exercise and CBT (Sandler et al., 2017; Zhang et al., 2018), physical activity and CBT (van Weert et al., 2010), and fatigue education through MI and counselling (Ream et al., 2015; Ream et al., 2006).

### Mode of Self-management Support Delivery

Modes of intervention delivery varied across studies and included web-based (n=9/50, 18%), face-to-face (n=20/50, 40%), home-based (n=2/50, 4%), telephone-based (n=7/50, 14%), and mixed hybrid approaches (n=12, 24%). Most interventions were delivered in a one-on-one format (n=33/50, 66%) (Abrahams et al., 2017; Armes et al., 2007; Barsevick et al., 2010; Barsevick et al., 2004; Bennett et al., 2007; Bruggeman-Everts et al., 2017; Dodd et al., 2010; Donnelly et al., 2011; Foster et al., 2016; Galiano-Castillo et al., 2016; Godino et al., 2006; Gokal et al., 2016; Golsteijn et al., 2018; Hoffman et al., 2017; Johnston et al., 2011; Lam et al., 2018; Lee et al., 2014; Mooney et al., 2017; Naraphong, 2015; Ream et al., 2015; Ream et al., 2006; Ritterband et al., 2012; Sandler et al., 2017; Smith et al., 2019; Stanton et al., 2005; Steel et al., 2016; Vallerand et al., 2018; van Waart et al., 2015; Wang et al., 2011; Wangnum et al., 2013; Willems et al., 2017; Yates et al., 2005; Yun et al., 2012). One study (2%) described two separate interventions for delivery at the group and individual level (van Weert et al., 2010), and another study (2%) did not specify the target type (Chan et al., 2011). The remaining studies were delivered either in a group (Fillion et al., 2008; Freeman et al., 2015; Grégoire et al., 2020; Johns et al., 2016; Johns et al., 2015; W. H. C. Li et al., 2018; Purcell et al., 2011; Reif et al., 2013; Savard et al., 2005; Schjolberg et al., 2014; Spahn et al., 2013; van der Lee & Garssen, 2012; Vargas et al., 2014) (n=13/50, 26%), or mixed one-onone-group format (n=2/50, 4%) (Bantum et al., 2014; Smith et al., 2019).

### Facilitation of Self-management Support

Interventions were supervised by health professionals (Barsevick et al., 2004; Bennett et al., 2007; Bruggeman-Everts et al., 2017; Chan et al., 2011; Fillion et al., 2008; Godino et al., 2006; Grégoire et al., 2020; Hoffman et al., 2017; Johns et al., 2016; Johns et al., 2015; Johnston et al., 2011; Lam et al., 2018; W. H. C. Li et al., 2018; Mooney et al., 2017; Purcell et al., 2011; Ream et al., 2015; Ream et al., 2006; Reif et al., 2013; Sandler et al., 2017; Savard et al., 2005; Schjolberg et al., 2014; Smith et al., 2019; Spahn et al., 2013; Stanton et al., 2005; Steel et al., 2016; van der Lee & Garssen, 2012; van Waart et al., 2015; van Weert et al., 2010; Wang et al., 2011; Wangnum et al., 2013; Yates et al., 2005; Zhang et al., 2018), were entirely self-directed (Foster et al., 2016; Golsteijn et al., 2018; Naraphong, 2015; Smith et al., 2019; Willems et al., 2017; Yun et al., 2012), or were facilitated by other cancer survivors (Bantum et al., 2014), or research staff (Armes et al., 2007). Facilitators were not recorded or were unclear in eight studies (Dodd et al., 2010; Donnelly et al., 2011; Galiano-Castillo et al., 2016; Gokal et al., 2016; Naraphong, 2015; Ritterband et al., 2012; Vallerand et al., 2018; Vargas et al., 2014); however, one of these studies indicated that participant family members were invited to assist in SMS coaching (Dodd et al., 2010).

### Theories guiding Self-management Support Intervention Development

Most studies (n=45/50, 90%) mentioned the use of behaviour change and self-efficacy enhancing techniques. Thirty-one (n=31/50, 62%) studies explicitly reported their use of a specific behavioural theory or model to guide intervention development. These included the Common Sense Model (Barsevick et al., 2010; Barsevick et al., 2004); Stress Coping Model (Bruggeman-Everts et al., 2017); Cognitive Behaviour change principles (Abrahams et al., 2017; Bruggeman-Everts et al., 2017; Foster et al., 2016; Ritterband et al., 2012; Sandler et al., 2017; Steel et al., 2016; Vargas et al., 2014; Willems et al., 2017; Yun et al., 2012; Zhang et al., 2018); Adult Learning Theory (Dodd et al., 2010); Transtheoretical Model (Donnelly et al., 2011; Golsteijn et al., 2018; Lee et al., 2014; van Waart et al., 2015; Yun et al., 2012); Selfefficacy Theory (Foster et al., 2016; Lam et al., 2018; Wang et al., 2011); Theory of Planned Behaviour (Gokal et al., 2016; Golsteijn et al., 2018); Social Cognitive Theory (Golsteijn et al., 2018; Johnston et al., 2011; Lam et al., 2018; Yun et al., 2012); Health Belief Model (Golsteijn et al., 2018; Purcell et al., 2011); Translational Care Model (Hoffman et al., 2017); Theory of Symptom Self-management (Hoffman et al., 2017); Integrative Medicine Theory (Johnston et al., 2011); Kolb's experiential learning theory (Lam et al., 2018; W. H. C. Li et al., 2018); Leventhal's self-regulatory theory (Ream et al., 2015); a Behaviour Change model for internet interventions (Ritterband et al., 2012); the Multi-process Action Control framework (Vallerand et al., 2018); elements of cognitive therapy and mindfulness-based principles (van der Lee & Garssen, 2012); Problem Solving Therapy (Willems et al., 2017), Green's PRECEDE model of health behaviour (Yates et al., 2005); and other self-management and patient empowerment approaches (Grégoire et al., 2020); social learning, health action processes, precaution adoption processes and theories of self-regulation (Golsteijn et al., 2018; Stanton et al., 2005).

### Self-management Support Components, Strategies and Approaches

SMS components and strategies of included interventions are summarised in Table 3, with specific Self-management Support Taxonomy components used in each study presented in Appendix 6. SMS approaches varied across studies and included information provision (n=32), goal setting (n=27), problem solving (n=21), self-monitoring (n=18), coaching (n=16), action planning (n=14), sharing experiences (n=13), positive feedback (n=13), stress management (n=10), peer modelling (n=7), counselling (n=7), and motivational interviewing (n=3). The average number of Self-management Support Taxonomy components used across all studies was 6.1 out of a possible 14 Taxonomy components.

### Self-management Support Facilitator Training

Outside of education associated with professional qualification or registration, 30% (n=15/50) of studies described additional skills development for the cancer survivors or health

professionals who delivered interventions. Most of these 15 studies reported a positive intervention effect on CRF (n=11/15, 73%) and other behavioural outcomes (n=5/6, 83%), for at least one timepoint (immediate post intervention or follow-up). Study authors provided intervention facilitators with counselling (Barsevick et al., 2004), motivational interviewing (Bennett et al., 2007; Ream et al., 2015), muscle relaxation (Chan et al., 2011), CBT (Fillion et al., 2008; Zhang et al., 2018), and mindfulness-based stress reduction training, (Johns et al., 2016; Johns et al., 2015; van der Lee & Garssen, 2012) along with other study specific training (Armes et al., 2007; Bantum et al., 2014; Lam et al., 2018; Reif et al., 2013; Stanton et al., 2005; van Weert et al., 2010). Further detail on facilitator training is presented in Table 3.

Self-management Support for Cancer Survivors Experiencing Fatigue							
Key Components of Self- management Support for Cancer-related Fatigue (CRF)	Strategies Used to Accomplish Components						
Providing visual, written, or verbal information about CRF and/its management	<ul> <li>Providing information on:</li> <li>the definition of CRF</li> <li>the causes, presenting characteristics, and potential effects/ interferences on daily living</li> <li>recognised coping strategies (i.e., sleep hygiene, cognitive restructuring, physical activity, activity management, stress management, etc.).</li> <li>the benefits of proposed CRF strategies (i.e., physical activity, cognitive restructuring) on wellbeing.</li> </ul>						
Providing visual, written, or verbal information about managing the psychosocial consequences of cancer and CRF	<ul> <li>Information on how psychological and psychosocial factors contribute to and are exacerbated by CRF.</li> <li>Information and solution-focused counselling on the management of symptoms such as fear of cancer recurrence, anxiety, depression, stress, uncertainty.</li> <li>Information on coping approaches for:         <ul> <li>processing and communicating emotional experiences with others</li> <li>dealing with difficult emotions and thoughts</li> <li>establishing emotional boundaries and personal needs</li> <li>talking with others about cancer experiences.</li> </ul> </li> </ul>						
Providing visual, written, or verbal information about available social support resources and navigating transitions.	<ul> <li>Providing information on re-entry into 'normal' life after cancer, return to work, and seeking social support.</li> <li>Providing counselling on how to communicate CRF and cancer to personal social networks and employers.</li> <li>Providing links to external resources.</li> </ul>						
Creating a CRF action plan in collaboration with the cancer survivor and/ or their social network	<ul> <li>Education on effective goal setting and planning (e.g., SMART goals).</li> <li>Collaborative goal setting and written action planning for improving lifestyle behaviours:         <ul> <li>sleep hygiene</li> <li>healthy food choices</li> <li>exercise and physical activity</li> <li>activity management</li> <li>general wellness.</li> </ul> </li> <li>Collaborative reformulation of goals as needed</li> <li>The inclusion of cancer survivor social networks (family, carers, friends) in action planning</li> <li>Discussion and establishment of long-term goals at the cessation of self-management support</li> </ul>						

 Table 3: Summary of self-management support components and strategies used across studies with positive intervention effects.

Providing regular clinical review of CRF, the uptake of CRF self- management behaviours, and any adverse events related to self-management of CRF.	<ul> <li>Using in-person or telehealth consultations, or logbook/ dairy reviews to monitor:         <ul> <li>CRF severity and CRF interference on day-to-day activities</li> <li>CRF associated symptoms (i.e., depression, anxiety, stress)</li> <li>CRF self-management progress (e.g., progress with exercise, diet modification, activity management; addressing any identified barriers; assessing self-management self-efficacy; discussion of perceived future issues, addressing questions).</li> </ul> </li> <li>Provision of additional counselling, management strategies, and management strategy modification if needed.</li> <li>Providing progress feedback (i.e., goal attainment, treatment modification, relevant counselling and information, links to external training modules or resources).</li> </ul>
Providing practical support to assist with self-monitoring of CRF, reporting of associated symptoms, CRF self- management adherence Providing equipment to enable cancer survivors to participate in CRF self-management activities in their personal environments	<ul> <li>Using diaries, logs, physical activity trackers, heart rate monitors, frequent telephone calls, regular consultations, automated email/ text message reminders, and home visits to assist CRF self-monitoring, symptom reporting (i.e., sleep habits, fatigue log, changes in energy level, factors that impacted fatigue) and intervention adherence (i.e., exercise, mediation habits, physical activity, dietary habits).</li> <li>Automated email and text message reminders (intervention adherence, reminder of goals, messages of encouragement)</li> <li>Providing equipment that enables cancer survivors to engage in physical activity or exercise in their homes</li> <li>Recording of self-management support coaching/consult sessions to facilitate home practice</li> </ul>
Providing access to CRF management advice or support on cancer survivor request	Availability of a health professional for symptom review, treatment modification and/ or counselling if symptom thresholds exceeded (i.e., additional CBT counselling on request, accessibility to care coordinators, nurse practitioners, and psychologists for concerns and additional support).
Training rehearsal to communicate CRF with healthcare professionals	<ul> <li>The provision of problem-solving and solution focused skills to support survivors to communicate about fatigue and other cancer symptoms with treatment teams (family members and employers). Examples below: <ul> <li>Providing examples of questions to ask healthcare professionals and examples of goals to prepare for discussions ("Tomorrow morning I will write a list of questions about fatigue to take with me to my appointment with the nurse next week.")</li> <li>Advising survivors to show fatigue diary to health teams, invite significant others to accompany them to consults and explain fatigue interference in terms of everyday life (e.g., difficulty climbing stairs, cooking etc.)</li> <li>Training on how to ask clarifying questions</li> </ul> </li> </ul>
Providing coaching for lifestyle modifications that support living with CRF	Tailored coaching on exercise (aerobic, resistance, yoga, balance), physical activity (daily walking, morning stretches) meditation, relaxation and breathing exercises, dietary changes, changing sleep habits, structuring activities, goal setting, etc.
Providing skills and problem- solving strategies for coping with CRF-related psychological symptoms.	<ul> <li>Coping with fear of cancer recurrence, anxiety, stress, depression         <ul> <li>cognitive restructuring, cognitive reframing, relaxation exercises, mind-body therapy, meditation, yoga, breathing exercises, activating positive emotions, overcoming depressive periods, guided imagery meditation</li> </ul> </li> </ul>

	<ul> <li>Coaching on interpersonal relationships, distress management, sharing experiences with family and other cancer survivors, and assignments for building mental and social activity.</li> </ul>						
Providing Social support	<ul> <li>Education and information available on social support (e.g., online modules, links to available resources and how to seek support, navigating relationships and social support, caregiver monitoring and support)</li> <li>Facilitating the involvement of the patient's support network (family members, friends, carers, significant others) in fatigue self-management activities</li> <li>Coaching and encouragement on how to communicate fatigue with others</li> <li>Modifying cancer survivor expectations of social support</li> <li>Facilitating the exchange of experiences and discussions between cancer survivors (via online discussion forums, face-to-face group formats, individual chat functions in an online environment, etc.).</li> </ul>						
Providing general health	Advice on sleep hygiene, exercise, diet, activity management/regulation, physical activity, stress management, maintaining general						
promotion and education on lifestyle adaptation strategies	health and wellbeing (preventing weight gain, increasing fluid intake, anger management) improving concentration, energy conservation, dealing with inter-personal relationships, pain control.						
Structuring self-management interventions to align with survivors' 'everyday life'	<ul> <li>Providing pre-intervention consultations/assessments (fatigue severity, fatigue interference, level of physical activity, diet, stage of change, readiness to change)</li> <li>Creation of a priority list of usual activities that are then used to formulate an action plan (e.g., energy conservation, physical activity, etc.)</li> <li>Planning daily and weekly activities based on survivor livelihood</li> <li>Modifying treatment/ self-management strategies based on cancer survivor feedback</li> <li>Engaging significant others to assist with action planning and structuring cancer survivor treatment activities</li> <li>Incorporating survivor lifestyle and culture into intervention activities</li> <li>Home visits to establish familiarity with cancer survivors</li> </ul>						
Other self-management support	Checking patient understanding of CRF self-management						
strategies	<ul> <li>Knowledge quiz to test understanding of CRF and CRF management</li> </ul>						
	Placing cancer survivors in charge of the support they receive to improve agency (fatigue self-efficacy)						
	<ul> <li>Allowing cancer survivors to choose their treatment type/ intensity</li> </ul>						
	Allowing cancer survivors, the option to choose intervention delivery method (in-clinic consult, telemedicine, home-visit)						
	Retrieving cancer survivor feedback during the intervention to iteratively improve self-management support provided						
	Social/ peer modelling to improve confidence to undertake fatigue self-management behaviours						
	<ul> <li>role model videos, patient stories, transcripts, cancer survivors delivering interventions</li> </ul>						
	Facilitating anonymous communication between cancer survivors						
	The provision of constant encouragement to improve fatigue self-management confidence						
	(During scheduled sessions, telephone calls, emails, text messages)						

Self-management Support Facilitator Training (support delivered to intervention facilitators)

**Training Skills/Content** 

- Information on CRF
- How to respond to cancer survivor comments
- Goal setting
- Motivational interviewing MI (overview of MI, developing core skills, using MI in chronic disease settings)
- Counselling
- Skills to facilitate experiential learning, common barriers to physical activity
- Principles of communication
- Training on adult education, lifelong learning, group leadership, group pedagogy, group therapy and communication skills
- Progressive muscle relaxation training
- Mindfulness-based stress reduction training

Mode of Training Delivery

- Didactic lectures
- Individual & group exercises
- Roleplay scenarios
- Case studies
- Skills assessment to test understanding

Training materials

- Written manuals/booklets
- In-person day training
- Multi-day in-person workshops
- Multiweek courses
- Online courses & content

Abbreviations: CBT - cognitive behaviour therapy; CRF - cancer-related fatigue; MI - motivational interviewing; SMART - specific, measurable

### Effectiveness of Self-management Support Programs

Fatigue and behavioural outcomes are summarised in Figures 6 and 7, with further details on study outcomes including outcome measures and follow-up periods presented in Appendix 7.

### **Fatigue outcomes**

Fatigue outcomes were assessed in all studies. No studies reported a negative intervention effect. Twenty-nine studies reported a positive intervention effect for fatigue immediately post intervention completion (Abrahams et al., 2017; Armes et al., 2007; Barsevick et al., 2004; Donnelly et al., 2011; Fillion et al., 2008; Freeman et al., 2015; Galiano-Castillo et al., 2016; Gokal et al., 2016; Golsteijn et al., 2018; Grégoire et al., 2020; Hoffman et al., 2017; Johns et al., 2015; Lam et al., 2018; Lee et al., 2014; Mooney et al., 2017; Ream et al., 2015; Ream et al., 2016; Ritterband et al., 2012; Sandler et al., 2017; Savard et al., 2005; van der Lee & Garssen, 2012; van Waart et al., 2015; van Weert et al., 2010; Wang et al., 2011; Wangnum et al., 2013; Willems et al., 2017; Yates et al., 2005; Yun et al., 2012; Zhang et al., 2018); and were mostly delivered post-cancer treatment (n=15/29, 52%), facilitated by health professionals (n=20/29, 69%), targeted at the individual level (n=22/29, 76%), guided by a specific behaviour change theory (n=21/29, 72%), and incorporated an in-person face-to-face component (n=19/29, 66%). The average number of Self-management Support Taxonomy components used across the 29 immediately effective studies was 6.6 (of possible number of 14). The average intervention duration was 10.5 weeks (median = 9 weeks).

Twenty-nine studies (n=29/50, 58%) completed a follow-up assessment of fatigue (follow-up period ranging two to three months post intervention). Of these, 10 studies (34%) (Donnelly et al., 2011; Galiano-Castillo et al., 2016; Golsteijn et al., 2018; Johns et al., 2015; Lam et al., 2018; Reif et al., 2013; Smith et al., 2019; van der Lee & Garssen, 2012; Vargas et al., 2014; Zhang et al., 2018) reported positive follow-up effects for fatigue, and comprised of physical activity programs (n=3), education-based programs (n=2), a mindfulness-based stress reduction program (n=1), an exercise program (n=1), mindfulness-based cognitive therapy (n=1), cognitive behavioural stress management (n=1), and an exercise with CBT intervention (n=1). The majority of these 10 studies were targeted at the individual level (n=6/10, 60%), facilitated by health professionals (n=5/10, 50%; three of ten studies were unclear), incorporated a face-to-face component (n=7/10, 70%), and were delivered solely post-cancer treatment (n=5/10; 50%). Five of the ten (50%) interventions were informed by a specific behaviour change theory or model. The average number of Self-management Support Taxonomy components used across the ten studies was 5.6, and the average intervention duration was 11.2 weeks (median = 10 weeks). One study (Li et al., 2018) reported a null

intervention effect immediately post intervention but reported a positive intervention effect six months after study completion at follow-up.

### Self-management Behavioural outcomes

Seventeen studies assessed physical activity outcomes (Armes et al., 2007; Bantum et al., 2014; Bennett et al., 2007; Fillion et al., 2008; Gokal et al., 2016; Golsteijn et al., 2018; Hoffman et al., 2017; Lam et al., 2018; W. H. C. Li et al., 2018; Naraphong, 2015; Purcell et al., 2011; Reif et al., 2013; Sandler et al., 2017; Vallerand et al., 2018; van Waart et al., 2015; Wang et al., 2011; Wangnum et al., 2013); eight studies assessed self-efficacy for physical activity (Bennett et al., 2007; Hoffman et al., 2017; Lam et al., 2018; W. H. C. Li et al., 2018; Naraphong, 2015; Reif et al., 2013; Wang et al., 2017; Lam et al., 2018; W. H. C. Li et al., 2018; Naraphong, 2015; Reif et al., 2013; Wang et al., 2011); three studies assessed diet outcomes (Bantum et al., 2014; Lee et al., 2014; Wangnum et al., 2013); two studies assessed self-efficacy for fatigue self-management (Foster et al., 2016; Hoffman et al., 2017), and single studies assessed chronic disease management self-efficacy (Smith et al., 2017), cancer self-efficacy (Yates et al., 2005), general self-efficacy (Reif et al., 2013), stage of change (motivational readiness) (Lee et al., 2014), and the ability to complete leisure and self-care activities (Purcell et al., 2011; Smith et al., 2019).

No studies reported a negative intervention effect for any behavioural outcome. Of the studies that assessed physical activity outcomes, ten (n=10/17, 59%) reported a positive intervention effect for physical activity immediately post intervention, and five (n=5/17, 29%) reported sustained positive interventions effects at follow-up (follow-up ranging from 1.5 to six months post intervention). Of the three studies that assessed dietary outcomes, all reported a positive intervention effect immediately post intervention. No follow-up assessments for dietary outcomes were conducted.

Of the studies that assessed physical activity self-efficacy, seven (n=7/8, 88%) reported a positive intervention effect immediately post-intervention, and three (n=3/8, 38%) reported a sustained intervention effect at follow-up (follow-up ranging from 1.5 to six months post intervention). Of the two studies that assessed fatigue self-management self-efficacy, one reported a positive intervention effect post-intervention while the other reported null intervention effects immediately post intervention and at follow-up. A positive intervention effect was reported for general self-efficacy six months post intervention, no results were reported immediate post intervention. Null intervention effects were reported for chronic disease self-efficacy, cancer self-efficacy, motivational readiness, and leisure and self-care activity participation at all time points.

			FATIGUI	EOUTCO	MES			
	IMMEDIATE POST INTERVENTION	FOLLOW-UP		IMMEDIATE POST INTERVENTION	FOLLOW-UP		IMMEDIATE POST INTERVENTION	FOLLOW-UP
Abrahams 2017	•	DID NOT COMPLETE	Gregoire 2020	•	DID NOT COMPLETE	Schjolberg 2014	4 Ø	Ø
Armes 2007	+	Ø	Hoffman 2017	+	DID NOT COMPLETE	Smith 2019	NR	+
Bantum 2014	Ø	DID NOT COMPLETE	Johns 2015	+	+	Spahn 2013	Ø	Ø
Barservick 2004	. 🕂	DID NOT COMPLETE	Johns 2016	Ø	Ø	Stanton 2005	Ø	Ø
Barsevick 2010	Ø	DID NOT COMPLETE	Johnston 2011	Ø	DID NOT COMPLETE	Steel 2016	Ø	DID NOT COMPLETE
Bennet 2007	NR	Ø	Lam 2018	+	+	Vallerand 2017	Ø	DID NOT COMPLETE
Bruggeman- Everts 2017	NR	+	Lee 2014	+	DID NOT COMPLETE	Van der Lee 2012	+	+
Chan 2011	•	Ø	Li 2014	Ø	DID NOT COMPLETE	Van Waart 2015	5 +	Ø
Dodd 2010	NR	Ø	Mooney 2016	+	DID NOT COMPLETE	Van Weert 2010	) +	DID NOT COMPLETE
Donnelly 2011	•	+	Naraphong 2015	5 Ø	DID NOT COMPLETE	Vargas 2014	NR	+
Fillion 2008	•	0	Purcell 2010	Ø	Ø	Wang 2011	÷	DID NOT COMPLETE
Foster 2016	Ø	Ø	Ream 2006	+	DID NOT COMPLETE	Wangnum 2013	3 +	DID NOT COMPLETE
Freeman 2015	•	Ø	Ream 2015	+	DID NOT COMPLETE	Willems 2017a & 2017b	÷	Ø
Galiano- Castillo 2016	•		Reif 2013	NR	•	Yates 2005	+	Ø
Godino 2006	0	DID NOT COMPLETE	Ritterband 2012	+	DID NOT COMPLETE	Yun 2012	+	DID NOT COMPLETE
Gokal 2015	•	DID NOT COMPLETE	Sandler 2017	•	Ø	Zhang 2018	+	+
Golsteijn 2018	<b>A</b>	<b>(</b>	Savard 2005	+	Ø		_	_

# Figure 6: Fatigue Outcomes

*DID NOT COMPLETE* (did not complete follow-up assessment); NR (not recorded); + (positive effect); Ø(null effect).

			BEHAVIOU	RAL OU	TCOMES	;		
	IMMEDIATE POST INTERVENTION	FOLLOW-UP		IMMEDIATE POST INTERVENTION	FOLLOW-UP		IMMEDIATE POST INTERVENTION	FOLLOW-UP
PHYSICAL ACTIVITY OUTCOMES		DIET OUTCOMES		SELF-EFFICACY FOR CHRONIC DISEASE MANAGEMENT				
Armes 2007	+	•	Bantum 2014	÷	DID NOT COMPLETE	Smith 2019	Ø	Ø
Bantum 2014	+	DID NOT COMPLETE	Lee 2014	+	DID NOT COMPLETE	CAI	NCER SELF-EFFI	CACY
Bennet 2007	NR	Ø	Wangnum 2013	+	DID NOT COMPLETE	Yates 2005	Ø	Ø
Fillion 2008	0	Ø	SELF-EFFICAC	FOR PHYSICA	LACTIVITY	GENE	GENERAL SELF-EFFICACY	
Gokal 2015	+	DID NOT COMPLETE	Bennet 2007	+	+	Reif 2013	NR	Ø
Golsteijn 2018	+	+	Hoffman 2017	+	DID NOT COMPLETE		STAGE OF CHANGE (MOTIVATIONAL READINESS)	
Hoffman 2017	+	+	Lam 2018	+	+	Lee 2014	+	DID NOT COMPLETE
Lam 2018	NR	+	Lee 2014	+	DID NOT COMPLETE	LEISURE AND	SELF-CARE PAF	RTICIPATION
Li 2014	+	+	Li 2014	+	+	Purcell 2010	Ø	Ø
Naraphong 2015	+	DID NOT COMPLETE	Naraphong 2015	+	DID NOT COMPLETE			
Purcell 2010	0	Ø	Reif 2013	NR	Ø			
Reif 2012	NR	+	Wang 2011	+	DID NOT COMPLETE			
Sandler 2017	Ø	Ø		FICACY FOR FA				
Vallerand 2017	+	DID NOT COMPLETE	Foster 2016	Ø	Ø			
Van Waart 2015	+	Ø	Hoffman 2017	+	DID NOT COMPLETE			
Wang 2011	+	DID NOT COMPLETE						
Wangnum 2013	Ø	DID NOT COMPLETE						

# Figure 7: Behavioural Outcomes

*DID NOT COMPLETE* (did not complete follow-up assessment); NR (not recorded); +(positive effect); Ø(null effect).

# Discussion

This review aimed to identify the optimal SMS intervention features and components that were associated with improved CRF and behavioural outcomes. A graphical summary of these findings are presented in Figure 8.

### **Cancer-related Fatigue Management Interventions**

Various approaches to CRF self-management have been used to date. Across studies reporting positive intervention effects for CRF, several management strategies were used (e.g., exercise, psychoeducation, mindfulness, cognitive behaviour therapy, etc.) with no strategy clearly presenting as the most favourable. Conversely, psychoeducation and education interventions reported no statistically significant effect on CRF more often than any other management strategy. This finding is consistent with those from a recent systematic review (Bennett et al., 2016), which identified that self-management educational interventions have little effects on the severity, interference, and intensity of CRF; but may potentially produce a more moderate effect on fatigue distress in those with non-advanced cancer. Despite this, it is recognised that education and psychoeducation may still be useful in the management of CRF, particularly when used in synergy with other effective strategies, or in the development of knowledge of self-management skills (Bennett et al., 2016). This review identified that most psychoeducation and education interventions that reported no effect on CRF, incorporated minimal clinician-survivor interactions and were of short duration; potentially indicating that general information provision delivered over a short period of time may be insufficient to facilitate the uptake of CRF self-management behaviours. Evidently, more research is needed to identify how these education and psychoeducation sessions are being delivered, and to determine the optimal education content and delivery approach (e.g., timing of intervention, duration, number of contact points, modality) depending on clinical context (e.g., early-stage or late-stage cancer, advanced/non-advanced cancer, baseline fatique severity level).

### Self-management Support Delivery

SMS delivered after cancer treatment, in one-on-one formats, facilitated by health professionals, while incorporating at least one in-person (i.e., face-to-face) contact was most likely to have positive effects on CRF and behavioural outcomes. Additionally, programs with positive intervention effects were longer in duration than the average (9.6 weeks vs 11.2 weeks).

Several studies have linked factors such as self-management intervention duration and frequency (number of contacts per week, number of minutes per contact), with improvements in health behaviours (e.g., diet and physical activity or exercise) (Barrett et al., 2021; Eakin et al., 2007; Stormacq et al., 2020). As CRF can persist long after treatment cessation (Weis & Horneber, 2014), it is conceivable that an extensive intervention incorporating multiple contact points that promotes an environment where partnerships can be developed between intervention facilitators, cancer survivors, and their caregivers would be favourable for durably

55

supporting CRF management (Banbury et. al, 2023). However, while extensive support with sustained maintenance, in-person guidance, and face-to-face contact might be beneficial, cancer survivors may appreciate the simplicity of receiving support from the comfort and familiarity of their own home. Additionally, survivors may not have the time to commit to prolonged interventions. As posited by Bennet and colleagues (Bennett et al., 2016), future intervention developers should seek the preferences of those receiving support, and qualitatively evaluate the comparative benefits of different delivery options.

### Facilitation of Self-management Support

Notably, this systematic review found that studies which reported the provision of additional training to intervention facilitators – see Table 3 – (e.g., training on motivational interviewing, goal setting, counselling, communication, etc.), were more likely to report a positive intervention effect for both CRF and behavioural outcomes. This finding indicates that supporting and upskilling those providing SMS – by ensuring they have the necessary capabilities – is essential to facilitate cancer survivor uptake of CRF self-management strategies (Kennedy et al., 2005).

### **Theories Underpinning Self-Management Support Programs**

Most SMS interventions reporting positive intervention effects for CRF (n=21/29, 72% at immediate post-intervention; and n=5/10, 50% at follow-up post-intervention) and/or behavioural outcomes (n=9/15; 60%) were guided by a behaviour theory. While several factors can influence intervention effectiveness, this potentially indicates that theory use may be associated with increased adoption by cancer survivors of CRF coping behaviours: a finding consistent with several studies identifying the superiority of theory-based self-management programs over non-theory-based programs in facilitating patient behaviour change (van Vugt et al., 2013). However, before this potential association can be explored, further examination of the specific impact of each behavioural theory must occur. Further, while a behaviour or self-management theory was indicated in several included studies, in many cases theoretical application was not described at all. An appropriate description of theory use should highlight the theoretical factors underpinning the causes of a behaviour (e.g., low self-efficacy), and explain how these causes were targeted using the appropriate theoretical change methods (whilst also ensuring theoretical methods were used accurately to maximise effectiveness) (Bluethmann et al., 2017; Webb et al., 2010). Studies have demonstrated that when such application occurs the production of behaviour change is enhanced (Bluethmann et al., 2017; Webb et al., 2010). As such, it is recommended that intervention developers move beyond description, and clarify how underpinning theories are applied in intervention development, implementation, and evaluation; to establish more consistent and robust evidence in this area and improve translatability to practice (Rimmer, Sharp, & on behalf of Ways Ahead study, 2021).

## Self-Management Support Components

The heterogeneity and complex multi-component nature of SMS interventions makes it difficult to determine if an optimal number of support components should be implemented to optimise intervention efficacy; and challenging to identify which specific SMS components are associated with intervention efficacy. Nonetheless, this review collated and summarised the specific SMS strategies and components used in studies that had positive intervention effects for CRF at any timepoint, and subsequently demonstrated that all components of the Selfmanagement Support Taxonomy were present. While efficacy of self-management interventions on cancer survivor fatigue can be attributed to several factors (e.g., age; type of active treatment received; type and stage of cancer diagnosis; frequency, duration and intensity of intervention; characteristics of interventionists etc.), and not all taxonomy components were tested to the same extent, it can be said that when implemented well, each component of the Self-management Support Taxonomy (as well as the other identified strategies not included in the taxonomy - Table 3) can potentially be used to effectively support cancer survivors to manage their CRF. This finding is further indicated in a recent qualitative study by Sheridan and colleagues (2017) in which patients and caregivers describe all 14 components of the Taxonomy as crucial for their SMS; although this was in the context of general chronic conditions and primary care.

Despite this, the review went further, and identified that several strategies, including the provision of lifestyle advice and support; training for lifestyle adaptations and everyday activities; training for psychological strategies; practical support with self-management adherence; goal setting and specific clinical action plans; information on psychosocial management, information about CRF and its management, the monitoring of CRF and its management; and the structuring of interventions to align with survivors 'everyday life', were the most frequently featured strategies in interventions with positive intervention effects for any outcome. This potentially indicates that these strategies are associated with improved CRF and uptake of associated management behaviours. Consensus on the best strategies and practices for CRF support have yet to be established in literature, and as such will be examined in the following chapter.

## EFFECTIVE SELF-MANAGEMENT SUPPORT FOR CANCER-RELATED FATIGUE

SELF-MANAGEMENT SUPPORT COMPONENTS							
Providing information about CRF and/its management			Providing information about managing the psychosocial consequences of cancer and cancer-related fatigue		Providing health promotion and education on lifestyle adaptation strategies		
Providing skills and coping strategies for social support		Providing access to CRF management advice or support on cancer survivor request		Creating a CRF action plan in collaboration with the cancer survivor and/ or their social network			
	ding coaching for difications that su living with CRF	pport	Structuring self-management interventions to align with survivors' 'everyday life'		Providing information about available social support resources and navigating transitions.		
	Providing skills and pro strategies for coping wit psychological sym		vith CRF-related	Providing practical support to assist with self-monitoring of CRF, reporting of associated symptoms, CRF self- management adherence			
	Providing equipment to survivors to participate management activities in environmen		ite in CRF self- in their personal	Providing regular clinical review of CRF, the uptake of CRF self-management behaviours, and adverse events related to self-management of CRF.			
	IANAGEMENT DRT DELIVERY		SELF-MANAGEMENT SUPPORT FACILITATOR TRAINING				
		TRAININ	TRAINING CONTENT T			TRAINING DELIVERY	
	Delivered post- primary treatment		Mindfulness-basec stress reduction training	Progressive m relaxation trai		Roleplay	Didactic
			Information on CRF	Goal settin	g	scenarios	lectures



Figure 8: Key Features of Effective Self-management Support for Cancer-related Fatigue

### **Strengths and Limitations**

Researchers, health professionals, and cancer survivors have long called for the support of CRF to be examined and defined, and for the essential support strategies and components being highlighted (Hilarius et al., 2011; Jones et al., 2021; Pearson et al., 2017). This review is the first to identify the features, components, and strategies used in effective CRF SMS programs – allowing health professionals and researchers to consider including these elements in future support interventions for CRF. Further, this review not only examined the impact of SMS behavioural interventions on CRF severity, but also investigated the effects of interventions on the adoption of fatigue self-management behaviours; an important factor that is often overlooked.

There are four main limitations to this review. Firstly, due to study heterogeneity a meta-analysis was unable to be conducted; however, this was not unique to this current review; with similar difficulties being demonstrated across the SMS literature (Boland et al., 2018; Hilarius et al., 2011; Jones et al., 2021). Secondly, as a meta-analysis could not be conducted, statistical significance in single studies was largely used to determine intervention efficacy. As statistical significance relates to covariance between variables, a null intervention effect does not necessarily indicate that a self-management intervention is futile or without impact, particularly as SMS is complex with varying outcomes outside of reduced CRF severity and changed behaviours. Thirdly, 22 out of the 51 included studies were scored as having an overall high risk of bias largely due to the inherent nature of interventions under examination: lack of intervention blinding and, lack of sensitivity analyses. As such findings of this review need to be interpreted with caution. Guidance around the choice of sensitivity analysis for non-pharmacological behavioural interventions is unclear, and randomised control trial reporting guidelines (e.g., SPIRIT, CONSORT) do not list the inclusion of a sensitivity analysis as a recommended item; potentially explaining why it was not often conducted. While practices of sensitivity analysis may be improved, the lack of blinding remains an expected challenge for these types of behavioural interventions. Finally, this review only included studies published in English; thus, potentially relevant self-management programs may not have been included.

# Conclusion

This review indicated that structured, comprehensive support that facilitates clinician-survivor exchange and is tailored to individual context and priorities can favourably enable CRF selfmanagement. The mapping and categorisation of SMS components used across CRF selfmanagement interventions provides novel insight into the management of one of the most distressing symptoms experienced by cancer survivors to date. This systematic review is crucial in determining the relevant future training and support requirements needed to enable the delivery of CRF care beyond the research context.

59

# 4.0 MODIFIED DELPHI STUDY (STUDY 2)

# **Chapter Introduction**

This chapter contributes to the overall aim of the doctoral thesis by presenting a framework of core practices required by health professionals to deliver effective SMS to cancer patients and survivors experiencing CRF, addressing a key barrier to the successful implementation of CRF management (which was highlighted in Chapter 1). Specifically, this chapter outlines the methodology, results, discussion, and implications of the modified Delphi study. This chapter incorporates the Flinders Capability Framework presented in Chapter 2 (e.g., using the 19 core capabilities detailing the essential knowledge, attitudes and skills needed by health professionals to support cancer survivors to self-manage) and the findings from the systematic review presented in Chapter 3 (e.g., co-creating a CRF action plan, etc.) to develop the best practice framework. This modified Delphi study was accepted for publication by the Journal of Cancer Survivorship and published online on the 24<sup>th</sup> of February 2023. The online published article can be accessed via the following link: <a href="https://doi.org/10.1007/s11764-023-01348-7">https://doi.org/10.1007/s11764-023-01348-7</a>.

# Background

Effective management strategies for CRF are well established with the existence of a plethora of clinical guidelines including the National Comprehensive Cancer Network (NCCN) Guidelines for CRF (Berger, 2019), the Canadian Association of Psychosocial Oncology (CAPO) Pan Canadian Guidelines for CRF (Howell et al., 2015), and the European Society for Medical Oncology (ESMO) Clinical Practice Cancer-related Fatigue Guidelines (Fabi et al., 2020). Although these clinical guidelines advise health professionals on what strategies they 'should' recommend to those experiencing CRF (e.g., *advise patients to engage in moderate intensity of physical activity 30 minutes per day, 5 days per week as tolerated* (Howell et al., 2015)); they do not provide guidance on 'how' health professionals can support individuals to undergo behavioural change and *adopt* these management strategies (e.g., by providing tools to assist with exercise, creating goals and actions plans that are regularly reviewed, etc.).

As mentioned in Chapter 1, engagement in fatigue self-management behaviours can be complex and can require individuals to recognise, track, self-monitor, self-report, and apply problem solving skills to manage their fatigue along with other comorbid conditions (Howell et al., 2021) – tasks that are not naturally part of an individual's regular behaviour. Additionally, external factors such as socioeconomic status, mental health, cognitive status, age, performance status, and/or other situational factors also make it challenging for those diagnosed with cancer to self-manage their fatigue (Gobeil-Lavoie et al., 2019; Howell et al.,

2021). To navigate the complexities associated with fatigue management, cancer survivors require access to comprehensive SMS that assists and enables them to manage their fatigue whilst accounting for their capacity, capability, confidence levels, and priorities.

Health professionals regularly report they frequently lack the confidence, knowledge, and ability to provide effective SMS to those experiencing CRF (Hilarius et al., 2011; James et al., 2015; Jones et al., 2021; Pearson et al., 2015). Clinicians often describe existing clinical guidance as lacking the relevant information for facilitating SMS (Hilarius et al., 2011; Jones et al., 2021; Pearson et al., 2017). There is need for specific guidance for health professionals to facilitate evidence-based fatigue self-management strategies (Berger & Mooney, 2016; Hilarius et al., 2011; Pearson et al., 2017; Ren et al., 2022). The aim of the current study is to establish a *best practice framework* for facilitating effective SMS for CRF.

This chapter details the research design including the methodology, population, sample, data collection, analytical methods and instruments used. Ethical implications are also discussed.

# Methodology and Research Design

# Methodology

The modified Delphi technique is a structured group consensus strategy that systematically uses high quality literature, the opinion of stakeholders, and judgment of industry specialists to reach agreement and achieve content validity (Trevelyan & Robinson, 2015). Further, this technique uses a structured, anonymous, and iterative feedback process to establish consensus on a specific topic of interest with an invited panel of 'experts' over a series of rounds.

In a *classic* Delphi study approach, open-ended questions or forums are used to generate ideas and concepts that may inform Delphi study content (Trevelyan & Robinson, 2015). A *modified* Delphi design differs, as initial concepts and ideas are generated through a review of relevant literature. A modified Delphi approach was used for this study to incorporate the extensive CRF SMS strategies identified in literature and to ensure framework comprehensiveness. Moreover, the modified Delphi study technique has been recommended over the classic Delphi study as it reduces complexity and minimises the misinterpretation of objectives; avoids biasing of panellists through the inclusion of inappropriate or unnecessary items; and increases the likelihood that consensus will be achieved in an efficient and timely manner (Custer et al., 1999; Trevelyan & Robinson, 2015). Further, the modified Delphi study method is commonly used when little evidence exists (or is lacking), and thus is considered suitable for this research objective. The validity and quality of a Delphi study process largely

depends on panel expertise, the availability of clearly defined content, and well-developed initial statements and thus, it is essential that all three components are well developed (Custer et al., 1999).

# **Research Design**

A modified Delphi study design was used to generate consensus on the essential clinician SMS practices required by health professionals to deliver effective support to people experiencing CRF. Although, quantifying the level of consensus is important, reaching consensus on all statements is not the general objective of a Delphi survey; therefore, attainment of consensus was not used as a criterion for ending the study (Linstone & Turoff, 2011; Ramos et al., 2016). Two rounds of consultation were chosen because unlike the traditional Delphi approach, initial framework items for consensus were not generated during a Delphi workshop round. Instead, the initial framework was informed by a review of literature; thus, one round was removed.

The use of the term 'expert' is contentious as it is difficult to define (e.g., professional registration or number of years practising does not necessarily indicate topic expertise (Baker et al., 2006)). To avoid misunderstanding and increase clarity, the term 'experts' was not used to describe Delphi study participants. Rather, the criteria used to select panel participants was detailed as per recommendations in literature (Baker et al., 2006; Trevelyan & Robinson, 2015).

## **Defining a Practice Framework**

Practice frameworks have been defined and used variably by different professions (e.g., health, education, research and evaluation) (Connolly, 2007; Stanley et al., 2021). Within healthcare, the term 'practice framework' or 'frameworks for practice' has increased in use, with contextual applications including (but not limited to) the provision of schematic diagrams and policies to improve practice and facilitate practice reform; the presentation of an organisation's values; and guidance on specific professional capabilities for development, or areas of knowledge needed to provide support. This study is informed by the definition used by Connolly (Connolly, 2007), and Stanley and colleagues (Stanley et al., 2021), where a 'practice framework' is summarised as a *"template not based on or informed by organisational imperatives (e.g., budgets or compliance edict) but designed through and informed by value-based practice research, and evidence; that offers a mapping out of what we do and why, offering a rationale for practice, while promoting a range of practice tools for assessments and intervention". In the context of this current study, 'value-based practice' refers to an approach to supporting clinical decision-making, which provides practical skills and tools for eliciting* 

individual values and negotiating these with respect to best available evidence (Petrova et al., 2006). More specifically, this proposed framework focuses on the health professional, and presents the 'practices' or tasks required to provide comprehensive SMS for cancer patients and cancer survivors experiencing acute and long-term CRF. Further, the proposed practice framework aims to reinforce what constitutes 'best' practice for CRF SMS.

# **Developing the Preliminary Practice Framework**

# Review of the Literature

To ensure framework comprehensiveness, minimise the number of rounds needed, and maximise the rate of survey completion, the initial statements used in Round 1 were derived from a qualitative synthesis of the systematic review of literature presented in Chapter 3 (Study 1), and the SMS Flinders Capability Framework (FCF) presented in Chapter 2. CRF management strategies from the National Comprehensive Cancer Network (NCCN) Fatigue Guidelines (Berger, 2019) were also incorporated. The development stages of the initial framework domains and components for consensus are shown in Figure 9.



# Initial Statement Development

# Figure 9: Preliminary Practice Framework Development Process

An iterative process was used to classify SMS components, strategies, and capabilities identified through research (Agbejule et al., 2022). Initially, a list of SMS strategies for CRF were compiled using information derived from Study 2 and the FCF Core Capabilities. CRF management strategies from the National Comprehensive Cancer Network's (NCCN) Fatigue Guidelines (Berger, 2019) were also incorporated.

A list of CRF SMS strategies were compiled from the information sources listed above, and then classified into core practices through an iterative process involving the creation, removal, and merging of categories to create mutually exclusive 'key practices' and 'practice components'. Key practices described the proposed activities health professionals are required to undertake to provide best practice SMS for CRF. Practice components describe the steps needed to complete a key practice. This process was reviewed by the research team who provided feedback, comment, and suggested changes. After addressing feedback, the practices were then tabulated by concept; analysed for similarities and inconsistencies between concept groups; before being categorised into domains, (e.g., Goal Setting, Action Planning  $\rightarrow$  Care Co-ordination). The final preliminary framework was sent to all research team members for feedback and approval. The research team comprised a clinical radiation therapist with expertise in cancer survivorship and self-management; a professor in cancer nursing with expertise in self-management and cancer related fatigue; an exercise scientist with expertise in cancer survivorship; and a health researcher in psychology. The resulting preliminary framework consisted of 47 practice items (14 Key Practices and 33 Practice Components) categorised under five domains: (1) Establishing Context and Defining the Problem; (2) Developing an Action Plan; (3) Improve Patient Knowledge; (4) Training Rehearsal (Strategy Building); and (5) Care Co-ordination and Maintenance (see Table 4).

## Table 4: Preliminary Practice Framework

## DOMAIN 1: ESTABLISHING CONTEXT AND DEFINING THE PROBLEM

**Key Practice 1.** Collect and use clinical and behavioural information to inform decision making about the patient's self-management of cancer-related fatigue

*Practice Component* 1a) Conduct a consultation and assessment with the patient, and if available, other key people in the patient's support network (e.g., carer's; family members) to collaboratively define key concerns, problem areas and priorities.

*Practice Component* 1c) Collect, record, and store key clinical, symptom and behavioural information.

Key Practice 2. Assess the patient's capacity for self-management.

*Practice Component* 2a) Identify factors within the patient, and the patient's physical and social environment, that they perceive may improve or worsen their cancer-related fatigue (e.g., more severe fatigue in the afternoon).

*Practice Component* 2b) Identify the patient's beliefs, attitudes, and knowledge about cancer-related fatigue, and identify their current coping strategies.

*Practice Component* 2c) Identify factors (i.e., cultural consideration, language literacy levels, availability of peer support network, pre-existing conditions) that may affect the patient's ability to participate in self-management activities.

## DOMAIN 2: DEVELOPING AN ACTION PLAN

Key Practice 3. Create a cancer-related fatigue management action plan in collaboration with the patient

*Practice Component* 3a) Reflect on the patient's capacity for self-management (including barriers), and the clinical and behavioural information gathered during pre-assessments, to refine the patient's priorities, needs, and goals concerning their cancer-related fatigue and general lifestyle.

*Practice Component* 3b) Plan a culturally and contextually relevant fatigue self-management care plan drawing on clinical and psychosocial information gathered during pre-assessments, the patient's capacity for self-management support and the health professional's applied knowledge of cancer-related fatigue management strategies.

Practice Component 3c) Incorporate the patient's support networks (family, friends, carers) into action planning.

## DOMAIN 3: IMPROVE PATIENT KNOWLEDGE

**Key Practice 4.** Provide information (visual, written, verbal) on cancer-related fatigue and common management strategies

*Practice Component* 4a) Inform the patient and their support network of the differences between cancer-related fatigue and 'normal' fatigue.

*Practice Component* 4b) Inform the patient and their support network of the causes, key risk factors, presenting characteristics, and the possible effects and interferences of cancer-related fatigue on daily living.

*Practice Component* 4c) Communicate tailored evidence-based information to the patient and their support network regarding the benefits of exercise and physical activity for managing cancer-related fatigue and have an awareness of the strength of such evidence.

*Practice Component* 4d) Communicate tailored evidence-based information to the patient and their support network on other management strategies for cancer-related fatigue (along with their risk and benefits) and have an awareness of the strength of such evidence.

**Key Practice 5.** Provide tailored evidence-based information on managing common psychosocial consequences of cancer and cancer-related fatigue

*Practice Component* 5a) Provide the patient and their support network with evidence-based information on how psychological and psychosocial factors (e.g., fear of cancer recurrence or progression, anxiety, depression, and stress) contribute to, and are exacerbated by cancer-related fatigue.

*Practice Component* 5b) Provide the patient with coping strategies for managing psychological and psychosocial factors

*Practice Component* 5c) Provide the patient with coping strategies for processing and communicating with others about cancer-related fatigue and cancer experiences.

Key Practice 6. Provide information about available social support

*Practice Component* 6a) Facilitate the involvement of the patient's support network (family members, friends, carers, significant others) in cancer-related fatigue self-management activities

*Practice Component* 6b) Provide the patient and their support network with education and information about how to seek further social support and inform the patient of relevant support services in their community.

Practice Component 6c) Facilitate the exchange of cancer and cancer-related fatigue experiences and/or

discussions between the patient and other cancer survivors.

DOMAIN 4: TRAINING REHEARSAL (STRATEGY BUILDING)

**Key Practice 7.** Provide the patient with problem solving and evidence-based solution-focused strategies to communicate with their systems of support (includes health professionals, non-health professionals, personal communities; and voluntary and community groups) about cancer-related fatigue.

*Practice Component* 7a) Provide the patient with the skills to self-advocate and communicate with health professionals, non-health professionals, personal networks, employees, and others about cancer-related fatigue.

Practice Component 7b) Provide coaching and counselling about navigating relationships and social support

**Key Practice 8.** Provide evidence-based coaching for lifestyle modifications that support living with cancer-related fatigue.

*Practice Component* 8a) Provide the patient and their support network with evidence-based tailored coaching and practical strategies for exercise (e.g., aerobic, resistance, yoga, balance) and physical activity (e.g., daily walking, morning stretches).

*Practice Component* 8b) Provide tailored coaching and practical strategies to the patient and their support network that supports the patient's everyday activities

*Practice Component* 8c) Refer the patient to relevant services and professionals for support when indicated (e.g., if you are unable to provide tailored coaching).

*Practice Component* 8d) Involve relevant services and professionals in the planning and decision making of the patient's care

**Key Practice 9.** Provide the patient with evidence-based problem-solving strategies for coping with the psychological effects or risk factors of cancer-related fatigue.

*Practice Component* 9a) Provide patients with strategies for coping with anxiety, fear of recurrence or progression, stress, depression, and managing interpersonal relationships.

Key Practice 10. Provide evidence-based general health promotion and education on lifestyle adaptation strategies.

*Practice Component* 10a) Provide relevant general lifestyle advice and counselling support to the patient and their support network.

DOMAIN 5: CARE CO-ORDINATION AND MAINTENANCE

**Key Practice 11.** Provide regular review of self-management goals and action plans in collaboration with the patient (and their social network).

*Practice Component* 11a) Reformulate previously established goals based off the patient's confidence, needs, and progress (i.e., goal attainment).

*Practice Component* 11b) Establish long term goals prior to the conclusion of support to facilitate continual patient self-care beyond the self-management support program.

*Practice Component* 11c) Provide scheduled reviews to monitor cancer-related fatigue (e.g., severity and lifestyle interference), associated symptoms (e.g., stress, depression), and review of progress with self-management behaviours (e.g., confidence, physical activity, activity management).

Key Practice 12. Provide practical support that facilitates ongoing self-management.

*Practice Component* 12a) Provide practical support to assist the self-monitoring of cancer-related fatigue symptom reporting (e.g., fatigue severity, energy levels, and other factors that impacted fatigue), and behaviour change (e.g., exercise, mediation habits, physical activity, and dietary habits).

Key Practice 13. Be able available on request to review the symptoms of cancer-related fatigue.

*Practice Component* 13a) Be available for symptom review, treatment modification or counselling if symptom thresholds are exceeded.

Key Practice 14. Be able available on request to review the symptoms of cancer-related fatigue

*Practice Component* 14a) Be available for symptom review, treatment modification and/ or counselling if symptom thresholds are exceeded

# **Participants**

Potential panel participants were identified through networks within the Multinational Association for Supportive Care in Cancer (MASCC) – including fatigue and self-management subgroups; cancer consumer networks including Cancer Voices Australia, Canadian Cancer Survivor Network, and the Guyana Cancer Foundation; and other relevant groups or individuals identified by the research team. Panel participants were included from the following groups: health practitioners and clinicians (e.g., general practitioners, primary care providers, cancer specialists, medical oncologists, radiation oncologists, surgeons, nurses, allied health); cancer survivorship researchers (academics), and cancer consumers (patients, cancer survivors, and caregivers).

While there is no standard sample size for a modified Delphi study panel, studies suggest a minimum of eight participants, with more members increasing the reliability and applicability of group judgment (Hallowell & Gambatese, 2010; Murphy et al., 1998). A minimum of 40 participants were aimed to be recruited to the modified Delphi process with adequate representation from each group (clinical/ academic and consumer). This sample size takes into consideration the 50% attrition rate commonly reported in literature (Jobst et al., 2013; Khodyakov et al., 2020), thus maximising the likelihood of a sufficient sample size in subsequent rounds of the study.

Criteria for inclusion into the modified Delphi study were as follows:

- Being a minimum of 18 years of age;
- Experience or involvement with cancer, or cancer survivorship;
- Being able to read and understand English to a proficient level;
- Access to a computer/phone/tablet and internet connection.

Potential panel participants received an invitation email seeking 'expressions of interest'. This invitation email comprised written study information; a study video detailing the background and aims of the study; and a link to an expression of interest form designed using the Qualtrics XM (Qualtrics, Provo, UT) survey software platform. This expression of interest form included background information and asked individuals to provide their email address, country of residence, and indicate whether they were a health professional, researcher, or consumer

(i.e., cancer patient, cancer survivor, caregiver). To reduce participant attrition, potential participants were informed of expected time commitments and were asked to confirm their interest to participate in all modified Delphi rounds (Khodyakov et al., 2020). Potential panel participants were encouraged to distribute the expression of interest form to peers and colleagues, approximating a snowball sampling technique. A mailing list of 92 individuals was created through the email addresses obtained through the completed expression of interest forms and was used to distribute the practice items and surveys for the modified Delphi rounds.

### Instruments

The online surveys for all modified Delphi study rounds were developed using the Qualtrics XM (Qualtrics, Provo, UT) survey software. The surveys were divided into three sections: (1) participant demographics, (2) participation instructions, and (3) the proposed core SMS components for managing CRF. In the first section, participants were asked to indicate their participant type (e.g., cancer survivor, caregiver, academic, health professional), sex, age-group, race, country of work, country of residence, occupation, years of experience in cancer survivorship care, cancer type, and years since cancer diagnosis.

In the 'participant instructions' section of the survey, participants were provided with task descriptions and detailed instructions on how to participate in the online process and use the online platform. A 'study instruction video' detailing the study aim, and instructions on how to complete the survey was also included. Section three of the survey listed a series of statements detailing the proposed core components and domains of SMS for CRF as determined through the systematic review (Study 2), a SMS capability framework for primary care providers, and the National Comprehensive Cancer Network guidelines for CRF. Participants were asked to rate their agreement with each statement using a five-point Likert scale (1 – strongly disagree; 2 – disagree; 3 – neither agree nor disagree; 4 – agree; 5 - strongly disagree). Open text boxes for additional written comments were also included. After the initial modified Delphi round, a second questionnaire was sent to panel participants in the subsequent round. The surveys were piloted tested by researchers and laypersons in the community to ensure coherence, and usability (e.g., clarity of participation instructions, statement wording, and rating criteria) (Khodyakov et al., 2020).

### **Round 1 Modified Delphi Study**

The preliminary framework was emailed to panel participants through an online survey, accompanied by a consent form, an explanation of study objectives, the consensus process, and instructions. Each panel participant was asked to rate their agreement - that each practice should in included in a final practice framework - on a 5-point Likert scale (from 'strongly disagree' to 'strongly agree'), indicating the degree that each practice statement should be considered best practice and included in the final practice framework. Panel participants were also given the option to provide comments and suggest additional items that may not have been included when initially developing practice items. Survey responses were then qualitatively and quantitatively analysed to appraise consensus among participants. As per recommendations in literature, panel participants were given two weeks to respond (Trevelyan & Robinson, 2015), with a 'reminder to complete' email sent to those who hadn't completed the survey after the first week..

## **Round 2 Modified Delphi Study**

Practice items that did not meet consensus in Round 1, or were newly created or modified based on Round 1 panel feedback, were sent back to panel participants for voting in Round 2. Individuals who did not complete Round 1 of the modified Delphi study were also invited to participate in Round 2 as evidence demonstrates this leads to better representation of originally invited panel participants opinions, reduces the chance of 'false consensus', and does not compromise the outcome of the Delphi process (Boel et al., 2021). Panel participants were provided with a document summarising the changes made from participant feedback in Round 1. Panel participants rated statements using the same methods as Round 1. Practice items that did not meet consensus were not included in the final framework. As with the first round, panel participants were given two weeks to respond, with a reminder email sent after the first week.

### Analysis

Survey responses were imported into the statistical software program Jamovi (Version 2.3) for analysis. Descriptive statistics such as frequency, mean, median, percentage, and range were used to describe demographic data. Statistics such as the median and IQR were derived and evaluated for each practice item (De Vet et al., 2005). Consensus on panel participant opinions was defined as IQR  $\leq 1 - a$  highly recommended rigorous and objective; widely accepted; and frequently used threshold for Delphi studies (Ahuja et al., 2018; De Vet et al., 2005; Ramos et al., 2016; Trevelyan & Robinson, 2015). As per recommendations in literature, median scores were used to summarise participant agreement with a statement (Ramos et al., 2016). A group median of 4–5 was considered to indicate agreement, and 1-2 disagreement. The percentage

of the panel participants responding to a given category was also recorded. Free-text responses were examined using basic thematic analysis methods (Braun & Clarke, 2006). Thematic analysis focused on concepts and categories used by participants in their free-text response. Analysis iteratively progressed from identifying specific ideas to conceptualising high-level explanations that constitute a patterned response, or 'theme'. These themes were then used to inform changes to the practice framework.

# **Ethical Considerations**

Panel participants were informed that their participation in the modified Delphi study was voluntary and were given the option to withdraw or refuse participation at any time. No identifiable data was collected. Low-risk ethical approval was obtained from the Flinders University Human Research Ethics Committee (Approval Ref #: HREC CIA4907-1).

# Results

# Demographics

# Round 1

Of the 92 individuals invited to participate, 52 completed the Round 1 survey (56.5% response rate). Demographic characteristics of participants in each round are presented in Table 5. Most participants identified as White or Caucasian (n=37/52; 71.2%); female (n=37/52; 71.2%); and indicated they were aged between the age groups 56-65 (n=15/52; 28.8%) or over 65 years (n=15/52; 28.8%). Twenty-four of 52 (46.2%) panel participants indicated their participant type as solely a cancer survivor/ patient, 11 (21.2%) indicated they were health professionals only, and nine panel participants as solely researchers/ academics (17.3%). The remaining participants selected more than one participant type with seven (13.5%) indicating they were health professionals and researchers, and one recording they were a health professional, cancer survivor/patient, and caregiver.

	Round 1 (n= 52) N,%	Round 2 (n=36) N,%
Sex		
Female	37 (71.2)	26 (72.2)
Male	16 (30.8)	10 (27.8)
Did not answer	1 (1.9)	
Age Group		
18-35 years	5 (9.6)	3 (8.3)
36-45 years	9 (17.3)	6 (16.7)
46-55 years	8(15.4)	5 (13.9)
56-65 years	15 (28.8)	11 (30.6)
Over 65 years	15 (28.8)	11 (30.6)

### **Table 5:** Demographics of panel participants

Race		
Asian	4 (7.7)	3 (8.3)
Multiracial or Biracial	1 (1.9)	
Black, African or African American	1 (1.9)	
White or Caucasian	37 (71.2)	23 (63.9)
Hispanic or Latino		1 (2.8)
A race/ethnicity not listed here	9 (17.3)	9 (25.0)
Participant Type	0 (17.0)	3 (20.0)
Cancer Survivor	25 (48.1)	16 (44.4)
Clinician	19 (36.5)	13 (36.1)
Researcher	16 (30.8)	11 (30.6)
Family/ Caregiver	1 (1.9)	3 (8.3)
Clinician/ Research Academic Data	(n= 28)	(n=21)
Occupation field	(= =•)	(== )
Medicine	14 (50.0)	8 (38.1)
Nursing	6 (21.4)	6 (28.6)
Allied Health	3 (10.7)	2 (9.5)
Psychology	2 (7.1)	
Primary Care	1 (3.6)	1 (4.8)
Social Work	1 (3.6)	1 (4.8)
Epidemiology	1 (3.6)	
Public Health		1 (4.8)
Researcher/ Advisor		1 (4.8)
Physical Medicine and Rehabilitation		1 (4.8)
Region of work		
Europe	14 (50.0)	9 (42.9)
North America	8 (28.6)	6 (28.6)
Oceania	4 (14.3)	5 (23.8)
Asia	2 (7.1)	1 (4.8)
Years of experience in cancer care	Ranged from 2 to 45	Ranged from 1 to 35
(research or clinical care)	years, with a median	years, with a median
	of 15 years.	of 14 years.
Cancer Survivor and Family/Caregiver	(n=25)	(n=17)
Data		
Primary Cancer site of Cancer Survivor		
Solid tumours	20 (80.0)	12 (70.6)
Haematological malignancies	5 (20.0)	4 (23.5)
Did not answer		1 (5.9)
Years since cancer diagnosis		
1-2 years	1 (4.0)	1 (5.9)
2-5 years	2 (8.0)	2 (11.8)
5-10 years	9 (36.0)	5 (29.4)
More than 10 years	13 (52.0)	8 (47.1)
Did not answer		1 (5.9)
Region of residence		
Oceania	16 (64.0)	11 (64.7)
North America	7 (28.0)	3 (17.6)
Europe	2 (8.0)	2 (11.8)
South America		1 (5.9)

Out of the 28 research academics and health professionals, most indicated their occupation was in the field of medicine (n=14/28; 50%), followed by nursing (n=6/28; 21.4%), with the
remainder in various allied health roles. Years of experience in research or clinical cancer care ranged from 2 to 45 years, with the median years of experience equalling 15 years. Region of residence varied across Europe (n=14/28; 50%), North America (n=8/28; 28.6%), Oceania (n=4/28, 14.3%), and Asia (n=2/28; 7.1%).

Out of the 25 cancer consumers (cancer patients/survivors and family member/care givers), most resided in Australia (n=16/25; 64%); indicated breast as the primary cancer site (n=9/25; 36%); and indicated it had been more than 10 years since first cancer diagnosis (n=13/25; 52%).

#### Round 2

Of the 92 invited individuals, 36 completed the Round 2 survey, resulting in a 39.1% response rate. Most participants identified as White or Caucasian (n=23/36; 63.8%), female (n=25/36; 69.4%); and indicated they were aged between the age groups 56-65 (n=11/36; 32.4%) or over 65 years (n=11/36; 32.4%). Twenty-one (n=21/36; 58.3%) participants indicated they were health professionals and/or research academics and 17 (n=17/36; 47%) were cancer patients/survivors or family/caregivers. Further demographic characteristics of Round 2 participants can be found in Table 5.

#### **Consensus Building**

#### Round 1

Quantitative consensus (IQR  $\leq$  1, and median of 4 to 5) on whether a key practice or practice component was to be included in the final framework was achieved for all practice items in Round 1. Themes identified from the panel participants free-text responses were also considered by the research team to ensure a co-creative process with participants, whereby the modification and inclusion of practice items were consistent with panel feedback.

In addition to feedback on grammar and formatting changes, priority areas for framework revisions from the panel's written feedback were: (1) the need to consider patient preferences for peer and familial support; (2) considering the needs of the patient's support network; (3) the importance of tailoring support information to different learning needs; (4) identifying risk factors for CRF; and (5) emphasis on referral to other health professionals for continual care. Identified themes, and subsequent responses/ amendments by the research team are presented in Appendix 8. As a result of panel feedback, 27 practice items were designated for inclusion in the final practice framework (Figure 10). Fourteen practice items were modified, and three new items were added and were included in Round 2 for rating. Five items were removed. Specific changes to practice items based on Round 1 panel feedback are displayed

in Appendix 9. Finally, in addition to the practice items, consensus was reached on the definition and components of a CRF SMS action plan Appendix 10.



Figure 10: Modified Delphi Study Process

## Round 2

All practice items submitted for feedback in Round 2 (n=17) reached consensus for inclusion into the final framework. Qualitative comments included suggestions for wording changes. The research team discussed the minor wording proposals from the panel and accepted or rejected them before finalising the framework. These changes did not go back to panel participants for endorsement through a third round, as they were not of sufficient scope for reappraisal (e.g., changes to grammar, spelling). Some panel participants also suggested that greater clarity and specificity was required for the practice components to be useful for implementation, and teaching and evaluating practice. To address this, a decision was made to re-incorporate the accompanying examples and contextual elements that were included in Round 1, back into

the main practice framework (and not just as supplemental material – see Appendix 8 for further detail). As a result of panel feedback from Round 2, all items were included in the final framework. Thus, the final practice framework consisted of 13 Key Practices and 31 Practice Components. Table 6 depicts the domains and key practices of the framework; the full practice framework is presented in Appendix 10.

Table 6: Domains and Key Practices of Self-management Support Practice Framework

Care of	coordination and Maintenance
1.	Collect and use clinical and behavioural information to inform decision making about the
	patient's self-management of cancer-related fatigue.
2.	Assess the patient's capacity for self-management
Devel	oping an Action Plan
3.	Create a cancer-related fatigue management action plan in collaboration with the patient that
	incorporates evidence-based coping strategies that are aligned with patient preferences.
Impro	ving Patient Knowledge
4.	Provide tailored evidence-based information on cancer-related fatigue and common
	management strategies in a diversity of formats to accommodate different learning styles
5.	Provide tailored evidence-based information on managing common psychologica
	consequences of cancer and cancer-related fatigue in a variety of formats to accommodate
	different learning styles
6.	Provide tailored evidence-based information about available social support in a variety of
	formats to accommodate different learning styles and check patient understanding
	ng Rehearsal (Strategy Building)
7.	Provide the patient with problem solving and evidence-based solution-focused strategies to
	communicate with their systems of support (includes health professionals, non-health
	professionals, personal communities; and voluntary and community groups) about cancer
	related fatigue
8.	Provide evidence-based coaching for lifestyle modifications that support living with cancer
	related fatigue
9.	Provide the patient with evidence-based problem-solving strategies for coping with the
4.0	psychological effects or risk factors of cancer-related fatigue
	. Provide evidence-based health promotion and education on lifestyle adaptation strategies
11	. Provide regular review of self-management activities, and self-management goals and action
	plans in collaboration with the patient, their support network (with the patient's consent), and
0.000	their health care team
	Co-ordination and Maintenance
	. Provide practical support that facilitates ongoing self-management
13	. Attend to requests to review the symptoms of cancer-related fatigue

# Discussion

This study is the first to develop a framework of core practices required by health professionals to deliver effective SMS to cancer survivors experiencing CRF. This framework had input from an international panel of cancer consumers, health professionals, and cancer researchers.

The modified Delphi study that was used established resounding consensus on the best clinician practices for facilitating CRF SMS at all phases of the cancer continuum.

For self-management to be effective, cancer survivors must be supported in managing their symptoms and conditions. Health professional guidance for fatigue management often lacks detail about effective SMS, contributing to the provision of inadequate and limited support (Agbejule et al., 2022; Hilarius et al., 2011; James et al., 2015; Jones et al., 2021) (e.g., information provision alone, normalisation of fatigue symptoms, advising individuals to simply rest and relax).

The practice framework presented in this study considers the complex nature of CRF management, by conceptualising self-management as an active process that requires an essential set of collaborative-and partnership building behaviours, skills, knowledge, and practices. Further, it presents the ideal practices needed to effectively facilitate the adoption of fatigue self-management behaviours. These include action planning, motivational interviewing, and assessment of self-management capacity to facilitate self-management through enhanced self-efficacy. This framework could be used by clinicians as a tool to guide their provision of CRF SMS. This framework may also allow clinicians to evaluate current practice, determine professional development needs, and support their understanding of the comprehensive nature of effective CRF SMS.

Although primarily directed to health professionals, this practice framework may have functionality across several settings (see Table 7). Clinical leaders and educators could use the framework to build awareness and knowledge among their clinical teams. Researchers could use the framework to synthesise evidence on CRF self-management, a deficiency to address the shortcomings described in Chapter 1 whereby several reviews identified that variability in fatigue self-management definitions, strategies, and intervention reporting impedes study synthesis and increases difficulty in determining the distinction of outcomes gained as a result of self-management practices. Consumers could refer to the framework to advocate for their care, and develop their own understanding on the various roles of SMS for CRF.

#### Table 7: Use of the Practice Framework

The following recommendations are designed to help individuals and organisations make the best possible use of the Practice Framework.

#### For the individual health professional

- Use the Practice Framework as a tool:
  - o to guide the provision of self-management support for cancer-related fatigue
  - o for determining your professional development needs
  - o for evaluating current practice when providing support to those managing cancer-related fatigue
  - Use the Practice Framework and associated learning resources to undertake self-directed learning.
- Refer colleagues new to working with people affected by cancer (i.e., cancer survivors, cancer patients) and/or cancer-related fatigue to the framework.
- Use the Practice Framework to develop an understanding about:
  - the extent of cancer-related fatigue impact on those affected by cancer and the importance of its management
  - the various roles of different health professionals in the delivery of self-management support for cancer-related fatigue

#### For the clinical leader/ educator

- Use the Practice Framework as a tool to:
  - develop clinician awareness and knowledge of evidence-based cancer-related management and assessment strategies
  - provide training/in-service programs to improve ability to undertake practices (how to use certain questionnaires, practice developing an action plan, etc.)
  - advocate for system-level changes to provide resources (time, space, and human) to deliver optimal support for cancer-related fatigue management

#### For the cancer-related fatigue self-management intervention/program developer

• Use the Practice Framework to aid development of a cancer-related fatigue self-management intervention/program (determining the specific components that are needed).

#### For the researcher

- Use the Practice Framework as a tool to:
  - o describe self-management support interventions for cancer-related fatigue
  - o synthesize evidence on cancer-related fatigue self-management
- For the consumer experiencing cancer-related fatigue (and their family/carer's)
  - Use the Practice Framework as a tool to:
    - develop understanding of the various roles of different health professionals in the delivery of selfmanagement support for cancer-related fatigue
    - Advocate for improved delivery of cancer-related fatigue management support.
    - advise your health care team(s) about the existence of the practice framework and teaching and learning resources in efforts to improve your care.

Although panel participants acknowledged framework components as best practice, comments identified a need for further clarity around healthcare professional responsibility (i.e., who should do what). CRF is multifactorial (Bower, 2014; Weis & Horneber, 2014), meaning that there are diverse factors that can contribute to, or cause it (e.g., cancer type, treatment type, anaemia, nutrition factors, psychological factors, etc.). It is therefore expected that the provision of CRF SMS will require a multidisciplinary approach. If a multidisciplinary approach is adopted, it will not be necessary for all healthcare professionals to be proficient and have the commitment to deliver all practice components, especially when these fall outside of a professional's expertise or scope of practice. However, this practice framework

does allow for healthcare professionals to identify areas of care provision that may be achievable within their clinical care domain, and where additional training or collaboration may be encouraged or required.

The execution of the key practices and practice components specified in the framework may require health professionals to apply, adapt, and integrate new and existing evidence-based knowledge or seek professional development opportunities. This framework does not provide, present, or describe the capabilities or competencies required by health professionals, as these are already described in clinical practice guidelines (Berger, 2019; Fabi et al., 2020; Howell. et al., 2013). Rather, the practice framework outlines the support tasks that health professionals and health care teams should undertake when supporting people affected by cancer to self-manage their fatigue.

The consumers involved in this study emphasised the importance of health professionals not merely providing information on self-managing fatigue, but delivering information in a way that promotes the understanding and knowledge of the consumer. Although not explicitly listed in each practice item, it is stressed that the execution of practices outlined in the framework should be underpinned by the presence of effective, person-centred, health professional communication which involves the ability to establish and develop mutual understanding, rapport, trust, respect, and cooperation with people affected by cancer using clear and plain language. This includes making appropriate adjustments (e.g., use of appropriate language and detail, use of appropriate verbal and non-verbal cues, confirming that the other person has understood) to meet the communication and information needs of patients and their support network (e.g., caregivers, family, friends) and providing opportunities for the patient and their support network to demonstrate their understanding.

#### **Future Work**

This study has identified the requisite practices needed to effectively deliver fatigue SMS. Although some feedback to enhance framework usability and implementation was received and incorporated, future work could involve further consultation with key stakeholders. This consultation could be used to enhance understanding of stakeholders' perspectives about the acceptability and relevance of the framework to specific clinical, educational, and cultural contexts, and among underserved or high-risk groups. Future work could also identify different stakeholders' needs in supporting the implementation of the framework in their local setting. This includes fine-tuning the language and presentation of the framework for different contexts (e.g., 'cheat sheets', communication tools, role play scenarios for training, flow diagrams, etc.) and determining educational and training requirements. Stakeholder consultation could also be used to further define the roles of different professional disciplines in providing SMS for

CRF. Such developments would fine-tune the framework to provide clinical and implementation guidance that encourages clear professional judgment and explicit decision making.

### **Strengths and Limitations**

Strengths of this study include its online anonymous nature, which allowed for unrestricted expression of panel opinions. This helped reduce the influence of dominant personalities and the effect of panellists' status on results (Ramos et al., 2016). However, the online forum limited the opportunity for robust discussion. Another strength of this work is that the modified Delphi study comprised representation from a diverse international panel of consumers, health professionals, and cancer researchers from varying continents, professional fields, and clinical settings. Although the resulting practice framework incorporated diverse international perspectives, the panel was not representative of participants from every country/region, culture, setting, or scope of practice. Cultural influences on health, fatigue, compliance, and attitudes towards care will need to be considered when adapting the framework to different contexts (Zhang et al., 2020). Limiting panel eligibility criteria to individuals proficient in English could have resulted in potential candidates and viewpoints being missed.

# Conclusion

This modified Delphi study presents a framework for health professionals that outlines the essential support practices needed to facilitate the uptake of CRF management strategies. Future work is needed to assess the clinical utility and implementation (including evaluation of such implementation) of the practice framework. The provision of comprehensive SMS by health care teams is key for the uptake and integration of evidenced-based fatigue management strategies into clinical practice and improving the outcomes of patients and cancer survivors. Future work is needed to investigate the practical implementation of the established best practices. The next chapter delves into the application of SMS for CRF in real clinical scenarios.

# 5.0 CONVERSATION ANALYSIS STUDY (STUDY 3)

#### **Chapter Introduction**

Enabling cancer survivors to adopt and sustain CRF management behaviours requires a collaborative partnership between cancer survivors, health care teams, and local and personal communities of support (community organisations, families, friends). The presence of a strong collaborative partnership or therapeutic alliance provides vital and consistent contributions to the outcomes (e.g., improved uptake of health management behaviours, self-management self-efficacy) of supportive care practices (Fitzpatrick et al., 2006). The Flinders SMS Capability Framework, introduced in Chapter 2, affirms that health professionals require an understanding of the impact of their interactions on cancer survivor choices and behaviours, and emphasises that to facilitate successful self-management, health professionals – with relevant expertise and resources – must foster productive interactions with cancer survivors (Lawn & Battersby, 2009). Difficulties in communication between cancer survivors and health professionals represent a key barrier to the implementation of CRF management strategies to practice, and is a large contributor to cancer survivors feeling dissatisfied with the CRF SMS they currently receive from clinicians (Jones et al., 2021; Jones et al., 2020).

The systematic review on SMS interventions for CRF presented in Chapter 3 (Study 1) identified a lack of research to understand exactly how SMS for fatigue has been and should be delivered and emphasised that more research is needed to examine ways communication can be effective in practice. In Chapter 4, a modified Delphi study (Study 2) was used to identify a framework of the requisite practices needed for health professionals to effectively provide CRF SMS to cancer survivors. Although the study participants acknowledged framework components as best practice, some clinician participants expressed hesitancy around the feasibility of undertaking the practices, citing interactional and logistical difficulties (e.g., issues with time availability). Additionally, some clinicians questioned how these support discussions (e.g., identifying barriers and facilitators to client self-management, providing coping strategies, goal setting, etc.), and creation of goals would occur in real-life clinical practice. The successful adoption of self-management CRF behaviours by cancer survivors (e.g., dietary changes, increased physical activity, etc.) is also greatly dependent on the quality of clinician-survivor interaction and the effective provision of efficacious SMS strategies. Further investigation is needed to help understand how the practice framework can be implemented while considering specific challenges that may arise during interactions. To understand the processes that may influence the quality and effectiveness of these critical interactions, it is important to identify, in detail, the communitive practices used by clinicians and cancer survivors during real-life encounters at the micro-level.

79

The purpose of this study is to investigate at the interactional level the communication practices employed by cancer nurse counsellors when providing support to cancer survivors in managing their CRF and the impact of these practices within consultations. Insights derived from this study complement the formation of the *Cancer-related Fatigue Self-management Support Practice Framework* by examining the application of best practices in routine clinical care at an interactional level.

This chapter describes the design, findings, discussion, and implications of a conversation analytic study to achieve the aims and objectives stated in Chapter 1: To gain an understanding of clinical interactions as part of SMS practices to manage CRF in cancer survivors. This study was conducted to gather insight on the following questions:

- 1. What SMS communication practices are used by cancer nurses during interventional consultations with cancer survivors experiencing CRF?
- 2. What are the interactional consequences of these practices?

This chapter describes the methodology used in the study and the research design; the participants of the study; the procedure used; how the data is analysed; and the ethical considerations of the research.

# **Methodology**

Conversation analysis (CA) is underpinned by a distinctive qualitative methodology that facilitates analysis of the constituent elements of interaction in fine-grain detail, focussing on how humans give meaning to each other in conversation through shared interactional methods (White, 2019). Established primarily from the work of the psychologists Sacks, Schegloff, and Jefferson, and rooted in sociology and ethnomethodology, CA seeks to examine and explain the interactional social order and organised reasoning procedures which inform the production of everyday naturally occurring social interaction (Bloor & Wood, 2006; Hutchby & Wooffitt, 2008).

Communication is an integral component of the delivery of safe and effective healthcare, with most healthcare provision performed through social interaction (e.g., clinical handover; interdisciplinary and multidisciplinary collaboration; and clinician-patient consultations). Further, the success of healthcare delivery largely depends on the quality of these interactions, particularly between health professionals, medical staff, and patients (Drew et al., 2001). Conversation analysis is a well-established approach for exploring the details of real-world healthcare communication (Barnes, 2019; Drew et al., 2001; Heritage & Maynard, 2006; Pilnick et al., 2009). Additionally, CA has advantages over other qualitative research, such as approaches that rely on interview data, due to its focus on unaltered naturalistic encounters

where the role and impact of the researcher is minimised (Potter & Hepburn, 2012). Despite the suitability of the CA methodology for examining communication in supportive care, there have been no CA studies on the SMS interactions of health professionals. Moreover, existing CA research has almost exclusively focused on doctor-patient interactions within primary care settings (Barnes, 2005; Drew et al., 2001; Gill & Roberts, 2013; Pilnick et al., 2009). As such, this study delivers novel insights for oncology, CRF, nurse-cancer survivor interactions, and SMS interventions.

CA consists of three fundamental theoretical assumptions: (1) talk is a form of action; (2) action is structurally organised; and (3) talk creates and maintains inter-subjectivity (Peräkylä, 2004; Willig & Stainton Rogers, 2017). The first theoretical assumption – talk is a form of action – focuses on what individuals 'do' with talk rather than on what they say, and places emphasis on the fundamental structure of talk-in-interaction such as turn-taking and the organisation of actions into sequences (Willig & Stainton Rogers, 2017). For example, a CA study by Toerien and colleagues (2018) found that neurologists and their cancer patients only agreed that a choice in cancer treatment had been offered in instances where the neurologist had distinctively listed options during consultation (Toerien et al., 2018). This finding was insightful as it identified why patients perceived a lack of choice in their treatment despite neurologists reporting they had provided different treatment options and that there was genuine scope for the patient to choose between them. Further, this example demonstrates how the action and organisation of talk-in-interaction is just as significant (if not more significant) than the content of the conversation itself (Sacks et al., 1974).

The second theoretical assumption – that interaction is structurally organised – delves into specific rules and technical practices that structure talk into sequences of action. The sequential organisation of talk can be categorised into three characteristics. First, when a current speaker provides a particular type of utterance (e.g., treatment options listing), this prompts particular types of responses relevant to next actions for another speaker (e.g., selection from the listed options). Second, speakers are informed by the context of a preceding interaction (most commonly, immediately preceding talk). Third, speakers demonstrate their understanding and interpretation of prior interaction through the production of next actions.

An example of sequential organisation is again observed in Toerien and colleagues' (2018) CA study. The authors analysed the immediate actions of neurology patients following treatment choice conversation, and found that during option listing, if neurologists did not explicitly invite patients to announce their views, the patient tended to trust the first listed option as the best option – when that was not necessarily what the physician was trying to convey (Toerien et al., 2018). This example also links to the final theoretical assumption – talk creates

and maintains intersubjectivity – which posits that conversation depends on the displayed understandings of prior talk. Moreover, due to this, "no detail of an interaction can be considered immaterial, unimportant or inconsequential within the context of interaction" (Peräkylä, 1997; pg. 89). This theoretical assumption underpins the analysis of data, which will be considered in the analysis section below.

# **Research Design**

## Telehealth Cancer-related Fatigue Self-management Support Clinic

Data for the CA study comprise audio or audio-visual recordings of nurse-cancer survivor consultations conducted between 2021-2022 in a Telehealth Cancer-related fatigue Selfmanagement support clinic (hereinafter referred to as the T-CRF SMS Clinic) in metropolitan Australia (Brisbane, Queensland). These data were collected as part of the Telehealth Cancer-Related Fatigue pilot randomised controlled trial (RCT) of which the full intervention protocol has been published elsewhere (Ladwa et al., 2022). Briefly, the T-CRF SMS Clinic is a 24week SMS program, informed by the Flinders SMS Capability Framework, that consists of three nurse-led telehealth clinic sessions between cancer survivors experiencing moderateto-severe fatigue (as determined by the Brief Fatigue Inventory instrument), and trained cancer nurse counsellors. During clinic sessions, nurse counsellors: 1) provide education on fatigue management addressing aspects such as physical activity, symptom-specific and general coping mechanisms; 2) collaboratively create a fatigue management plan with up to three goals; and 3) facilitate referrals to exercise specialists and other relevant support services. During the consultations nurse-counsellor interventionists use different behavioural strategies (i.e., goal setting, coaching, motivational interviewing, and cognitive behaviour therapies) to facilitate cancer survivor behaviour change (i.e., increasing physical activity) with the aim of improving cancer survivor CRF (see Figure 11).



Figure 11: The Telehealth Cancer-Related Fatigue (T-CRF) Nurse Consultation Model

## Context

The nurse counsellors delivering the T-CRF SMS Clinic, were from the consumer advocacy community group Cancer Council Queensland (CCQ). CCQ is a state-member of Cancer Council Australia, a national, non-profit organisation which aims to promote cancer-control and prevention policies and to improve the quality of life of people diagnosed with cancer in Australia (Cancer Council Australia, 2023). CCQ nurse counsellors have extensive experience caring for cancer survivors and have been trained in counselling and cognitive behaviour techniques. In addition to pre-existing training, prior to commencing the intervention nurse counsellors received written material on how to deliver the intervention; material on effective communication, motivational interviewing, and behavioural cognitive techniques; and attended a full-day workshop incorporating role-play activities (Ladwa et al., 2022).

### **Participants**

### Eligibility Criteria

Cancer survivors were included in T-CRF study if they met the following criteria:

- were at least 6-weeks post-completion of primary cancer treatment, or completed at least 3-months of maintenance treatment;
- were receiving care at Princess Alexandra Hospital outpatient clinics;
- were aged  $\geq$  18 years;
- had a definitive diagnosis of solid tumour or haematological cancer;
- had a score  $\geq$  4 on the Brief Fatigue Inventory;
- had a Eastern Cooperative Oncology Group performance status of ≤2;
- were not currently receiving specialist palliative care;
- had access to a telephone.

Cancer survivors were excluded from study participation if they displayed a presence of mental, cognitive, or physical conditions that would limit their ability to participate in the intervention; had a declared prognosis of less than six months survival expectancy at the discretion of the treating clinician; or for any other reason that they were deemed unsuitable to participate at the discretion of their treating clinician.

#### Procedure

#### Recruiting and Consenting Participants

Cancer survivors for the T-CRF pilot trial were recruited through the Princess Alexandra Hospital's (PAH) outpatient clinics and therapy units of the Division of Cancer Services and Surgical Services, Brisbane, Australia. Eligible cancer survivors were asked to provide written informed consent, which included the following declaration:

I understand that my information collected as part of this study may be used for secondary analysis for another research purpose.

Information on the audio and video recording of nurse-led clinic sessions was also provided in the written information sheet that accompanied the informed consent form. Prior to the beginning of each nurse-led clinic, intervention nurse counsellors sought additional verbal consent from participants to record sessions using the Microsoft Teams recording function.

#### Data Collection

Audio and video recordings of telehealth nurse-led clinic sessions were performed using the Microsoft Teams recording function. As part of standard practice, following participant consent, CCQ nurse counsellors used Microsoft Teams to conduct and record the clinic sessions with cancer survivors. Audio and video data were transcribed verbatim using a professional service that stored the data securely and did not retain a copy of the recording beyond the period required for transcription. Specific cases or verbatim transcripts were then transcribed and annotated using the conversation analytic transcription system developed by Jefferson (Hepburn, 2012; Jefferson, 2004) (see Figure 12). Data was collected and analysed for intervention consultations that occurred between January 2021 and September 2022. At the time of analysis, 41 distinct telehealth consultations had occurred. All 41 consultations were included in the analysis. A description of the sample can be found in the Data section.

#### Analysis

Video and audio data were analysed using the following standard conversation analytic approach (Schegloff, 1996, 1999; Sidnell, 2012; White, 2019). Unlike many other qualitative research methods, in conversation analysis, observation is commonly used as a basis of theorising (Sacks, 1984; Sidnell, 2012). Accordingly, audio and video recordings were first examined to identify recurrent ways in which talk about self-management of CRF appear to occur. This process involved the general observation of data without seeking specific technical details or preformulated phenomena (Sacks, 1984; White, 2019). Observations such as general changes in social interaction as well as other conduct such as gestures, eye contact, gaze, and body organisation were noted. Instances of phenomena of interest were then transcribed using Jefferson's transcription system described in Figure 12, to facilitate detailed analysis.

Transcription Notation					
Symbol	Correspondence to features of talk				
(.)	A dot in a bracket indicates a pause of less that tenths of a second				
Word (0.7) Word	A number within parentheses refers to silence, which is measured to the nearest tenth of a second and can occur either as a pause within a current speakers turn or a gap between two speaker's turns				
Wh <u>at</u>	Underlining indicates stress or emphasis in speech				
Wo:::rd	Colons indicates the stretching of the immediately preceding sound with multiple colons representing prolonged stretching				
W <u>o:::</u> :rd	Underlining followed by one or more colons indicates a shift in pitch during the pronunciation of a sound, with rising pitch on the underlined component followed by falling pitch on the colon component that is not underlined.				
Wo[rd] [Wo]rd	Square brackets mark speaker overlap, with left square bracket indicating overlap onset and right square brackets indicating overlap offset.				
Word-	A dash following a word indicates a cut-off sound at the end of the talk				
=	Equals sign indicates continuous talk between speakers				
↑Word↑	An utterance encased with upwards hours indicates that the talk is produced at a higher pitch than the surrounding talk.				
↓Word↓	An utterance encased with downward arrows indicates that the talk is produced at a lower pitch then the surrounding talk.				
°Word°	Words encased in degree symbols indicate utterances produced at a lower volume than surrounding talk.				
Word.	A period indicates falling intonation at the end of a unit of talk				
Word,	A comma indicates slightly rising intonation				
Word?	A question mark indicates rising intonation				
>Word<	Words encased with greater-than followed by less than symbols indicate talk produced at a faster pace than surrounding talk.				

Figure 12: Jeffersonian's Conversation Analysis Transcription Convention

The Jeffersonian transcribed segments were then analysed in fine-grain detail to systematically idenitfy and understand the organisation of phenomena. This process can be undertaken in several ways, but generally involves the in-depth analysis of each identified segment of data to understand the organisation of conversational structure, (such as the way in which turns at talk are organised into sequences of action) (ten Have, 2007).

During this analytic process, particular features of interest become apparent and lead to the collection of instances displaying the same phenomenon. This process of building collections allows the researcher to identify the position and composition of the focal phenomenon. Further, during this approach, boundary or deviant cases (cases that partially or completely go against the trend or pattern that is typical in the collection) should not be ignored as they

may prove the 'rule' or systematic practice observed and may provide further context for understanding that practice (Sidnell, 2012; White, 2019). Deviant case analysis may also warrant revision or refinement of analysis (Maynard & Clayman, 2003). The ultimate goal of collection-based analysis, including deviant case analysis, is to make generalisations across cases, but not at the expense of the applicability of the analysis to individual cases (Stivers & Sidnell, 2012).

The analysis reported below will provide key insight into how SMS interactions occur between health professionals and cancer survivors in clinical practice. More specifically, analysis will seek to identify the methods and communication strategies used by health professionals and cancer survivors to initiate (or avoid) discussions about managing CRF and will explore the interactional impact of these methods on cancer survivor uptake of CRF management strategies. Further, the CA study will provide insight into the state of health professional SMS practices for individuals experiencing CRF and examine whether current practices align with best practice recommendations.

#### Ethics and Limitations

Low risk negligible ethics was sought for all recording and analysis as part of the ethics submission for T-CRF RCT, which was approved by the Human Research Ethics Committee (HREC) of Metro South Health (MSH: HREC/2020/QMS/63495). Administrative approval was also provided by the Queensland University of Technology (QUT) HREC (Approval No. 2000000546). The study adheres to the principles of the National Health and Medical Research Statement on Research Involving Human Subjects. All names and identifiers included in transcript fragments are pseudonyms.

#### Data

Forty-one distinct audio-recorded and audio-visual recordings of telehealth consultations between three different nurse counsellors, and 23 different cancer survivors were examined. Clinic consultation times ranged from 46 to 75 minutes, with an average duration of 52.5 minutes. The age range of cancer survivors in the sample was between 23 and 76 years, with an average age of 53 years. The sample consisted of 14 females and 9 males, all diagnosed within the last 8 years (between 2015 and 2021). Cancer diagnoses comprised of 4 bowel cancer, 9 breast cancer, 4 lung cancer, and a mix of others, including melanoma, testicular cancer, pancreatic cancer, renal cancer, leukaemia, and prostate cancer. Of the 23 cancer survivors, 14 reported moderate fatigue at baseline, while the remaining 9 reported severe fatigue as per the BFI tool. Fragment identification codes are as follows: NC = nurse counsellor, CS=cancer survivor, 0:00 = minute:seconds)].

# Analysis

The primary agenda of the nurse-led consultations analysed in this study was to collaboratively establish coping strategies that support the cancer survivor to manage their CRF; in other words, to provide SMS. To facilitate this process, nurse counsellors typically undertake a selection of activities. First, the nurse counsellor begins a process of information gathering to make an assessment of the cancer survivor's fatigue. This can include assessing the cancer survivor's clinical status, determining how fatigue impacts the cancer survivor's day-to-day living, establishing if there is a pattern of fatigue, and identifying potential barriers or enablers to possible fatigue support strategies (e.g., inability to lift items, sleep disturbances). Next, a fatigue management plan is established in collaboration with the cancer survivor, in which behaviour change goals to mitigate CRF are created (e.g., walking continuously for 20 minutes each day of the week). During creation of the fatigue management plan, the cancer survivor is provided with information about CRF as well as coping strategies and a referral to an exercise specialist (i.e., physiotherapist, exercise physiologist) if indicated. Throughout this process the nurse counsellor may also use support strategies such as motivational interviewing, and support practices based on cognitive behavioural techniques. In subsequent consultations the goals developed in the fatigue management plan are reviewed and adjusted according to cancer survivor feedback and needs.

The SMS provided by the nurse counsellor is a complex multi-component process. Although a range of important topics relating to the psychosocial, physiological, and physical impact of the cancer survivor's cancer are discussed during the SMS consultation (e.g., clinical symptoms; psychological and social wellbeing; etc.), the primary agenda – providing support to help cancer survivors manage their CRF – remains focal throughout the intervention. This conversation analysis study demonstrates that a recurrent way this focus is achieved and maintained is through the use of conversational practices such as formulations and establishing the agenda from the outset of the consultation. The use of these conversational practices by the nurse counsellors involved in this study provides scope for holistic consideration of the cancer survivor's circumstances, while also maintaining a core focus on CRF management.

# **Findings**

This section presents instances of nurse counsellors 'establishing the agenda from the outset' and creating formulations during consultations. The interactional consequences of instances where such techniques are not employed are also examined. The analysis finds that structuring SMS consultations using these practices promotes intersubjectivity (i.e., shared understanding) between cancer survivors and nurse counsellors, and sustains a primary focus on the agenda of managing CRF.

# Establishing a focus on fatigue self-management support from the outset of the consultation

SMS consultations differ to the usual types of medical encounters that cancer survivors may have previously experienced. For example, in a typical encounter with a general practitioner (GP), a patient generally states their medical problem, or the reason for their visit, and may provide additional information if prompted by their GP, who then delivers a clinical judgement and provides a treatment recommendation if indicated (Robinson, 2003). In contrast to general practice, most individuals do not have experience with SMS counselling sessions and may not know what to expect, what to do, and how to act in a consultation. This analysis identified that establishing the agenda of consultation from the outset of a session provides an initial opportunity for clinicians to focus the conversation on SMS for CRF and manage the expectations of the cancer survivor.

An example of a nurse counsellor establishing the primary agenda at the outset of the consultation can be seen in the fragment below, which begins five seconds into the recording of the consultation. In Fragment 1, the nurse counsellor begins the consultation by announcing her name and role (lines 2-3). On lines 9 to 13, she establishes what her expectations are from the cancer survivor, (*"Importantly, I'm interested to hear from you today and to find out more about the experience of fatigue…"*). She then sets the agenda of the consultation through the arrangement proposed on lines 14-26.

#### Fragment 1 [NC1CS1/0:05]

01 02 03 04 05 06	NUR:	So, (.) lovely to have that little chat before we started the recording and (.) so just to um >introduce myself,< my name is Sarah I'm one of our nurse counsellorshhh (0.5) at the Cancer Centrehhh and uh >I understand< you've been refer <u>r:ed</u> uh to uh:s, because you've been (.) experiencing hh a lot of fatiguehh(.) an:d I wanna thank you for completing that questionnaire?=
07	CLI:	Yep. ((Single head nod downward))
08	NUR:	=I've hadda (.) read of that this morning, (0.3) to get >a bit of a
09		sense<[ of ] what's been happening. Importantly, I'm I'm interested
10	CLI:	[(Cough)] ((short repeated nods, eye contact kept with NUR))
11	NUR:	to hear from you toda::y and to find out mo:re (0.3) about the experience
12		of fati:gue.=
13	CLI:	((short repeated nods, eye contact kept with NUR))
14	NUR:	=.hhh uh so I <u>won</u> dered to get us started, if >would it be oka::y< if
15		(0.3) uh:: I start with some questions? [an::]d to open up our discussion?
16	CLI:	[ yep] ((Single head movement
17		upwards then downwards))
18	NUR:	an::d feel free at any point to add in anythi[ng that you think °might° be^
19		of relevance,=
20	CLI:	[((short repeated nods, eye contact kept with NUR))
21	CLI:	Yeh-
22	NUR:	=or that you want to share?
23	CLI:	((short repeated nods, eye contact kept with NUR)) Yep
24	NUR:	an::d then what we can do is talk aboww:t (.) a plan, (.) of how we might
25		be able to provi:de some support=
26	NUR:	=[to help in managing the fatigue hhh=I thni-
27	CLI:	[((short repeated nods, eye contact kept with NUR)) $^{\circ}$ Okeh $^{\circ}$

28 NUR: Does that aw:ll sound okay: for you? ((head nodding while gazing at cancer survivor with head tilted sideways)) 30 NUR: [yeh- that (0.2)chat?] 31 CLI: [yeah, that's good. ]((double head movement downward, high amplitude head movement)) 33 CLI: Yep= 34 NUR: =Yep (.) great

Through this proposed arrangement the nurse counsellor orients the cancer survivor to the structure of the consultation, by establishing her role (i.e., to conduct an initial assessment of the cancer survivor's fatigue by asking questions; to assist the cancer survivor to come up with a plan to help manage his fatigue), and the role and responsibility of the cancer survivor (i.e., to answer questions about his fatigue, provide information that he thinks is relevant to the creation of his fatigue management plan, and to co-develop a plan to manage his fatigue) (Patterson, 1985). Importantly, in this early phase of the consultation, the nurse counsellor explicitly informs the cancer survivor of the agenda - to provide SMS for CRF ("... then what we can do is talk about a plan, of how we might be able to provide some support to help in managing the fatigue", lines 24-26). After providing her proposed arrangement, on line 28 the nurse counsellor produces a pursuit of response ("Does that all sound okay for you?"). described by conversation analysts as response solicitation, whereby the arrangement (described on lines 14-26) is contingent upon the recipient's (in this case the cancer survivor's) acceptance (Jefferson, 1981). The cancer survivor's verbal confirmation ("yeah that's good", line 31) and head nod (line 32) in response to the nurse counsellor's response solicitation, accepts the nurse counsellor's proposed plan for the consultation and shows that the cancer survivor and the nurse counsellor have a shared understanding of the tasks and goals of the session; whereby the nurse counsellor is trying to establish a course of activity, and the cancer survivor is aligning with what the nurse counsellor is seeking to undertake.

The alignment and collaborative partnership between the cancer survivor and nurse counsellor is further evidenced by the verbal and embodied conduct used by the cancer survivor throughout the fragment. Specifically, in Fragment 1, the cancer survivor's use of verbal (i.e., yep, yeah) and physical (i.e., head movements or 'nods') response tokens throughout the nurse counsellor's introductory sequence implies they are an attentive recipient of the nurse counsellor's talk, and subsequently possess an understanding on the focus of the session (Kidwell, 1997). Gardner (2001) describes response tokens as actions and utterances that provide feedback to the speaker on the listener's engagement with, and understanding of the conversation (Gardner, 2001). For example, the cancer survivor's head nod and verbal response on line 16 to the nurse counsellor's previous yes/no question (*"is it okay if I start with some questions?"*) confirms the cancer survivor is happy start the consultation with questions. Similarly, the cancer survivor's short, repeated nods while the nurse counsellor is speaking on

lines 10, 13, 23, and 23, (in addition to his use of continuers such as 'yep' on line 23) indicate that he claims to understand what the nurse counsellor is saying. In this fragment, the cancer survivor's verbal and embodied conduct along with the nurse counsellors' utterances (e.g., line 34) claim alignment about the focus for the ensuing session; fatigue and fatigue self-management. As the next fragment shows, although the detail of the opening sequences of the consultation may differ, what is common is that fatigue SMS is presented as the agenda from the outset of the session.

### Fragment 2 [NC2CS2/ 0:15]

In contrast to Fragment 1, which was conducted via videoconferencing, Fragment 2 provides an instance of establishing a focus on fatigue from the outset of a consultation conducted only via the audio channel of the Microsoft Teams videoconferencing software, demonstrating that alignment can also be produced without relying on embodied conduct. Directly prior to the beginning of this fragment, the nurse counsellor has introduced her name and role, and has asked the cancer survivor if she consents to being recorded.

```
01 NUR: Um hh (.) so what we'll do to<u>day</u> is:s (.) we'll talk a little bit about
          how your fatigue is (.) impacting you? Um and the ways [in which]=
02
03
    CLI:
                                                                  [ mm hmm ]
    NUR: =its impacting you, and then at the end of the session we'll come up
04
          with some hhh (.) uh:m some goals (0.2) of how to [minimise]=
05
06
    CLI:
                                                             [ okay
    NUR: =that fatigue or start working towards uhhh (1.0) you know improving
07
80
          that fatigue hhhh so Maryan-
09 CLI: Yep
```

As with Fragment 1, after introducing herself (data not shown), the nurse counsellor provides a proposed arrangement that incorporates the agenda ("*so what we'll do today, we'll talk a little bit about how your fatigue is impacting you*", lines 1-2... "*we'll come up with some goals of how to minimise that fatigue*..." lines 4-7). By setting the agenda from the outset, the nurse counsellor establishes the joint role of the cancer survivor and themself in a collaborative partnership to develop a plan to help manage the cancer survivor's CRF. Like Fragment 1, the shared understanding of the focus of the consultation can be indicated through the cancer survivor's use of response tokens such as "mmhm" (line 3), "okay" (line 6), and "yep" (line 9) throughout this opening sequence. These tokens display the cancer survivor's attentive participation (Gardner, 2001) and claim alignment about the focus for the ensuing session. Fragments 1 and 2 are examples of interactional practices that were recurrently observed in the analysed data, and are examples of fatigue and fatigue self-management being presented from the outset of a consultation (usually directly after introducing name and role), facilitating displays of alignment between the cancer survivor and nurse counsellor about the focus of the consultation.

The way in which a therapy session is initiated is crucial for maintaining a collaborative partnership between clinicians and cancer survivors and ensuring a productive therapeutic session (DiMatteo, 1998; Ekberg et al., 2016; Ha & Longnecker, 2010). The analysis identified that consultations where the nurse counsellor did not establish a focus on fatigue self-management from the outset were liable to problems in understanding. An example of this is seen in Fragment 3 below.

#### Fragment 3 [NC3CS3/ 0:00

In contrast to Fragments 1 and 2, after stating her name (data not shown) the nurse counsellor does not immediately present the agenda of the consultation (i.e., which is to provide fatigue SMS through the joint creation of a fatigue management plan). Rather, she refers to the agenda vaguely, describing it as 'this process' on line 2 (*"thank you for being a part of this process I suppose"*), and begins the consultation by posing broad questions to the cancer survivor on lines 3 to 4 (*"How do you feel about it ?"; "Do you know what to expect?"; "What are your thoughts about it?"*).

01 NUR: Oh, I'm (0.7) >so pleased< to be able to get to talk to yo:u. U::m (0.6) Thank 02 you for (0.5) agree:ing to be part of this um, this process? I suppose, Noah. Uh::mm hahhheh How do you feel about it? Do you know what to expe:ct\_or? What's-03 04 what are your th[oughts] about it [hh] 05 CLI: [No:: ] [I ]have absolutely no- nn no- >nothing well I'mma bout< (.) to (.) find out. 06 07 NUR: hahhheh  $\underline{Ok}$ ay well, I think it's- the idea is that it's a bit of a joint thing 80 between (0.3) the two of us. 09 (0.7)10 NUR: uh:m [so:o]= 11 CLI: [Yeah] 12 NUR: =I I think we know that a lot of people (0.3) have a problem with fatigue? (0.5)13 when they've got a cancer diagnosis?  $\underline{an}::d$ , I gather from the team (0.3) that (.) you've been experiencing a fair amount? Does that sound right? 14 15 (1.5)16 CLI: A fair amount of? 17 NUR: FATIGUE. 18 (1.0)19 CLI: Oh yes, y<u>e:ah.</u> 20 NUR: Yeah? (3.0)21 22 CLI: Yeah. 23 (0.8)NUR: 24 So:? 25 Uh:m, (0.2) yeah probably tiyed, ya know those sorts of things? CLI:  $_{\uparrow} Yea:h_{\uparrow}. And and if it's okay, what I might do, >just to get started< is (.) to$ 26 NUR: 27 ask you a couple of questions (0.3) about fatigue. [Bu- ] 28 CLI: [Sure.] NUR: 29 I also just want to check in and make su:re (0.5) I know um ((clears throat)) 30 what's been happening with you, and what's important to you, cos (0.2) the idea 31 is m[aybe there-] >look at a bit of a< plan to (0.4) hopefully help= 32 CLI: mmhmmm ] ſ 33 NUR: =support you, uh::m pt 34 CLI: yeah. 35 maybe (.) maybe get back into some of the >day to day< things? that you (0.4) you NUR: 36 might miss doing? Does that sound all right? 37 CLI: Yeah. Well, (mbfg) (0.2) um because of the chemo I'm on 38 (0.7)39 NUR: mmmm? 40 CLI: It- (0.2) it does restrict me going out. 41 (2.0)42 NUR: Why is that (0.2) Noah.

The cancer survivor's reply on line 6 ("nothing, well I'm about to find out") in response to the nurse counsellors open-ended queries, indicates that he claims uncertainty about what to expect in the session, highlighting some of the expectations clinicians may need to manage at the outset of an SMS consultation.

Throughout the opening sequence presented in Fragment 3, the cancer survivor's verbal responses, vocal hesitations, pauses, and gaps at his projected turn at talk (i.e., lines 18, 21, and 23) within the conversation possibly indicate a lack of alignment as to the agenda and purpose of the consultation. For example, in response to the nurse counsellor's query on lines 12-17, the cancer survivor produces an extended pause (1 second at line 18) and thus delays his answer; he then answers with (*"Oh, yes yeah"*, line 19). Work by Heritage (1998) suggests that the 'oh' that prefaces the cancer survivor's response can indicate that the questions being posed to him are problematic in terms of their pre-suppositions and relevance, and that there

is some lack of fit between the nurse counsellor's question and cancer survivor's answer. In other words, the cancer survivor claims a lack of clarity regarding the purpose and relevance of the questions being posed by the nurse counsellor, as well as the level of detail expected in his responses. Additionally, the absence of any explanation of the cancer survivor's experience of fatigue other than him merely confirming its presence ("Oh, yes yeah", line 19) appears to not align with the nurse counsellor's expected response, showcasing the 'lack of fit'. This misalignment is solidified by the nurse counsellor's additional prompts for further explanation from the client (line 20).

Difficulty in establishing a shared understanding of the consultation persists throughout the fragment. On line 19, the nurse counsellor prompts the cancer survivor for a further response when she says "yeah?" - switching the 'turn at talk' back to him. Again, the cancer survivor does not immediately respond to this prompt and a long gap in the conversation re-emerges (line 21). When the cancer survivor does respond, it is with a single emphasised word response ("Yeah", line 22) which in this context has been shown to imply insufficient agency and commitment to a course of action being assented to (Heritage & Raymond, 2012). Put simply, the cancer survivor is unaware of how to respond to the nurse counsellor's queries and is no longer dedicated to that line of questioning. Further, the cancer survivor's difficulty in providing a response that aligns with the nurse counsellor's need for a more detailed answer and preferred response type is influenced by problems in understanding (Heritage & Raymond, 2012; Pomerantz, 1975). The nurse counsellor again prompts the cancer survivor for a further response on line 24 ("So?"). The nurse counsellor's persistent pursuit displays an understanding that there is something insufficient about the client's response (Jefferson, 1981). Further, work by Raymond (2004) finds that a stand-alone "so" is a prompt often produced in environments of 'misalignment', when silences or other conduct from a recipient (in this case the cancer survivor) do not align with the response the speaker (in this case the nurse counsellor) expects should be- or could be produced (Raymond, 2004). Moreover, when the nurse counsellor produces a "so" prompt, she invites the cancer survivor to reassess his prior turn at talk ("yeah", line 22) and the action it accomplishes. The cancer survivor responds to this prompt with a drawn-out vocal hesitation ("Uhm") and briefly pauses on line 25, which also delays his eventual answer ("probably, tired you know those sorts of things?"). The use of the word "probably" at the beginning of the cancer survivor's answer indicates he is unsure if he is answering the nurse counsellor correctly (in a way that the nurse counsellor expects).

Conversational gaps convey meaning, with studies highlighting that repeated extended silences can be influenced by content, and speaker understanding. The pauses and vocal hesitations exhibited by the cancer survivor in this third fragment indicates he is having

difficulty producing an answer to, and thus dealing with, the nurse counsellor's gueries and prompts. Pomerantz (1975) observes that features which delay the production of a response, such as vocal hesitations (e.g., umm, ah, I guess, probably), pauses, and gaps usually accompany a dis-preferred response (where the cancer survivor either does not understand the projection, or wishes to say something different to what the nurse counsellor is projecting (Pomerantz, 1975). The cancer survivor's difficulty may also stem from his struggle to respond relevantly to a question where the agenda has not been set. As demonstrated by Ekberg and colleagues (2016), when projecting the proposed arrangement from the outset of a consultation, therapists help clients to understand how they should contribute to the therapeutic process; as without adequate expectation management, cancer survivors have little structure to appreciate how they can contribute (Ekberg et al., 2016). Similarly, in this fragment it is evident that the cancer survivor does not have a clear understanding of his role in the consultation because he has not been given a framework to understand how he can contribute (e.g., what is the consultation about?, is his role to answer questions? what sort of information is he supposed to provide? what level of detail is required in his responses?). The client's clear confusion highlights the importance of clearly establishing the agenda of the consultation at the outset for fatigue self-management consultations. Further, this fragment is one of several instances in the data corpus that exhibit similar findings in establishing a focus on support for fatigue management. Each of these are similar insofar as they do not involve establishing the focus of the session from the outset of the consultation. In these cases, nurse counsellors often struggle to bring the ensuing conversation back to the fatigue management. This analysis has shown that managing expectations at the outset promotes shared understanding between cancer survivors and nurse counsellors and sustains a primary focus on the agenda of managing CRF. Once a primary focus for SMS has been established, a challenge for participants (i.e., nurse counsellor and cancer survivor) can be maintaining that focus. The next section examines one practice through which maintaining a focus on the agenda is accomplished.

#### Formulations

The provision of SMS for CRF is a complex multi-component process that involves clinicians undertaking several support tasks (Agbejule et al., 2023) (e.g., identifying facilitators and barriers to fatigue management; providing information and coping strategies; providing motivational interviewing; establishing goals) along with navigating the physical and psychological risk factors and side effects of fatigue that may also arise during conversation. These multiple tasks must usually be completed within a set timeframe. Accordingly, to allow for timely completion of these therapeutic tasks, it is crucial to ensure that the fatigue SMS agenda – established from the outset of the consultation – is maintained throughout an entire

94

SMS session. This analysis found that using '*formulations*' enabled nurse counsellors to maintain a focus on matters relevant for self-management fatigue planning and provided an opportunity to tie divergent conversation back to support for CRF.

Although conceptualisations vary, in simple terms, a formulation is where one speaker summarises talk up to that point (Antaki, 2008; Peckitt & Smart, 2018). It is commonly used in clinical psychology and aims to develop an understanding of a cancer survivor's (or other service user's) problems and perspectives (Johnstone & Dallos, 2014). There are generally two types of formulation; "gist", which provides the sense of talk a up to a certain point, and "upshot" which provides an inference from prior talk (Heritage & Watson, 1980). Formulative techniques are commonly used in clinical scenarios as a method to direct a conversation into a clinician's or therapist's preferred direction of talk (Heritage & Watson, 1980). In the current context, formulations are a way to introduce change or new understanding for the cancer survivor. Fragments 4, and 5 pertain to the same nurse counsellor and cancer survivor uses formulation practices to collect relevant information from the cancer survivor.

#### Using formulations to focus on matters relevant for fatigue self-management planning

This section provides examples of how formulation practices were used to maintain focus on matters relevant for fatigue self-management planning.

#### Fragment 4 [NC1CS1/ 14:03]

At this point in the consultation, the nurse counsellor is attempting to determine potential barriers or facilitators to the cancer survivor undertaking fatigue self-management activities (e.g., physical activity). This assessment of the cancer survivor's self-management capacity will inform the creation of his fatigue self-management plan. Through her use of formulations, the nurse counsellor selects parts of the cancer survivor's talk that are ostensibly relevant to her current agenda, and presents this to the cancer survivor in a way that it could be integrated into his fatigue management plan. Prior to the beginning of Fragment 4, the cancer survivor has disclosed that he perceives he is less fatigued around 11 o'clock at night. He expresses that he finds this frustrating as he is a musician, and he thinks he cannot play music at this time.

```
01
   CLI: I'm a musician, (0.6) [you know I can't] play music at [eleven o'clock].
                              [ Ah:h oka:ay ]
02
   NUR:
                                                              [Yeah,
                                                                      yeah? ]
03 NUR: okay, yeah yeah. what >sort of< um (0.2) musician, (.) what <instrume::nt>=
04 CLI: oh-
05 NUR: =or:r >instruments< or:r? he-
06 CLI: There's- there's pretty much not an instrument I can't pla:y.
07
        (0.4)
08 NUR: Yeah oka[y, wow]
09 CLI:
                [I've ] been doing this for years, [so- ]
```

```
10 NUR:
                                                    [Yeah,] (.) yeah
11 CLI: and I miss that.
12
         (0.5)
13 NUR: Yeah
   CLI: Um:mm (0.3)
14
15 NUR: Yep
16 CLI: It's not so much because I don't have the energy?
17
         (0.3)
18 NUR: mmm
19
   CLI: but because my hands have been damaged? [through the immunotherapy], and it makes
20 NUR:
                                                        mm:mhm mmhm
   CLI: it very hard (1.0) to use my fingers and hands,=
21
22 NUR: = [ mmmmm. ]
23 CLI: =[I can't ] explain they're=
24 NUR: mmm.
25
         (.)
26 NUR: mmm.
27
         (0.9)
28 CLI: =some-times almost unusable. [Like ] I can barely pick a cup.
29 NUR:
                                      [↑Okav↑]
30
   NUR: Yeah, (.) okay, (.) yeah
   CLI: And so (.) that's pretty common.
31
32 NUR: mmhm mm-
33 CLI: \uparrowTemperature\uparrow (0.5) has a >hell of a lot< to do with everything.
34 NUR: mmhmm mmhm[mm]
                   [If] I get cold, (2.0) and that can happen in minutes.
35
   CLI:
36 NUR: ↓mmmm↓
37 CLI: Um like if I were to go outside now, (1.0) and uh \geqin the wind<
38 NUR: uh huh?
39 CLI: ee- five minutes (0.2) and I would pay for it for about three- three-
40 three or four hours.
41
   NUR: Okay. Yeah. Ye[ah.
                                  1
                       [(An' not-)] Temperature changes, (1.5) that's a bi:g one. (0.3) Yeah
42
   CLT:
43 NUR: Okay, so with temperature change, (.) what does that trigger? What tends
44
        to happen wh[en you] pay for i- for that-
45 CLI:
                     [pain ]
46 NUR: So pain.
47
        (.) [okay, yeah]
48 CLI: [pain, yeah]
49 NUR: Yeah, yeah, okay, [yes ]
50 CLI:
                           [lots] of it [yeah]
                                        [yeah]
51 NUR:
52
   NUR: Okay pain? So, you've really got to consider your environment [and whether]to when=
53
   CLI:
                                                                       [absolutely ]
54 NUR: =to go [outside,] okay
55 CLI:
               [Yes
                        1
    ((20 seconds omitted, cancer survivor describes how spending 5 or 10 minutes
    in an air-conditioned shop, can cause him four hours pain))
58 NUR: So Ben, it's like you're doing a balancing act.
59
   CLI: Yeah=
60 NUR: =You're weighing up, is it worth (.) while doing that=
61 CLI: yep
62 NUR: =or going out [because of the] ramifications.
                       [that's exactly it]
63
   CLI:
64
   CLI: yeah
65 NUR: and and that's really tough when [you're needing to do that].
66 CLI:
                                          [(
                                                                   ) ]
67 NUR: Yeah yeah, And I can hear you're motivated. You've got this real
68
         motivation to engage with life, with people.
```

On lines 19 to 28, the cancer survivor shares that he has been unable to play musical instruments because of difficulty using his fingers and hands. He then relays that this inability to use his hands is triggered by temperature changes when he goes outside (lines 37 to 42). After prompting from the nurse counsellor on lines 43 to 44 (*"ok, with the temperature change,* 

what does that trigger? What tends to happen when you...") the cancer survivor states the underlying factor is that temperature changes cause him pain. On lines 52 to 54, the nurse counsellor produces an upshot formulation ("So, you've really got to consider your environment and whether to- when to go outside"). This formulation reframes and creates a version of the cancer survivor's preceding talk on lines 21 to 50. In presenting the formulation to the cancer survivor, the nurse counsellor gives him the option to agree or disagree with this reframed summary. Work by Heritage and Watson shows that there is a preference for confirmation and acceptance of formulations, as disconfirmation might be seen to jeopardise "the sense of the talk thus far" (Heritage & Watson, 1979).

The formulation produced by the nurse counsellor is accepted by the cancer survivor on line 55 ("Yes"). Notably, in her formulation, the nurse counsellor does not use any of the cancer survivor's exact words, but rather selects the components of his talk that ostensibly align with her SMS agenda (to determine potential barriers to the cancer survivor undertaking fatigue self-management activities and use this information to inform his fatigue management planas indicated by the nurse counsellor's consultation checklist). Evidently, the nurse counsellor's use of the word "*you've*" shows that she is attempting to incorporate the cancer survivor's experience into her formulation. This practice has been recurrently observed in clinical interactions (Antaki, 2008; Heritage & Watson, 1980), and enables the nurse counsellor to reformulate the design and terminology of the cancer survivor's preceding talk to ostensibly suit her agenda and progress the consultation

Another example of formulation practices being used to focus on matters relevant for fatigue management planning occurs within Fragment 4. On lines 58 to 65, the nurse counsellor produces another formulation in response to the cancer survivor's temperature-mediated pain ("So Ben, it's like you're doing a balancing act. You're weighing up, is it worthwhile doing that or going out because of the ramifications and that's really tough when you're needing to do that"). Again, the nurse counsellor uses this formulation to reframe the cancer survivor's symptoms into a shape suitable for creating his fatigue self-management plan (Antaki, 2008). Further, as with the previous instance, the cancer survivor agrees with the nurse counsellor's formulation on lines 61 to 63 (*"yep, that's exactly it"*). This is consistent with evidence that shows formulations generally project agreement, which the speaker must either actively provide (as seen on lines 49 and 53-55) or combat in the next turn (Heritage and Watson, 1979; Antaki, 2008).

# Using formulations to close down matters not relevant for fatigue self-management planning

The previous section showed how nurse counsellors used formulations to sustain talk that is ostensibly relevant in co-creating the cancer survivor's fatigue self-management plan. When providing specific symptom support (such as support for CRF), there are situations where a clinician may need to refocus conversations to maintain focus on the primary agenda of fatigue support. This section demonstrates how formulations can be used to momentarily redirect talk that is not ostensibly relevant (at least, not at the moment) in the creation of the fatigue management plan; and subsequently progress the conversation in direction that aligns with the agenda of providing SMS for fatigue.

#### Fragment 5 [NC1CS1/ 19:55]

Like Fragment 4, at this point in the session the nurse counsellor is attempting to determine potential barriers or facilitators to the cancer survivor undertaking fatigue self-management activities, to inform the creation of goals for his fatigue self-management plan. Prior to the beginning of this fragment, the cancer survivor has relayed to the nurse that "it doesn't matter whether it has been a good day or a bad day, I must complete something" (data not shown). The cancer survivor explains that he writes lists for himself that consist of tasks that he must complete each day. He describes that these daily lists used to be "big" but he "found that would just make [him] depressed" as he couldn't complete everything. Directly prior to the beginning of Fragment 5, the cancer survivor has recounted that something as simple as cleaning his fish tank which is 15 min work, is difficult for him and no matter how hard he tries he just cannot

do it.

01 02 03 04 05	CLI: NUR:	I just can't do it, I can't. And I wonder if, (0.3) let's say for instance, myself or someone who's close to you was to see that list and see what you've achieved, that we would notice that uhh (0.2) you know, given (.) what you're going through, what you've experienced and the impact of all of this, that (.) that was actually really	
06	~	significant what you achieved in your day.	
07	CLI:	Yea:h	
08	NUR:	But I can hear what you're saying=	
09	CLI:	Ye <u>ah</u>	
10	NUR:	=at the same time it's it's at ti:mes. It can be: (0.5), I guess, confronting	
11	OT T .	for you (.) or: uh=	
	CLI:	1	
13		=or disappointing or-	
•			
•	. to undertake are no longer "getting done". This causes him to feel that he didn't		
•		ibute or achieve anything.))	
17	CLI:	it's sort of like each day is worth nothing.	
18	NUR:		
19	CLI:	I didn't achieve anything. I didn't (1.0) I dunno I guess I didn't	
20		<u>con</u> tribute.	
21	NUR:	mmm mmm	
22	CLI:	I dunno (1.0) um	
23	NUR:	mmm whereas- whereas, at the same time, I can uh really hear that those lists	
24		have been really helpful=	
25	CLI:	oh yeah	
26	NUR:	=because they give you a goal. And sometimes small achievable=	

27 CLI: Yeah
28 NUR: =tasks and goals are most helpful.
29 CLI: Yep

On lines 8 to 13 the nurse counsellor begins a formulation, and in doing so reframes the cancer survivor's description of his frustration ("I just cant do it, I cant.", line 1) into "But I can hear what you're saying at the same time, I guess it can be confronting for you or disappointing-". In this instance the nurse counsellor uses her formulation to acknowledge the cancer survivor's distress and redefine a topic of concern to both parties – that the client is frustrated at not being able to complete daily tasks and that this is confronting for him. As with the previous fragment, the nurse counsellor's use of the phrases 'you're saying' and 'confronting for you' demonstrates attentiveness to the cancer survivor's talk and incorporates his experience into her formulation. The cancer survivor subsequently accepts this formulation on line 12 ("it can be yeah"); however, he continues to voice frustration at not being able to undertake what he considers to be simple tasks (e.g., "it's sort of like each day is worth nothing... I didn't achieve anything... I didn't contribute" ... "sometimes I'll look at what I achieved and just think it's so pathetic", line 17, lines 19-20). Instead of immediately launching into an inquiry of the cancer survivor's comments, the nurse counsellor continues the formulation she began on line 8, and focuses on the identified facilitator ("whereas, at the same time, I can really hear that those lists have been really helpful, ... because they give you a goal. And sometimes small achievable tasks are most helpful", lines 23-24; lines 26-28). The cancer survivor's account of his apprehensions is valuable to the nurse counsellor as it provides insight into how the fatigue is impacting his quality of life (Berger, 2019). As the nurse counsellor has already acknowledged and noted the cancer survivor's apprehensions in her formulation on lines 8 to 13, she uses the remainder of her formulation to momentarily close that direction of talk as spending extra time discerning it (during this history taking stage), is not ostensibly useful to the creation of the fatigue management plan. Notably, the cancer survivor agrees with the nurse counsellor's formulation (on lines 25, 27, 29) and does not continue with his talk about his perceived inability to cope, which subsequently allows the conversation to progress in a different direction towards the intended goal of the consultation and allows the nurse counsellor to continue providing support for CRF.

This 'conversation management' (i.e., the technique of effectively managing the flow and direction of communication) was particularly useful for nurse counsellor's to obtain a comprehensive view of the cancer survivor's fatigue first (i.e., determining his fatigue severity; fatigue interference on daily living; risk factors for fatigue occurrence; presenting characteristics; self-management capabilities; etc.) and then still have the time to collaboratively provide solutions to matters raised. Moreover, when conducting her

99

formulation, the nurse counsellor does not completely dismiss the cancer survivor's concerns but addresses his apprehensions at later stages in the consultation. For instance, later in the session she offers counselling support, refers the cancer survivor to external psychological services, and also considers his frustrations when creating goals for his fatigue management plan. Antaki (2008) describes this practice as 'formulating the trouble away' and suggested that its purpose was to keep a client's descriptions 'non-therapizable' until the assessment stage of a session was complete. Similarly, in this instance the nurse counsellor is using her formulations to control the flow and progress of the consultation by redirecting certain avenues of talk, and choosing the parts of the cancer survivors talk that ostensibly aligns with her agenda of providing fatigue SMS (e.g., setting small realistic goals gives the cancer survivor a sense of achievement); while seemingly being attentive to something in the cancer survivor's own words ("I always look to make sure to put a line through at least one [list item], And that way I feel like I've done something" [data not shown]). Analysis of the entire data corpus found that in consultations where this 'conversation management' did not occur, it appeared nurse counsellors often became overwhelmed with cancer survivor recounts (e.g., extended personal accounts about family life and upcoming holidays; prolonged treatment-related talk about chemotherapy and/or opinions on other past their medical teams etc.), which left insufficient time for the provision of supportive management strategies and action planning. This fragment demonstrates how the nurse counsellor used formulation practices to maintain focus on fatigue support provision, allowing her the time to acknowledge the other themes or activities that arise in the consultation in a more productive way.

# Discussion

This research responds to the need for evidence-based explanations of how SMS practices for CRF can be, or have been integrated and delivered in clinical care at the interactional level. This study is the first attempt to identify, in detail, the communicative practices used by nurse counsellors and cancer survivors during real-life SMS sessions for CRF. Overall, this study presents two observable practices nurse counsellors use in their work with cancer survivors to provide SMS for CRF: (1) introducing the agenda at the outset of the consultation, and (2) formulations (see Figure 13).

SMS encounters in cancer care are particularly complex as the organisation of supportive care visits and tasks are generally unfamiliar to cancer survivors, meaning they must navigate the challenges of understanding what a clinician is doing and what is expected of them (i.e., what to say and when) (Sterponi et al., 2019). In their recent realist review on factors influencing clinician engagement in SMS, Kantilal and colleagues (2022) identified that in order for productive SMS to occur, clinicians and patients must be clear about their respective roles in

self-management and create a sense of shared responsibility. The findings of this conversation analysis study offer a potential resolution to this complexity by identifying that in instances where clinicians focus discussion on fatigue SMS through clearly introducing the agenda of the consultation at the outset, cancer survivors have a clear understanding of their role, and the tasks and goals of a session. Further, the analysis indicated that managing expectations in this manner produced shared alignment, consequently strengthening the clinician-cancer survivor collaborative partnership. Studies have shown that lack of explanation or poor management of expectations can hinder patient understanding of treatment; and can result in a lack of consensus between clinicians and cancer survivors, as well as therapeutic failure (Ardito & Rabellino, 2011; DiMatteo, 1998; Ha & Longnecker, 2010). Fitzpatrick and colleagues (2006) provide further insight into the mechanisms of a collaborative partnership in their own qualitative study where they identify that establishing shared understanding of therapeutic tasks and goals in the early stages of support provision, produces an upward spiral which launches cancer survivors into an exploratory process that supports more productive therapeutic work (Fitzpatrick et al., 2006). This is consistent with qualitative findings which indicated that cancer survivors often withhold information (i.e., reporting of CRF severity and CRF impact on daily living) from their health care teams if they feel they are rushed or if they perceive there is little time to discuss CRF with their treating clinician (Jones et al., 2021).

Sterponi and colleagues suggest several other significant complexities of self-management encounters in cancer care; 1) much of the supportive care tasks conducted during a consultation depends on information that the clinician acquires from the cancer survivor during the course of the session, and as such support tasks are unpredictable; 2) the extent of attention to be devoted to each support task cannot be fully determined in advance but requires clinicians to exercise ongoing judgement; and 3) the timeframe of supportive care encounters are generally limited (Sterponi et al., 2019). Qualitative research on CRF support provision has demonstrated that clinicians often attempt to bypass these challenges by not responding at all to patient inquires or accounts they consider irrelevant in an attempt to provide quick support, and may even avoiding initiating CRF support discussions with their patients altogether (Bootsma et al., 2020; Borneman et al., 2011; Jones et al., 2021; Jones et al., 2020). This contributes to cancer survivors feeling misunderstood and discouraged to discuss CRF with their healthcare teams, and has been shown to contribute to the breakdown of the patient-provider partnership that is essential to successfully manage CRF (Bootsma et al., 2020; Jones et al., 2021). This conversation analysis study makes a significant contribution in addressing the above challenges, by identifying that employing formulation practices allows clinicians to: select ostensibly useful parts of a cancer survivor's accounts as they occur and reframe them into a shape suitable for supportive tasks whilst remaining attentive; maintain a focus on matters relevant for support tasks and self-management fatigue planning; tie divergent conversation back to support for CRF; and subsequently provide support within limited time frames. During the provision of symptom support, there may be several instances where clinicians need to redirect and refocus conversations to ensure productive and timely supportive care is provided; formulations act to facilitate this process in a less abrupt manner compared the other reported methods above. As suggested by Antaki (2008), closing, or redirecting conversation using formulation techniques "promotes the sense that one has listened to the other speaker and has extracted something that they themselves might have said" and allows clinicians to maintain respectful and attentive communication.

# **Limitations and Strengths**

This study analysed interactions between cancer survivors and specialist nurses who have received training in supportive care counselling procedures. Notably, not all health professionals are readily trained in this practice. As such, a limitation of this work is that it did not observe the extent to which clinicians who are not specialist trained can operationalise these practices (i.e., introducing the agenda at the outset, producing formulations) successfully. Despite this, these findings have significant implications for research and training. For example, while beginning clinical interactions with open-ended question formats can sometimes be valuable to solicit a patient's chief complaint (e.g., in a primary care consultation) (Robinson et al., 2016), the analysis indicated that in situations that are taskfocused and that have a specific agenda – such as in SMS consultation – opening with broad questions can be problematic as these question types can communicate different things; and thus can be understood, and responded to, differently (Robinson, 2006). This is also supported by findings from Jones and colleagues' recent qualitative study in which cancer survivors reported that being asked general, open-ended questions was insufficient when discussing CRF with their healthcare teams (Jones et al., 2021). Several studies have indicated that fostering communication and consultation skills can improve clinician confidence in engaging cancer survivors in support discussions about symptom self-management (Kantilal et al., 2022). This conversation analytic study is an initial step in identifying the required training to facilitate this process.

# Conclusions

A breakdown of clinician-provider partnerships (i.e., not identifying shared responsibility) has been identified as a significant barrier to the translation of CRF management strategies to clinical practice and contributes to the underreporting and under-management of CRF (Jones et al., 2021; Kantilal et al., 2022), (as cancer survivors are discouraged to report their fatigue to their health teams). During supportive care sessions, where fatigue management is being targeted, clinicians should focus discussion on fatigue SMS early, by clearly introducing the agenda from the outset of the consultation and asking for client agreement. Continually formulating or summarising patient's talk allows clinicians to maintain a focus on matters relevant for self-management fatigue planning; tie divergent conversation back to support for CRF; and subsequently provide support within limited time frames.

# 6.0 DISCUSSION AND CONCLUSIONS

# Chapter Introduction

This doctoral thesis has explored the efficacy of support programs for CRF management, and identified the components of effective SMS interventions (Study 1); developed a best practice clinical framework that outlined the essential support practices needed for health professionals to successfully support cancer survivors with their fatigue (Study 2); and analysed how SMS can be delivered in clinical practice at an interactional level (Study 3). The doctoral thesis contributes to understanding and enhancing self-management behaviours, CRF, and SMS. The doctoral thesis also addresses several national health and research priorities (e.g., the establishment of SMS CRF best practices, investigation of SMS communication practices and techniques and consultation skills) and has the potential to impact the education and training of the health workforce, improve the adoption of fatigue management behaviours by cancer survivors, and ultimately reduce CRF severity and improve cancer survivor quality of life. This final chapter synthesises the key research findings from the three studies that comprise this doctoral thesis. This chapter also outlines the significance and contributions of this program of research, possible directions for future research, and overall conclusions that can be made from the doctoral thesis.

# **Overview, Key Research Findings, and Implications**

## **Establishment of CRF SMS Best Practices**

Cancer survivor self-management is focal in the management of CRF; thus, the importance of empowering survivors to self-manage through SMS cannot be overstated. The literature review presented in Chapter 1 emphasised that evidence-based fatigue management strategies are rarely implemented in practice, with CRF remaining undertreated and underreported. Additionally, cancer survivors routinely report they feel discouraged by their health care teams and that they receive inadequate management support. Health professionals report they do not have confidence in providing fatigue SMS to cancer survivors and have called for clear guidance in this area (Hilarius, 2011). Unfortunately, little work had been done to understand the components of SMS for CRF. Inconsistences in selfmanagement conceptualisations and terminologies, along with limited identification of CRF SMS strategies hinder the synthesisation, advancement, uptake, and translation of CRF selfmanagement strategies to practice. The systematic review (Study 1) presented in Chapter 3, is the first published study to identify potentially optimal SMS intervention features and components that are associated with improved CRF and behavioural outcomes. This original contribution to knowledge is unique in that it presented a definitive concept of comprehensive SMS for fatigue and examined the impact of SMS practices on the adoption of fatigue selfmanagement behaviours, as well as CRF severity - an important factor that is often

overlooked. The examination of behavioural outcomes in Study 1 was particularly important because SMS interventions for CRF target cancer survivor behaviour change and support the reduction of fatigue severity through behavioural mechanisms.

Study 1 identified several factors that potentially influence the effectiveness of a fatigue SMS program. Most notably, the review found that studies in which SMS was delivered after cancer treatment, in one-on-one formats, facilitated by health professionals, while incorporating at least one in-person (i.e., face-to-face) contact most frequently reported positive effects on CRF and behavioural outcomes. Additionally, studies which reported the provision of additional training to intervention facilitators most frequently reported positive intervention effects for both CRF and associated fatigue management behavioural outcomes (including self-efficacy to undertake fatigue self-management behaviours). Findings from Study 1 highlighted that guidance and prioritisation of SMS clinical strategies are needed to (1) ensure health professionals have the required CRF SMS capabilities; and (2) to enable the sustained translation of CRF management strategies into practice – a finding that is consistent with key action areas emphasised in national research and health priorities.

The Australian Government identified education and training of SMS as a national health priority, and emphasised the importance of equipping the health workforce with support capabilities. Key national action areas include embedding self-management as core learning, developing health professional roles and practices to better support self-management, and implementing evaluation mechanisms to assess SMS across health services (Australian Health Ministers' Advisory Council, 2017; Nichols et al., 2020). Additionally, the Clinical Oncology Society of Australia (COSA) recently identified the top research priorities for cancer survivorship in Australia. Their report clearly outlined intervention development for CRF among the top three physiological outcome priorities; and the development, implementation, and dissemination of self-management as top health service priority (Crawford-Williams et al., 2022).

The CRF SMS Practice Framework presented in Chapter 4 (Study 2: Modified Delphi Study) makes a significant contribution to the progress of the priorities and key action areas described above. The modified Delphi study was the first published research to develop and clearly define the core practices needed for health professionals to deliver SMS to cancer survivors experiencing CRF; a widely acknowledged critical action point that has not been addressed until now (Australian Health Ministers' Advisory Council, 2017; Howell et al., 2017; Nichols et al., 2020; Rimmer, Sharp, & on behalf of Ways Ahead study team, 2021). Rimmer and colleagues (2021) recently emphasised the importance of defining the core components of SMS programs to ensure the mechanisms of the support are understood and that

implementation is replicable (Rimmer, Sharp, & on behalf of Ways Ahead study team, 2021). Notably, the CRF SMS Practice Framework not only presents key components, it also considers the complex nature of CRF SMS, by conceptualising support as an active process that requires an essential set of collaborative-and partnership building strategies, knowledge, and practices. As mentioned by some panel members in the modified Delphi study, the identification of best practices promotes consistent and evidenced-based approaches to CRF care, as clinicians can use the practice framework as a reference tool for support strategies, recommendations, and information, which can promote continuous learning – another one of the key priority areas highlighted above. Further, the establishment of key practices can streamline the implementation process (Agbejule et al., 2021) and promote best practice, as clinicians who are knowledgeable of support practices are more likely to conduct support conversations with their patients (Kantilal et al., 2022). Additionally, the practice framework has functionality for several different users across a variety of settings, including health professionals, program evaluators; clinical leaders and educators; cancer consumers; clinical researchers; and intervention developers (see Table 7).

#### Exploration of how SMS Practice Could Be Delivered in Clinical Care

Although a best practice framework is useful, it is also equally important to explore the implementation of a framework in the real-world. This doctoral thesis therefore incorporates research designed to understand how SMS for CRF was accomplished by health professionals in clinical practice and potentially provide insight into where to direct efforts to ensure clinical practice meets best practice recommendations. In the modified Delphi study presented in Chapter 4, some clinician panel participants expressed hesitancy around the feasibility of delivering the established best practice support items in clinical settings. Clinicians questioned how support discussions (e.g., identifying barriers and facilitators to client self-management, providing coping strategies, goal setting, etc.), and creation of goals would occur in real-life clinical practice. Chapter 5 presented a conversation analysis study (Study 3) that provided insight into how SMS CRF practices have been executed in clinical care in a specific local setting (T-CRF SMS Clinic) and provided an opportunity to examine communication practices of SMS in more detail and highlight communication techniques another national health policy action area. The literature review in Chapter 1 and the Flinders SMS Capability Framework introduced in Chapter 2, highlighted how successful adoption of self-management CRF behaviours by cancer survivors (e.g., dietary changes, increased physical activity, etc.) is greatly dependent on the quality of clinician-survivor interaction and the effective provision of efficacious SMS strategies. Australian policymakers have emphasised the importance of equipping the health workforce with effective communication techniques to support patient self-management. Policy and research priorities stipulate the

need for further investigation and clarification of how communication could be optimised when providing support to cancer survivors (Australian Health Ministers' Advisory Council, 2017; Crawford-Williams et al., 2022). Kantilal and colleagues' recent review emphasised this when they highlighted that the communication, mutual trust, and shared responsibility between clinicians and patients, as well as the presence of sound clinician consultation skills, are key factors influencing whether health professionals engage in SMS tasks and conversations with their patients (Kantilal et al., 2022). The conversation analysis study (Study 3) presented in Chapter 5, generated direct observational evidence for understanding the communication practices used by cancer nurse counsellors when providing SMS to cancer survivors experiencing CRF, and is the first study to examine the impact of these practices within consultations. Findings from Study 3 demonstrated that in instances where clinicians focus discussion on fatigue SMS through clearly introducing the agenda of the consultation at the outset, cancer survivors have a clear understanding of their role, and the tasks and goals of a session. Additionally, study findings suggest that using formulation practices could allow clinicians to maintain a focus on fatigue during consultations and potentially provide support for fatigue in limited time frames. The conversation analytic study contributes to a limited evidence base by providing much needed insight into the communicative practices of clinicians providing SMS for CRF which can support recommendations for future training and practice; contribute to identifying effective strategies for discussing CRF and successfully initiating and maintaining supportive discussions; and support efforts to improve outcomes for cancer survivors.

## Limitations

Limitations of each study (Studies 1 - 3) have been presented in their respective chapters. This limitations section presents overarching limitations of the doctoral thesis.

#### **Exclusion of Concurrent Symptoms and Psychological Outcomes**

The absence of reporting psychological outcomes in addition to CRF and behavioural outcomes in Study 1, is a potential limitation of this doctoral thesis. It is well reported that symptoms often do not occur alone, and can present as symptom clusters. Psychological outcomes such as depression and anxiety are known to contribute to and potentially exacerbate CRF and impede self-management, forming a complex interplay between these factors. This doctoral thesis took an *a prior* decision to focus on CRF as an unresolved, complex, debilitating symptom experienced by cancer survivors. For future research, including data on concurrent symptoms including psychological outcomes alongside the measured outcomes would have been beneficial. The inclusion of measures could have provided valuable insights into the understanding of CRF SMS. Future studies should consider
incorporating comprehensive assessments of psychological outcomes to capture the multifaceted nature of CRF and its potential comorbidities, thereby enhancing the overall understanding of the condition and its management.

### **Specialised Settings**

The high international representation within the panel of the modified Delphi study may pose a potential risk to the specificity and contextual relevance of the CRF SMS Practice Framework within the Australian setting. While the broad global applicability of the study findings is valuable, it may result in a trade-off with the depth of impact on the local context. Another limitation was that Study 3 assessed CRF SMS in a specific CRF-focused setting by nurse counsellors with extensive experience in counselling. It is worth reflecting on the suitability of the framework to other settings such as busy routine care environments like primary care or hospital settings that are non-specialist fatigue clinics. The effectiveness and feasibility of implementing the framework in these diverse contexts may present challenges and require further investigation. Factors such as resource availability, staff expertise, and workflow dynamics could significantly impact the practicality and outcomes of implementing the framework outside the specialised CRF clinic. Thus, it is crucial to acknowledge that the generalisability and adaptability of the framework to different clinical settings might be limited, warranting additional research to evaluate its feasibility and efficacy in these non-specialist environments.

#### Focus on the Health Professional System of Support

The exclusive focus on only health professional SMS could be considered an overarching limitation of this doctoral thesis. So far, this research has emphasised the importance of the clinician-survivor relationship in improving the ability of cancer survivors to adapt health behaviours and manage their CRF. Both the FCF and the CCM introduced in Chapter 2, assert that improved health outcomes are dependent on productive interactions between patients and their health care team, and perceived self-efficacy is highlighted as a significant factor underpinning the effectiveness of CRF self-management behaviours, as well as several self-management theories and models (Chan et al., 2016; Foster & Fenlon, 2011; Girgis, 2020; Hoffman et al., 2009; Lawn & Battersby, 2009; Wagner et al., 1996). Although, these understandings of self-management have strong theoretical basis and empirical support, they do not take into consideration systems of support beyond survivor and health professional interactions (Mills et al., 2017; Vassilev et al., 2010). In fact, a frequent criticism is that current understandings of SMS are often constructed from the health provider point of view, and are often thought to be as a result of an individual's motivation or self-efficacy (Rogers et al.,

2011). Further, in traditional theories, factors such as the contexts, resources, practices, priorities, and networks of cancer survivors (which are impacted by wider determinants of health) are often not acknowledged (Koetsenruijter et al., 2014; Vassilev et al., 2010). The exploration of such factors are important, as they provide a comprehensive insight into how self-management is actually completed by individuals in the context of their everyday lives, and allows for SMS strategies and interventions to be better targeted for sustained individual completion (Vassilev et al., 2010).

There is increased recognition of the need to move beyond the sole focus on health professionals, and consider the social systems of support of individuals (also referred to a 'communities of practice' and 'systems of support') - in the self-management of long-term conditions (Koetsenruijter et al., 2014; Mills et al., 2017; Rogers et al., 2011; Thoits, 2011; Vassilev et al., 2010). Moreover, evidence has shown that SMS programs which actively engage patients in social relationships, in particular personal (e.g., family, friends) and community (e.g. social support groups) systems of support, and address the context in which a patient's life and illness are enacted, are well received and lead to sustained change in the management of chronic conditions (Kendall et al., 2012). Further, as most survivor self-management occurs outside of clinical health care settings, social systems of support outside of the health professional (see Figure 13) are integral and cannot be ignored (Rogers et al., 2011). The current doctoral thesis provides important understanding of aspects of how chronic conditions are managed but cannot provide comprehensive understanding of such a complex phenomenon.



Figure 13: Systems of Support (Rogers et al., 2011)

# **Future Directions**

## **Research for Contextual Evidence**

Several avenues for future research have been identified from this doctoral thesis. One of the key points for future work is the need to contextualise the CRF SMS Practice Framework to allow for successful implementation into routine care. Puddy and Wilkins' present a model that offers a way of conceptualising evidence in three overlapping components, the best available research evidence; experiential evidence; and contextual evidence (Puddy & Wilkins, 2011). These components are defined as follows:

- Best available research evidence is based on empirical studies or research reviews in which data has been collected, analysed, and documented (e.g., systematic review and synthesis of RCTs on the effectiveness of support programs for CRF).
- **Experiential evidence** is based on an expert's long-term engagement in a field, offering insights into what has worked, what seems to work, and other knowledge and expertise that can sometimes be understood, without being stated *(e.g., consultation)*

with cancer consumers; health professionals who have experience in cancer care; fatigue researchers and supportive cancer researchers, via modified Delphi study).

• **Contextual evidence** is based on information about whether a strategy or practice is perceived as useful, relevant, feasible to implement and is acceptable to specific groups.

This doctoral thesis incorporates these tenets of evidence to inform the practice framework (see Figure 14). Although the development of the practice framework integrated feedback on usability from panel participants of the modified Delphi study (Study 2), and the conversation analysis study (Study 3) examined the local context of CRF SMS communication practices in a specific clinical care setting, additional contextual evidence is needed to investigate and support implementation of the practice framework in routine care. As briefly mentioned in Chapter 4, the CRF SMS Practice Framework could be significantly enhanced through additional consultation with key stakeholders. Moreover, contextual evidence could be further honed by conducting primary research evaluations that provide insight on feasibility (i.e., ability to use the practice framework to undertake to SMS tasks given the resources available and the economic, social, geographic, and historical aspects of the current setting); acceptability (i.e., whether or not the practice framework is useful or appropriate for the needs of the people in the current setting) (Puddy & Wilkins, 2011).

### Social Systems of Support



Figure 14: Conceptualising Evidence

SMS for cancer survivors is a multi-level process that involves several essential support systems. A core finding from Study 1 was the need to structure SMS advice and fatigue selfmanagement activities in the context of everyday life. The systematic review conducted for Study 1 found that studies which reported this essential component (i.e., structuring selfmanagement interventions to align with survivors' 'everyday life') most frequently reported reduced participant fatigue and sustained uptake of fatigue management behaviours (Agbejule et al., 2022). This concept was incorporated into the CRF SMS Practice Framework and received positive comments particularly from consumer participants of the modified Delphi study (Appendix 8). Consumer participants reiterated the importance of their personal network as well as other peer support in their perceived confidence to self-manage their CRF. In fact, studies show that a cancer survivor's first response to experiencing CRF is usually communicating their support needs with their personal network (Bootsma et al., 2021). Future qualitative research could examine in more detail the needs, preferences, and views of cancer survivors concerning the support of their CRF, to determine an optimal model of care. Further, future research could explore how voluntary and community groups, non-health professionals, and personal communities are currently involved in CRF management support; could best be recognised, supported, or formally incorporated in support provision; and potentially whether or how these groups could utilize the practice framework to provide supportive care for CRF. By understanding and defining the role of these support networks, SMS for those experiencing CRF can be refined to ensure that cancer survivors truly receive comprehensive support.

## Conclusions

The World Health Organization (World Health Organization, 2003) asserts that increasing the effectiveness of clinician SMS will have far greater impact on the health of a population than any improvement in special medical treatments. Enabling health professionals to provide effective SMS for CRF – a symptom that is routinely reported as being unaddressed and unmet – is crucial. This doctoral thesis moves beyond the investigation of intervention effectiveness to the identification, dissemination, and implementation of evidence based CRF self-management strategies into clinical care through the establishment of effective SMS practices.

# BIBLIOGRAPHY

- Abdalrahim, M. S., Herzallah, M. S., Zeilani, R. S., & Alhalaiqa, F. (2014). Jordanian nurses' knowledge and attitudes toward cancer-related fatigue as a barrier of fatigue management. *The Journal of American Science*, *10*(2), 191-197.
- Abrahams, H. J. G., Gielissen, M. F. M., Donders, R. R. T., Goedendorp, M. M., van der Wouw, A. J., Verhagen, C. A. H., & Knoop, H. (2017). The efficacy of Internet-based cognitive behavioral therapy for severely fatigued survivors of breast cancer compared with care as usual: A randomized controlled trial. *Cancer*, *123*(19), 3825-3834. <u>https://doi.org/10.1002/cncr.30815</u>
- Agbejule, O. A., Hart, N. H., Ekberg, S., & Chan, R. J. (2023). Development of a selfmanagement support practice framework for addressing cancer-related fatigue: a modified Delphi study [Advanced online publication]. *Journal of Cancer Survivorship*. <u>https://doi.org/10.1007/s11764-023-01348-7</u>
- Agbejule, O. A., Hart, N. H., Ekberg, S., Crichton, M., & Chan, R. J. (2022). Selfmanagement support for cancer-related fatigue: A systematic review. *International Journal of Nursing Studies*, *129*(5), e104206. <u>https://doi.org/10.1016/j.ijnurstu.2022.104206</u>
- Agbejule, O. A., Hart, N. H., Ekberg, S., Koczwara, B., Ladwa, R., Simonsen, C., Pinkham, E., P., & Chan, R. J. (2021). Bridging the research to practice gap: a systematic scoping review of implementation of interventions for cancer-related fatigue management. *BMC Cancer*, *21*(1). <u>https://doi.org/10.1186/s12885-021-08394-3</u>
- Ahuja, M., Aseltine, R., Warren, N., Reisine, S., Williams, P. H., & Cislo, A. (2018).
   Challenges faced with the implementation of Web-Based Data Query Systems for population health: development of a questionnaire based on expert consensus. *Pilot and Feasibility Studies*, *4*(1), e113. <u>https://doi.org/10.1186/s40814-018-0307-3</u>
- Antaki, C. (2008). Formulations in psychotherapy. In (pp. 26-42). Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511490002.003</u>
- Araújo-Soares, V., Hankonen, N., Presseau, J., Rodrigues, A., & Sniehotta, F. F. (2019). Developing Behavior Change Interventions for Self-Management in Chronic Illness: An Integrative Overview. *European Psychologist*, 24(1), 7-25. https://doi.org/10.1027/1016-9040/a000330
- Ardito, R., & Rabellino, D. (2011). Therapeutic Alliance and Outcome of Psychotherapy: Historical Excursus, Measurements, and Prospects for Research. *Frontiers in Psychology*, 2(270). <u>https://doi.org/10.3389/fpsyg.2011.00270</u>
- Armes, J., Chalder, T., Addington-Hall, J., Richardson, A., & Hotopf, M. (2007). A randomized controlled trial to evaluate the effectiveness of a brief, behaviorally oriented intervention for cancer-related fatigue. *Cancer*, *110*(6), 1385-1395. https://doi.org/10.1002/cncr.22923
- Australian Health Ministers' Advisory Council. (2017). National Strategic Framework for Chronic Conditions. Canberra: Australian Government
- Australian Institute of Health and Welfare. (2021). Cancer in Australia 2021. Canberra: AIHW
- Australian Institute of Health and Welfare. (2022). *Australian Burden of Disease Study 2022*. AIHW.
- Baker, J., Lovell, K., & Harris, N. (2006). How expert are the experts? An exploration of the concept of 'expert' within Delphi panel techniques. *Nurse researcher*, *14*(1), 59-70. https://doi.org/10.7748/nr2006.10.14.1.59.c6010
- Bandura, A. (1977). Self-efficacy: Toward a unifying theory of behavioral change. *Psychological Review*, 84(2), 191-215. <u>https://doi.org/10.1037/0033-295X.84.2.191</u>
- Banbury, A., Taylor, M., Caffery L., Der Vartanian, C., Haydon, H., Mendis, R., Ng, Kawai., & Smith, A. (2023) Consumers' experiences, preferences, and perceptions of effectiveness in using telehealth for cancer care in Australia. *Asia-Pacific Journal of Clinical Oncology*. e1-10. <u>https://doi.org/10.1111/ajco.14002</u>

- Banipal, R. P. S., Singh, H., & Singh, B. (2017). Assessment of Cancer-related Fatigue among Cancer Patients Receiving Various Therapies: A Cross-sectional Observational Study. *Indian Journal of Palliative Care*, 23(2), 207-211. <u>https://doi.org/10.4103/IJPC.IJPC\_135\_16</u>
- Bantum, E. O., Albright, C. L., White, K. K., Berenberg, J. L., Layi, G., Ritter, P. L., Laurent, D., Plant, K., & Lorig, K. (2014). Surviving and thriving with cancer using a Webbased health behavior change intervention: randomized controlled trial. *Journal of Medical Internet Research*, *16*(2), e54. <u>https://doi.org/10.2196/jmir.3020</u>
- Barnes, R. (2005). Conversation analysis: a practical resource in the health care setting. *Medical Education*, 39(1), 113-115. <u>https://doi.org/10.1111/j.1365-2929.2004.02037.x</u>
- Barnes, R. K. (2019). Conversation Analysis of Communication in Medical Care: Description and Beyond. *Research on Language and Social Interaction*, *52*(3), 300-315. <u>https://doi.org/10.1080/08351813.2019.1631056</u>
- Barrett, S., Begg, S., O'Halloran, P., Howlett, O., Lawrence, J., & Kingsley, M. (2021). The effect of behaviour change interventions on changes in physical activity and anthropometrics in ambulatory hospital settings: a systematic review and meta-analysis. *International Journal of Behavioral Nutrition and Physical Activity*, *18*(1), e7. https://doi.org/10.1186/s12966-020-01076-6
- Barsevick, A., Beck, S. L., Dudley, W. N., Wong, B., Berger, A. M., Whitmer, K., Newhall, T., Brown, S., & Stewart, K. (2010). Efficacy of an intervention for fatigue and sleep disturbance during cancer chemotherapy. *Journal of Pain and Symptom Management*, 40(2), 200-216. <u>https://doi.org/10.1016/j.jpainsymman.2009.12.020</u>
- Barsevick, A. M., Dudley, W., Beck, S., Sweeney, C., Whitmer, K., & Nail, L. (2004). A randomized clinical trial of energy conservation for patients with cancer-related fatigue. *Cancer*, *100*(6), 1302-1310. <u>https://doi.org/10.1002/cncr.20111</u>
- Bennett, J. A., Lyons, K. S., Winters-Stone, K., Nail, L. M., & Scherer, J. (2007). Motivational interviewing to increase physical activity in long-term cancer survivors: a randomized controlled trial. *Nursing Research*, *56*(1), 18-27. https://www.cochranelibrary.com/central/doi/10.1002/central/CN-00574610/full
- Bennett, S., Pigott, A., Beller, E. M., Haines, T., Meredith, P., & Delaney, C. (2016). Educational interventions for the management of cancer-related fatigue in adults. *Cochrane Database of Systematic Reviews*(11), Article CD008144. <u>https://doi.org/10.1002/14651858.CD008144.pub2</u>
- Berger. (2019). NCCN clinical practice guidelines in oncology (NCCN Guidelines®) cancerrelated fatigue.
- Berger, Mitchell, Jacobsen, & Pirl. (2015). Screening, evaluation, and management of cancer-related fatigue: Ready for implementation to practice? CA: A Cancer Journal for Clinicians, 65(3), 190-211. <u>https://doi.org/10.3322/caac.21268</u>
- Berger, Mooney, K., Alvarez-Perez, A., Breitbart, W. S., Carpenter, K. M., Cella, D., Cleeland, C., Dotan, E., Eisenberger, M. A., Escalante, C. P., Jacobsen, P. B., Jankowski, C., LeBlanc, T., Ligibel, J. A., Loggers, E. T., Mandrell, B., Murphy, B. A., Palesh, O., Pirl, W. F., . . . Smith, C. (2015). Cancer-Related Fatigue, Version 2.2015. *13*(8), e1012. <u>https://doi.org/10.6004/jnccn.2015.0122</u>
- Berger, A., M., & Mooney, K. (2016). Dissemination and Implementation of Guidelines for Cancer-Related Fatigue. *Journal of the National Comprehensive Cancer Network*, 14(11), 1336-1338. <u>https://doi.org/10.6004/jnccn.2016.0144</u>
- Biering, K., Frydenberg, M., Pappot, H., & Hjollund, N. H. (2020). The long-term course of fatigue following breast cancer diagnosis. *Journal of Patient-Reported Outcomes*, 4(1), 37. <u>https://doi.org/10.1186/s41687-020-00187-9</u>
- Bin Hasan, M. Z., Bin Hossain, T., & Aminul, I. (2014). Factors Affecting Self-Efficacy Towards Academic Performance: A Study on Polytechnic Students in Malaysia. *Advances in Environmental Biology*, *8*.
- Blase, K., & Fixsen, D. L. (2013). Core Intervention Components: Identifying and Operationalizing What Makes Programs Work. *ASPE Research Brief*, 1-21.

https://aspe.hhs.gov/reports/core-intervention-components-identifyingoperationalizing-what-makes-programs-work-0

- Bloor, M., & Wood, F. (2006). Conversation Analysis. In SAGE (Ed.), *Keywords in Qualitative Methods* (pp. 39-34). SAGE Publications Ltd. <u>https://doi.org/10.4135/9781849209403</u>
- Bluethmann, S. M., Bartholomew, L. K., Murphy, C. C., & Vernon, S. W. (2017). Use of Theory in Behavior Change Interventions. *Health Education & Behavior*, 44(2), 245-253. <u>https://doi.org/10.1177/1090198116647712</u>
- Boel, A., Navarro-Compán, V., Landewé, R., & van der Heijde, D. (2021). Two different invitation approaches for consecutive rounds of a Delphi survey led to comparable final outcome. *Journal of Clinical Epidemiology*, *129*, 31-39. https://doi.org/10.1016/j.jclinepi.2020.09.034
- Bohlius, J., Tonia, T., Nüesch, E., Jüni, P., Fey, M. F., Egger, M., & Bernhard, J. (2014). Effects of erythropoiesis-stimulating agents on fatigue- and anaemia-related symptoms in cancer patients: systematic review and meta-analyses of published and unpublished data. *British Journal of Cancer*, *111*(1), 33-45. https://doi.org/10.1038/bjc.2014.171
- Boland, L., Bennett, K., & Connolly, D. (2018). Self-management interventions for cancer survivors: a systematic review. *Supportive Care in Cancer*, *26*(5), 1585-1595. https://doi.org/10.1007/s00520-017-3999-7
- Boogaard, L., Gater, L., Mori, M., Trincao, A., & Smith-Turchyn, J. (2016). Efficacy of selfmanagement programs in managing side effects of breast cancer: A systematic review and meta-analysis of randomized control trials. *Rehabilitation Oncology*, 34, 14-26. <u>https://doi.org/10.1097/01.REO.0000475835.78984.41</u>
- Bootsma, T. I., Schellekens, M. P. J., van Woezik, R. A. M., Slatman, J., & van der Lee, M. L. (2021). Forming new habits in the face of chronic cancer-related fatigue: An interpretative phenomenological study. *Supportive Care in Cancer*, *29*(11), 6651-6659. https://doi.org/10.1007/s00520-021-06252-3
- Bootsma, T. I., Schellekens, M. P. J., van Woezik, R. A. M., van der Lee, M. L., & Slatman, J. (2020). Experiencing and responding to chronic cancer-related fatigue: A metaethnography of qualitative research. *Psycho-Oncology*, *29*(2), 241-250. <u>https://doi.org/10.1002/pon.5213</u>
- Borneman, T., Koczywas, M., Sun, V., Piper, B. F., Smith-Idell, C., Laroya, B., Uman, G., & Ferrell, B. (2011). Effectiveness of a clinical intervention to eliminate barriers to pain and fatigue management in oncology. *Journal of Palliative Medicine*, *14*(2), 197-205. https://doi.org/10.1089/jpm.2010.0268
- Bower, J. E. (2014). Cancer-related fatigue—mechanisms, risk factors, and treatments. *Nature Reviews Clinical Oncology*, *11*(10), 597-609. <u>https://doi.org/10.1038/nrclinonc.2014.127</u>
- Bower, J. E., Bak, K., Berger, A., Breitbart, W., Escalante, C. P., Ganz, P. A., Schnipper, H. H., Lacchetti, C., Ligibel, J. A., Lyman, G. H., Ogaily, M. S., Pirl, W. F., & Jacobsen, P. B. (2014). Screening, Assessment, and Management of Fatigue in Adult Survivors of Cancer: An American Society of Clinical Oncology Clinical Practice Guideline Adaptation. *Journal of Clinical Oncology*, *32*(17), 1840-1850. https://doi.org/10.1200/JCO.2013.53.4495
- Bower, J. E., Ganz, P. A., Desmond, K. A., Bernaards, C., Rowland, J. H., Meyerowitz, B. E., & Belin, T. R. (2006). Fatigue in long-term breast carcinoma survivors. *Cancer*, 106(4), 751-758. <u>https://doi.org/10.1002/cncr.21671</u>
- Bower, J. E., Ganz, P. A., Desmond, K. A., Rowland, J. H., Meyerowitz, B. E., & Belin, T. R. (2000). Fatigue in breast cancer survivors: Occurrence, correlates, and impact on quality of life. *Journal of Clinical Oncology*, *18*(4), 743-753. https://doi.org/10.1200/JCO.2000.18.4.743
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. Qualitative Research in Psychology, 3(2), 77-101. <u>https://doi.org/10.1191/1478088706qp063oa</u>

- Bruggeman-Everts, F. Z., Wolvers, M. D. J., van de Schoot, R., Vollenbroek-Hutten, M. M. R., & Van der Lee, M. L. (2017). Effectiveness of Two Web-Based Interventions for Chronic Cancer-Related Fatigue Compared to an Active Control Condition: Results of the "Fitter na kanker" Randomized Controlled Trial. *Journal of Medical Internet Research*, 19(10), e336. <u>https://doi.org/10.2196/jmir.7180</u>
- Cancer Australia. (2017). *Cancer incidence, mortality and survival in Australia*. Retrieved 25 from http://edcan.org.au/edcan-learning-resources/using-edcan-resources
- Chan, C. W. H., Richardson, A., & Richardson, J. (2011). Managing Symptoms in Patients with Advanced Lung Cancer During Radiotherapy: Results of a Psychoeducational Randomized Controlled Trial. *Journal of Pain and Symptom Management*, 41(2), 347-357. <u>https://doi.org/10.1016/j.jpainsymman.2010.04.024</u>
- Chan, R., Yates, P., & McCarthy, A. L. (2016). Fatigue Self-Management Behaviors in Patients With Advanced Cancer: A Prospective Longitudinal Survey. *Oncology Nursing Forum*, *43*(6), 762-771. <u>https://doi.org/10.1188/16.0nf.762-771</u>
- Charalambous, A., & Kouta, C. (2016). Cancer Related Fatigue and Quality of Life in Patients with Advanced Prostate Cancer Undergoing Chemotherapy. *BioMed research international*, 2016, e3989286. <u>https://doi.org/10.1155/2016/3989286</u>
- Chen, H.-L., Liu, K., & You, Q.-S. (2018). Self-efficacy, cancer-related fatigue, and quality of life in patients with resected lung cancer. *European Journal of Cancer Care*, *27*(6), e12934. <u>https://doi.org/10.1111/ecc.12934</u>
- Chiba, I., Sasahara, T., & Mizuno, M. (2019). Factors in Cancer-Related Fatigue Self-Management Behaviors of Outpatients Undergoing Chemotherapy. Asia-Pacific Journal of Oncology Nursing, 6(3), 209-211. https://doi.org/10.4103/apjon.apjon 1 19
- Chin, C.-H., Tseng, L.-M., Chao, T.-C., Wang, T.-J., Wu, S.-F., & Liang, S.-Y. (2021). Selfcare as a mediator between symptom-management self-efficacy and quality of life in women with breast cancer. *PLOS ONE*, *16*(2), e0246430. https://doi.org/10.1371/journal.pone.0246430
- Connolly, M. (2007). Practice frameworks: Conceptual maps to guide interventions in child welfare. *British Journal of Social Work*, 37(5), 825-837.
- Corbett, T. K., Groarke, A., Devane, D., Carr, E., Walsh, J. C., & McGuire, B. E. (2019). The effectiveness of psychological interventions for fatigue in cancer survivors: systematic review of randomised controlled trials. *Systematic Reviews*, *8*(1), e324(2019). <u>https://doi.org/10.1186/s13643-019-1230-2</u>
- Corbin, J., & Strauss, A. (1985). Managing chronic illness at home: three lines of work. *Qualitative Sociology*, *8*(3), 224-247.
- Crawford-Williams, F., Koczwara, B., Chan, R. J., Vardy, J., Lisy, K., Morris, J., Iddawela, M., Mackay, G., & Jefford, M. (2022). Defining research and infrastructure priorities for cancer survivorship in Australia: a modified Delphi study. *Supportive Care in Cancer*, *30*(5), 3805-3815. <u>https://doi.org/10.1007/s00520-021-06744-2</u>
- Curt, G. A., Breitbart, W., Cella, D., Groopman, J. E., Horning, S. J., Itri, L. M., Johnson, D. H., Miaskowski, C., Scherr, S. L., Portenoy, R. K., & Vogelzang, N. J. (2000). Impact of cancer-related fatigue on the lives of patients: new findings from the Fatigue Coalition. *Oncologist*, *5*(5), 353-360. <u>https://doi.org/10.1634/theoncologist.5-5-353</u>
- Custer, R., Scarcella, J. A., & Stewart, B. R. (1999). The Modified Delphi Technique A Rotational Modification. *Journal of Career and Technical Education*, *15*, 50-58. <u>https://files.eric.ed.gov/fulltext/EJ590767.pdf</u>
- Cuthbert, C. A., Farragher, J. F., Hemmelgarn, B. R., Ding, Q., McKinnon, G. P., & Cheung, W. Y. (2019). Self-management interventions for cancer survivors: A systematic review and evaluation of intervention content and theories. *Psycho-Oncology*, *28*(11), 2119-2140. <u>https://doi.org/10.1002/pon.5215</u>
- de Silva, D. (2011). Helping people help themselves: a review of the evidence considering whether it is worthwhile to support self-management. The Health Foundation. https://www.health.org.uk/sites/default/files/HelpingPeopleHelpThemselves.pdf

- De Vet, E., Brug, J., De Nooijer, J., Dijkstra, A., & De Vries, N. K. (2005). Determinants of forward stage transitions: a Delphi study. *Health Education Research*, 20(2), 195-205. <u>https://doi.org/10.1093/her/cyg111</u>
- DiMatteo, M. R. (1998). The role of the physician in the emerging health care environment. *The Western Journal of Medicine*, *168*(5), 328-333.
- Dineen-Griffin, S., Garcia-Cardenas, V., Williams, K., & Benrimoj, S. I. (2019). Helping patients help themselves: A systematic review of self-management support strategies in primary health care practice. *PLOS ONE*, *14*(8), e0220116. https://doi.org/10.1371/journal.pone.0220116
- Dodd, M. J., Cho, M. H., Miaskowski, C., Painter, P. L., Paul, S. M., Cooper, B. A., Duda, J., Krasnoff, J., & Bank, K. A. (2010). A randomized controlled trial of home-based exercise for cancer-related fatigue in women during and after chemotherapy with or without radiation therapy. *Cancer Nursing*, 33(4), 245-257. https://doi.org/10.1097/NCC.0b013e3181ddc58c.
- Donnelly, C., Blaney, J., Rankin, J., Lowe-Strong, A., Campbell, A., & Gracey, J. (2011). A randomised controlled trial testing the feasibility of a physical activity intervention in managing fatigue with gynaecological cancer survivors. *Physiotherapy (united kingdom)*, *97*, eS294-eS295.

https://www.cochranelibrary.com/central/doi/10.1002/central/CN-01726200/full

- Drew, P., Chatwin, J., & Collins, S. (2001). Conversation analysis: a method for research into interactions between patients and health-care professionals. *Health Expectations*, *4*(1), 58-70. https://doi.org/10.1046/j.1369-6513.2001.00125.x
- Dunbar, S. B., Clark, P. C., Quinn, C., Gary, R. A., & Kaslow, N. J. (2008). Family influences on heart failure self-care and outcomes. *The Journal of Cardiovascular Nursing*, 23(3), 258-265. <u>https://doi.org/10.1097/01.JCN.0000305093.20012.b8</u>
- Duprez, V., Beeckman, D., Verhaeghe, S., & Van Hecke, A. (2017). Self-management support by final year nursing students: A correlational study of performance and person-related associated factors. *International Journal of Nursing Studies*, 74, 120-127. <u>https://doi.org/10.1016/j.ijnurstu.2017.06.010</u>
- Duprez, V., Beeckman, D., Verhaeghe, S., & Van Hecke, A. (2018). Are person-related and socio-structural factors associated with nurses' self-management support behavior? A correlational study. *Patient Education and Counseling*, 101(2), 276-284. https://doi.org/https://doi.org/10.1016/j.pec.2017.08.011
- Eakin, E. G., Lawler, S. P., Vandelanotte, C., & Owen, N. (2007). Telephone Interventions for Physical Activity and Dietary Behavior Change: A Systematic Review. *American Journal of Preventive Medicine*, 32(5), 419-434. <u>https://doi.org/10.1016/j.amepre.2007.01.004</u>
- Ekberg, S., Barnes, R. K., Kessler, D. S., Malpass, A., & Shaw, A. R. G. (2016). Managing clients' expectations at the outset of online Cognitive Behavioural Therapy (CBT) for depression. *Health Expectations*, *19*(3), 557-569. <u>https://doi.org/10.1111/hex.12227</u>
- Elissen, A., Nolte, E., Knai, C., Brunn, M., Chevreul, K., Conklin, A., Durand-Zaleski, I., Erler, A., Flamm, M., Frølich, A., Fullerton, B., Jacobsen, R., Saz-Parkinson, Z., Sarria-Santamera, A., Sönnichsen, A., & Vrijhoef, H. (2013). Is Europe putting theory into practice? A qualitative study of the level of self-management support in chronic care management approaches. *BMC Health Services Research*, *13*(1), e117. <u>https://doi.org/10.1186/1472-6963-13-117</u>
- Fabi, A., Bhargava, R., Fatigoni, S., Guglielmo, M., Horneber, M., Roila, F., Weis, J., Jordan, K., & Ripamonti, C. I. (2020). Cancer-related fatigue: ESMO Clinical Practice Guidelines for diagnosis and treatment. *Annals of Oncology*, *31*(6), 713-723. <u>https://doi.org/10.1016/j.annonc.2020.02.016</u>
- Fillion, L., Gagnon, P., Leblond, F., Gelinas, C., Savard, J., Dupuis, R., Duval, K., & Larochelle, M. (2008). A brief intervention for fatigue management in breast cancer survivors. *Cancer Nursing*, 31(2), 145-159. <u>https://doi.org/10.1097/01.NCC.0000305698.97625.95</u>

- Finnegan-John, J., Molassiotis, A., Richardson, A., & Ream, E. (2013). A systematic review of complementary and alternative medicine interventions for the management of cancer-related fatigue. *Integrative Cancer Therapies*, *12*(4), 276-290. <u>https://doi.org/10.1177/1534735413485816</u>
- Fitch, M. I., Mings, D., & Lee, A. (2008). Exploring patient experiences and self-initiated strategies for living with cancer-related fatigue. *Canadian Oncology Nursing Journal*, 18(3), 124-140. <u>https://doi.org/10.5737/1181912x184124131</u>
- Fitzpatrick, M. R., Janzen, J., Chamodraka, M., & Park, J. (2006). Client critical incidents in the process of early alliance development: A positive emotion–exploration spiral. *Psychotherapy Research*, 16(4), 486-498. https://doi.org/10.1080/10503300500485391
- Foster, C., & Fenlon, D. (2011). Recovery and self-management support following primary cancer treatment. *British Journal of Cancer*, *105*(Suppl 1), S21-S28. https://doi.org/10.1038/bjc.2011.419
- Foster, C., Grimmett, C., May, C., Ewings, S., Myall, M., Hulme, C., Smith, P., Powers, C., ra, Calman, L., Armes, J., Breckons, M., Corner, J., Fenlon, D., Batehup, L., Lennan, E., May, C., Morris, C., Neylon, A., . . . Turner, L. (2016). A web-based intervention (RESTORE) to support self-management of cancer-related fatigue following primary cancer treatment: a multi-centre proof of concept randomised controlled trial. *Supportive Care in Cancer*, *24*(6), 2445-2453. <u>https://doi.org/10.1007/s00520-015-3044-7</u>
- Freeman, L. W., White, R., Ratcliff, C. G., Sutton, S., Stewart, M., Palmer, J. L., Link, J., & Cohen, L. (2015). A randomized trial comparing live and telemedicine deliveries of an imagery-based behavioral intervention for breast cancer survivors: reducing symptoms and barriers to care. *Psycho-Oncology*, 24(8), 910-918. <u>https://doi.org/10.1002/pon.3656</u>
- Galiano-Castillo, N., Cantarero-Villanueva, I., Fernández-Lao, C., Ariza-García, A., Díaz-Rodríguez, L., Del-Moral-Ávila, R., & Arroyo-Morales, M. (2016). Telehealth system: A randomized controlled trial evaluating the impact of an internet-based exercise intervention on quality of life, pain, muscle strength, and fatigue in breast cancer survivors. *Cancer*, *122*(20), 3166-3174. <u>https://doi.org/10.1002/cncr.30172</u>
- Gardner, R. (2001). When Listeners Talk: Response Tokens and Listener Stance. https://doi.org/10.1075/pbns.92
- Gernier, F., Joly, F., Klein, D., Mercier, M., Velten, M., & Licaj, I. (2020). Cancer-related fatigue among long-term survivors of breast, cervical, and colorectal cancer: a French registry–based controlled study. *Supportive Care in Cancer, 28*(12), 5839-5849. <u>https://doi.org/10.1007/s00520-020-05427-8</u>
- Gill, V. T., & Roberts, F. (2013). Conversation Analysis in Medicine. In *The Handbook of Conversation Analysis* (pp. 575-590).
- Girgis, A. (2020). The role of self-management in cancer survivorship care. *The Lancet Oncology*, 21(1), 8-9. <u>https://doi.org/10.1016/S1470-2045(19)30715-6</u>
- Gobeil-Lavoie, A.-P., Chouinard, M.-C., Danish, A., & Hudon, C. (2019). Characteristics of self-management among patients with complex health needs: a thematic analysis review. *BMJ Open*, *9*(5), e028344. <u>https://doi.org/10.1136/bmjopen-2018-028344</u>
- Godino, C., Jodar, L., Durán, Á., Martínez, I., & Schiaffino, A. (2006). Nursing education as an intervention to decrease fatigue perception in oncology patients. *European Journal of Oncology Nursing*, *10*(2), 150-155. <u>https://doi.org/10.1016/j.ejon.2005.03.004</u>
- Goedendorp, M. M., Andrykowski, M. A., Donovan, K. A., Jim, H. S., Phillips, K. M., Small, B. J., Laronga, C., & Jacobsen, P. B. (2012). Prolonged impact of chemotherapy on fatigue in breast cancer survivors: a longitudinal comparison with radiotherapy-treated breast cancer survivors and noncancer controls. *Cancer*, *118*(15), 3833-3841. <a href="https://doi.org/10.1002/cncr.26226">https://doi.org/10.1002/cncr.26226</a>
- Gokal, K., Wallis, D., Ahmed, S., Boiangiu, I., Kancherla, K., & Munir, F. (2016). Effects of a self-managed home-based walking intervention on psychosocial health outcomes for

breast cancer patients receiving chemotherapy: a randomised controlled trial. *Supportive Care in Cancer*, *24*(3), 1139-1166. <u>https://doi.org/10.1007/s00520-015-2884-5</u>

- Goldberg, J. I., Schulman-Green, D., Hernandez, M., Nelson, J. E., & Capezuti, E. (2019). Self-Management Interventions for Psychological Distress in Adult Cancer Patients: A Systematic Review. *Western Journal of Nursing Research*, *41*(10), 1407-1422. <u>https://doi.org/10.1177/0193945919845104</u>
- Golsteijn, R. H. J., Bolman, C., Volders, E., Peels, D. A., de Vries, H., & Lechner, L. (2018). Short-term efficacy of a computer-tailored physical activity intervention for prostate and colorectal cancer patients and survivors: a randomized controlled trial. *International Journal of Behavioral Nutrition and Physical Activity*, *15*(1), e106. https://doi.org/10.1186/s12966-018-0734-9
- Gong, J., Hu, C., Chen, M., Cao, Q., & Li, Q. (2021). Interventions to Improve Self-Efficacy in Colorectal Cancer Patients and/or Caregivers: A Systematic Review and Meta-Analysis. *Journal of Oncology*, *2021*, 4553613. https://doi.org/10.1155/2021/4553613
- Grady, P. A., & Gough, L. L. (2014). Self-management: a comprehensive approach to management of chronic conditions. *American Journal of Public Health*, *104*(8), e25-e31. <u>https://doi.org/10.2105/AJPH.2014.302041</u>
- Grégoire, C., Faymonville, M.-E., Vanhaudenhuyse, A., Charland-Verville, V., Jerusalem, G., Willems, S., & Bragard, I. (2020). Effects of an intervention combining self-care and self-hypnosis on fatigue and associated symptoms in post-treatment cancer patients: A randomized-controlled trial. *Psycho-Oncology*, *29*(7), 1165-1173. https://doi.org/10.1002/pon.5395
- Grey, M., Knafl, K., & McCorkle, R. (2006). A framework for the study of self- and family management of chronic conditions. *Nursing Outlook*, *54*(5), 278-286. https://doi.org/10.1016/j.outlook.2006.06.004
- Grover, A., & Joshi, A. (2014). An overview of chronic disease models: a systematic literature review. *Global Journal of Health Science*, 7(2), 210-227. https://doi.org/10.5539/gjhs.v7n2p210
- Gudgeon, M., Wilkinson, A., & Hale, L. (2022). Healthcare professional students' perceptions of supporting patient self-management: A mixed method study. *Chronic Illness*, *19*(2), 395-408. <u>https://doi.org/10.1177/17423953211073367</u>
- Ha, J. F., & Longnecker, N. (2010). Doctor-patient communication: a review. *The Oschsner Journal*, *10*(1), 38-43. <u>https://www.ochsnerjournal.org/content/ochjnl/10/1/38.full.pdf</u>
- Hallowell, M. R., & Gambatese, J. A. (2010). Qualitative Research: Application of the Delphi Method to CEM Research. *Journal of Construction Engineering and Management*, 136(1), 99-107. <u>https://doi.org/10.1061/(ASCE)CO.1943-7862.0000137</u>
- Hammer, M. J., Ercolano, E. A., Wright, F., Dickson, V. V., Chyun, D., & Melkus, G. D. E. (2015). Self-management for Adult Patients With Cancer: An Integrative Review. *Cancer Nursing*, 38(2). <u>https://doi.org/10.1097/NCC.00000000000122</u>
- Hepburn, A. B., G.B. (2012). The Conversation Analytic Approach to Transcription. In *The* Handbook of Conversation Analysis (pp. 57-76). https://doi.org/10.1002/9781118325001.ch4
- Heritage, J., & Maynard, D. W. (2006). Problems and Prospects in the Study of Physician-Patient Interaction: 30 Years of Research. *Annual Review of Sociology*, *3*2(1), 351-374. <u>https://doi.org/10.1146/annurev.soc.32.082905.093959</u>
- Heritage, J., & Raymond, G. (2012). Navigating epistemic landscapes:: Acquiescence, agency and resistance in responses to polar questions. In J. P. de Ruiter (Ed.), *Questions: Formal, Functional and Interactional Perspectives* (pp. 179-192). Cambridge University Press. https://doi.org/10.1017/CBO9781139045414.013
- Heritage, J., & Watson, R. (1980). Formulations as conversational objects. *Semiotica*, 3, 245-262.
- Hernandez Silva, E., Lawler, S., & Langbecker, D. (2019). The effectiveness of mHealth for self-management in improving pain, psychological distress, fatigue, and sleep in

cancer survivors: a systematic review. *Journal of Cancer Survivorship*, *13*(1), 97-107. <u>https://doi.org/10.1007/s11764-018-0730-8</u>

- Hilarius, D. L., Kloeg, P. H., van der Wall, E., Komen, M., Gundy, C. M., & Aaronson, N. K. (2011). Cancer-related fatigue: clinical practice versus practice guidelines. *Supportive Care in Cancer*, *19*(4), 531-538. <u>https://doi.org/10.1007/s00520-010-0848-3</u>
- Hilfiker, R., Meichtry, A., Eicher, M., Nilsson Balfe, L., Knols, R. H., Verra, M. L., & Taeymans, J. (2018a). Exercise and other non-pharmaceutical interventions for cancer-related fatigue in patients during or after cancer treatment: a systematic review incorporating an indirect-comparisons meta-analysis. *British Journal of Sports Medicine*, 52(10), 651-658. <u>https://doi.org/10.1136/bjsports-2016-096422</u>
- Hilfiker, R., Meichtry, A., Eicher, M., Nilsson Balfe, L., Knols, R. H., Verra, M. L., & Taeymans, J. (2018b). Exercise and other non-pharmaceutical interventions for cancer-related fatigue in patients during or after cancer treatment: a systematic review incorporating an indirect-comparisons meta-analysis. *British Journal of Sports Medicine*, 52(10), e651. <u>https://doi.org/10.1136/bjsports-2016-096422</u>
- Hoffman, A. J., Brintnall, R. A., Given, B. A., von Eye, A., Jones, L. W., & Brown, J. K. (2017). Using Perceived Self-efficacy to Improve Fatigue and Fatigability In Postsurgical Lung Cancer Patients: a Pilot Randomized Controlled Trial. *Cancer Nursing*, *40*(1), 1-12. <u>https://doi.org/10.1097/NCC.00000000000378</u>
- Hoffman, A. J., von Eye, A., Gift, A. G., Given, B. A., Given, C. W., & Rothert, M. (2009). Testing a Theoretical Model of Perceived Self-efficacy for Cancer-Related Fatigue Self-management and Optimal Physical Functional Status. *Nursing Research*, *58*(1). https://doi.org/10.1097/NNR.0b013e3181903d7b
- Hoffmann, T. C., Glasziou, P. P., Boutron, I., Milne, R., Perera, R., Moher, D., Altman, D. G., Barbour, V., Macdonald, H., Johnston, M., Lamb, S. E., Dixon-Woods, M., McCulloch, P., Wyatt, J. C., Chan, A.-W., & Michie, S. (2014). Better reporting of interventions: template for intervention description and replication (TIDieR) checklist and guide. *BMJ*, *348*, e1687. <u>https://doi.org/10.1136/bmj.g1687</u>
- Holman, H., & Lorig, K. (2004). Patient Self-Management: A Key to Effectiveness and Efficiency in Care of Chronic Disease. *Public Health Reports*, *119*(3), 239-243. <u>https://doi.org/10.1016/j.phr.2004.04.002</u>
- Horneber, M., Fischer, I., Dimeo, F., Rüffer, J. U., & Weis, J. (2012). Cancer-Related Fatigue. *109*(9), 161-172. <u>https://doi.org/10.3238/arztebl.2012.0161</u>
- Howell, D., Harth, T., Brown, J., Bennett, C., & Boyko, S. (2017). Self-management education interventions for patients with cancer: a systematic review. *Supportive Care in Cancer*, *25*(4), 1323-1355. <u>https://doi.org/10.1007/s00520-016-3500-z</u>
- Howell, D., Keshavarz, H., Broadfield, L., Hack, T., Hamel, M., Harth, T., Jones, J., McLeod, D., Olson, K., Phan, S., Sawka, A., Swinton, N., & Ali, M. (2015). A Pan Canadian Practice Guideline for Screening, Assessment, and Management of Cancer-Related Fatigue in Adults Version 2-2015.
- Howell, D., Mayer, D. K., Fielding, R., Eicher, M., Verdonck-de Leeuw, I. M., Johansen, C., Soto-Perez-de-Celis, E., Foster, C., Chan, R., Alfano, C. M., Hudson, S. V., Jefford, M., Lam, W. W. T., Loerzel, V., Pravettoni, G., Rammant, E., Schapira, L., Stein, K. D., & Koczwara, B. (2021). Management of Cancer and Health After the Clinic Visit: A Call to Action for Self-Management in Cancer Care. *Journal of the National Cancer Institute*, *113*(5), 523-531. <u>https://doi.org/10.1093/jnci/djaa083</u>
- Howell, D., Richardson, A., May, C., Calman, L., Fazelzad, R., Moradian, S., & Foster, C. (2019). Implementation of self-management support in cancer care and normalization into routine practice: a systematic scoping literature review protocol. *Systematic Reviews*, 8(1), e37. <u>https://doi.org/10.1186/s13643-019-0952-5</u>
- Howell., Keller-Olaman, S., Oliver, T. K., Hack, T. F., Broadfield, L., Biggs, K., Chung, J., Gravelle, D., Green, E., Hamel, M., Harth, T., Johnston, P., McLeod, D., Swinton, N., Syme, A., & Olson, K. (2013). A pan-Canadian practice guideline and algorithm:

screening, assessment, and supportive care of adults with cancer-related fatigue. *Current Oncology*, *20*(3), e233-e246. <u>https://doi.org/10.3747/co.20.1302</u>

- Huang, J., Han, Y., Wei, J., Liu, X., Du, Y., Yang, L., Li, Y., Yao, W., & Wang, R. (2019). The effectiveness of the Internet-based self-management program for cancer-related fatigue patients: a systematic review and meta-analysis. *Clinical Rehabilitation*, *34*(3), 287-298. <u>https://doi.org/10.1177/0269215519889394</u>
- Hunter, J., Smith, C., Delaney, G. P., Templeman, K., Grant, S., & Ussher, J. M. (2019). Coverage of cancer services in Australia and providers' views on service gaps: findings from a national cross-sectional survey. *BMC Cancer*, *19*(1), e570. <u>https://doi.org/10.1186/s12885-019-5649-6</u>

Hutchby, I., & Wooffitt, R. (2008). Conversation Analysis. Wiley.

- Institute of Medicine Committee on Identifying Priority Areas for Quality Improvement. (2003). *Priority Areas for National Action: Transforming Health Care Quality*. National Academies Press (US)
- Institute of Medicine National Research Council. (2006). *From Cancer Patient to Cancer Survivor: Lost in Transition*. The National Academies Press. https://doi.org/10.17226/11468
- James, S., Wright, P., Scarlett, C., Young, T., Jamal, H., & Verma, R. (2015). Cancer-related fatigue: results from patient experience surveys undertaken in a UK regional cancer centre. *Supportive Care in Cancer*, *23*(7), 2089-2095. <u>https://doi.org/10.1007/s00520-014-2565-9</u>
- Jefferson, G. (2004). Glossary of transcript symbols with an introduction. . In G. H. Lerner (Ed.), *Conversation Analysis: Studies from the first generation* (pp. 13-34). John Benjamins Publishing Company.
- Jefferson, G. D. (1981). The abominable ne? : an exploration of post-response pursuit of response.
- Jobst, A., Kirchberger, I., Cieza, A., Stucki, P. D. G., & Stucki, A. (2013). Content Validity of the Comprehensive ICF Core Set for Chronic Obstructive Pulmonary Diseases: An International Delphi Survey. *The Open Respiratory Medicine Journal*, 7, 33-45. <u>https://doi.org/10.2174/1874306401307010033</u>
- Johns, S., Brown, L., Beck-Coon, K., Talib, T., Monahan, P., Giesler, R., Tong, Y., Wilhelm, L., Carpenter, J., Ah, D., Wagner, C., Groot, M., Schmidt, G., Monceski, D., Danh, M., Alyea, J., Miller, K., & Kroenke, K. (2016). Randomized controlled pilot trial of mindfulness-based stress reduction compared to psychoeducational support for persistently fatigued breast and colorectal cancer survivors. *Supportive Care in Cancer*. https://doi.org/10.1007/s00520-016-3220-4
- Johns, S. A., Brown, L. F., Beck-Coon, K., Monahan, P. O., Tong, Y., & Kroenke, K. (2015). Randomized controlled pilot study of mindfulness-based stress reduction for persistently fatigued cancer survivors. *Psycho-Oncology*, *24*(8), 885-893. <u>https://doi.org/10.1002/pon.3648</u>
- Johnston, M. F., Hays, R. D., Subramanian, S. K., Elashoff, R. M., Axe, E. K., Li, J. J., Kim, I., Vargas, R. B., Lee, J., Yang, L., & et al. (2011). Patient education integrated with acupuncture for relief of cancer-related fatigue randomized controlled feasibility study. *BMC Complementary and Alternative Medicine*, *11*, e49. <u>https://doi.org/10.1186/1472-6882-11-49</u>
- Jones, G., Gollish, M., Trudel, G., Rutkowski, N., Brunet, J., & Lebel, S. (2021). A perfect storm and patient-provider breakdown in communication: two mechanisms underlying practice gaps in cancer-related fatigue guidelines implementation. *Supportive Care in Cancer*, *29*(4), 1873-1881. <u>https://doi.org/10.1007/s00520-020-05676-7</u>
- Jones, G., Rutkowski, N., Trudel, G., St-Gelais, C., Ladouceur, M., Brunet, J., & Lebel, S. (2020). Translating Guidelines to Practice: A Training Session about Cancer-Related Fatigue. *Current Oncology*, *27*(2), 163-170. <u>https://doi.org/10.3747/co.27.5681</u>
- Jongerden, I. P., Slev, V. N., van Hooft, S. M., Pasman, H. R., Verdonck-de Leeuw, I. M., de Veer, A. J. E., van Uden-Kraan, C. F., & Francke, A. L. (2019). Self-Management

Support in Patients With Incurable Cancer: How Confident Are Nurses? *Oncology Nursing Forum*, *46*(1), 104-112. <u>https://doi.org/10.1188/19.Onf.104-112</u>

- Jordan, J. E., Briggs, A. M., Brand, C. A., & Osborne, R. H. (2008). Enhancing patient engagement in chronic disease self-management support initiatives in Australia: the need for an integrated approach. *The Medical Journal of Australia, 189*(S10), S9-S13. <u>https://doi.org/10.5694/j.1326-5377.2008.tb02202.x</u>
- Kantilal, K., Hardeman, W., Whiteside, H., Karapanagiotou, E., Small, M., & Bhattacharya, D. (2022). Facilitating healthcare practitioners to deliver self-management support in adult cancer survivors: A realist review. *Research in Social and Administrative Pharmacy*, *18*(11), 3870-3883. <u>https://doi.org/10.1016/j.sapharm.2022.05.011</u>
- Kendall, E., Foster, M. M., Ehrlich, C., & Chaboyer, W. (2012). Social Processes That Can Facilitate and Sustain Individual Self-Management for People with Chronic Conditions. *Nursing Research and Practice*, 2012, e282671. <u>https://doi.org/10.1155/2012/282671</u>
- Kenioua, M., & El-Kadder, B. (2016). Self-efficacy, Achievement motivation and Anxietyof Elite Athletes. *IOSR Journal of Sports and Physical Education*, *3*, 2347-6745. <u>https://doi.org/10.9790/6737-03044548</u>
- Kennedy, A., Gask, L., & Rogers, A. (2005). Training professionals to engage with and promote self-management. *Health Education Research*, *20*(5), 567-578. <u>https://doi.org/10.1093/her/cyh018</u>
- Khodyakov, D., Grant, S., Denger, B., Kinnett, K., Martin, A., Peay, H., & Coulter, I. (2020). Practical Considerations in Using Online Modified-Delphi Approaches to Engage Patients and Other Stakeholders in Clinical Practice Guideline Development. *The Patient - Patient-Centered Outcomes Research*, *13*(1), 11-21. https://doi.org/10.1007/s40271-019-00389-4
- Kidwell, M. (1997). Demonstrating Recipiency: Knowledge Displays as a Resource for the Unaddressed Participant. *Issues in Applied Linguistics, 8*(2), 85-96. https://doi.org/10.5070/L482005258
- Kim, S. H., Kim, K., & Mayer, D. K. (2017). Self-Management Intervention for Adult Cancer Survivors After Treatment: A Systematic Review and Meta-Analysis. Oncology Nursing Forum, 44(6), 719-728. <u>https://doi.org/http://dx.doi.org/10.1188/17.ONF.719-728</u>
- Koetsenruijter, J., van Lieshout, J., Vassilev, I., Portillo, M. C., Serrano, M., Knutsen, I., Roukova, P., Lionis, C., Todorova, E., Foss, C., Rogers, A., & Wensing, M. (2014).
   Social support systems as determinants of self-management and quality of life of people with diabetes across Europe: study protocol for an observational study. *Health and Quality of Life Outcomes*, *12*(1), e29. <u>https://doi.org/10.1186/1477-7525-12-29</u>
- Kuipers, S. J., Cramm, J. M., & Nieboer, A. P. (2019). The importance of patient-centered care and co-creation of care for satisfaction with care and physical and social wellbeing of patients with multi-morbidity in the primary care setting. *BMC Health Services Research*, e19(1), e13. <u>https://doi.org/10.1186/s12913-018-3818-y</u>
- Ladwa, R., Pinkham, E. P., Teleni, L., Hanley, B., Lock, G., Nixon, J., Agbejule, O. A., Crawford-Williams, F., Jones, L., Pinkham, M. B., Turner, J., Yates, P., McPhail, S. M., Aitken, J. F., Escalante, C. P., Hart, N. H., & Chan, R. J. (2022). Telehealth cancer-related fatigue clinic model for cancer survivors: a pilot randomised controlled trial protocol (the T-CRF trial). *BMJ Open*, *12*(5), e059952. <u>https://doi.org/10.1136/bmjopen-2021-059952</u>
- Lake, A. J., & Staiger, P. K. (2010). Seeking the views of health professionals on translating chronic disease self-management models into practice. *Patient Education and Counseling*, *79*(1), 62-68. <u>https://doi.org/10.1016/j.pec.2009.07.036</u>
- Lam, K. K. W., Li, W. H. C., Chung, O. K., Ho, K. Y., Chiu, S. Y., Lam, H. S., & Chan, G. C. F. (2018). An integrated experiential training programme with coaching to promote physical activity, and reduce fatigue among children with cancer: a randomised

controlled trial. *Patient Education and Counseling*. <u>https://doi.org/10.1016/j.pec.2018.07.008</u>

- Lawn, S., & Battersby, M. (2009). Capabilities for supporting prevention and chronic condition self-management: a resource for educators of primary health care professionals. Flinders University, Adelaide: Australian Government Department of Health and Ageing.
- Lawn, S., & Lawton, K. (2011). Chronic condition self-management support within a respiratory nursing service. *Journal of Nursing and Healthcare of Chronic Illness*, 3(4), 372-380. <u>https://doi.org/10.1111/j.1752-9824.2011.01109.x</u>
- Lee, M. K., Yun, Y. H., Park, H.-A., Lee, E. S., Jung, K. H., & Noh, D.-Y. (2014). A Webbased self-management exercise and diet intervention for breast cancer survivors: Pilot randomized controlled trial. *International Journal of Nursing Studies*, *51*(12), 1557-1567. <u>https://doi.org/10.1016/j.ijnurstu.2014.04.012</u>
- Li, W. H., Ho, K. Y., Lam, K. K., Lam, H. S., Chui, S. Y., Chan, G. C., Cheung, A. T., Ho, L. L., & Chung, O. K. (2018). Adventure-based training to promote physical activity and reduce fatigue among childhood cancer survivors: a randomized controlled trial. *International Journal of Nursing Studies*, 83, 65-74. https://doi.org/10.1016/j.ijnurstu.2018.04.007
- Linstone, H. A., & Turoff, M. (2011). Delphi: A brief look backward and forward. *Technological Forecasting and Social Change*, *78*(9), 1712-1719. <u>https://doi.org/10.1016/j.techfore.2010.09.011</u>
- Lis, C. G., Rodeghier, M., Grutsch, J. F., & Gupta, D. (2009). Distribution and determinants of patient satisfaction in oncology with a focus on health related quality of life. *BMC Health Services Research*, 9, e190. <u>https://doi.org/10.1186/1472-6963-9-190</u>
- Lisy, K., Langdon, L., Piper, A., & Jefford, M. (2019). Identifying the most prevalent unmet needs of cancer survivors in Australia: A systematic review. Asia-Pacific Journal of *Clinical Oncology*, 15(5), e68-e78. <u>https://doi.org/10.1111/ajco.13176</u>
- Lorig, K. R., & Holman, H. (2003). Self-management education: history, definition, outcomes, and mechanisms. *Annals of Behavioral Medicine*, *26*(1), 1-7. https://doi.org/10.1207/s15324796abm2601\_01
- Luthy, C., Cedraschi, C., Pugliesi, A., Di Silvestro, K., Mugnier-Konrad, B., Rapiti, E., & Allaz, A.-F. (2011). Patients' views about causes and preferences for the management of cancer-related fatigue—a case for non-congruence with the physicians? *Supportive Care in Cancer*, *19*(3), 363-370. https://doi.org/10.1007/s00520-010-0826-9
- Matthews, T., & Trenoweth, S. (2015). Nurses' perceptions of self-management in renal care. *British Journal of Nursing*, *24*(19), 956-961. https://doi.org/10.12968/bjon.2015.24.19.956
- Maynard, D., & Clayman, S. (2003). Ethnomethodology and conversation analysis.
- Miller, K. D., Nogueira, L., Mariotto, A. B., Rowland, J. H., Yabroff, K. R., Alfano, C. M., Jemal, A., Kramer, J. L., & Siegel, R. L. (2019). Cancer treatment and survivorship statistics, 2019. CA: A Cancer Journal for Clinicians, 69(5), 363-385. https://doi.org/10.3322/caac.21565
- Mills, S. L., Brady, T. J., Jayanthan, J., Ziabakhsh, S., & Sargious, P. M. (2017). Toward consensus on self-management support: the international chronic condition selfmanagement support framework†. *Health Promotion International*, 32(6), 942-952. <u>https://doi.org/10.1093/heapro/daw030</u>
- Minton, O., Berger, A., Barsevick, A., Cramp, F., Goedendorp, M., Mitchell, S. A., & Stone, P. C. (2013). Cancer-related fatigue and its impact on functioning. *119*(S11), 2124-2130. <u>https://doi.org/10.1002/cncr.28058</u>
- Molassiotis, A., Yates, P., Li, Q., So, W. K. W., Pongthavornkamol, K., Pittayapan, P., Komatsu, H., Thandar, M., Yi, M., Titus Chacko, S., Lopez, V., Butcon, J., Wyld, D., Chan, R. J., Doolan, M., Litam, M. E., Onofre, R., Lluch, C., Nacion, R., . . . Xu, Y. (2017). Mapping unmet supportive care needs, quality-of-life perceptions and current symptoms in cancer survivors across the Asia-Pacific region: results from the

International STEP Study. *Annals of Oncology*, *28*(10), 2552-2558. <u>https://doi.org/10.1093/annonc/mdx350</u>

- Mooney, K. H., Beck, S. L., Wong, B., Dunson, W., Wujcik, D., Whisenant, M., & Donaldson, G. (2017). Automated home monitoring and management of patient-reported symptoms during chemotherapy: results of the symptom care at home RCT. *Cancer Medicine*, 6(3), 537-546. <u>https://doi.org/10.1002/cam4.1002</u>
- Mudge, S., Kayes, N., & McPherson, K. (2015). Who is in control? Clinicians' view on their role in self-management approaches: a qualitative metasynthesis. *BMJ Open*, 5(5), e007413. <u>https://doi.org/10.1136/bmjopen-2014-007413</u>
- Murphy, M. K., Black, N. A., Lamping, D. L., McKee, C. M., Sanderson, C. F., Askham, J., & Marteau, T. (1998). Consensus development methods, and their use in clinical guideline development. *Health Technology Assessment*, *2*(3), 1-88.
- Mustian, K. M., Alfano, C. M., Heckler, C., Kleckner, A. S., Kleckner, I. R., Leach, C. R., Mohr, D., Palesh, O. G., Peppone, L. J., Piper, B. F., Scarpato, J., Smith, T., Sprod, L. K., & Miller, S. M. (2017). Comparison of Pharmaceutical, Psychological, and Exercise Treatments for Cancer-Related Fatigue: A Meta-analysis. *JAMA Oncology*, *3*(7), 961-968. https://doi.org/10.1001/jamaoncol.2016.6914
- Náfrádi, L., Nakamoto, K., & Schulz, P. J. (2017). Is patient empowerment the key to promote adherence? A systematic review of the relationship between self-efficacy, health locus of control and medication adherence. *PLOS ONE*, *12*(10), e0186458. https://doi.org/10.1371/journal.pone.0186458
- Naraphong, W. (2015). Effects of a Culturally Sensitive Exercise Program on Fatigue, Sleep, Mood, and Symptom Distress among Thai Women with Breast Cancer Receiving Adjuvant Chemotherapy: A Pilot Randomized Controlled Trial. University of Cincinnati].
- Narayanan, V., & Koshy, C. (2009). Fatigue in cancer: a review of literature. *Indian Journal* of Palliative Care, 15(1), 19-25. <u>https://doi.org/10.4103/0973-1075.53507</u>
- National Health Service. (2019). *NHS Long Term Plan*. <u>https://www.longtermplan.nhs.uk/wp-content/uploads/2019/08/nhs-long-term-plan-version-1.2.pdf</u>
- NCCN. (2017). Clinical Practice Guidelines in Oncology: Cancer-related fatigue. Version 2. https://www.nccn.org/professionals/physician\_gls/pdf/fatigue.pdf.
- Nichols, T., Calder, R., Morgan, M., Lawn, S., Beauchamp, A., Trezona A, Byambasuren, O., Bowman, J., Clinton-McHargT, Willis, K., Kearns, R., Harris-Roxas, B., Duggan M, Wardle J, Litt J, Menzies D, Dawda P, Benrimoj S, Dineen-Griffin S, . . . B, K.-P. (2020). Self-care for health: a national policy blueprint. Policy paper 2020-01. Mitchell Institute, Victoria University, Melbourne
- Norris, M., & Kilbride, C. (2014). From dictatorship to a reluctant democracy: stroke therapists talking about self-management. *Disability and Rehabilitation*, *36*(1), 32-38. <u>https://doi.org/10.3109/09638288.2013.776645</u>
- O'Leary, A. (1992). Self-efficacy and health: Behavioral and stress-physiological mediation. *Cognitive Therapy and Research*, *16*(2), 229-245. <u>https://doi.org/10.1007/BF01173490</u>
- O'Connell, S., Mc Carthy, V. J. C., & Savage, E. (2018). Frameworks for self-management support for chronic disease: a cross-country comparative document analysis. *BMC Health Services Research*, *18*(1), 583. <u>https://doi.org/10.1186/s12913-018-3387-0</u>
- Page, M. J., McKenzie, J. E., Bossuyt, P. M., Boutron, I., Hoffmann, T. C., Mulrow, C. D., Shamseer, L., Tetzlaff, J. M., Akl, E. A., Brennan, S. E., Chou, R., Glanville, J., Grimshaw, J. M., Hróbjartsson, A., Lalu, M. M., Li, T., Loder, E. W., Mayo-Wilson, E., McDonald, S., . . . Moher, D. (2021). The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. *BMJ*, *372*, e71. <u>https://doi.org/10.1136/bmj.n71</u>
- Panagioti, M., Richardson, G., Small, N., Murray, E., Rogers, A., Kennedy, A., Newman, S., & Bower, P. (2014). Self-management support interventions to reduce health care utilisation without compromising outcomes: a systematic review and meta-analysis. BMC Health Services Research, 14, e356. <u>https://doi.org/10.1186/1472-6963-14-356</u>

Patterson, C. H. (1985). *The therapeutic relationship: foundations for an eclectic psychotherapy*. Brooks/Cole.

- Pearce, G., Parke, H. L., Pinnock, H., Epiphaniou, E., Bourne, C. L., Sheikh, A., & Taylor, S. J. (2016). The PRISMS taxonomy of self-management support: derivation of a novel taxonomy and initial testing of its utility. *Journal of Health Services Research & Policy*, *21*(2), 73-82. <u>https://doi.org/10.1177/1355819615602725</u>
- Pearson, E., Morris, M., & McKinstry, C. (2017). Cancer related fatigue: Implementing guidelines for optimal management. *BMC Health Services Research*, 17. <u>https://doi.org/10.1186/s12913-017-2415-9</u>
- Pearson, E. J., Morris, M. E., & McKinstry, C. E. (2015). Cancer-related fatigue: a survey of health practitioner knowledge and practice. *Supportive Care in Cancer*, 23(12), 3521-3529. <u>https://doi.org/10.1007/s00520-015-2723-8</u>
- Peckitt, K., & Smart, C. (2018). Conversation Analysis of Psychological Formulation Discussions in Adult Learning Disabilities Teams. In C. Smart & T. Auburn (Eds.), Interprofessional Care and Mental Health: A Discursive Exploration of Team Meeting Practices (pp. 147-167). Springer International Publishing. <u>https://doi.org/10.1007/978-3-319-98228-1\_7</u>
- Peräkylä, A. (2004). Reliability and validity in research based on naturally occurring social interaction. *Qualitative Research: Theory, Method, and Practice*, 2, 283-304.
- Pertl, M. M., Quigley, J., & Hevey, D. (2014). 'I'm not complaining because I'm alive': barriers to the emergence of a discourse of cancer-related fatigue. *Psychol Health*, 29(2), 141-161. <u>https://doi.org/10.1080/08870446.2013.839792</u>
- Peters, M., Potter, C. M., Kelly, L., & Fitzpatrick, R. (2019). Self-efficacy and health-related quality of life: a cross-sectional study of primary care patients with multi-morbidity. *Health and Quality of Life Outcomes*, *17*(1), 37. <u>https://doi.org/10.1186/s12955-019-1103-3</u>
- Petrova, M., Dale, J., & Fulford, B. (2006). Values-based practice in primary care: easing the tensions between individual values, ethical principles and best evidence. *British Journal of General Practice*, *56*(530), 703.
- Pilnick, A., Hindmarsh, J., & Gill, V. T. (2009). Beyond 'doctor and patient': developments in the study of healthcare interactions. *Sociology of Health & Illness*, *31*(6), 787-802. https://doi.org/10.1111/j.1467-9566.2009.01194.x
- Pomerantz, A. M. (1975). Second Assessments: a study of some features of agreements/disagreements [Ph.D., University of California]. ProQuest Dissertations & Theses Global. Ann Arbor.
- Potter, J., & Hepburn, A. (2012). The SAGE Handbook of Interview Research: The Complexity of the Craft. In (2 ed.). SAGE Publications, Inc. https://doi.org/10.4135/9781452218403
- Puddy, R. W., & Wilkins, N. (2011). Understanding Evidence: A Guide to the Continuum of Evidence of Effectiveness. Best available research evidence. National Center for Injury Prevention and Control, Division of Violence Prevention.
- Purcell, A., Fleming, J., Burmeister, B., Bennett, S., & Haines, T. (2011). Is education an effective management strategy for reducing cancer-related fatigue? *Supportive Care in Cancer*, *19*(9), 1429-1439. <u>https://doi.org/10.1007/s00520-010-0970-2</u>
- Ramos, D., Arezes, P., & Afonso, P. (2016). Application of the Delphi Method for the inclusion of externalities in occupational safety and health analysis. *DYNA*, *83*, 14-20. <u>https://doi.org/10.15446/dyna.v83n196.56603</u>
- Raymond, G. (2004). Prompting Action: The Stand-Alone "So" in Ordinary Conversation. Research on Language and Social Interaction, 37(2), 185-218. https://doi.org/10.1207/s15327973rlsi3702\_4
- Ream, E., Gargaro, G., Barsevick, A., & Richardson, A. (2015). Management of cancerrelated fatigue during chemotherapy through telephone motivational interviewing: modeling and randomized exploratory trial. *Patient Education and Counseling*, *98*(2), 199-206. <u>https://doi.org/10.1016/j.pec.2014.10.012</u>

- Ream, E., Richardson, A., & Alexander-Dann, C. (2006). Supportive Intervention for Fatigue in Patients Undergoing Chemotherapy: A Randomized Controlled Trial. *Journal of Pain and Symptom Management*, *31*(2), 148-161. https://doi.org/10.1016/j.jpainsymman.2005.07.003
- Reif, K., de Vries, U., Petermann, F., & Görres, S. (2013). A patient education program is effective in reducing cancer-related fatigue: a multi-centre randomised two-group waiting-list controlled intervention trial. *European Journal of Oncology Nursing*, 17(2), 204-213. <u>https://doi.org/10.1016/j.ejon.2012.07.002</u>
- Ren, L.-I., Tian, X.-b., He, Z.-c., Song, E.-h., & Tang, T.-t. (2022). Cancer-related fatigue in hospitalised patients treated for lymphoma and its burden on family caregivers. *European Journal of Cancer Care*, 31(1), e13547. https://doi.org/10.1111/ecc.13547
- Rimmer, B., Sharp, L., & on behalf of Ways Ahead study, t. (2021). Implementation of Self-Management Interventions in Cancer Survivors: Why Are We Not There Yet? *Journal* of Cancer Education. <u>https://doi.org/10.1007/s13187-021-02021-2</u>
- Rimmer, B., Sharp, L., & on behalf of Ways Ahead study team. (2021). Implementation of Self-Management Interventions in Cancer Survivors: Why Are We Not There Yet? *Journal of Cancer Education*, *36*(6), 1355-1358. <u>https://doi.org/10.1007/s13187-021-02021-2</u>
- Ritterband, L. M., Bailey, E. T., Thorndike, F. P., Lord, H. R., Farrell-Carnahan, L., & Baum, L. D. (2012). Initial evaluation of an Internet intervention to improve the sleep of cancer survivors with insomnia. *Psychooncology*, *21*(7), 695-705. https://doi.org/10.1002/pon.1969
- Robinson, J. D. (2003). An Interactional Structure of Medical Activities During Acute Visits and Its Implications for Patients' Participation. *Health Communication*, *15*(1), 27-59. https://doi.org/10.1207/S15327027HC1501\_2
- Robinson, J. D. (2006). Soliciting patients' presenting concerns. In D. W. Maynard & J. Heritage (Eds.), *Communication in Medical Care: Interaction between Primary Care Physicians and Patients* (pp. 22-47). Cambridge University Press. https://doi.org/10.1017/CBO9780511607172.004
- Robinson, J. D., Tate, A., & Heritage, J. (2016). Agenda-setting revisited: When and how do primary-care physicians solicit patients' additional concerns? *Patient Education and Counseling*, *99*(5), 718-723. <u>https://doi.org/10.1016/j.pec.2015.12.009</u>
- Rodríguez Antolín, A., Martínez-Piñeiro, L., Jiménez Romero, M. E., García Ramos, J. B., López Bellido, D., Muñoz del Toro, J., García García-Porrero, A., & Gómez Veiga, F. (2019). Prevalence of fatigue and impact on quality of life in castration-resistant prostate cancer patients: the VITAL study *BMC Urology*, *19*(1), 92. <u>https://doi.org/10.1186/s12894-019-0527-8</u>
- Rodriguez, M. A., & Foxhall, L. E. (2018). *Handbook of Cancer Survivorship Care*. Springer Publishing Company.
- Rogers, A., Vassilev, I., Sanders, C., Kirk, S., Chew-Graham, C., Kennedy, A., Protheroe, J., Bower, P., Blickem, C., Reeves, D., Kapadia, D., Brooks, H., Fullwood, C., & Richardson, G. (2011). Social networks, work and network-based resources for the management of long-term conditions: a framework and study protocol for developing self-care support. *Implementation Science*, *6*(1), 56. <u>https://doi.org/10.1186/1748-5908-6-56</u>
- Ryan, P., & Sawin, K. J. (2009). The Individual and Family Self-Management Theory: background and perspectives on context, process, and outcomes. *Nursing Outlook*, 57(4), 217-225.e216. <u>https://doi.org/10.1016/j.outlook.2008.10.004</u>
- Sacks, H. (1984). Notes on Methodology. S. 21-27 in: J. Maxwell Atkinson/John Heritage (Hrsg.), Structures of Social Action. In: Cambridge: Cambridge University Press.
- Sacks, H., Schegloff, E., & Jefferson, G. (1974). A Simple Systematic for the Organisation of Turn Taking in Conversation. *Language*, *50*, 696-735. <u>https://doi.org/10.2307/412243</u>
- Saito, M., Hiramoto, I., Yano, M., Watanabe, A., & Kodama, H. (2022). Influence of Self-Efficacy on Cancer-Related Fatigue and Health-Related Quality of Life in Young

Survivors of Childhood Cancer. International Journal Environmental Research Public Health, 19(3). https://doi.org/10.3390/ijerph19031467

- Sandler, C. X., Goldstein, D., Horsfield, S., Bennett, B. K., Friedlander, M., Bastick, P. A., Lewis, C. R., Segelov, E., Boyle, F. M., Chin, M. T. M., Webber, K., Barry, B. K., & Lloyd, A. R. (2017). Randomized Evaluation of Cognitive-Behavioral Therapy and Graded Exercise Therapy for Post-Cancer Fatigue. *Journal of Pain and Symptom Management*, 54(1), 74-84. https://doi.org/10.1016/j.jpainsymman.2017.03.015
- Sarkar, U., Ali, S., & Whooley, M. A. (2007). Self-efficacy and health status in patients with coronary heart disease: findings from the heart and soul study. *Psychosomatic medicine*, 69(4), 306-312. https://doi.org/10.1097/PSY.0b013e3180514d57
- Savage, J. (2009). Models of care for chronic disease. In *Models of Access and Clinical* Service Delivery Project. Australasian Society for HIV Medicine.
- Savard, J., Simard, S., Ivers, H., & Morin, C. M. (2005). Randomized study on the efficacy of cognitive-behavioral therapy for insomnia secondary to breast cancer, part I: Sleep and psychological effects. *Journal of Clinical Oncology*, *23*(25), 6083-6096. https://doi.org/10.1200/jco.2005.09.548
- Schegloff, E. A. (1996). Confirming Allusions: Toward an Empirical Account of Action. American Journal of Sociology, 102(1), 161-216. https://doi.org/10.1086/230911
- Schegloff, E. A. (1999). Discourse, Pragmatics, Conversation, Analysis. *Discourse Studies*, 1(4), 405-435. <u>https://doi.org/10.1177/1461445699001004002</u>
- Schjolberg, T. K., Dodd, M., Henriksen, N., Asplund, K., Cvancarova Småstuen, M., & Rustoen, T. (2014). Effects of an educational intervention for managing fatigue in women with early stage breast cancer. *European Journal of Oncology Nursing*, *18*(3), 286-294. <u>https://doi.org/10.1016/j.ejon.2014.01.008</u>
- Schmidt, M. E., Bergbold, S., Hermann, S., & Steindorf, K. (2021). Knowledge, perceptions, and management of cancer-related fatigue: the patients' perspective. *Supportive Care in Cancer*, 29(4), 2063-2071. <u>https://doi.org/10.1007/s00520-020-05686-5</u>
- Schulman-Green, D., Jaser, S., Martin, F., Alonzo, A., Grey, M., McCorkle, R., Redeker, N. S., Reynolds, N., & Whittemore, R. (2012). Processes of Self-Management in Chronic Illness. *Journal of Nursing Scholarship*, 44(2), 136-144. <u>https://doi.org/10.1111/j.1547-5069.2012.01444.x</u>
- Servaes, P., Verhagen, C., & Bleijenberg, G. (2002). Fatigue in cancer patients during and after treatment: prevalence, correlates and interventions. *European Journal of Cancer Care*, *38*(1), 27-43. <u>https://doi.org/10.1016/s0959-8049(01)00332-x</u>
- Sheridan, N., Kenealy, T., Kuluski, K., McKillop, A., Parsons, J., & McKillop, C. A. (2017). Are patient and carer experiences mirrored in the Practice Reviews of Self-Management Support (PRISMS) provider taxonomy? *International Journal of Integrated Care*, *17*(2). <u>https://doi.org/10.5334/ijic.3180</u>
- Sidnell, J. (2012). Basic Conversation Analytic Methods. In *The Handbook of Conversation Analysis* (pp. 77-99). <u>https://doi.org/10.1002/9781118325001.ch5</u>
- Slev, V. N., Pasman, H. R. W., Eeltink, C. M., van Uden-Kraan, C. F., Verdonck-de Leeuw, I. M., & Francke, A. L. (2017). Self-management support and eHealth for patients and informal caregivers confronted with advanced cancer: an online focus group study among nurses. *BMC Palliative Care*, *16*(1), 55. <u>https://doi.org/10.1186/s12904-017-0238-4</u>
- Smith-Turchyn, J., Morgan, A., & Richardson, J. (2016). The Effectiveness of Group-based Self-management Programmes to Improve Physical and Psychological Outcomes in Patients with Cancer: a Systematic Review and Meta-analysis of Randomised Controlled Trials. *Clincal Oncology (Royal College of Radiologists)*, 28(5), 292-305. <u>https://doi.org/10.1016/j.clon.2015.10.003</u>
- Smith, S. K., MacDermott, K., Amarasekara, S., Pan, W., Mayer, D., & Hockenberry, M. (2019). Reimagine: a randomized controlled trial of an online, symptom selfmanagement curriculum among breast cancer survivors. *Supportive Care in Cancer*, 27(5), 1775-1781. <u>https://doi.org/10.1007/s00520-018-4431-7</u>

- Sood, A., Barton, D. L., Bauer, B. A., & Loprinzi, C. L. (2007). A Critical Review of Complementary Therapies for Cancer-Related Fatigue. *Integrative Cancer Therapies*, *6*(1), 8-13. <u>https://doi.org/10.1177/1534735406298143</u>
- Spahn, G., Choi, K.-E., Kennemann, C., Lüdtke, R., Franken, U., Langhorst, J., Paul, A., & Dobos, G. J. (2013). Can a Multimodal Mind–Body Program Enhance the Treatment Effects of Physical Activity in Breast Cancer Survivors With Chronic Tumor-Associated Fatigue? A Randomized Controlled Trial. *Integrative Cancer Therapies*, 12(4), 291-300. <u>https://doi.org/10.1177/1534735413492</u>
- Stanley, T., Baron, S., & Robertson, P. (2021). Examining Practice Frameworks Mapping out the Gains. *Practice*, 33(1), 21-35. https://doi.org/10.1080/09503153.2020.1713312
- Stanton, A. L., Ganz, P. A., Kwan, L., Meyerowitz, B. E., Bower, J. E., Krupnick, J. L., Rowland, J. H., Leedham, B., & Belin, T. R. (2005). Outcomes From the Moving Beyond Cancer Psychoeducational, Randomized, Controlled Trial With Breast Cancer Patients. *Journal of Clinical Oncology*, *23*(25), 6009-6018. https://doi.org/10.1200/jco.2005.09.101
- Steel, J. L., Geller, D. A., Kim, K. H., Butterfield, L. H., Spring, M., Grady, J., Sun, W., Marsh, W., Antoni, M., Dew, M. A., Helgeson, V., Schulz, R., & Tsung, A. (2016). Web-based collaborative care intervention to manage cancer-related symptoms in the palliative care setting [Article]. *Cancer*, 122(8), 1270-1282. <u>https://doi.org/10.1002/cncr.29906</u>
- Sterne, J. A. C., Savović, J., Page, M. J., Elbers, R. G., Blencowe, N. S., Boutron, I., Cates, C. J., Cheng, H.-Y., Corbett, M. S., Eldridge, S. M., Emberson, J. R., Hernán, M. A., Hopewell, S., Hróbjartsson, A., Junqueira, D. R., Jüni, P., Kirkham, J. J., Lasserson, T., Li, T., . . . Higgins, J. P. T. (2019). RoB 2: a revised tool for assessing risk of bias in randomised trials. *BMJ*, *366*, el4898. <u>https://doi.org/10.1136/bmj.14898</u>
- Sterponi, L., Zucchermaglio, C., Fatigante, M., & Alby, F. (2019). Structuring times and activities in the oncology visit. *Social Science & Medicine*, *228*, 211-222. <u>https://doi.org/10.1016/j.socscimed.2019.03.036</u>
- Stivers, T., & Sidnell, J. (2012). *The Handbook of Conversation Analysis*. https://doi.org/10.1002/9781118325001.ch1
- Stone, P., Richardson, A., Ream, E., Smith, A. G., Kerr, D. J., & Kearney, N. (2000). Cancerrelated fatigue: inevitable, unimportant and untreatable? Results of a multi-centre patient survey. Cancer Fatigue Forum. *Annals of Oncology*, *11*(8), 971-975. <u>https://doi.org/10.1023/a:1008318932641</u>
- Stormacq, C., Wosinski, J., Boillat, E., & Van den Broucke, S. (2020). Effects of health literacy interventions on health-related outcomes in socioeconomically disadvantaged adults living in the community: a systematic review. *JBI Evidence Synthesis*, *18*(7). <u>https://doi.org/</u> 10.11124/JBISRIR-D-18-00023
- Tahmassian, K., & Jalali Moghadam, N. (2011). Relationship between self-efficacy and symptoms of anxiety, depression, worry and social avoidance in a normal sample of students. *Iranian Journal of Psychiatry and Behavioral Sciences*, *5*(2), 91-98. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3939966/
- Tan, F. C. J. H., Oka, P., Dambha-Miller, H., & Tan, N. C. (2021). The association between self-efficacy and self-care in essential hypertension: a systematic review. BMC Family Practice, 22(1), e44. <u>https://doi.org/10.1186/s12875-021-01391-2</u>
- ten Have, P. (2007). Doing Conversation Analysis. SAGE Publications.
- Thoits, P. A. (2011). Mechanisms Linking Social Ties and Support to Physical and Mental Health. *Journal of Health and Social Behavior*, *5*2(2), 145-161. https://doi.org/10.1177/0022146510395592
- Thornton, C. P., Li, M., Yeh, C. H., & Ruble, K. (2021). Self-efficacy in symptom management for adolescents and young adults with cancer: a systematic review. *Supportive Care in Cancer*, 29(6), 2851-2862. <u>https://doi.org/10.1007/s00520-020-05960-6</u>

- Toerien, M., Reuber, M., Shaw, R., & Duncan, R. (2018). Generating the perception of choice: the remarkable malleability of option-listing. *Sociology of Health & Illness*, 40(7), 1250-1267. <u>https://doi.org/10.1111/1467-9566.12766</u>
- Tomlinson, D., Robinson, P. D., Oberoi, S., Cataudella, D., Culos-Reed, N., Davis, H., Duong, N., Gibson, F., Götte, M., Hinds, P., Nijhof, S. L., van der Torre, P., Cabral, S., Dupuis, L. L., & Sung, L. (2018). Pharmacologic interventions for fatigue in cancer and transplantation: a meta-analysis. *Current Oncology*, 25(2), e152-e167. https://doi.org/10.3747/co.25.3883
- Trevelyan, E. G., & Robinson, P. N. (2015). Delphi methodology in health research: how to do it? *European Journal of Integrative Medicine*, *7*(4), 423-428. https://doi.org/10.1016/j.eujim.2015.07.002
- Turner, J., Yates, P., Kenny, L., Gordon, L. G., Burmeister, B., Thomson, D., Hughes, B., McCarthy, A. L., Perry, C., Chan, R. J., Fraser, A., Skerman, H., Batstone, M., & Carswell, K. (2014). The ENHANCES study—Enhancing Head and Neck Cancer patients' Experiences of Survivorship: study protocol for a randomized controlled trial. *Trials*, 15(1), 191. <u>https://doi.org/10.1186/1745-6215-15-191</u>
- Vallerand, J. R., Rhodes, R. E., Walker, G. J., & Courneya, K. S. (2018). Feasibility and preliminary efficacy of an exercise telephone counseling intervention for hematologic cancer survivors: a phase II randomized controlled trial. *Journal of Cancer Survivorship*, 12(3), 357-370. <u>https://doi.org/10.1007/s11764-018-0675-y</u>
- van der Lee, M. L., & Garssen, B. (2012). Mindfulness-based cognitive therapy reduces chronic cancer-related fatigue: a treatment study. *Psycho-Oncology*, *21*(3), 264-272. <u>https://doi.org/10.1002/pon.1890</u>
- van Vugt, M., de Wit, M., Cleijne, W. H., & Snoek, F. J. (2013). Use of behavioral change techniques in web-based self-management programs for type 2 diabetes patients: systematic review. *Journal of Medical Internet Research*, *15*(12), e279-e279. <u>https://doi.org/10.2196/jmir.2800</u>
- van Waart, H., Stuiver, M. M., van Harten, W. H., Geleijn, E., Kieffer, J. M., Buffart, L. M., de Maaker-Berkhof, M., Boven, E., Schrama, J., Geenen, M. M., Meerum Terwogt, J. M., van Bochove, A., Lustig, V., van den Heiligenberg, S. M., Smorenburg, C. H., Hellendoorn-van Vreeswijk, J. A., Sonke, G. S., & Aaronson, N. K. (2015). Effect of Low-Intensity Physical Activity and Moderate- to High-Intensity Physical Exercise During Adjuvant Chemotherapy on Physical Fitness, Fatigue, and Chemotherapy Completion Rates: Results of the PACES Randomized Clinical Trial. *Journal of Clinical Oncology*, *33*(17), 1918-1927. https://doi.org/10.1200/jco.2014.59.1081
- van Weert, E., May, A. M., Korstjens, I., Post, W. J., van der Schans, C. P., van den Borne, B., Mesters, I., Ros, W. J. G., & Hoekstra-Weebers, J. E. H. M. (2010). Cancer-Related Fatigue and Rehabilitation: A Randomized Controlled Multicenter Trial Comparing Physical Training Combined With Cognitive-Behavioral Therapy With Physical Training Only and With No Intervention. *Physical Therapy*, *90*(10), 1413-1425. <u>https://doi.org/10.2522/ptj.20090212</u>
- Vargas, S., Antoni, M. H., Carver, C. S., Lechner, S. C., Wohlgemuth, W., Llabre, M., Blomberg, B. B., Glück, S., & DerHagopian, R. P. (2014). Sleep quality and fatigue after a stress management intervention for women with early-stage breast cancer in southern Florida. *International Journal of Behavioral Medicine*, 21(6), 971-981. <u>https://doi.org/10.1007/s12529-013-9374-2</u>
- Vassilev, I., Rogers, A., Sanders, C., Kennedy, A., Blickem, C., Protheroe, J., Bower, P., Kirk, S., Chew-Graham, C., & Morris, R. (2010). Social networks, social capital and chronic illness self-management: a realist review. *Chronic Illness*, 7(1), 60-86. <u>https://doi.org/10.1177/1742395310383338</u>
- Vogelzang, N. J., Breitbart, W., Cella, D., Curt, G. A., Groopman, J. E., Horning, S. J., Itri, L. M., Johnson, D. H., Scherr, S. L., & Portenoy, R. K. (1997). Patient, caregiver, and oncologist perceptions of cancer-related fatigue: results of a tripart assessment survey. The Fatigue Coalition. *Seminars in Hematology*, *34*(3 Suppl 2), 4-12.

- Wagner, E. H., Austin, B. T., & Von Korff, M. (1996). Organizing care for patients with chronic illness. *Milbank Quarterly*, 74(4), 511-544.
- Wang, T., Molassiotis, A., Chung, B. P. M., & Tan, J.-Y. (2018). Unmet care needs of advanced cancer patients and their informal caregivers: a systematic review. *BMC Palliative Care*, 17(1), e96. <u>https://doi.org/10.1186/s12904-018-0346-9</u>
- Wang, Y. J., Boehmke, M., Wu, Y. W., Dickerson, S. S., & Fisher, N. (2011). Effects of a 6week walking program on Taiwanese women newly diagnosed with early-stage breast cancer. *Cancer Nursing*, *34*(2), E1-13. <u>https://doi.org/10.1097/NCC.0b013e3181e4588d</u>
- Wangnum, K., Thanarojanawanich, T., Chinwatanachai, K., Jamprasert, L., Maleehuan, O., & Janthakun, V. (2013). Impact of the multidisciplinary education program in self-care on fatigue in lung cancer patients receiving chemotherapy. *Journal of the Medical Association of Thailand*, *96*(12), 1601-1608.
- Webb, T. L., Joseph, J., Yardley, L., & Michie, S. (2010). Using the internet to promote health behavior change: a systematic review and meta-analysis of the impact of theoretical basis, use of behavior change techniques, and mode of delivery on efficacy. *Journal of Medical Internet Research*, 12(1), e4. https://doi.org/10.2196/jmir.1376
- Weis, J., & Horneber, M. (2014). Cancer-Related Fatigue. Springer Healthcare Iberica.
- White, L. L., Cohen, M. Z., Berger, A. M., Kupzyk, K. A., & Bierman, P. J. (2019). Self-Efficacy for Management of Symptoms and Symptom Distress in Adults With Cancer: An Integrative Review. Oncology Nursing Forum, 46(1), 113-128. <u>https://doi.org/10.1188/19.Onf.113-128</u>
- White, S. J. (2019). Conversation analysis: An introduction to methodology, data collection, and analysis. In *Handbook of Research Methods in Health Social Sciences* (pp. 471-490). <u>https://doi.org/10.1007/978-981-10-5251-4\_107</u>
- Willems, R. A., Bolman, C. A. W., Mesters, I., Kanera, I. M., Beaulen, A. A. J. M., & Lechner, L. (2017). Short-term effectiveness of a web-based tailored intervention for cancer survivors on quality of life, anxiety, depression, and fatigue: randomized controlled trial. *Psycho-Oncology*, 26(2), 222-230. https://doi.org/10.1002/pon.4113
- Willig, C., & Stainton Rogers, W. (2017). Conversation Analysis. In *The SAGE Handbook of Qualitative Research in Psychology*. SAGE Publications.
- Xu, A., Wang, Y., & Wu, X. H. (2019). Effectiveness of e-health based self-management to improve cancer-related fatigue, self-efficacy and quality of life in cancer patients: systematic review and meta-analysis. *Journal of Advanced nursing*, 75(12), 3434– 3447. https://doi.org/10.1111/jan.14197
- Yarcheski, A., Mahon, N. E., Yarcheski, T. J., & Cannella, B. L. (2004). A Meta-Analysis of Predictors of Positive Health Practices. *Journal of Nursing Scholarship*, *36*(2), 102-108. <u>https://doi.org/10.1111/j.1547-5069.2004.04021.x</u>
- Yates, P., Ar, a, S., Margraves, M., Mirolo, B., Clavarino, A., McLachlan, S., & Skerman, H. (2005). Randomized controlled trial of an educational intervention for managing fatigue in women receiving adjuvant chemotherapy for early-stage breast cancer. *Journal of Clinical Oncology*, *23*(25), 6027-6036. https://doi.org/10.1200/JCO.2005.01.271
- Yun, Y. H., Lee, K. S., Kim, Y. W., Park, S. Y., Lee, E. S., Noh, D. Y., Kim, S., Oh, J. H., Jung, S. Y., Chung, K. W., Lee, Y. J., Jeong, S. Y., Park, K. J., Shim, Y. M., Zo, J. I., Park, J. W., Kim, Y. A., Shon, E. J., & Park, S. (2012). Web-based tailored education program for disease-free cancer survivors with cancer-related fatigue: a randomized controlled trial. *Journal of Clinical Oncology*, *30*(12), 1296-1303. <u>https://doi.org/10.1200/jco.2011.37.2979</u>
- Zhang, Q., Li, F., Zhang, H., Yu, X., & Cong, Y. (2018). Effects of nurse-led home-based exercise & cognitive behavioral therapy on reducing cancer-related fatigue in patients with ovarian cancer during and after chemotherapy: A randomized controlled trial. *International Journal of Nursing Studies*, *78*, 52-60. https://doi.org/10.1016/j.ijnurstu.2017.08.010

- Zhang, S., Li, J., & Hu, X. (2022). Peer support interventions on quality of life, depression, anxiety, and self-efficacy among patients with cancer: A systematic review and metaanalysis. *Patient Education and Counseling*, *105*(11), 3213-3224. https://doi.org/10.1016/j.pec.2022.07.008
- Zhang, X., Meng, K., & Chen, S. (2020). Competency framework for specialist critical care nurses: A modified Delphi study. *Nursing in Critical Care*, *25*(1), 45-52. https://doi.org/10.1111/nicc.12467

# **APPENDICES**

# Appendix 1: A Systematic Scoping Review of Implementation of Interventions for Cancer-Related Fatigue Management

## Authors.

Oluwaseyifunmi Andi Agbejule BRadTherapy<sup>1</sup>\*, Nicolas H. Hart PhD<sup>1,3,4</sup>, Stuart Ekberg PhD<sup>1</sup>, Bogda Koczwara BMBS MBioethics<sup>5</sup>, Rahul Ladwa MBChB, BSc, MPhil<sup>2,6</sup>, Camilla Simonsen, RN BAppSci<sup>1,2</sup>, Elizabeth P Pinkham MPhty, BAppSci<sup>1,2,7</sup>, Raymond Javan Chan RN, PhD<sup>1,2</sup>

- 1. Cancer and Palliative Care Outcomes Centre, School of Nursing, Queensland University of Technology (QUT), Kelvin Grove, Queensland 4059, Australia
- 2. Princess Alexandra Hospital, Metro South Hospital and Health Services, Woolloongabba, Queensland 4102, Australia
- 3. Exercise Medicine Research Institute, Edith Cowan University, Perth, Western Australia, 6027 Australia
- 4. Institute for Health Research, University of Notre Dame Australia, Fremantle, Western Australia, 6959 Australia
- 5. Flinders University and Flinders Medical Centre, Flinders Drive, Bedford Park, South Australia 5048, Australia
- 6. School of Medicine, University of Queensland, St Lucia, Queensland 4072, Australia
- 7. School of Health and Behavioural Science, University of Queensland, St Lucia, Queensland 4072, Australia

\*Corresponding Author. Oluwaseyifunmi Andi Agbejule,

Address: Level 3, School of Nursing, N Block, Kelvin Grove Campus, Queensland University of

Technology, Queensland 4059, Australia.

Email: agbejulo@qut.edu.au

#### Title

# Bridging the Research to Practice Gap: A Systematic Scoping Review of Implementation of Cancer-Related Fatigue Management

#### Abstract

**Background.** Cancer-related fatigue (CRF) is one of the most common and distressing symptoms in people with cancer. Although efficacy of interventions for CRF have been extensively investigated, less has been done to ensure successful translation into routine clinical practice. The aim of this systematic scoping review was to synthesise knowledge surrounding the implementation of CRF interventions, summarise the processes and outcomes of implementation strategies used, and identify opportunities for further research.

**Methods.** PubMed, Cochrane CENTRAL, EMBASE and CINAHL databases were searched (up to December 2020). The Cochrane Effective Practice and Organisation of Care (EPOC) Group taxonomy and the RE-AIM Framework were used to guide the evaluation of implementation strategies and outcomes, respectively.

**Results.** Six studies were included. Three used an implementation framework (PARIHS, KTA, Cullens & Adams' Implementation Guide) to guide implementation. Overall, the implementation strategies used across all studies were reported to have directly resulted in immediate changes at the clinician level (e.g., increased clinician behaviours, self-efficacy, attitudes, knowledge of CRF management). No clear relationship was found between the use of implementation models and the number or type of implementation strategies used. For outcomes, Effectiveness and Implementation were the most highly reported RE-AIM measures followed by Reach then Maintenance. Adoption was the least reported.

**Conclusions.** Despite the high prevalence of CRF and evidence-based interventions for managing CRF, there is limited evidence informing the sustainable implementation of these interventions. This systematic scoping review emphasises the lack of quality CRF implementation studies presently available in the literature leading to a disconnect between effective CRF interventions, routine clinical care, and cancer survivors at present. This review highlights the need for robust study designs guided by established frameworks to methodically design and evaluate the implementation of CRF management interventions in the future.

#### Keywords

cancer-related fatigue; exercise; implementation science; oncology; physical activity; survivorship

#### 1.0 Background

Cancer-related fatigue (CRF) is experienced by over 60% of cancer survivors depending on their cancer diagnosis and associated treatments, with two-thirds reporting severe CRF extending beyond 6 months, and one-third reporting persistent CRF over many years [1]. While CRF is known as one of the most distressing and prevalent symptoms experienced by people with cancer [2, 3], it has no current universal definition, with the National Comprehensive Cancer Network (NCCN) describing it as "a persistent, subjective sense of physical, emotional and/or cognitive exhaustion related to cancer or cancer treatment that is not proportional to recent activity and interferes with usual functioning" [4]. CRF greatly diminishes patients' physical, mental, occupational, emotional and social wellbeing during and after treatment [2, 5, 6]. Other than reduced quality of life, CRF can also lead to difficulties in decision making, daily living disruption and an increased dependency on others [7]. Such impacts on quality of life (QoL) have been widely reported by a broad range of cancer survivor populations [4, 6-13].

Various interventions for managing CRF have been investigated. These include physical activity and exercise (e.g., aerobic, resistance), pharmacological interventions (e.g., erythropoietin, methylphenidate, modafinil), psychological interventions (e.g., cognitive behaviour therapy), and integrative therapies (e.g., acupuncture, massage). There is level one evidence [14-18] supporting the benefits of physical activity, exercise and psychological interventions for managing CRF. In a recent meta-analysis, Mustian and colleagues [16] identified that exercise (weighted effect size [WES], 0.30; 95% CI, 0.25–0.36; *P* < .001) and psychological interventions (WES, 0.27; 95% CI, 0.21–0.33; *P* < .001) produced significant moderate positive effects on CRF improvement, with authors suggesting that both management strategies be prescribed as first line therapy. A plethora of research has focused on the efficacy of CRF interventions; however, there is much to learn about how these interventions can be incorporated into healthcare and 'real-world' settings.

While the discipline of implementation science is gaining momentum, less than half of interventions found to be effective in disease management and prevention are ever adopted into clinical use and routine practice [19-21]. Over recent years, cancer care and health service leaders are increasingly

concentrating their efforts on facilitating the systematic uptake of research findings into routine care to improve service and patient outcomes [22]. There are numerous systematic reviews on the clinical efficacy of CRF interventions; however, to our knowledge there is no comprehensive review focussing on the implementation of CRF management interventions and programs. To address this gap, this systematic scoping review of the CRF literature was conducted to answer the following key questions: (1) What current efforts have been made to implement CRF interventions in clinical care?; (2) What implementation frameworks, strategies, theories or models have been used when implementing CRF interventions in clinical care?; and (3) What were the outcomes of identified CRF interventions and implementation efforts?

#### 2.0 Methods

This review sought to examine implementation in CRF literature and identify possible knowledge gaps, thus a scoping methodology was adopted [23]. This systematic scoping review was conducted according to the Preferred Reporting Items for Systematic Reviews and Meta-Analyses Extension for Scoping Reviews (PRISMA-ScR) guidelines [24].

#### **Eligibility Criteria**

The population for this review were cancer survivors (regardless of age, gender, tumour and treatment type) at any stage of their cancer trajectory that have experienced fatigue as a result of their cancer or cancer treatment. The taxonomy of implementation strategies developed by the Cochrane Effective Practice and Organisation of Care (EPOC) Group [25] were used to determine the definition and inclusion of implementation studies in the review (see Additional File 1). These EPOC implementation strategies were developed for interventions that targeted and produced changes at the healthcare organisation level, healthcare professional level and the health service level and thus were considered relevant for this review.

For inclusion, studies were required to meet the following criteria: 1) have the implementation of an intervention/program/guideline as a primary goal; 2) have cancer-related fatigue as a primary symptom

of interest; 3) incorporate at least one of the EPOC implementation strategies; 4) be published in English; and 5) have full-text available.

No restrictions were placed on types of study designs eligible for inclusion. As the key interest of our systematic scoping review is to describe implementation outcomes of CRF management interventions, we included original research articles (i.e., randomised controlled trials, observational studies, qualitative studies, mixed methods, abstracts, program evaluations) and other grey literature (e.g., evaluations of modules, online programs and institutional/government interventions). Descriptive articles (i.e., commentaries, editorials, recommendation reports/articles) were excluded.

#### **Search Strategy**

Four databases (PubMed, Cochrane CENTRAL, EMBASE and CINAHL) were searched (up to December 2020) as outlined in Additional File 2. Free text terms and relevant subject headings (i.e., MeSH, EMTREE) for "cancer-related fatigue" (cancer fatigue, fatigue) and "interventions" were used. These terms were also combined with implementation study terms (e.g., "implementation", "translation", "program development", etc.) using the Boolean logic operators (or, and). Reference and citation lists of relevant articles were also hand searched for eligible studies that met the inclusion criteria. Titles and abstracts of articles retrieved from the search strategy were independently screened by two authors (CS, OAA). The same two authors then assessed the eligibility of relevant full-text articles for inclusion in the review. Disagreements were resolved through consensus among the two authors, with a third author (RC) as arbiter where required.

#### **Data Extraction**

Data extraction was conducted by one author (OAA) and checked for accuracy by a second author (RC). Key information extracted included: study characteristics (i.e., author, publication year, study design, purpose, participants, sample size); intervention characteristics (i.e., setting, context, model of care, resources used, intervention description); implementation framework or theory used; implementation strategies used; and implementation outcomes. A model or framework was considered specific to implementation if it described the process of translating research into practice, explained the influences of implementation outcomes, or evaluated implementation processes [26]. Implementation strategies were categorised using the components from the EPOC taxonomy (see Additional File 1).

The RE-AIM framework [27] was used to catalogue the outcomes of strategies, methods or techniques designed to change clinician or patient behaviours related to CRF. RE-AIM was initially developed to balance emphasis on internal and external validity and to expand on assessments of interventions beyond efficacy [28-30]. The RE-AIM dimensions include reach (R), effectiveness/efficacy (E), and maintenance (M)–which operate at the individual-level (i.e., rate or participation, intervention success rate, and endurance of individual behaviour respectively); and adoption (A), implementation (I), and maintenance (M), which focus on the organisation level [31]. In our review, data were extracted using a widely used [28, 30] RE-AIM coding sheet for systematic reviews published on the RE-AIM website (http://www.re-aim.org/resources-and-tools/measures-and-checklists/ - Additional File 3). Due to the heterogeneity of included studies, a narrative synthesis was conducted. As the purpose of this review was to provide an overview of existing evidence regardless of methodological quality or risk of bias, no quality assessment was conducted, consistent with the PRISMA-ScR guidelines.

#### 3.0 Results

Database searches resulted in 561 potentially eligible records. Of these, eight articles representing six implementation studies [32-37] met the inclusion criteria and were included in the review (See PRISMA Flow Chart: Figure 1).





From: Moher D, Liberati A, Tetzlaff J, Altman DG, The PRISMA Group (2009). Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS Med 6(7): e1000097. doi:10.1371/journal.pmed1000097

#### **Characteristics of Included Studies and Programs**

Included studies are described in Table 1. Of the six studies included, three were conducted in the USA [34, 36, 37], two in China [32, 35], and one in Canada [33]. Study design varied and consisted of one clinical audit implementation study [32], one qualitative program evaluation [37], three pre-test, posttest single group observation studies [33, 35, 36], and one longitudinal 3-group quasi-experimental comparative study [34]. Three studies [32, 33, 35] examined the impact of implementation efforts on the adoption of CRF evidenced-based guidelines, of which two [32, 33] reported the impact of implementation strategies at the organisational and health professional level, and one [35] described the impact of implementation efforts on oncology nurses and patients. The remaining three studies [34, 36, 37] investigated the efficacy of CRF interventions implemented at the patient level. Participants in each of the three efficacy studies [34, 36, 37] had mixed tumour types (i.e., breast, colon, lung, gastrointestinal, prostate, ovarian, uterine, myeloma, non-Hodgkin and Hodgkin lymphoma cancers). CRF interventions described in these three studies were physical activity and exercise [36], clinician and patient education [34], or combined exercise and education [37]. Of the two studies with education components [34, 37], only 'information giving' education strategies were described (e.g., information sessions, printed material). Types of physical activity described were aerobic exercise (e.g., walking, treadmill, Nu-Step), resistance training (e.g., resistance bands), balance and stretching, yoga and aquatic exercises [36, 37]. Of the three studies that implemented CRF interventions [34, 36, 37], two [34, 36] reported the impact of CRF interventions on patient fatigue outcomes (e.g., reduced fatigue).

#### **Implementation Models and Frameworks**

Only three of six studies were informed by an implementation framework or model. Huether and colleagues [36] utilised Cullens and Adams' Implementation Strategies for Best Practice Guide. Jones and colleagues [33] used the Knowledge to Action (KTA) implementation framework; and Tian and colleagues [35] used the Promoting Action on Research Implementation in Health Services (PARIHS) framework to guide dissemination and implementation of CRF guidelines into clinical practice.

#### **Implementation Strategies**

Distinct EPOC implementation strategies used by all included studies are recorded in Table 1.

#### Educational Meetings and Materials

Coaching and training strategies were the most used strategies (five of six studies) [32-36] to promote uptake of CRF interventions. Health professional education was delivered in the form of regular staff meetings, staff training sessions and workshops, formal presentations, unit in-services, role-play sessions (i.e., participation in mock fatigue assessments & patient education sessions) and print materials (i.e., 'pocket' fatigue guidelines & tip sheets, flipcharts, newsletters, education booklets). Education content across each study varied but generally included background information on fatigue, fatigue management, fatigue assessment procedures and tools, and referral processes.

#### Local Opinion Leaders and Stakeholder Engagement

Stakeholder engagement were used in five of six studies [32, 33, 35-37]. Stakeholder groups commonly included clinicians (i.e., psychiatrists, radiation and medical oncologists, physical therapists, surgeons, professors, specialist nurses), cancer survivors, research staff (i.e., research assistants, postdoctoral fellows), and community support professionals. The use of a key opinion leader was described in only one study [35], engaging a nurse who trained and persuaded fellow clinicians to accept implementation efforts.

#### Use of Clinical Guidelines and Local Consensus Processes

Three studies [36-38] focused efforts on implementing existing clinical fatigue guidelines, including the National Comprehensive Cancer Network (NCCN) Fatigue Guidelines [32, 34] and the Canadian Association of Psychosocial Oncology (CAPO) guidelines for CRF [33]. Tian and colleagues developed and implemented the Clinical Practice Guideline: Nursing Care of Cancer-Related Fatigue in Adults with Cancers [35], whereas Jones and colleagues specifically described the adaptation of guidelines to the local context after stakeholder consensus [33].

#### Audit and Feedback

Specific audit and feedback strategies were described in two studies [32, 35]. One study [35] described periodic audit completion rounds on nursing units, and routine discussions with staff to gather concerns, challenges, suggestions and distribute feedback; however, study authors did not report the specific details of audit content. The second study [32] described health professional training; fatigue screening at patient admission; fatigue screening at regular intervals throughout care; delivery of comprehensive fatigue assessments; and the provision of management strategies (i.e., physical activity and other strategies) to patients as components that were audited. In addition, interviews with patients and reviewed patient records were used to measure clinician compliance with audit components. A third study did [36] report the use of audit and feedback strategies, however they did not specify processes in further detail.

#### Managerial Supervision and Managing Performance

Managerial supervision and monitoring of performance strategies were utilised in two studies [32, 36] in the form of regular reporting to senior leaders, routine communication with nurses, and ongoing monitoring of staff to ensure program compliance. Wang and colleagues [32] reported that intervention leaders worked within the nursing unit to monitor nursing practice to ensure adequate nurse education on CRF, nurse provision of CRF assessment upon patient admission, and nurse delivery of patient education for CRF management strategies (i.e., physical activity, and other strategies). Unfortunately, Huether and colleagues [36] reported using managerial supervision and monitoring of performance strategies but did not provide specific detail on performance outcomes. '*Continuous Quality Improvement* 

Continuous quality improvement strategies were described by three studies [35-37] and generally involved protocol revisions and program modification based on clinician or patient feedback. Of these studies, the 'Energy Through Motion' CRF program [36] reported the formation of a dedicated quality improvement program; however, details of this quality improvement program were not provided.

#### Tailored Interventions

Four studies [32-35], described tailored interventions that were based on assessments of barriers to change. Wang and colleagues [32] conducted a clinical audit to determine CRF management barriers; Borneman and colleagues [34] identified barriers at the patient, professional and system level during the first phase of their quasi-experimental study; Tian and colleagues [35] identified barriers and facilitators through focus group discussions, surveys and observation; and Jones and colleagues [33] conducted semi-structured focus group interviews with stakeholder groups. Common barriers identified were lack of CRF knowledge, inconsistent application of CRF guidelines, insufficient knowledge of CRF screening and assessment, resistant attitudes towards program adoption, busy environments, heavy workloads, and time restrictions.

#### Reminders

Only one study [36] reported the regular use of practice reminders to reinforce the intervention to staff members. Reminders were distributed by the project leader through intervention 'tips of the week'; however, the distribution method used (e.g., newsletter, text, email) remains unclear.

#### Organisational Culture

Efforts to change organisational culture were reported across all studies [32-37] using practical methods including the formation of fatigue specific referral and clinical feedback systems; creation, and incorporation of CRF assessment flowcharts and assessment tools; addition of fatigue management processes to organisation protocol; and the development of CRF information documentation for both staff and patients.

 Table 1: Summary of Included Articles

Author, year [Program name], country	Study design /Setting/ Sample Size	Purpose	Participants	Models of Care	Resources Used/ Described		Intervention/ Program Description	Implementation Model/ Framework/ Theories
Abbott et al. 2017 [65] & Huether et al., 2016 [40] / Energy Through Motion ©]/Iowa United States	Pre-, post-test single group/ Ambulatory / (n=39) Pre-, post-test single group / 2 Survivorship clinics/ (n=50)	Feasibility (Abbott et al., 2017) Effectiveness (Huether et al., 2016)	Adult Cancer Survivors	Nurse-led survivorship clinics Home-based exercise	Activity trackers, Resistance Bands, Verbal instruction, Printed material (i.e., exercise & calorie guides), Pedometers, Activity logs, Bottle		3-month physical activity program that incorporated education, a specialised kit (including info on PA benefits, exercise equipment, sleep strategies logbook, home workouts) and ongoing patient support. The program also included an intricate text message system provide information and encouragement and promote adherence.	(Cullen & Adams, 2002) Implementation Strategies for Best Practice Guide
EPOC Implementation Strategies System/health professional level Reminders: update practice reminders provided by project leader to regularly reinforce the program. Local Opinion leaders: DNP student, clinical nurse specialist, nurse practitioners informed content and development. Continuous Quality Improvement: Senior leaders, existing and new team members were regularly updated on intervention progress. • Dedicated quality improvement program. Protocol revisions based on feedback from clinicians, patients, or family. Managerial supervision & Monitoring Performance: Regular reports to senior leaders. Interprofessional Education & Educational meetings: Presentations on evidence on CRF, physical activity and the Energy Through Motion program at staff meetings, unit in-services, Educational materials: pocket guides provided for clinicians						Outcomes         Effectiveness         • Results         • Fatigue: Decreased by an average of 2 points compared to an increase of 0.69 in usual care arm (p=0.0006)         • Activity Levels: Increased activity levels by a mean of 2.59 points (p=0.0016) compared to usual care (decreased levels by a mean of 1.07)         • QOL: Improved in all measured areas from 1.24-2.41 points (0-10 scale) compared to decrease (0.69 - 1.14 points) in usual care.         • Program evaluation: Participants reported that the program was helpful and beneficial. Connecting activity trackers to the computer and follow-along activity videos considered least helpful (Information videos also reported as not used regularly).         • Percent attrition: 10%         Implementation         • Completion rate: 90%         Maintenance (Setting Level)         Program still maintained at the University of Iowa Holden Comprehensive Cancer Centre		
Author, year [Program name], country	Study design /Setting/ Sample Size			Models of Care	De	urces Used/ escribed	Intervention/ Program Description	Implementation Model/ Framework/ Theories
--	--	--	---	---	---	---	--	--
Borneman et al., 2010 [66] & 2011 [38]/ [Passport to Comfort]/ California, United States	Quasi- experimental comparative study/Ambulat ory/ (Phase 1 [usual care] n=83, Phase 2 [intervention] n=104, Phase 3 [Disseminatio n] n=93)	Phase 1 &2: Effectiveness Phase 3: Dissemination	Patients with breast, colon, lung and prostate cancer (at least 1 month after diagnosis).	Nurse-led	consisti	g Packet ng of written onal materials.	Phase 1, 2: Psycho-educational intervention. Each patient received 4 (approx. 60min) educational sessions as well as written information material or 'tip' sheets in a 'teaching packet'. Tip sheets provided education on exercise, nutrition, emotional issues and sleep disturbance. During sessions, information on pain assessment, fatigue assessment and fatigue management was provided. A month after last educational session participants received bi- weekly follow-up phone calls every 2 weeks for 3 months.	None Described
Local Consens Comprehensive on NCCN pain Educational M presentations by to practitioners quarterly to gain Organisationa total of 38 in-see fatigue assessm supportive care into Spanish. • posters placed a Audit and feed and NPs based of	Professional Leve us Processes & C Cancer Network ( guidelines. eetings & Materi national experts ( for ongoing educa n clinician input fr l Culture (Phase rvices. • Pain and ent added to outpa departments for p Patient education round clinic to rer back: Clinical fee on chart audits wit rentions: Strategie	OC Implementation <u>i</u> <b>linical Guidelines:</b> (NCCN). A 'Patien <b>al:</b> Regular meetir to oncologists and r tion and communic om researchers inv <b>3):</b> Pain and fatigue fatigue information tient clinic vital sig ain and fatigue. • F materials made ava nind staff and patie baback reports comp h specific feedback es created to addres	Intervention inf t Pain Knowledg ngs with nurses. nurse practitioner cation. • Internal olved in the inter e education prov n provided at key gn flow sheet. • I Patient education iilable on employ ents to discuss fat pleted for patient c for pain and fat	<ul> <li>Pain and fatigue</li> <li>Pain and fatigue</li> <li>rs. Monthly new</li> <li>Advisory Board a</li> <li>rvention.</li> <li>ided to all clinicia</li> <li>meetings. Rou</li> <li>Roue</li> <li>Increased referrals</li> <li>materials were tra-</li> <li>ree Intranet. Ad</li> <li>Ad</li> <li>tigue. Increased to an and provided to a</li> </ul>	ed based evsletter met uns at a titine s to anslated vocacy MDs	Effectiveness Phas usua more Phas Fatig dem level QOI <u>Maintenance</u>	se 1 & 2: <b>Fatigue -</b> fatigue management barriers were sig al care group than in the intervention group. The usual care e fatigue (beta = -0.155). se 3: <b>Fatigue -</b> Significant immediate and sustained effe gue Barriers Scale (FBS) for the intervention group. The onstrated a significant delayed effect in Physical QOL – ls of QOL throughout the study when we would normally L. Statistically significant differences between QOL meas	re group had significantly cts were shown on the intervention group maintained baseline v expect a decrease in

Author, year [Program name], country	Study design /Setting/ Sample Size	Purpose	Participants	Models of Care		urces Used/ escribed	Intervention/ Program Description	Implementation Model/ Framework/ Theories
Jones et al., 2020 [37]/ Canada	Prospective Cohort/(n=18)	Implementation Intervention Acceptability and Feasibility	Healthcare professional and community support workers	N/A	summar	es, screening, essment	A one-time in person 2-hour training session offered to health care practitioners and community support providers about the CAPO CRF guidelines. First hour provided information on practice gaps reported in literature, CAPO CRF guidelines, communication skills and motivational interviewing principles. Second hour involved role-play and group discussions.	Knowledge to Action (KTA) Model
CRF used to inf Local Consense amongst stakeho Local opinion I with stakeholde professionals, p created. Educational ma	lines: Canadian A form education ses us Processes: Clin olders. eaders & Tailore r groups (patients, edagogy expert) to	OC Implementation ssociation of Psych sions. nical guidelines ada ed Intervention: For health care profess o identify barriers to cipants provided wi	osocial Oncolog pted to the Ottav ocus interviews a ionals and comr o change; subseq	wa context after co and program devel nunity support juent strategies the	onsensus lopment en	guid • CRI • Self- (d = • Self- (d = • Inte large	<u>on Impact</u> gram was effective in increasing knowledge, self-efficacy elines. <b>F Knowledge:</b> $-3.959(14)$ , p = 0.001) with a large effect <b>Efficacy in CRF Assessment</b> (t = 2.621(13), p = 0.021) 0.88). <b>efficacy to intervene for CRF</b> (t = 2.924(13), p = 0.012 1.13). <b>nt to apply Clinical Guidelines in Practice:</b> t = 4.786(1 e effect size (d = 1.35). <b>sibility:</b> mean satisfaction score (52.27 ± 6.97 out of 60 p <u>on</u>	size (d = 0.98). with a large effect size ) with a large effect size 3), $p = 0.000$ ) with a

Author, year [Program name], country	Study design /Setting/ Sample Size	Purpose	Participants	Models of Care		rces Used/ escribed	Intervention/ Program Description	Implementation Model/ Framework/ Theories
Tian et. al 2017 [39]/ China	Pre-test, post- test/ Radiotherapy Unit & Medical Oncology Unit/N/A	Implementation – Intervention Translating guidelines into practice	Nursing Staff	N/A	CRF edu booklet,	ality control	Study outlined the creation of a 'CRF Nursing Guideline' using a steering group (consisting of clinical experts). The resulting guidelines were implemented into practice through an evidenced- based project utilising training and education for nurses, changes to nursing procedures (screening and assessment and quality review) and the provision of staff resources. Impact of the project was measured at the organisational, staff and patient level.	The Promoting Action on Research Implementation in Health Services (PARIHS) framework
Continuous Qu periodically col procedure) was Educational m nursing staff. Educational m elementary train care. Seminars of Clinical Practio Nursing Care of interventionists. Local opinion I oncology, radio leader identified role model for f Tailored Interv	etings: Creation for a constraint of the constraints of the constraint	OC Implementation ent & Audit and For e whether further sp acation booklet and courses on CRF num pased nursing practi- practice concerning fratigue in Adults with of a steering group al Chinese Medicin negative opinion of o Integration of exist cus groups and discon n created.	eedback: Feedback: feedback: feedback: feedback: feedback: other training provide the standard specific to and specific to g CRF managem cesses: The "Cl th Cancer was d (six directors from e, Psychiatry de the project, train ting staff into fa	or modification (to rint materials give established, includ raining on CRF m nent. inical Practice Gu eveloped by om nursing, medic partments. • Opin n other nurses and cilitation team.	o nursing en to ling ursing ideline: cal nion act as a	beha • Patio	Outcomes on Impact se Outcomes: After implementation of the project, knowl aviour scores were all higher than at baseline. ent Outcomes: • No differences were detected between the baseline a "self-efficacy questionnaire for CRF management" ( • Patients adopted more effective CRF management st rested to alleviate fatigue) • CRF scores lower after intervention than prior to int 6.50 (1.90); t = 2.22, p = 0.04].	nd final scores of the (SQFM) scale. trategies (previously just

Author, year [Program name], country	Study design /Setting/ Sample Size	Purpose	Participants	Models of Care		rces Used/ escribed	Intervention/ Program Description	Implementation Model/ Framework/ Theories				
Van Gerpen & Becker, 2013 [41]/ [LifeSpring]/ United States	Program Evaluation Article /Wellness Centre/N/A	Program Evaluation	Adult Cancer Survivors	Physical therapist and exercise physiologist led	erapist and T-shirts kercise logo hysiologist Written l		Exercise and education program. 12-week program consisting of a bi-weekly exercise session and a weekly education session. Exercise component consists of: 20-30-minutes of aerobic exercise (5-min intervals of stationary bike, treadmill, walking on the indoor track, recumbent stepper (Nu-SteP), or upper- body ergometer) and 20-30-minutes of group exercises (resistance training, balance/flexibility/stretching exercises, aquatic exercises and relaxation, Pilates or BODYFLOW <sup>TM</sup> exercises). Educational sessions are led by content experts and include topics such as exercise and cancer, healing, communication and coping, spirituality sleep.					
Local Opinion oncologists, gen leaders from oth design education Continuous Qu (originally only were limited to	EPOC Implementation Str ystem/Health Professional Level Local Opinion leaders: Program development by physic ncologists, general surgeons, nurses, cancer survivors. • eaders from other programs to provide additional insight esign education session topics, screening tools, etc. Continuous Quality Improvement: Program was modified originally only breast cancer survivors) after results from vere limited to 12 participants after previous larger class individualised support.			rviews with interv lent, retention, pro de all cancer types gram evaluation. S	ention ogram s essions	quality of life • Fatig • Slee • Qual • Pain • Depr <u>Implementati</u> <b>Participatior</b> program and <u>Maintenance</u>	eported improvements in their fatigue, pain, sleep disturb , (demonstrated from their pre-, mid-, and post program s gue: [5.58 (2.11) vs. 3.55 (1.86); $p < 0.0001$ ] p [4.77 (2.5) vs. 3.26 (2.27); $p < 0.0001$ ], lity of life [3.63 (2.27) vs. 2.08 (1.86); $p < 0.0001$ ], [2.52 (2.31) vs. 1.85 (1.85); $p < 0.001$ ], ression [2.72 (2.21) vs. 1.65 (1.49); $p < 0.0001$ ].	scores).				

Author, year [Program name], country	Study design /Setting/ Sample Size	Purpose	Participants	Models of Care		irces Used/ escribed	Intervention/ Program Description	Implementation Model/ Framework/ Theories
Wang et al., 2018 [36]/ China	Clinical Audit/ Hospital/ N/A	Implementation Project	Nursing Staff	N/A	Educatio	onal materials	This article first determined the current state of CRF management in the oncology department by undertaking an initial audit. Strategies (listed below) were then implemented to improve practice and address the barriers identified. A follow-up audit was conducted to evaluate the impact of changes made.	None described. Researchers used the JBI Getting Research into Practice (GRiP) tool to identify barriers and practice gaps.
	EF	OC Implementation	n Strategies				Outcomes	
development an Clinical Guidel guidelines. Organisational CRF manageme assessment step (including BFI) Educational me assessment deliv sessions comple Monitoring Per and monitoring Tailored interv adjusted to accor reallocation of v tasks, balancing	d promote good c lines: Content of a <b>Culture &amp; Educ</b> ent strategies deve s created and disp created and distri eetings: Formal 2 vered to all nurses ted by nurses (une formance & Ma of nurses to ensur rentions: Clinical ount for changes in work tasks, manag resources).	ational materials: ational materials: loped for patients and layed in nursing un- buted for use. -hr education session. • 'Practice fatigue der supervision). nagerial Supervisi e compliance. audit conducted to e environment, work ement support, time	rials derived fro Information bro nd staff. • Flow it. • Paper-based ons on CRF back e assessments' a <b>on:</b> Ongoing dis address barriers cload and time re e management v	m the NCCN CRI ochures and poster chart detailing CF d CRF assessment ground, managen nd patient education scussions, commu- to change. • Proc estrictions (i.e., ia prioritisation of	F rs about RF tools nent and on nication, edures	<ul> <li>Heal audi</li> <li>CRF audi</li> <li>Focu audi</li> <li>Patie 78%</li> <li>Patie - 0%</li> </ul>	with best practice audit criteria (compliance rates) - lth professional received education and training: Baseline t - 97% F assessment upon admission and at regular intervals thro t - 0% Follow up audit - 86% used assessment of fatigue undertaken in patients who scr t - 0% Follow up audit - 64% ent education about physical activities: Baseline audit - 3	ughout care: Baseline reen positively: Baseline <b>3%,</b> Follow up audit - d fatigue: Baseline audit

<u>Abbreviations:</u> **BFI** = Brief Fatigue Inventory; **CAPO** = Canadian Association for Psychosocial Oncology; **CRF** = Cancer-related Fatigue; **DNP** 

= Doctor of Nursing Practice; **JBI** = Joanna Briggs Institute; **MD** = Doctor of Medicine; **NCCN** = National Comprehensive Cancer Network;

N/A= Not applicable; NP = nurse practitioner; PA=physical activity; QOL = Quality of Li

#### Implementation and Intervention Outcomes (RE-AIM)

Implementation outcomes of the included studies are outlined in Table and Additional File 3. Overall, Effectiveness and Implementation were the most highly reported dimensions followed by Reach. Adoption and Maintenance were the least reported dimensions.

### **Reach of CRF interventions**

Reach is defined as the number, proportion and representativeness of individuals who are willing to participate in a given initiative or intervention [31]. Descriptions of target population (including demographic information), inclusion criteria and sample size were reported in five [33-37] of six studies. Only one study [35] reported the representativeness or characteristics of participants and non-participants by comparing the sample with broader populations. Program participation rate was reported by two studies [33, 34].

# **Efficacy of CRF interventions**

Efficacy describes the impact of CRF interventions on identified outcomes (e.g., fatigue). Fatigue and behavioural outcome measures were reported in four of six studies [34-37]. Of these studies, all reported reduced CRF severity as a result of the intervention. The 'Energy Through Motion' CRF intervention [36] resulted in decreased fatigue severity by an average of two points compared to an increase of 0.69 points in the usual care arm (p=0.0006). Pre- and post-program scores from Van Gerpen and Becker's 'LifeSpring' CRF intervention [37] demonstrated statistically significant improvements in fatigue (5.58 (SD 2.11) vs. 3.55 (SD 1.86); p < 0.0001). Tian and colleagues' [35] also produced lower patient CRF scores after their CRF intervention (p = 0.04). Lastly, the 'Passport to Comfort' CRF intervention [34] produced significant and beneficial effects on fatigue barriers (p=0.001) and patient fatigue management knowledge (p=0.002). No studies reported on cost-effectiveness.

# Adoption (setting and staff level) of CRF implementation efforts

Adoption is defined as the number, proportion, and representativeness of settings and intervention agents who are willing to initiate a program [31]. Indicators for adoption were the least reported

outcomes in the included studies. Further, indicators such as the description of targeted locations, inclusion/exclusion criteria of settings and staff, method to identify settings and staff, setting and staff participation rate, representativeness of staff and settings, number of staff participating in intervention delivery, and measures of intervention cost were not reported by any study.

#### Implementation

According to Glasgow and colleagues [31], implementation at the setting level refers to the cost of implementation, and whether the intervention was delivered as intended. At the individual level, implementation refers to clients' use of the intervention and implementation strategies. Intervention completion rates were reported by four studies [32, 33, 36, 37] and ranged from 80 to 90%. No study described methods to ensure fidelity of the intervention. Additionally, only the 'Energy Through Motion' CRF program [36] detailed the ongoing implementation cost of the intervention (intervention kits valued at \$21.75 USD per patient).

Results of implementation efforts varied across all studies. Implementation strategies utilised in Wang and colleagues' study [32] resulted in increases in nurse CRF education, nurse assessment of patient CRF upon admission and at regular intervals throughout treatment, and nurse provision of patient education on exercise and other management strategies for CRF. Jones and colleagues [33] reported that their two-hour health and community professional training session resulted in large to very large increases in clinician CRF knowledge (d=0.98), self-efficacy in CRF assessment (d=0.88), self-efficacy to intervene for CRF (d=1.13), and intent to apply CRF guidelines (d=1.35). Tian and colleagues [35] dissemination of CRF guidelines led to increased clinician knowledge, attitude and CRF management behaviours, and the increased adoption of effective CRF management strategies amongst patients. Borneman and colleagues [36] strategies to address professional and system barriers (e.g., formal fatigue presentations to staff, monthly newsletters, ongoing meetings with nurse practitioners) resulted in organisational change (e.g., routine fatigue assessment added to outpatient clinic sheet, increased supportive care referrals). Although Huether and colleagues [36] and Van Gerpen and Becker [37] described the use of implementation strategies in their respective programs, outcomes of their implementation efforts were not reported.

#### Maintenance of CRF interventions and implementation efforts

Maintenance is defined as the extent to which individual behaviour is sustained six months or more after the intervention; and whether a program or policy is institutionalised as part of routine organisational practice [31]. Maintenance indicators at both the individual and setting level were not fully reported in any study and only partially reported across four of the six studies [34-37]. Borneman and colleagues [34] reported individual follow-up and attrition (three months) after program completion; however, did not provide follow-up data at  $\geq 6$  months post-intervention Tian and colleagues [35] stated that innovations and strategies developed from their implementation efforts were maintained for two months after project completion, but were discontinued due to lack of staff time and funding. Borneman and colleagues [34] noted their dissemination of the intervention was conducted at the study institution and that plans were underway to disseminate the intervention into other community centres.

At the time of writing this review, the 'LifeSpring' CRF intervention described in Van Gerpen and Becker's evaluation [37] is currently maintained and institutionalised at the US Bryan Health Medical Center. Additionally, the 'Energy Through Motion' CRF intervention [36] appears to be institutionalised at survivorship clinics offered by the University of Iowa Holden Comprehensive Cancer Centre in the United States; however, little detail of the current program and its integration can be found. None of the studies provided details on costs associated with maintenance, however Van Gerpen and colleagues [37] state that ongoing funding for the program was provided by the medical centre's foundation, and Huether and colleagues report that after project completion of the 'Energy Through Motion' CRF intervention, "continuation of funding for patient supplies was obtained through requests from a regular benefactor of the cancer centre" [36].

#### 4.0 Discussion

Efforts to sustainably implement evidenced-based CRF management strategies into routine clinical care are urgently needed, owing to the high incidence, prevalence, and burden of CRF in cancer survivors severely impacting health-related quality of life [1]. Despite established CRF management strategies

[1, 16], sustainable models of care connecting cancer survivors to effective CRF interventions have yet to be satisfactorily investigated [38]. Our systematic scoping review was able to identify only six studies evaluating the implementation of interventions designed for individuals experiencing CRF. Further, all studies had limited external validity and lacked methodological rigor (e.g., poor reporting of exclusion criteria, study design, data analysis; limited to no follow-up periods; absence of frameworks and theories to guide implementation, etc).

Only three studies used specific implementation models to guide the dissemination process. When applied accurately, implementation theories and frameworks have been shown to enhance dissemination into practice by improving interpretability of study findings and increasing the use of essential implementation strategies [27]. Given only three studies adopted an implementation framework in our review, it is difficult to establish which model is the most helpful for future CRF implementation.

Despite limited use of implementation models, a range of strategies were used across included studies. While most studies in this review demonstrated immediate changes at the clinician, organisational and patient level through their use of various implementation strategies, it was difficult to determine the impact (and impact strength) of individual strategies on implementation outcomes. Further research to identify the preferred strategy from clinicians, patients, and other stakeholders in CRF interventions is likely to be helpful in ascertaining the usefulness, relevance, and effectiveness of specific implementations strategies that will improve implementation efforts [39].

Maintenance and Adoption were the least reported RE-AIM indicators, while Reach, Effectiveness, and Implementation were highly reported across the studies. For all domain indicators, reporting was exceptionally higher for aspects of internal validity (e.g., inclusion criteria, sample size) than external validity (e.g., representativeness of participants, description of settings and staff, intervention fidelity). This is consistent with previous reviews of health interventions across a variety of populations [32, 40-43]. Of note, the level of reporting on Adoption was poor with indicators at the staff and setting level amongst the lowest reported. Details of intervention settings and delivery staff are critical as they allow for the assessment of intervention applicability (and its effect) to different conditions [43]. In five of six included studies, intervention facilitators were employed solely to deliver the CRF intervention or implementation effort, and often had high levels of specific training and supervision, a situation which is not indicative of "real-life practice". Thus, to assist the replication and translation of CRF guidelines and management interventions into routine practice; information regarding intervention setting and staff characteristics, and level of staff skill and training is vital.

Cost was another implementation outcome that was under-reported yet is essential when establishing sustainable models of care for cancer survivors. Cost effectiveness, including start-up and ongoing costs of intervention delivery, have been identified as key factors in determining the translation of research findings into practice [44]. However, these costs were rarely reported, with only the 'Energy Through Motion' CRF intervention providing an explicit cost of intervention materials (\$21.75 USD per patient) [36]. Cost concerns are associated with reduced stakeholder willingness to implement evidenced-based interventions and represent the most significant barrier to evidenced-based practice implementation and program sustainability [45-47]. Reporting costs in future implementation efforts for CRF management is critical.

Assessment of intervention maintenance and sustainability has been identified as a neglected area in clinical research [40-43, 48] with results of this review in agreement. Across all studies, the same common barriers to program continuation were reported: lack of clinician knowledge and skills in the management of CRF, shortage of clinician human resources, lack of program and staff funding, and lack of clinician time. These barriers have been repeatedly highlighted across the CRF literature [49-51]. Berger and Mooney [51] emphasise the lack of access to, and re-imbursement for, integrated supportive cancer programs and services remains the largest challenge to effectively implementing CRF guidelines into routine clinical practice. Further, they conclude that without additional time and reimbursement, clinicians cannot be expected to adequately provide effective or targeted clinical care to individuals experiencing CRF.

### **Implications for Future Research and Practice**

Implementation research in CRF management is severely lacking, highlighting the need for focussed research in this area. In Box. 1, we provide key findings and recommendations of our systematic scoping review. Although feasibility, acceptability and effectiveness outcomes are widely reported across CRF

literature, a greater focus on other pertinent implementation outcomes such as adoption and program maintenance are paramount to translate CRF guidelines and interventions into real-world settings. While we acknowledge that these studies have different aims, and may not comprehensively cover all dimensions outlined in the RE-AIM framework, it is suggested that CRF implementation studies incorporate several stepwise iterative phases to provide opportunities to trial, assess and refine elements; determine resource needs and costs; and gather evidence of implementation impact [49].

Most studies included in this review described CRF implementation at the health professional level, or in acute health care settings. However, the physical, psychological, and psychosocial needs of cancer survivors after active treatment require continuous long-term support which is often only provided by primary and community health care teams [52]. As such, there is a need to extend CRF implementation and translation efforts to community and primary care settings where they will be more accessible to larger population of cancer survivors in the community.

Box .1 Key Findings and Implications for Future Research and Practice

Future research should:

- put greater emphasis on reporting aspects of external validity such as representativeness, setting characteristics, staff level characteristics, and implementation cost.
- be underpinned or guided by an implementation framework.
- utilise rigorous pragmatic designs with adequately powered samples and longer follow-up periods.
- report the impact of implementation at the system, health professional and cancer survivor level.

Intervention developers should:

• consider sources of ongoing funding and endeavour to use existing resources (staff, equipment, infrastructure, etc.) to deliver implementation efforts.

Clinical leaders should:

- endeavour to build clinician awareness and knowledge of evidenced-based CRF management and assessment strategies through the provision of educational training and resources.
- seek regular engagement with clinical staff and relevant stakeholder groups to identify potential/existing enablers or barriers to clinical change and to tailor implementation efforts to specific contexts.
- place emphasis on the allocation of clinician resources within settings, the provision of time management support to clinicians (e.g., reallocation of work tasks, adjustment of procedures to fit clinician schedule, adjustment of clinician schedule to fit procedures) and the identification of 'clinical champions' or opinion leaders, to encourage peer behaviour change and compliance with the recommended CRF management and assessment practices.

#### Limitations

This review has two main limitations. First, the inconsistencies of what constitutes an implementation study (e.g., implementation processes, terminologies, definitions, intention to treat, inclusion/exclusion criteria) made it difficult to detect a distinct relationship between the use of implementation models and strategies, and implementation outcomes. However, this is not unique to our review with similar inconsistencies frequently reported across the implementation science literature [52-54]. Second, our review was limited to studies in English, potentially resulting in some level of publication bias limiting the generalisability of results.

## 5.0 Conclusion

This systematic scoping review is the first to examine models, strategies, and outcomes of studies reporting on the implementation of interventions for individuals experiencing cancer-related fatigue. Our review found that various implementation strategies have been used to promote uptake of CRF management interventions and guidelines at the organisational, clinician, and patient level. However, lack of consistent reporting of external indicators (e.g., ongoing and start-up costs of intervention, setting and staff representativeness) and factors such as lack of clinician time, insufficient clinician and intervention funding, and unsustainable maintenance costs, are potential barriers to study translatability and CRF program implementation. This review emphasises the absence of quality CRF implementation studies to bridge this important knowledge-practice gap.

## References

- 1. Fabi A, Bhargava R, Fatigoni S, Guglielmo M, Horneber M, Roila F, Weis J, Jordan K, Ripamonti CI. Cancer-related fatigue: ESMO Clinical Practice Guidelines for diagnosis and treatment. Ann Oncol. 2020;31(6):713-723.
- 2. Horneber M, Fischer I, Dimeo F, Rüffer JU, Weis J: Cancer-related fatigue: epidemiology, pathogenesis, diagnosis, and treatment. Dtsch Arztebl Int. 2012, 109(9):161-172.
- 3. Weis J, Horneber M. Definition and Prevalence of Cancer-Related Fatigue. In: Cancer-Related Fatigue. edn. Edited by Weis J, Horneber M. Tarporley: Springer Healthcare Ltd.; 2015: 1-10.
- 4. Berger AM, Mooney K, Alvarez-Perez A, Breitbart WS, Carpenter KM, Cella D, Cleeland C, Dotan E, Eisenberger MA, Escalante CP et al. Cancer-Related Fatigue, Version 2.2015. J Natl Compr Canc Netw. 2015;13(8):1012.
- 5. Bower JE. Cancer-related fatigue—mechanisms, risk factors, and treatments. Nat Rev Clin Oncol. 2014;11(10):597-609.
- 6. Dagnelie PC, Pijls-Johannesma MCG, Lambin P, Beijer S, De Ruysscher D, Kempen GIJM. Impact of fatigue on overall quality of life in lung and breast cancer patients selected for high-dose radiotherapy. Ann Oncol. 2007;18(5):940-944.
- Charalambous A, Kouta C. Cancer Related Fatigue and Quality of Life in Patients with Advanced Prostate Cancer Undergoing Chemotherapy. Biomed Res Int. 2016;2016:3989286-3989286.
- 8. Borneman T. Assessment and Management of Cancer-Related Fatigue. 2013, 15(2):77-86.
- 9. Bower JE, Ganz PA, Desmond KA, Rowland JH, Meyerowitz BE, Belin TR. Fatigue in breast cancer survivors: Occurrence, correlates, and impact on quality of life. J Clin Oncol. 2000;18(4):743-753.
- 10. Curt GA, Breitbart W, Cella D, Groopman JE, Horning SJ, Itri LM, Johnson DH, Miaskowski C, Scherr SL, Portenoy RK et al. Impact of cancer-related fatigue on the lives of patients: new findings from the Fatigue Coalition. Oncologist. 2000;5(5):353-360.
- 11. Minton O, Berger A, Barsevick A, Cramp F, Goedendorp M, Mitchell SA, Stone PC. Cancerrelated fatigue and its impact on functioning. 2013;119(S11):2124-2130.
- 12. Minton O, Stone P.: How common is fatigue in disease-free breast cancer survivors? A systematic review of the literature. Breast Cancer Res Treat. 2008;112(1):5-13.
- Stone P, Hardy J, Broadley K, Tookman AJ, Kurowska A, A'Hern R:. Fatigue in advanced cancer: A prospective controlled cross-sectional study. Br J Cancer. 1999;79(9-10):1479-1486.
- 14. Finnegan-John J, Molassiotis A, Richardson A, Ream E. A systematic review of complementary and alternative medicine interventions for the management of cancer-related fatigue. Integr Cancer Ther. 2013;12(4):276-290.
- 15. Corbett TK, Groarke A, Devane D, Carr E, Walsh JC, McGuire BE The effectiveness of psychological interventions for fatigue in cancer survivors: systematic review of randomised controlled trials. Systematic Reviews. 2019;8(1):324.

- Mustian KM, Alfano CM, Heckler C, Kleckner AS, Kleckner IR, Leach CR, Mohr D, Palesh OG, Peppone LJ, Piper BF et al. Comparison of Pharmaceutical, Psychological, and Exercise Treatments for Cancer-Related Fatigue: A Meta-analysis. JAMA Oncol. 2017;3(7):961-968.
- Hilfiker R, Meichtry A, Eicher M, Nilsson Balfe L, Knols RH, Verra ML, Taeymans J.Exercise and other non-pharmaceutical interventions for cancer-related fatigue in patients during or after cancer treatment: a systematic review incorporating an indirect-comparisons meta-analysis. Br J Sports Med. 2018;52(10):651.
- 18. Blackhall L, Petroni G, Shu J, Baum L, Farace E. A pilot study evaluating the safety and efficacy of modafinal for cancer-related fatigue. J Palliat Med. 2009;12(5):433-439.
- 19. Bauer MS, Damschroder L, Hagedorn H, Smith J, Kilbourne AM. An introduction to implementation science for the non-specialist. BMC Psychol. 2015;3(1):32-32.
- 20. Morris ZS, Wooding S, Grant J. The answer is 17 years, what is the question: understanding time lags in translational research. J R Soc Med. 2011;104(12):510-520.
- 21. Green LW, Glasgow RE. Evaluating the relevance, generalization, and applicability of research: issues in external validation and translation methodology. Eval Health Prof. 2006;29(1):126-153.
- 22. Eccles MP, Mittman BS. Welcome to Implementation Science. Implementation Science. 2006;1(1):1.
- 23. Munn Z, Peters MDJ, Stern C, Tufanaru C, McArthur A, Aromataris E: Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach. Med Res Methodol. 2018; 18(1):143.
- 24. Tricco AC, Lillie E, Zarin W, O'Brien KK, Colquhoun H, Levac D, Moher D, Peters MDJ, Horsley T, Weeks L et al. PRISMA Extension for Scoping Reviews (PRISMA-ScR): Checklist and Explanation. Ann Intern Med.. 2018;169(7):467-473.
- 25. Effective Practice and Organisation of Care (EPOC). EPOC Taxonomy; 2015. epoc.cochrane.org/epoc-taxonomy (accessed 03 July 2020).
- 26. Nilsen P. Making sense of implementation theories, models and frameworks. Implementation Science. 2015;10(1):53.
- 27. Glasgow RE, Estabrooks PE. Pragmatic Applications of RE-AIM for Health Care Initiatives in Community and Clinical Settings. Prev Chronic Dis. 2018;15:E02-E02.
- Harden SM, Gaglio B, Shoup JA, Kinney KA, Johnson SB, Brito F, Blackman KC, Zoellner JM, Hill JL, Almeida FA et al. Fidelity to and comparative results across behavioral interventions evaluated through the RE-AIM framework: a systematic review. Syst Rev. 2015;4:155.
- 29. Kwan BM, McGinnes HL, Ory MG, Estabrooks PA, Waxmonsky JA, Glasgow RE. RE-AIM in the Real World: Use of the RE-AIM Framework for Program Planning and Evaluation in Clinical and Community Settings. Front Public Health. 2019;7(345).
- Zeiler M, Kuso S, Nacke B, Klesges LM, Waldherr K. Evaluating reach, adoption, implementation and maintenance of Internet-based interventions to prevent eating disorders in adolescents: a systematic review. Eur J Public Health.2019;30(1):179-188.

- 31. Glasgow RE, Harden SM, Gaglio B, Rabin B, Smith ML, Porter GC, et al. RE-AIM Planning and Evaluation Framework: Adapting to New Science and Practice With a 20-Year Review. Front Public Health. 2019;7(64).Wang N, Yang Z, Miao J, Mi X, Liu S, Stern C, Porritt K, Zhang L. Clinical management of cancer-related fatigue in hospitalized adult patients: a best practice implementation project. JBI Database System Rev Implement Rep. 2018;16(10):2038-2049.
- 32. Jones G, Rutkowski N, Trudel G, St-Gelais C, Ladouceur M, Brunet J, Lebel S. Translating guidelines to practice: a training session about cancer-related fatigue. Curr Oncol. 2020;27(2):e163-e170.
- 33. Borneman T, Koczywas M, Sun V, Piper BF, Smith-Idell C, Laroya B, Uman G, Ferrell B. Effectiveness of a clinical intervention to eliminate barriers to pain and fatigue management in oncology. J Palliat Med. 2011;14(2):197-205.
- 34. Tian L, Yang Y, Sui W, Hu Y, Li H, Wang F, Qian K, Ji J, Tao M. Implementation of evidence into practice for cancer-related fatigue management of hospitalized adult patients using the PARIHS framework. PLoS One. 2017;12(10):e0187257.
- 35. Huether K, Abbott L, Cullen L, Cullen L, Gaarde A. Energy Through Motion<sup>©</sup>: An Evidence-Based Exercise Program to Reduce Cancer-Related Fatigue and Improve Quality of Life. Clin J Oncol Nurs. 2016;20(3):E60-E70.
- 36. Van Gerpen RE, Becker BJ. Development of an evidence-based exercise and education cancer recovery program. Clin J Oncol Nurs. 2013;17(5):539-543.
- 37. Kennedy MA, Bayes S, Newton RU, Zissiadis Y, Spry NA, Taaffe DR, Hart NH, Davis M, Eiszele A, Galvão DA. We have the program, what now? Development of an implementation plan to bridge the research-practice gap prevalent in exercise oncology. Int J Behav Nutr Phys Act. 2020;17(1):128.
- 38. Kitson A, Harvey G, McCormack B. Enabling the implementation of evidence based practice: a conceptual framework. Qual Health Care. 1998;7(3):149.
- 39. Brien J, Finch CF. The Implementation Of Team Ball Sport Injury Prevention Exercise Programmes: A Systematic Review Employing The Re-Aim Framework. Br J Sports Med. 2014;48(7):645.
- McGoey T, Root Z, Bruner MW, Law B. Evaluation of physical activity interventions in children via the reach, efficacy/effectiveness, adoption, implementation, and maintenance (RE-AIM) framework: A systematic review of randomized and non-randomized trials. J Prev Med. 2016;82:8-19.
- Ellis LA, Augustsson H, Grødahl AI, Pomare C, Churruca K, Long JC, Ludlow K, Zurynski YA, Braithwaite J. Implementation of e-mental health for depression and anxiety: A critical scoping review. J Community Psychol. 2020;48(3):904-920.
- 42. Craike M, Hill B, Gaskin CJ, Skouteris H. Interventions to improve physical activity during pregnancy: a systematic review on issues of internal and external validity using the RE-AIM framework. Bjog. 2017;124(4):573-583.
- 43. Cidav Z, Mandell D, Pyne J, Beidas R, Curran G, Marcus S. A pragmatic method for costing implementation strategies using time-driven activity-based costing. Implement Sci.2020;15(1):28.

- 44. Aarons GA, Wells RS, Zagursky K, Fettes DL, Palinkas LA. Implementing Evidence-Based Practice in Community Mental Health Agencies: A Multiple Stakeholder Analysis. Am J Public Health. 2009;99(11):2087-2095.
- 45. Scheirer MA, Dearing JW: An Agenda for Research on the Sustainability of Public Health Programs. Am J Public Health.2011; 101(11):2059-2067.
- 46. Kennedy MA, Bayes S, Galvão DA, Singh F, Spry NA, Davis M, Chee R, Zissiadis Y, Hart NH, Taaffe DR et al. If you build it, will they come? Evaluation of a co-located exercise clinic and cancer treatment centre using the RE-AIM framework. Eur J Cancer Care. 2020;29(4):e13251.
- 47. Walugembe DR, Sibbald S, Le Ber MJ, Kothari A. Sustainability of public health interventions: where are the gaps? Health Res Policy Syst. 2019;17(1):8.
- 48. Pearson EJM, Morris ME, McKinstry CE. Cancer related fatigue: implementing guidelines for optimal management. BMC Health Serv Res. 2017;17(1):496.
- 49. Piper BF, Borneman T, Sun VC-Y, Koczywas M, Uman G, Ferrell B, James RL. Cancerrelated fatigue: role of oncology nurses in translating National Comprehensive Cancer Network assessment guidelines into practice. Clin J Oncol Nurs. 2008;12(5 Suppl):37-
- 50. Berger A, Mooney K. Dissemination and Implementation of Guidelines for Cancer-Related Fatigue. J Natl Compr Canc Netw. 2016;14(11):1336-1338.
- 51. Leach CR, Weaver KE, Aziz NM, Alfano CM, Bellizzi KM, Kent EE, et al. The complex health profile of long-term cancer survivors: prevalence and predictors of comorbid conditions. J Cancer Surviv. 2015;9(2):239–51.
- 52. Flodgren G, Parmelli E, Doumit G, Gattellari M, O'Brien MA, Grimshaw J, et al. Local opinion leaders: effects on professional practice and health care outcomes. Cochrane Database Syst. Rev.. 2011(8).
- 53. Colquhoun H, Leeman J, Michie S, Lokker C, Bragge P, Hempel S, et al. Towards a common terminology: a simplified framework of interventions to promote and integrate evidence into health practices, systems, and policies. Implement Sci. 2014;9(1):781.
- 54. McKibbon KA, Lokker C, Wilczynski NL, Ciliska D, Dobbins M, Davis DA, et al. A crosssectional study of the number and frequency of terms used to refer to knowledge translation in a body of health literature in 2006: a Tower of Babel? Implement Sci. 2010;5:16.

# Additional File Legend

Additional File 1: Modified EPOC Implementation Strategies

Additional File 2: Search Strategy

Additional File 3: RE-AIM Indicators & Reporting of RE-AIM Indicators Across all Included Studies

# Additional File 1: Modified EPOC Implementation Strategies

STRATEGY	CRITERIA
Organisational culture	Did the intervention utilise any strategies to change organisational culture?
Audit and Feedback	Is performance feedback provided in either a verbal, written or electronic format? Are recommendations for clinical action provided?
Communities of practice	Does the intervention utilise communities in practice? (Groups of people with a common interest who deepen their knowledge and expertise in an area by interacting on an ongoing basis)
Continuous Quality Improvement	Does the article or intervention describe/undertake any iterative processes to review and improve care? (This can include the involvement of healthcare teams, analysis of a process or system, a structured process improvement method or problem-solving approach, and use of data analysis to assess changes).
Monitoring the performance of the delivery of healthcare	Is the intervention/program/ service routinely monitored. Has it been compared with an external standard?
Educational games	Does the intervention use games as an educational strategy to improve standards of care
Educational materials	Does the article or intervention describe/ incorporate the distribution of knowledge and educational materials to support clinical care? For example, this may be facilitated by the internet, learning critical appraisal skills; skills for electronic retrieval of information, diagnostic formulation; question formulation, etc.
Educational meetings	Does the intervention or article describe or use courses, workshops, conferences or other educational meetings?
Educational outreach visit, or academic detailing	Does the article/ intervention describe or utilise personal visits by a trained person to facilitators/intervention participants/ in their own settings (to provide information with the aim of changing practice).
Clinical Practice Guidelines/ Local Consensus Processes	Is the development of the intervention informed by any clinical practice guidelines, frameworks or models? Does the article promote the implementation of clinical guidelines? Was there any formal or informal consensus process to choose or adapt guidelines for use?
Inter-professional Education	Does the intervention use/provide/incorporate continued education of health professionals?
Local opinion leaders	Were local opinion leaders identified and used to inform intervention development and to promote good clinical practice?
Patient-mediated interventions	Was the intervention informed by information provided by patients (survey, interview, other patient interactions)?
Managerial supervision	Does the article/ intervention incorporate routine supervision visits with health?
Routine patient reported outcomes	Does the article or intervention describe/allow for the facilitation of frequent routine patient reported outcome measures (e.g., logbook recording fatigue severity, impact on QOL, etc.)?
Reminders	Does the intervention use or describe any manual or computerised mechanisms that prompt individuals to perform an action?
Tailored interventions	Is the intervention based off any assessment of barriers to change? Was the intervention informed by any assessments like interviews or surveys?

Summary of EPOC Modifications

- Implementation strategies put into question format to assist screening process.
- EPOC taxonomy contains more items. Only the taxonomy relevant to the study were included (see summary of EPOC modifications)

The following strategies were not included as they were not considered relevant to the study (outside the scope of the study or a different scope of the study): Public release of performance data, Clinical incident reporting and the "Interventions targeted at specific types of practice, conditions or settings" section.

# Additional File 2: Search Strategy

# <u>CINAHL</u>

- 1. (MH "Systems Implementation") OR (MH "Program Implementation") OR (MH "Implementation Science") OR (MH "Program Development") [53,225]
- TI ( implement\* OR disseminat\* OR translat\* OR ("Systems Implementation") OR ("Program Implementation") OR ("Implementation Science") OR (Program Development) ) OR AB ( implement\* OR disseminat\* OR translat\* OR ("Systems Implementation") OR ("Program Implementation") OR ("Implementation Science") OR ("Program Development") ) [266,712]
- 3. S1 OR S2 [299,074]
- 4. (MM "Cancer Fatigue") [959]
- 5. (MH "Cancer Fatigue") [1,590]
- 6. TI ( ("cancer fatigue" OR "cancer related fatigue" OR cancer related fatigue) ) OR AB ( ("cancer fatigue" OR "cancer related fatigue" ) [1,380]
- 7. S4 OR S5 OR S6 [2,435]
- 8. (S3 AND S7) [159]

# <u>EMBASE</u>

- 1. 'implementation science'/exp [1510]
- 2. implement\*:ab,ti OR translat\*:ab,ti OR 'program development':ab,ti [1,060,964]
- 3. #1 OR #2 1,061,142]
- 4. 'cancer fatigue'/exp/mj [1,226]
- 5. 'cancer fatigue'/exp [3,179]
- 6. 'cancer fatigue':ab,ti OR 'cancer related fatigue':ab,ti [2,270]
- 7. #4 OR #5 OR #6 [4,395]
- 8. #3 AND #7 [251]

COCHRANE LIBRARY

- 1. MeSH descriptor: [Implementation Science] explode all trees [34]
- 2. MeSH descriptor: [Health Plan Implementation] explode all trees [176]
- 3. MeSH descriptor: [Translational Medical Research] explode all trees [116]
- 4. MeSH descriptor: [Program Development] explode all trees [710]
- 5. (implement\* OR translation OR "program development"):ti,ab,kw [42,849]
- 6. ("cancer related fatigue" OR "cancer fatigue"):ti,ab,kw [1,016]
- 7. (#1 OR #2 OR #3 OR #5) AND #6 [52]

<u>PUBMED</u>: (((("implementation science"[MeSH Terms] OR "translational medical research"[MeSH Terms] OR "program development"[MeSH Terms] OR knowledge translation[MeSH Terms] OR health plan implementation[MeSH Terms] OR (implement\*[Title/Abstract]) OR (translation[Title/Abstract])))) AND ((("cancer fatigue"[Title/Abstract]) OR "cancer related fatigue"[Title/Abstract])) [99]

# Additional File 3: RE-AIM Indicators & Reporting of RE-AIM Indicators Across all Included Studies

REACH	
R1. Described Target Population & Demographics	A brief description of the broader target population & Gender, age, educational attainment, occupation, SES, behavioural outcomes
R2. Method to Identify Target Population	Describe the process by which the target population was identified for participation in the study.
R3. Recruitment Strategies	Describe the methods used to recruit participants into the study.
R4. Inclusion Criteria	Description of inclusion criteria for individual participants
R5. Exclusion Criteria	Description of exclusion criteria for individual participants
R6. Representativeness of participants	Comparison of characteristics (% females, mean age, age range) between participants and non- participants
R7. # Eligible & Invited (exposed) to Recruitment	The total number of eligible participants contacted for participation.
R8. Sample Size	The number of people who agree to participate
R9. Participation Rate	Sample size divided by the target population denominator
R10. Cost of Recruitment	The cost of recruitment can reflect monetary and/or time units.
EFFICACY / EFFECTIVENESS	
E1. Measures and results for post-intervention assessment	Description of outcome measures
E2. Intention-to-treat analysis utilised	Analysing participants in the groups in which they were randomised regardless of adherence
E3. Imputation procedure	Description of imputation procedure used for Intention-to-treat analysis
E4. Quality of Life measure included	Description of use and results for QoL
E5. Measure of satisfaction with / acceptability of programme	Description of included measures regarding satisfaction and acceptability
E6. Cost Effectiveness	Code as reported if specific mention and amounts are
E7. Effects at follow-up	provided for the cost of the intervention.
er. Ellects at follow-up	At least one follow-up assessment (after post- intervention assessment) included
E8. Attrition at program completion ADOPTION	Description of type and % of attrition
A1. Setting Adoption rate	The proportion of sites eligible and contacted that participated.
A2. Description of Targeted Location	Characteristics that would be considered an ideal location for the intervention.
A3. Inclusion/Exclusion Criteria of Setting	Description of eligibility criteria for settings
A4. Description of Intervention Location	The explicit statement of characteristics of the location of the intervention
A5. Method to Identify Setting	Describe the process by which the location was identified for participation in the study
A6. Average # of Persons Served Per Setting	Calculated average number of participants at each site.
A7. Staff Participation Rate	The proportion of the staff that was eligible and contacted and participated

A8. Method to Identify Target Delivery Agent	Describe the process by which the target delivery agent was identified for participation in the study.
A9. Description of Delivery Agent	Characteristics of Staff
A10.Level of Expertise of Delivery Agent	Description of expertise (e.g. training) of delivery agents
A11. Inclusion/Exclusion Criteria of Delivery Agent	The explicit statement of characteristics of the delivery agents that were used to determine if a potential delivery agent is eligible to participate.
A12. Measures of Cost Adoption	Description of costs for intervention set-up
IMPLEMENTATION	
I1. Theories	Explicit statement of theories or principles used to develop the intervention
<b>I2.</b> Intervention Number of Contacts	Total number of encounters with participants
I3. Timing of Contacts	Describe when the intervention contacts occur over the course of the intervention.
I4. Duration of contacts	Length of each intervention contact
I5. Extent to which intervention was delivered as intended	Description of individual adherence, compliance, attendance and/or staff adherence
I6. Consistency of intervention delivery	Description of consistency across staff, time, settings and subgroups of participants
<b>I7.</b> Participant Attendance/ Completion rates	The proportion of the intervention that the participants received, on average.
18. Costs of delivery (ongoing costs)	Description of costs (time, money)
MAINTENANCE	
M1. Assessed outcomes ≥ 6 months	Follow-up periods ≥ 6 months post-intervention
M2. Drop-out rate to last follow-up	% drop-out (if follow-up assessment ≥ 6 months)
M3. Current status of programme	Description of current status after end of study
M4. Use of qualitative methods to measure individual or organisational level maintenance	
M5. Is the program still in place	Description of program continuation after completion of the research study.
M6. Adaptations made	Description of adaptations made to foster sustainability of programme
M7. Costs of maintenance	Description of costs (money, time) related to sustainability of programme

					RE	ACH						E	FFEC	TIV	ENE	SS						Å	DOP	тю	N					IMPLEMENTATION						MAINTENANCE								
	R1	R2	R3	R4	R5	R6	R7	R8	R9	R10	E1	E2	E3	E4	E5	E6	E7	A1	A2	A3	A4	A5	A6	A7	A8	A9	A10	A11	A12	11	12	13	14	15	16	17	18	M1	M2	M3	M4	M5	M6	M7
Abbott et al. 2017 & Huether et al., 2016	+	-	+	+	+	-	-	+	-	-	+	-	-	+	+	-	+	+	-	-	+	-	-	-	-	+	+	-	-	+	+	+	+	-	+	+	+	-	-	+	-	Y	-	-
Borneman et al., 2010 & 2011	+	-	-	+	-	-	-,+	+	-,+	-	+	-	+	+	-	-	+,-	-	-	-	+	-	+	-	-	+	+	-	-	-	+	+	+	-	-	-	-	-	-	+	-	-	-	-
Jones et al., 2020	+	-	-	+	-	-	-	+	+	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	+	+	-	-	+	+	+	+	-	-	+	-	-	-	-	-	-	-	-
Tian et. al 2017	+	-	-	+	-	-	-	+	-	-	+	-	-	-	-	-	-	-	-	-	+	-	+	-	-	-	-	-	-	+	-	-	-	-	-	-	-	-	-	+	-	-	-	-
Van Gerpen & Becker, 2013	+	-	+	+	-	-	-	+	-	-	+	-	-	+	+	-	-	-	-	-	-	-	+	-	-	+	+	-	-	-	+	+	+	-	-	+	-	-	-	+	-	Y	+	-
Wang et al., 2018	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	+	-	-	-	-	-	-	-	-	-	+	-	-	-	-	+	-	-	-	-	-	-	-	-

# Reporting of RE-AIM Indicators Across all Included Studies

(N.A) Not Applicable, (+) reported, (-) not-reported, (Y) yes, (N) no.

# Appendix 2: Population Intervention Comparison Outcome (PICO) Selection Criteria (Study 1)

	Inclusion	Exclusion
Participants	Cancer survivors (regardless of age, gender, tumour & treatment type) at any stage of their cancer trajectory that experience fatigue.	
	<ul> <li>Self-described self-management interventions OR self-management models including at least one of the following intervention components: <ul> <li>information about cancer-related fatigue and/its management</li> <li>information about managing psychosocial consequences of illness</li> <li>information about available social support resources and navigating transitions</li> <li>provision of /agreement on goal setting and specific clinical action plans and/or urgent problems</li> <li>regular clinical review</li> <li>monitoring of condition with feedback</li> <li>practical support with adherence (medication or behavioural)</li> <li>provision of access to advice or support when needed</li> <li>training rehearsal to communicate with healthcare professionals</li> <li>training rehearsal for lifestyle adaptations and for everyday activities</li> <li>training rehearsal for psychological strategies</li> <li>social support</li> <li>lifestyle advice and support</li> </ul> </li> </ul>	No intervention or any intervention other than those specified in the inclusion criteria.
Control	No intervention, a clearly defined usual or standard care or an alternative intervention.	
Outcomes	Must include one of the following primary outcomes: Primary: 1. Fatigue (Severity, Interference, Distress)	Lack of at least one relevant pre-specified outcome
	<ul> <li>Secondary:</li> <li>1. Behavioural Outcomes (e.g., physical activity, diet, exercise, sleep, fatigue management self-efficacy, other self-efficacy outcomes - cancer self-efficacy, general self-efficacy, physical self-efficacy, etc)</li> </ul>	
Study Design	Randomised controlled trials	Studies reporting non- randomised studies.

# Appendix 3: Search Strategy SMS Systematic Review (Study 1) - Updated 20/06/2021

Search Strategy for CINAHL

- TI ( "self manage\*" OR "self care" OR "self efficacy" OR "self medication" OR"self admin\*" OR "self concept" OR "self help" OR "self monitor\*" OR "symptom management" ) OR AB ( "self manage\*" OR "self care" OR "self efficacy" OR "self medication" OR"self admin\*" OR "self concept" OR "self help" OR "self monitor\*" OR "symptom management" ) OR (MH "Self-Management") OR (MH "Self Care") OR (MH "Self-Efficacy") OR (MH "Self Concept") OR (MH "Self Administration" (MH "Self-Management") OR (MH "Self Care") OR (MH "Self Care") OR Concept") OR (MH "Self Administration")
- TI ( "cancer fatigue" OR "cancer related fatigue" OR "cancer-related fatigue") OR AB ( "cancer fatigue" OR "cancer related fatigue" OR "cancer-related fatigue") OR (MM "Cancer Fatigue") OR (MH "Cancer Fatigue") (MM "Cancer Fatigue")
- 3. TI ( S1 AND S2 ) OR AB ( S1 AND S2 ) [275]

Search Strategy for EMBASE

- 'self manage\*':ab,ti OR 'self care':ab,ti OR 'self efficacy':ab,ti OR 'self medication':ab,ti OR 'self admin\*':ab,ti OR 'self concept':ab,ti OR 'self monitor\*':ab,ti OR 'self help':ab,ti OR 'self care'/exp
- 2. 'cancer fatigue':ab,ti OR 'cancer related fatigue':ab,ti OR 'cancer fatigue'/exp/mj OR 'cancer fatigue'/exp
- 3. #1 AND #2 [199]

# Search Strategy for PUBMED

(((((((("self management"[MeSH Terms] OR "self care"[MeSH Terms]) OR "self administration"[MeSH Terms]) OR "self efficacy"[MeSH Terms]) OR "self medication"[MeSH Terms]) OR "self concept"[MeSH Terms]) OR "self medication"[Title/Abstract]) OR "self management"[Title/Abstract]) OR "self care"[Title/Abstract]) OR "self administration"[Title/Abstract]) OR "self efficacy"[Title/Abstract]) OR "self concept"[Title/Abstract]) OR "self medication"[Title/Abstract]) OR "self

Search Strategy for COCHRANE CENTRAL

- 1. MeSH descriptor: [Self Medication] explode all trees
- 2. MeSH descriptor: [Self Efficacy] explode all trees
- 3. MeSH descriptor: [Self-Management] explode all trees
- 4. #1 OR #2 OR #3 OR "self management" OR "self care" OR "self efficacy" OR "self medicat\*"
- 5. "cancer related fatigue" OR "cancer fatigue"
- 6. #4 AND #5 **[118]**

# Appendix 4: Summary of Study Characteristics for Systematic Review (Study 1)

Citation & Country	Study Design	Intervention Aim	Population	Cancer Type	Timing	Inclusion Fatigue Cut-off Score	Sample Size	Control Name & Description	Intervention Name & Description	Mode of Delivery	Intervention Delivery agent	Intervention Timing and Duration	# of Self- management Support Taxonomy Components
Abrahams et al., 2017 Netherlands	2-arm RCT	To improve severe fatigue	Severely fatigued breast cancer survivors (who had completed treatment with curative intent at least 3 months prior).	Breast	Post Tx	CIS-FS≥35	N=132 INT: 66 CONT :66	Participants placed on 6- month waiting list for F2F CBT, included oncological follow-up examinations and referral for psychosocial care. No restriction on the use of fatigue interventions.	Internet based cognitive behavioural therapy. 3 F2F sessions and 8 web-based treatment modules. 2 initial F2F sessions (goal setting, & treatment plan). Therapists use assessment tools to assess which treatment modules each survivor should complete (minimum of 3). Completion of online modules (Modules 1 and 8 compulsory) – therapists provide feedback answer questions via email/ video consult. Final F2F session (discussion fatigue severity outcomes) after the completion of online modules.	Web-based & F2F Individual	HP: Licensed cognitive behavioural therapists experienced in working with fatigued cancer survivors.	Max duration 6 months. 2 initial F2F sessions. 8 online modules. Bi-weekly email correspondence from therapists (also option for 2 video calls max).	8
Armes et al., 2007 UK	2-arm RCT	To improve CRF related thoughts and behaviour.	Adult patients with significant fatigue undergoing active chemo.	Any	During Tx	Self-reported fatigue	N=55 INT: 28 CONT:27	Usual care. CRF not assessed routinely & management advice given in an ad hoc manner.	Behaviour oriented intervention that provided cognitive and physical strategies for managing CRF.	F2F Individual	Research fellow trained to deliver INT using study manual.	6-9 weeks. 3 sessions, 60- min. Delivered every 3 to 4 weeks.	6
Bantum et al, 2014 USA	2-arm RCT	Determine the effectiveness of a web- based behaviour program on fatigue.	Adult cancer survivors who had at completed primary treatment at least 4 weeks prior.	Any	Post Tx	Not assessed	N=303 INT: 156 CONT:147	Waitlist control	Six session online education course/ workshop. 20-25 survivors per cohort (9 cohorts total). Each week of course content includes approx. 30-35 webpages of didactic content. Program content includes skill building, problem solving, action planning to build self- efficacy, etc.	Web-based Enables group discussion via forums	Peers: Cancer survivors (2 per cohort) with intensive online training about intervention materials and how to respond to user comments and goals	6 weeks. New topic completed weekly	9
Barsevick et al., 2004 USA	2-arm RCT	Evaluate efficacy of energy conservation and activity management on CRF	Adult cancer survivors initiating chemo, RT, or concurrent therapy.	Breast, lung, colorecta l, prostate, gynaecol ogic, bladder, testicular cancer, or lymphom a (mostly breast)	During Tx	Not assessed	N=396 INT: 200 CONT:196	Attention control. Information on nutrition and healthy diet.	Energy Conservation and Activity Management (ECAM). Three oncology nurse-led telephone sessions. After session 1 participants complete fatigue monitoring journal and assess activity patterns. Journal used to create energy conservation plan in session 2. In session 3 the plan was evaluated and revised.	Telephone Individual	HP: Oncology nurse with 8 hours of counsellor training and individual case supervision.	Total duration varied, approx. 3 weeks. Three phone sessions during the first 3 weeks of chemo or weeks 3-5 of radiotherapy. Session 1 & 2: 30 minutes Session 3: 15 minutes	6

Barsevick et al., 2010 USA	2-arm RCT	Evaluate efficacy of energy and sleep enhancement on fatigue and sleep disturbance and functional status in chemo pts.	Adult cancer survivors commencing chemo.	Breast, lung, colorecta l, prostate, gynaecol ogic, bladder, testicular cancer, or lymphom a (mostly breast)	During Tx	Not assessed	N=292 INT:153 CONT:139	Attention control. Information on nutrition and healthy diet	Energy and sleep enhancement. Three oncology nurse-led telephone sessions and a handbook relating to the intervention (consisting of and examples of energy conservation and sleep management strategies) detailing information on CRF, and coping skills.	Telephone Individual	HP: Oncology nurse	Total duration varied. Three phone sessions during the second, third and fourth weeks after chemotherapy.	6
Bennet et al., 2007 USA	2-arm RCT	Evaluate effect of motivational interviewing on increasing self-efficacy, physical activity, aerobic fitness and CRF.	Adult cancer survivors who completed primary treatment at least 6 months prior to enrolment.	Any	Post Tx	Self-reported fatigue	N=56 INT: 28 CONT: 28	Usual care. Maintenance of current physical activities	One initial F2F MI counselling session & 3 subsequent MI phone calls to assist with problem solving.	F2F & Telephone Individual	HP: Physical activity counsellor with 8hrs group training and 6hrs individual training in MI by an experienced MI trainer.	4.5 months. One 30- min counselling session. 20 min telephone call at 2 weeks, 2 months, and 4.5 months after initial F2F session.	5
Bruggeman- Everts et al., 2017 Netherlands	3-arm RCT	To reduce chronic cancer related fatigue	Adult participants who had completed curative treatment at least 3 months prior	Any	Post-Tx (hormo nal treatme nt allowed )	CIS-FS ≥35	N=167 AAF: 62 eMBCT:55 CONT: 50	Active control. Psychoeducatio n. Received automatic 'no- reply' emails detailing possible causes of fatigue and information on sleep hygiene, energy balance and coping with worry.	Trial investigating the effectiveness of 2 web-based interventions: Ambulant Activity Feedback therapy (AAF) – (physical activity coaching) and electronic Mindfulness-based Cognitive Therapy eMBCT. <b>AAF:</b> ambulant activity coaching system, supported weekly by a physiotherapist through email. Participants work to achieve personal goals created in collaboration with the physiotherapist. Patients have access to real-time feedback, can monitor their past personal activity, gain access to their activity pattern, and then increase/ balance daily activity to improve energy levels. <b>eMBCT:</b> psychologist guided online program that provides audio mindfulness exercises and information about a different mindfulness exercise each week. Patients can write down their experiences in a log following their exercises which a therapist then responds to weekly.	Web-based Individual	HP eMBCT: psychologist (provided guidance remotely). AAF: home based physiotherapist	AAF: 3hrs per week for nine weeks eMBCT: 4hrs per week for nine weeks	8
Chan et al., 2011 China	2-arm RCT	Examine effectiveness of psychoeducati	Stage III or IV Lung cancer survivors	Lung (III -IV)	During Tx	Not assessed	N=140 INT: 70 CONT: 70	Usual care. 5-7 min briefing of radiotherapy procedure and	Psychoeducational intervention consisting of education sessions on symptom management and	F2F Unclear if group or individual	HP: Registered nurses with at least 2 years of clinical	4 weeks. 40-min education delivered one week prior to	6

		on intervention on anxiety, breathlessness and fatigue	aged 16 years and older who were scheduled to receive palliative RT.					skin care side effects; and optional group talk session on general care	progressive muscle relaxation coaching.		experience. 2-day training session on INT and progressive muscle relaxation provided.	commencing RT, and repeated three weeks after beginning RT.	
Dodd et al., 2010 USA	3-arm RCT	Evaluate effectiveness of home-based exercise on CRF management.	Adult female cancer survivors beginning chemo treatment.	Breast, colorecta 1, ovarian	Group 1 (INT): During & Post Tx Group 2: Post Tx	Not assessed	N= 119 (Group 1): 44 (Group 2): 36 CONT: 39	Usual Care. No exercise prescription.	Pro-Self Fatigue Control Program. Home-based exercise training intervention consisting of three main components: CRF knowledge, self-management skills to manage CRF, coaching to support the continuation of positive behaviours. Included a tailored exercise prescription that was adjusted through weekly follow-up phone calls. Group 1: INT provided before & after chemo	Home based Individual	Unclear – nurse?	Group 1: 1 year Group 2: 6 months	6
Donnelly et al., 2011 UK	2-arm pilot RCT	Evaluate efficacy of a physical activity behavioural change intervention in managing CRF among gynaecologica l cancer survivors	Adult gynaecologi cal cancer survivors currently undergoing treatment or post- treatment (<3 years of diagnosis)	Gynaecol ogical cancer (stage I- III)	Post Tx	Moderate (4- 6) or severe (7-10) on a scale from 1 to 10	N=33 INT=16 CONT=17	Usual care (not specified).	Home-based physical activity intervention (walking and strengthening exercises). Consisted of initial physiotherapist consultation, subsequent weekly phone calls and a final F2F consultation.	F2F & Telephone Individual	Unclear – Physiotherapist?	12 weeks. At least 5 times a week (30 minutes)	7
Fillion et al., 2008 Canada	2-arm RCT	Effectiveness of stress- management and psychoeducati on program on fatigue, energy level and QOL	Adult female breast cancer survivors who had completed treatment at least 2 years prior	Breast	Post Tx	Not assessed	N=87 INT:44 CONT:43	Usual care. General information	Stress management psychoeducation and supervised physical activity. Participants also given home-based relaxation assignments for stress/fatigue management.	F2F & Home- based Group	HP: Kinesiologist or research nurse supervised walking training. 2 oncology nurses delivered psychoeducation sessions. (Trained & supervised by a health psychologist in CBT – 10hours & 6hours reading)	4 weeks. 4 weekly group meetings of 2.5hrs and one telephone booster session (5–15 min)	8
Foster et al., 2016 UK	2-arm RCT	Increase self- efficacy to manage CRF	Adult patients experiencing CRF who have completed primary treatment at least 5 years prior	Any	Post Tx	≥ 4 on an 11-point rating scale	N=163 INT: 83 CONT: 76	Usual Care. Fatigue education leaflet	Self-management support web program. Five sessions (1. Intro. 2. Goal setting. 3. Exercise, diet, sleep, work, home. 4. Thought & feelings. 5. Talking to others). Additional activities: patient stories (video & text) used to provide examples of CRF management; links to relaxation and mindfulness	Web-based Individual	Self-directed (online)	Participants given 6-week access. Five (2 compulsory, 3 non- compulsory), 30-min sessions, delivered weekly.	7

									training and information on financial support.				
Freeman et al., 2015 USA	3-arm RCT	Evaluate the benefit of imagery-based education on QOL	Adult breast cancer survivors a least 6 weeks post treatment	Breast	Post Tx	Not assessed	N= 118 Live Delivery: 48 Telemed: 23 CONT:47	Waitlist control	Imagery-based group behavioural intervention. First 4 sessions separated into three modules consisting of didactic education, group interaction and guided imagery. Brief 10 min weekly phone calls to encourage home-based practice (continued for 3 months post treatment). Study compared live and telemedicine deliveries of intervention.	F2F or telemedicine Group	HP: Licensed professional counsellor, and a family medicine physician	5 weeks. Five 4hr sessions delivered weekly.	6
Galiano- Castillo et al., 2016 Spain	2-arm RCT	To improve adverse effects in breast cancer survivors.	Adult female breast cancer survivors who had completed adjuvant therapy	Breast (I -IIIA)	Post Tx	Not assessed	N=81 INT: 40 CONT: 41	Usual care- written exercise information	Internet-based exercise program tailored to each participant. Each program session consisted of warm-up, resistance & aerobic training, and cool-down. Program staff monitored participant performance and feedback remotely.	Telemedicine Web-based Individual	Unclear: e-cuidate research staff?	8 weeks. Three 90-min training sessions per week (non- consecutive days)	5
Godino et al., 2006 Spain	2-arm RCT	Investigate the impact and satisfaction of a nursing education program on fatigue severity	Gastric and Colon cancer survivors between 30 and 75 years – before, during and after chemotherap y treatment	Gastric & Colon	Post and During Tx	Not assessed	N=40 INT: 23 CONT: 17	Usual care. Written information	Patient education program that included training, counselling, and educational materials.	F2F Individual	HP: Oncology nurses	Three sessions. Session 1 – first cycle of chemo, session 2, second cycle f chemo, session 3, 1-month after treatment.	4
Gokal et al., 2015 UK	2-arm RCT	To improve psychosocial health outcomes among breast cancer patients	Breast cancer survivors aged 18 to 75 about to commence chemo.	Breast (I- III)	During Tx	Not assessed	N=50 INT: 25 CONT: 25	Usual care (not specified)	Home-based walking intervention consisting of moderate intensity walking and guidance recommendations (intervention booklet).	Home-based Individual	Unclear – research staff?	12 weeks	4
Golsteijn et al., 2018 Netherlands	2-arm RCT	To increase physical activity in prostate and colorectal survivors.	Adult prostate and colorectal cancer survivors undergoing curative treatment or who have completed primary treatment up to a year prior.	Prostate & Colorecta 1	During and Post Tx	Not assessed	N=510 INT: 266 CONT:244	Waitlist control	Computer-tailored physical activity program. Participants received physical activity advice (via a secure website and mail) automatically generated from a message library, questionnaire data and computer data-driven decision rules. Also had access to interactive content on website comprising of home exercise instruction videos, expert consultation with physical therapist, discussion groups, and other background information.	Web-based Individual	No delivery agent: guidance automatically generated using computer decision making	3-4 months. Three computer- tailored advice messages delivered each month. Three sessions.	6
Gregoire et al., 2020 Belgium	2-arm RCT	To improve fatigue and associated symptoms	Adult cancer survivors who have completed active treatment	Any	Post Tx	(Score of at least 4 out of 10 on 1 of these 6 items: physical	N=95 INT: 48 CONT: 47	Waitlist control (received intervention 4 months later).	Group intervention combining self-care and self-hypnosis. 8 sessions comprising of relaxation exercises, information and 15-min self- hypnosis (under therapist	F2F Group	HP: Anaesthetist and an international expert in hypnosis.	8-week duration. 8 Weekly, 2-hour sessions.	4

			since less than a year.			fatigue, moral fatigue, depression, anxiety, fear of recurrence, ruminations)			supervision) at the end of each session. Participants competed self-care tasks at home in between sessions.				
Hoffman et al., 2017 USA	2-arm pilot RCT	To improve self-efficacy for fatigue self- management	NSCLC survivors at least 21 years old who were undergoing primary care surgery	NSCLC	During Tx	Not assessed	N=87 INT: 47 CONT:40	Usual care standard medical care from health care providers	Perceived self-efficacy - enhancing (PSE) CRF self- management education. Content based on the NCCN CRF guidelines. Nurse home visit scheduled when pt. ready to begin exercise. Exercise program conducted at home via Nintendo Wii Fit Plus (nurse installed console at participants home) and involved incremental increase of walking minutes to result in 30 minutes of walking per day by end of the intervention. Balance exercises also completed 5 days a week. Nurse adjusted exercise prescription each week.	F2F & Phone Home- based Individual	HP: Primary registered nurse	6 weeks duration. End goal to achieve 30 min of walking for 5 days during the week.	8
Johns et al 2015 USA	2-arm pilot RCT	Examine efficacy of mindfulness- based stress reduction for CRF	Adult cancer survivors who reported CRF for longer than 8 week and were at least 3 months post treatment.	Any	Post Tx	FSI ≥4	N=35 INT: 18 CONT: 17	Waitlist control	Mindfulness based stress reduction. Guided training during class and through audio recordings outside of class on mindfulness meditation practices (i.e., body scan, sitting meditation, hatha yoga, walking meditation, and compassion meditation). Brief psychoeducation related to CRF.	F2F Group	HP: Instructor with 6 years MSBR teaching experience (MSBR Teacher Certification)	7 weeks. 2hrs each week.	6
Johns et al., 2016 USA	2-arm pilot RCT	Examine efficacy of mindfulness- based stress reduction for CRF compared to an active comparator	Adult breast and colorectal patients with non- metastatic disease more than 3 months and less than 5 years post- treatment (RT, chemo) experiencing severe fatigue.	Breast & Colorecta 1 (Stages 0-III)	Post Tx	FSI≥4	N=71 INT:35 CONT:36	Attention Control. Psychoeducatio nal support. 8- week group class 2hr weekly. Education on fatigue management, group discussion weekly readings.	Mindfulness based stress reduction. Guided training during class and through audio recordings outside of class on mindfulness meditation practices (i.e., body scan, sitting meditation, hatha yoga, walking meditation, and compassion meditation). Brief psychoeducation related to CRF.	F2F Group	HP: Physician and a doctoral-level clinical health psychologist with 9 and 3 years of MBSR teaching experience, respectively.	8 weeks. 2hrs each week.	7
Johnston et al., 2011 USA	2-arm pilot RCT	To relieve CRF	Breast cancer survivors between 18 and 65 who had completed	Breast	Post Tx	BFI≥4	N=13 INT: 6 CONT: 7	Usual care. Received normal practice from their personal physicians which included	Education + acupuncture. Education component involved teaching patients self-care techniques alongside an eight-session program that involved administering acupuncture. Acupuncture	F2F Individual	HP: Acupuncturist with over 10 years of cancer experience.	8 weeks. 4- weekly self-care sessions. 8 weekly acupuncture sessions.	6

Lam et al., 2018 China	2-arm RCT	To promote physical activity, reduce fatigue,	primary treatment and were cancer free. Individuals aged 9 -18 years diagnosed	Any	During Tx	Not assessed	N= 70 INT: 37 CONT: 33	pharmacological and non- pharmacologic options. Attention control: 15- minute health education talk,	administered according to STRICTA Acupuncture protocol. Participants also received usual care. Integrated physical programme with 28 home visits from nurse coaches over a 6-month period. Coaches provided physical	F2F, Home visit Individual	HP student: Nursing student (n=16) (enrolled in bachelor's	6-months. Coaches visited their assigned participants	5
		enhance self- efficacy muscle strength and QOL in children with cancer	with cancer within previous 2- months					medical follow up, home visits mimicking the INT, & placebo INT activities (playing chess/cards, diet & influenza advice).	activity management strategies and health information.		degree year 2 or above. Attended a 1-day training workshop and were required to pass 2 exams).	twice per week for the first 4 weeks, and then once per week for 20 weeks. Each visit = 1hr.	
Lee 2014 South Korea	2-arm pilot RCT	To promote exercise and dietary behaviours for breast cancer survivors.	Adult breast cancer survivors who completed cancer within 12 months prior and were diagnosed within 2 months prior	Breast (0-III)	Post Tx	Not assessed	N=59 INT: 30 CONT: 29	Attention Control: 50pg educational booklet on exercise and diet	Web-based self-management exercise and diet intervention. 5 modules comprising of exercise and dietary education, assessments, action planning and automatic feedback.	Web-based Individual	Self -directed (no interaction with any health professionals)	12 weeks. Pts encouraged to access the platform twice weekly.	5
Li et al., 2018 Hong Kong/ China	2-arm RCT	To promote physical activity, reduce fatigue and enhance self-efficacy	Cancer survivors aged 9-16 years who had completed cancer treatment at least 6- months previously	Any	Post Tx	FS-C≥25	N=222 INT: 117 CONT:105	Attention control. 4 days engaged in different leisure activities organised by a community centre over a 6- month period (e.g., cartoons, chess, handcraft, health education on influenza and diet, theme parks, museums)	Adventure-based training program conducted at campsite. Comprised a 40-min briefing session covers brief health education components, and adventure activities with increasing levels of difficulty.	F2F Group (12 participants session)	HP: 2 adventure- based trainers & registered nurse	4 separate training days delivered 2 months apart (2 weeks after randomisation and at 2, 4 and 6 months)	2
Mooney et al., 2016 Canada	2-arm RCT	To reduce chemo related symptoms	Adult cancer survivor beginning a course of chemo (excl. concurrent RT)	Any	During Tx	Not assessed	N=358 INT: 180 CONT:178	Attention control. Enhanced usual care – Automated telephone interactive voice response symptom reporting system without nurse care. Reminded to call practitioner for concerns.	Telephone-based interactive voice response system included daily home monitoring of chemo symptoms with nurse provider alerts for poorly controlled symptoms. System used patient-reported data to provide tailored automated algorithm-based self-care management messages. Messages comprised of coaching content based on symptom severity level and other habits. Poorly managed symptoms (fatigue at level	Telephone based symptom reporting system Individual	HP: Nurse practitioner & automated service	Symptoms reported daily during chemo for mean of 77 days	5

Naraphong et al., 2015 Thailand	2-arm pilot RCT	To improve fatigue related symptoms (fatigue, symptom distress, physical fitness, sleep, mood disturbance)	Adult female breast cancer survivors receiving adjuvant chemo	Breast	During Tx	Not assessed	N=23 INT: 11 CONT: 12	Attention control. Typical support services provided through the cancer centre, plus 2 booklets on 'general issues about breast cancer' and self-care activities for chemo, DVD on self-care after breast surgery & weekly phone calls.	<ul> <li>≥4/10 for 3 days) triggered calls from nurses who then provided more intensive care.</li> <li>Program comprised of exercise interventions tailored to relevant to Thai physical activities.</li> <li>12-week self-directed culturally sensitive exercise program.</li> <li>Initial training session included and an exercise kit containing written instructional materials, a pedometer, and exercise logs.</li> <li>Subsequent intervention sessions were scheduled during regular clinic visits.</li> </ul>	F2F Individual	Not recorded who provided exercise prescription Peer: Family member invited as a coach to help the participant identify appropriate activities, self- directed	12 weeks. Initial session 45min, subsequent sessions 30- 40min. Weekly phone calls to monitor progress	6
Purcell et al, 2010 Australia	Factorial RCT (4- arm RCT)	Reduce severity of fatigue in RT pts.	Adult cancer survivors scheduled for 20 or more days of curative RT.	Any	During Tx	Not assessed	N=110 Group 1 (Pre- and post- RFES): 27 Group 2 (Pre- RFES): 28 Group 3 (Post- RFES): 27 Group 4 (No RFES): 28	Usual care. One-on-one verbal nursing education & flyer on generic fatigue information.	Radiotherapy fatigue education and support (RFES). Content on RT processes, treatment side effects, and behavioural strategies to reduce fatigue. Education delivered via PPT presentation 1 week prior to RT planning and/or 1-2 weeks after RT completion.	F2F Group	HP: Nurse	60min education session.	5
Ream et al., 2006 UK	2-arm RCT	Evaluate a supportive/ educative approach to managing CRF	Cancer survivors aged 18-70 due to commence first cycle of chemo	Non- Hodgkin' s lymphom a, gastroint estinal, non- small cell lung, colorecta l, breast, or unknown primary cancer.	During Tx	Not assessed	N=103 INT: 48 CONT: 55	Usual care. Assessment of fatigue and education provided in ad hoc manner.	Assessment/monitoring of fatigue; education on fatigue; coaching in self-care; and provision of emotional support. Self-care counselling on energy conservation and management of activity and functioning.	F2F, Home visit Individual	HP: Cancer nurse with counselling qualifications and CRF knowledge	Intervention provided over the first three treatment cycles (3 months).	7
Ream et al., 2015 UK	2-arm pilot RCT	Reduce fatigue severity	Adult cancer survivors scheduled for first course of intravenous chemo	Breast, colorecta l, lymphom a	During Tx	Moderate fatigue or higher (NCCN definition - a score of 4 or above on a numeric rating scale from 1 to 10	N=44 INT: 23 CONT: 21	Usual care- screening for fatigue, limited self-care advice and referral.	Resource pack containing 'Coping with Fatigue' booklet, fatigue diary & handbook. Motivational interviewing delivered via telephone.	Telephone Individual	HP: Cancer Nurse received 10-week module on motivational interviewing	Three calls delivered over 3 treatment cycles. First call approx. 40min, final two calls 20min	6

Reif et al., 2013 Germany	2-arm RCT	To reduce perceived fatigue in cancer survivors	Adult cancer survivors diagnosed with malignant tumours (excl. brain cancer pts.) who have completed active treatment.	Malignan t cancers excl. brain metastase s	Post Tx	Moderate (4- 6) or severe (7-10) on a scale from 0 to 10 (NCCN CRF guidelines)	N=261 INT: 132 CONT:129	Waitlist control	Patient education program. Content included information on causes and fatigue management, time, energy, and sleep management, coping with emotions and adopting appropriate management strategies.	F2F Group (groups of 8)	HP: Nurses and psychologists (authors note the intervention can be carried out by any HP). Trainers attended 2-day workshop.	6 weeks. 6 sessions, 90min each delivered weekly.	7
Ritterband et al., 2012 USA	2-arm pilot RCT	To improve insomnia in cancer survivors	Cancer survivors aged 21 or older, in remission at least 1 month post treatment	Any	Post Tx	Self-reported experiences of fatigue (no questionnair e)	N=28 INT: 14 CONT: 14	Waitlist control	Internet CBT program with education on sleep hygiene, sleep restriction stimulus control, and relapse prevention. Program incorporates tailored feedback.	Web-based Individual	Unclear. 'research staff'	Can be completed over 6-9 weeks. 6 modules (approx. 45- 60min to complete)	8
Sandler et al., 2017 Australia	2-arm RCT	To improve post-cancer fatigue	Breast & colon cancer survivors who have completed primary treatment between 3 and 12 months prior	Breast & colon	Post Tx	SOMA≥3	N=46 INT: 22 CONT: 24	Attention control. Education package comprising of written CBT booklet & and single session with exercise physiologist and clinical psychologist	CBT and graded exercise therapy program. Comprised of 5 consultations with exercise physiologist and 6-8 sessions with a clinical psychologist. Compulsory modules on activity pacing, graded exercise therapy, sleep-wake cycle management, neurocognitive disturbance management. Optional modules on coping enhancement, and depression and anxiety management.	F2F Individual	HP: Exercise physiologist Clinical psychologist	12 weeks	10
Savard et al., 2005 Canada	2-arm RCT	To improve chronic insomnia	Adult female breast cancer survivors with insomnia (caused/ aggravated by cancer) who had completed primary treatment at least 1 month prior	Breast (I -III)	Post Tx	Not assessed	N=57 INT: 27 CONT: 30	Waitlist control	CBT program that combined the use of stimulus control, sleep restriction, cognitive therapy, sleep hygiene, and fatigue management. Multimodal approach combined behavioural, cognitive, and educational strategies.	F2F Group	HP: Master-level psychologist (experience with INT protocol).	8 weeks. 8 sessions approx. 90 min delivered weekly	5
Schjolberg et al, 2014 Norway	2-arm RCT	To manage fatigue in women with early breast cancer	Adult female cancer survivors who had competed a form of curative treatment for early-stage breast cancer	Breast (I- II)	Post Tx	NRS≥2.5	N = 160 INT: 79 CONT: 81	Usual care. Standard F2F care (info on general issues associated with living with cancer) provided by nurses and radiotherapists in outpatient clinic	Educational intervention to improve knowledge and skills. PPT presentation, patient booklet, group discussions. Education comprised of strategies to reduce fatigue, promote balance between activity, guide energy conservation and promote restorative activities.	F2F Group (approx. 10 per group)	HP: 2 cancer nurses experienced in education.	3 weeks. Three 2hr sessions delivered weekly	6

Smith et al., 2019 USA	2-arm RCT	To manage pain and associated symptoms in cancer survivors	Adult breast cancer survivors with chronic pain	Breast	During & Post Tx Any time point	Not assessed	N=89 INT:37 CONT: 52	Waitlist control	Online symptom self- management curriculum. Comprises of online videos, cognitive reframing exercises, mind-body exercises, compulsory online introductory group meeting.	Web-based Individual (one online group meeting)	Self-directed – asynchronous. Live online group meeting led by master's level therapist	10 weeks. 10 online sessions delivered weekly	1
Spahn et al., 2013 Germany	2-arm RCT	To improve chronic fatigue symptoms of women with stage I-IIIA breast cancer	Adult female breast cancer survivors who have completed primary treatment at least 3 months prior.	Breast (I- IIIA)	Post Tx	>40mm on 100mm VAS	N=64 INT: 32 CONT: 32	Attention control – home based walking treatment	Walking treatment and multimodal mind-body medicine treatment (MMMT) program. MMMT included nutrition counselling, relaxation exercises, physical exercises, stress reduction, cognitive restructuring, and hydrotherapy. Supervises walking sessions.	F2F Group (10 -20 per group)	HP: Sports therapist delivered walking sessions	10 weeks. 10 walking sessions and one 6-hr mind body session completed weekly.	5
Stanton et al., 2005 USA	3-arm RCT	To improve fatigue, energy and cancer specific distress.	Adult female breast cancer survivors 6 weeks post- surgery	Breast (I -II)	Post Tx	Not assessed	N=558 INT 1:187 INT 2: 184 CONT:187	Usual care – 43pg 1994 National Cancer Institute- Facing <i>Forward</i> publication	<ul> <li>INT 1: Videotape – 23 min film addressing re-entry challenges in four life domains: physical health, emotional well-being, interpersonal relations, and life perspectives.</li> <li>INT 2: psychoeducational counselling. One F2F and one telephone session incorporating a review of pt. cancer-related concerns in four domains, setting, action planning, Video tape (INT 1), 60pg manual.</li> </ul>	INT 1: self- directed, video INT 2: F2F, telephone Individual	HP: Trained cancer educators (social worker, psychologist) masters or PhD level education received full day training session (guided by INT manual).	INT 2 F2F session: 80min; telephone session: 30 min (delivered 2 weeks after F2F session)	6
Steel et al., 2016 USA	2-arm RCT	To reduce depression, pain, fatigue in an advanced cancer population	Advanced cancer survivors and family caregivers aged at least 21 years	hepatocel lular, gallbladd er, neuroend ocrine, pancreati c carcinom a, cholangi ocarcino ma, or other primary cancers that had metastasi zed to the liver	Post Tx	Not specified – authors expected survivors would experience symptoms over the course of the INT	N=261 INT: 144 CONT:117	'Enhanced usual care. Usual care provided by medical care + contact by care- coordinator & appropriate referral if symptoms for pain and depression were high	Web-based stepped collaborative care intervention comprising of psychoeducational website & collaborative care coordinator. Coordinator provided CBT, and recommendation for symptom management. Website included symptom tracking, symptom info, diary taking, peer chat room, audio-visual library, resource library.	Web-based, Telephone & F2F Individual	HP: Therapist trained in CBT and with experience in cancer care (master's or PhD level)	6 months. Telephone contact with care coordinator fortnightly, F2F contact every 2 months.	7
Vallerand et al., 2017 Canada	2-arm RCT (phase II)	Increase aerobic exercise behaviour in haematologic cancer survivors	haematologi c cancer survivors aged 18-80	leukaemi a, non- Hodgkin lymphom a, Hodgkin	Unclear	Not assessed	N=51 INT: 26 CONT: 25	Attention control. Self- directed exercise (Canada's Physical Activity	Exercise telephone counselling. Consisting of exercise prescription, physical activity guidelines and counselling sessions incorporating action planning, goal setting, coping	Telephone Individual	Unclear	12 weeks. Weekly telephone counselling sessions.	7

		(content on fatigue)		lymphom a				guideline + exercise goal (60- 300min/week of weekly moderate-to- vigorous aerobic exercise)	planning, self-monitoring, and self-regulation activities.				
Van der Lee et al, 2012 Netherlands	2-arm RCT	Reduce severe chronic fatigue in cancer survivors	Adult cancer survivors who had completed treatment at least 1 year prior.	Any	Post Tx	CIS-FS ≥35	N=100 INT: 72 CONT: 28	Waitlist control	Mindfulness-based cognitive therapy (MBCT). Comprised of information on the stress-coping model, and strategies for coping with pain and fatigue, recognising unpleasant experiences, coping with negative emotions and communication.	F2F Group	HP: Therapist. Had received mindfulness- based stress reduction training courses with Kabat Zinn	9 weeks. Eight weekly sessions of 2.5 h and one 6h session, plus one 2.5 h follow-up session 2 months after the ninth session. Total hrs 28.5h.	6
Van Waart et al., 2015 Netherlands	3-arm RCT	To maintain or enhance physical fitness and minimise fatigue in patients undergoing adjuvant chemo.	Adult cancer survivors scheduled to undergo chemo.	Breast, colon cancer	During Tx	Not assessed	N=230 OnTrack: 76 Onco- Move: 77 CONT: 77	Usual care. Varied depending on hospital setting. Did not involve exercise	Onco-Move. Low intensity home-based physical activity program. OnTrack - moderate-to-high intensity, combined resistance and aerobic exercise program supervised by physical therapists.	Onco-Move: Home based Individual OnTrack- F2F, individual	HP Onco-Move specially trained nurses OnTrack: specially trained physical; therapist	From first cycle of chemo to 3 weeks after last chemo cycle. Onco-Move: 30 min of physical activity per day, 5 days per week. OnTrack: 2 60min sessions	3
Van Weert et al., 2010 Netherlands	3-arm RCT	To reduce cancer related fatigue	Adult cancer survivors who received treatment at least 3 months prior	Any	Post Tx	Reporting increased fatigue to physician	N=213 (PT+CBT) :76 PT: 71 CONT: 66	Waitlist control	Rehabilitation program. PT: 24 hrs of individual physical training and 24hrs of group sports & activities. Training based on pt. goals. Information on exercise & fatigue provided. PT+CBT: Physical training combined with CBT. See above for PT component. CBT consisted of information and coping strategies for stress, exercise, physiology, and relaxation. Problem solving, goals setting, decision making skills taught.	PT: F2F Individual, Group CBT: F2F Individual	HP PT: 2 physical therapists with cancer survivor experience. CBT: 2 psychologists Trained with a manual.	per week. 12 weeks. PT: 2 times per week for 1hr. CBT: once a week, 2 hours per session	7
Vargas et al., 2014 USA	2-arm RCT	To improve sleep quality and fatigue	Adult female early breast cancer survivors who had undergone lumpectomy or mastectomy	Breast (I- III)	During Tx	Not assessed	N=240 INT: 120 CONT:120	Attention control. Standard care + 1 day 6h psycho- education session	Cognitive behavioural stress management (CBSM) program comprising of didactic presentations, instruction in CBT techniques and relaxation exercises, and home practice exercises in participant workbook.	F2F Group (up to 8)	Unclear	10 weeks. Ten 2hr sessions delivered weekly.	3
Wang et al., 2011 Taiwan	2-arm RCT	To improve fatigue QOL, sleep disturbances,	Female cancer survivors aged 18 to	Breast (I- II)	During Tx	Not assessed	N=72 INT: 35 CONT: 37	Usual care – not specified	Low-to-moderate intensity walking program. Incorporated heart rate ring monitor, pedometer, weekly phone call,	F2F, phone Individual	HP: Nurses	6 weeks. Sessions and phone calls delivery weekly.	6

Wangnum et al., 2013	2-arm RCT	exercise self- efficacy, exercise capacity and exercise behaviours To improve fatigue	72 with early breast cancer who had undergone surgery and were scheduled for chemo Lung cancer survivors aged	Lung (III-IV)	During Tx	Not assessed	N=60 INT: 30 CONT: 30	Unclear if usual care. Cancer nurse provided	meetings & exercise diary, and a role model story. Multidisciplinary education program in self-care fatigue. Sessions comprised of breathing	F2F Individual	HP: Physical therapist, nutritionist,	9 weeks. Four F2F sessions delivered every	7
Thailand			between 45 and 65 years who had received at least one round of 47 platinum- based chemo					service & information. Pts received 30min training on how to exercise during chemo sessions, and instructions to take home and review.	exercises, physical activity coaching and exercises, nutritional assessment and advice and information & exercises for coping with depression.		psychological nurse	three weeks. Session 1 was approx. 90min and remaining three sessions were 60min each.	
Willems et al., 2017a & 2017b Netherlands	2-arm RCT	To manage psychosocial and lifestyle- related issues	Adult cancer survivors who had completed primary treatment at least 4 weeks prior, but longer than 56 weeks.	Any	Post Tx	Not assessed	N=518 INT: 265 CONT:253	Waitlist control	Web-based psychosocial education program. 8 modules covering return to work, fatigue, anxiety and depression, social relationships and intimacy issues, physical activity, diet, and smoking cessation.	Web-based Individual	Self-directed online	Given access to the program for 6 months.	9
Yates et al., 2005 Australia	2-arm RCT	To improve CRF	Adult female breast cancer survivors commencing adjuvant chemo.	Breast (I- II)	During Tx	Not assessed	N=110 INT: 57 CONT: 53	Attention control. General cancer education sessions with oncology nurse. One F2F session, two telephone sessions. Delivered weekly.	Fatigue psychoeducational support program. Identification of pt. needs and coping strategies provided accordingly. Preparation of fatigue management plan/ Patient booklet (published by Oncology Nursing Society) to supplement sessions.	F2F. telephone Individual	HP: Nurses	3 weeks. Three 10-20min sessions delivered weekly. Session 1 F2F, Session 2&3 via telephone.	7
Yun et al., 2012 South Korea	2-arm RCT	To improve CRF	Cancer survivors aged 20 to 65 years who have completed primary treatment within the past 24 months	Any (I- III)	Post Tx	BFI ≥ 4 (moderate- severe fatigue)	N=273 INT: 136 CONT:137	Waitlist control	Internet-based education CRF program based on NCCN program guidelines that covered a general introduction to CRF energy conservation, physical activity, nutrition, sleep hygiene, pain control, and distress management.	Web-based Individual	Self-directed	12 weeks.	8
Zhang et al., 2018 China	2-arm RCT	Reduce fatigue severity	Female ovarian cancer survivors between 18-	Ovarian	During Tx	Moderate (4- 6) or severe (7-10) on a scale from 0 to 10	N=72 INT: 36 CONT: 36	Usual care. Drug education, balanced diet recommendatio n, education on	Nurse-led home-based exercise and cognitive behavioural therapy. Comprising of online CBT sessions, self-help education manual, exercise	F2F, web- based, telephone, home based	HP: 5 nurses with master's level education received INT and CBT training	12 weeks. 12 online CBT sessions delivered once weekly (approx.	9

80 years	(	(NCCN CRF	chemo for	education manual, telephone	Individual	from specialist	1hr). Weekly
who had	1	guidelines)	ovarian cancer.	counselling on fatigue		cancer nurse with	nurse telephone
completed				information and exercise		over 10 years'	for exercise
surgery and				methods. Option to receive		experience in	component.
were				online interventions at home in		gynaecology.	
receiving				or clinic.			
adjuvant							
chemo.							

Abbreviations: BFI - brief fatigue inventory; CBT - cognitive behavioural therapy; Chemo - chemotherapy; CIS-FS - Checklist of individual strength fatigue severity subscale; CONT - control; F2F - face-to-face; FS-C - Fatigue scale-child; FSI - fatigue symptom inventory; HP - health professional; INT - intervention; NCCN - national comprehensive caner network; NRS - numeric rating scale; pg. - page; PPT - Microsoft PowerPoint; PT - physical training; Pts. - patients; QOL - quality of life; RCT - randomised controlled trial; RT - radiation therapy; SOMA - somatic fatigue subscale; Telemed - telemedicine; Tx - treatment; VAS - visual analogue scale.

# Appendix 5: Risk of Bias for each Included Study (Study 1)





- D1 Randomisation process
- D2 Deviations from the intended interventions
- D3 Missing outcome data
- D4 Measurement of the outcome
- D5 Selection of the reported result
|                       |   | _             |   |   |          |            |
|-----------------------|---|---------------|---|---|----------|------------|
| Naraphong<br>2015     | + |               | ! | ! | Ŧ        |            |
| Purcell               |   |               |   |   | <u> </u> |            |
| 2010                  | + | !             | ! |   | !        |            |
| Ream 2006             | ! |               | ! | ! | !        | •          |
| Ream 2015             | Ŧ | +             | ! | ! | +        | !          |
| Reif 2013             | • | !             | Ŧ | ! | !        | •          |
| Ritterband            |   | 1             | 4 | Ŧ | 1        | (!)        |
| 2012                  | + | •             | - | + | •        | <u>·</u>   |
| Sandler               |   | 1             |   | 1 |          |            |
| 2017                  | • | •             |   | • | •        |            |
| Savard<br>2005        | ! |               |   | ! |          | •          |
| Schjolberg<br>2014    | ! |               |   | ! | !        | •          |
| Smith 2019            | Ŧ | !             | Ŧ | ! | !        | !          |
| Spahn 2013            | + |               |   | ! | !        | •          |
| Stanton et<br>al 2005 | + | !             | • | ! | !        | •          |
| Steel et al.,<br>2016 | ÷ | !             | + | ! | !        | !          |
| Vallerand<br>2017     | ! | !             |   | ! | !        | •          |
| Van der Lee<br>2012   | ! | !             | ! | ! | !        | !          |
| Van Waart<br>2015     | • | !             | ! | ! | !        | !          |
| Van Weert<br>2010     | • | !             |   | + | !        | •          |
| Vargas<br>2014        | ! |               | ! | ! | !        | •          |
| Wang 2011             | ! |               |   | ! | !        | •          |
| Wangnum               | + | Ŧ             |   | 1 | Ŧ        |            |
| 2013                  |   | $\overline{}$ | - |   |          | $\bigcirc$ |
| Willems               |   |               |   |   |          | _          |
| 2017a &<br>2017b      | + | +             | ! | ! | +        | !          |
| Yates 2005            | + | •             | ! | Ŧ | !        | •          |
| Yun 2012              | • | !             | ! | ! | !        | !          |
| Zhang 2018            | ÷ | !             | ! | ! | !        | !          |

## Appendix 6: SMS Taxonomy Components in each study (Study 1)

Study	1. Information about CRF and/its management	2. Information about managing psychosocial consequences of illness.	3. Information about available social support resources and navigating transitions.	4. Provision of /agreement on goal setting and specific clinical action plans and/or urgent problems.	5. Regular clinical review.	6. Monitoring of condition with feedback.	7. Practical support with adherence (medication or behavioural)	8. Provision of equipment	9. Provision of access to advice or support when needed	10. Training rehearsal to communicate with healthcare professionals	11. Training rehearsal for lifestyle adaptations and for everyday activities	12. Training rehearsal for psychological strategies	13. Social support	14. Lifestyle advice and support	# of SMS taxonomy components per study
Abrahams 2017	✓	✓		✓ ✓						<b>√</b>	✓	✓	✓	✓	8
Armes 2007	$\checkmark$			✓		$\checkmark$					$\checkmark$	<ul> <li>✓</li> </ul>		✓	6
Bantum 2014	<ul> <li>✓</li> </ul>	✓		✓		✓	✓			$\checkmark$		✓	✓	$\checkmark$	9
Barsevick 2004	$\checkmark$			✓		$\checkmark$	$\checkmark$				$\checkmark$			$\checkmark$	6
Barsevick 2010	✓			✓		$\checkmark$	✓				$\checkmark$			✓	6
Bennet 2007				✓	✓		✓				$\checkmark$			✓	5
Bruggeman-Everts 2017	✓	$\checkmark$		✓	✓	$\checkmark$	✓					$\checkmark$		✓	8
Chan 2011	$\checkmark$	✓					✓				$\checkmark$	✓		✓	6
Dodd 2010	$\checkmark$				$\checkmark$	$\checkmark$	$\checkmark$				$\checkmark$			$\checkmark$	6
Donnelly 2011	-			✓	· ✓	$\checkmark$	✓				$\checkmark$	✓		$\checkmark$	7
-					-										
Fillion 2008	$\checkmark$	$\checkmark$		$\checkmark$		$\checkmark$	$\checkmark$				✓	~		$\checkmark$	8
Foster 2016	$\checkmark$	$\checkmark$		✓		✓	✓						✓	$\checkmark$	7
Freeman 2015					✓	✓	✓				✓	✓	✓		6
Galiano-Castillo 2016						$\checkmark$	✓		$\checkmark$		$\checkmark$			$\checkmark$	5
	$\checkmark$					✓	✓		-					✓ <b>·</b>	4
Godino 2006				<b>√</b>		•	<ul> <li>✓</li> </ul>				<ul> <li>✓</li> </ul>			✓ ✓	
Gokal 2015															4
Golsteijn 2018				✓		✓	✓				✓		✓	✓	6
Gregoire 2020		✓									✓	✓		✓ ✓	4
Hoffman 2017	✓ ✓			✓	✓	$\checkmark$	✓	✓			✓			✓ ✓	8
Johns 2015	✓ ✓	✓					✓ ✓				✓ ✓	✓		✓ ✓	6
Johns 2016	✓ 	✓					✓				✓	✓	✓	✓ ✓	7
Johnston 2011	✓	✓		✓		$\checkmark$						✓		✓ ✓	6
Lam 2018				✓			✓				✓ ✓	✓		✓ ✓	5
Lee 2014				✓		$\checkmark$	✓				✓			✓	5
Li 2014											✓ ✓			✓ ✓	2
Mooney 2016					✓	✓ ✓			✓		✓ ✓			✓ ✓	5
Naraphong 2015				✓		✓	✓ ✓				✓		✓	✓ ✓	6
Purcell 2010	✓			✓			✓ ✓					✓ ✓		✓ ✓	5
Ream 2006	✓ ✓				<ul> <li>✓</li> <li>✓</li> </ul>	✓	✓				✓	V		✓ ✓	7
Ream 2015	•		✓	✓	✓		V							V	6
Reif 2012	✓	✓					✓ ✓				✓ ✓	✓ ✓	✓	✓ ✓	7
Ritterband 2012	✓ ✓	✓		$\checkmark$	✓	✓ ✓	✓ ✓				✓ ✓	✓ ✓		✓ ✓	8
Sandler 2017	✓ ✓	✓		✓	✓	✓	✓		✓		✓ ✓	$\checkmark$		✓ ✓	10
Savard 2005	$\checkmark$	✓			✓						$\checkmark$	<b>v</b>		✓ ✓	5
Schjolberg 2014	¥			✓	v						¥	✓	✓	✓	6
Smith 2019		✓			✓						✓	✓ ✓		$\checkmark$	5
Spahn 2013		<ul> <li>▼</li> <li>✓</li> </ul>	✓	✓	•						•	•	✓	▼ ✓	5
Stanton et al 2005	$\checkmark$	<ul> <li>▼</li> <li>✓</li> </ul>	•	•	✓		✓		$\checkmark$			✓ ✓	v √	•	6
Steel et al., 2016	<ul> <li>▼</li> <li>✓</li> </ul>	•		✓	•		▼ ✓		•	$\checkmark$	<ul> <li>✓</li> </ul>	•	v √	$\checkmark$	7
Vallerand 2017	✓ ✓	✓		<b>▼</b>			*			· ·	▼ ✓	✓	v √	✓ ✓	7
Van der Lee 2012	<b>*</b>	•					✓				▼ ✓	•	Y	✓ ✓	6
Van Waart 2015	$\checkmark$			✓		✓	<ul> <li>✓</li> </ul>				▼ ✓	<ul> <li>✓</li> </ul>		✓ ✓	3 7
Van Weert 2010	¥	✓		<b>v</b>		*	•				✓ ✓	✓ ✓		Y	
Vargas 2014		*		✓	✓		✓				✓ ✓	*	✓	$\checkmark$	3
Wang 2011	$\checkmark$			✓ ✓	✓ ✓	<ul> <li>✓</li> </ul>	✓ ✓				*	✓	¥	✓ ✓	6
Wangnum 2013	✓ ✓	✓	✓	✓ ✓	<b>v</b>	*	•			✓	✓ <b>·</b>	✓ ✓	✓	✓ ✓	7
Willems 2017a & 2017b Yates 2005	✓ ✓	<ul> <li>▼</li> <li>✓</li> </ul>	•	✓ ✓		<ul> <li>✓</li> </ul>				▼ ✓	•	<ul> <li>▼</li> <li>✓</li> </ul>	•	✓ ✓	9 7

Yun 2012	$\checkmark$	$\checkmark$				$\checkmark$			$\checkmark$		$\checkmark$	$\checkmark$	✓	$\checkmark$	8
Zhang 2018	$\checkmark$	$\checkmark$		$\checkmark$	✓	$\checkmark$	$\checkmark$		$\checkmark$			$\checkmark$		$\checkmark$	9
Total # of studies	32	23	3	30	16	26	34	1	6	5	37	31	16	46	

## Appendix 7: Reported outcomes of included studies, post-intervention, and follow-up (Study 1)

Study name	Type of study, INT vs CONT length	Relevant outcome measures	Immediate post-intervention results	Follow-up	Post intervention follow-up results
Abrahams 2017	2-arm RCT: Internet CBT (ICBT) vs waiting list for CBT, varied – max duration 6 months	Fatigue severity (CIS- FIS)	Lower fatigue scores in ICBT (MD 11.5; 95%CI 7.7-15.3, p<.0001). Large INT effect size (Cohen's d = 1.0).	No follow-up assessment completed	N/A
Armes 2007	2-arm RCT: behaviour intervention vs usual care, 9-12 weeks	Fatigue severity (VAS-F), fatigue- related distress (FOM), physical functioning (EORTC- QLQc30)	Sig. lower fatigue severity (MD: -15.9; 95%CI 2-29.8, p=0.03) and higher physical functioning (MD: 19.7;95%CI: -30.2 to -9.2; p=0.001) in behaviour intervention. No sig. difference between groups for fatigue-related distress (U test score: 183; p=0.25).	9 months post- recruitment; 6-7 months post-CTX (? post INT)	No sig. difference between groups for fatigue-related distress (U test score: 149; p=0.90) or fatigue severity (MD: -17.1; 95%CI: -1.5 to 35.7; p=0.07). Higher physical function in behaviour intervention (MD: 17.1; 95%CI: -31.7 to -2.5; p=0.02).
Bantum 2014	2-arm RCT delayed treatment design: online education vs waitlist control, 6 weeks	Fatigue severity (BFI), exercise (Godin Exercise questionnaire), fruit/vegetable intake (Block Food Frequency questionnaire)	Treatment effect was not sig. (p=0.56, effect size 0.17) for fatigue severity. No sig. effect on fruit/vegetable intake (p=0.24, effect size= 0.21), strenuous or moderate aerobic exercise (p=0.45, effect size 0.29), moderate aerobic exercise (p=0.49, effect size= 0.10), mild aerobic exercise (p=0.28, effect size= 0.10). Increased strenuous exercise (p=0.01, effect size 0.36) and stretching exercise (p=0.01, effect size 0.12) compared to control.	No follow-up assessment completed	N/A
Barsevick 2004	2-arm RCT: energy conservation and activity management vs attention control, varied	Fatigue impact (GFS), fatigue intensity (POMS-F), physical and mental fatigue (SCFS-P)	ECAM group experienced a greater decrease in fatigue impact, intensity, and physical and mental fatigue over time compared with the control group (GFS: F= 4.5; p < 0.01) (POM-F: F=3.7.	No follow-up assessment completed	N/A

	according to chemo schedule		p<0.05), (SCFS-P: F=3.1. p<0.05) respectively.		
Barsevick 2010	2-arm RCT: energy sleep enhancement vs attention control, varied according to chemo schedule	Fatigue severity(GFS, POMS-F)	No sig. group-by-time effects for fatigue severity (GFS: F=0.02; p>0.005; & POMS-F: F=0.47; p>0.05).	No follow-up assessment completed	N/A
Bennet 2007	2-arm RCT: motivational interviewing vs usual care, 4.5 months	Fatigue (Schwartz Cancer Fatigue Scale), <u>Physical</u> <u>Activity (</u> CHAMPS Physical Activity questionnaire), <u>Self-</u> <u>Efficacy for physical</u> <u>activity</u> (6-item questionnaire)	Not measured.	1.5 months post INT	No sig. difference between INT and CONT for fatigue (d=0.14) and aerobic fitness (d=0.09). High-efficacy individuals in INT increased their physical activity levels faster over 6 months than did low-efficacy individuals in the intervention group (B = 121.35, p= 0.05). In the control group, self-efficacy did not affect levels of physical activity.
Bruggeman-Everts 2017	3-arm RCT: ambulant feedback therapy (AAF) vs electronic mindfulness therapy (eMBCT) vs ambulant physical activity coaching vs psychoeducational educational emails, 9 weeks.	Fatigue Severity (CIS- FS)	Specific results not reported.	3 months post INT	Fatigue severity decreased sig. more in the AAF and eMBCT groups compared to the psychoeducational control group over time. No sig. difference between AAF and eMBCT.
Chan 2011	2-arm RCT: psychoeducation vs usual care (radiotherapy effects education); 4 weeks	Fatigue severity (Revised PFS - Chinese ver.)	Lower fatigue severity in INT. Sig. between-group difference (p=0.011) in the pattern of change in fatigue, with a small effect size (partial eta squared = 0.033).	(?) 3 months post baseline (? post INT)	No sig. difference was found in the pattern of change in fatigue overtime (baseline to 3 months post INT) (p=0.034).
Dodd 2010	3-arm RCT: exercise during & after cancer treatment vs exercise after cancer treatment	Fatigue severity (PFS)	Results not reported.	Group 1: (6 months post INT/treatme	The linear change in fatigue severity over time by group showed no sig. differences (p=0.084) among the groups.

	vs usual care, varied between groups.			nt completion)	
Donnelly 2011	2-arm pilot RCT: home based physical activity program vs usual care (not specified), 12 weeks	Fatigue severity (MFSI-SF), Clinically sig. fatigue measure (FACIT-F)	Sig. decrease in fatigue severity with INT (MFSI-SF) (p=0.046; Cohen's d=0.13). No between-group differences for FACIT-F fatigue severity (p=0.34).	6 months post INT	Sig. decrease in fatigue severity with INT (MFSI-SF) (p=0.01; Cohen's d=0.20). No between-group differences for FACIT-F fatigue severity (p=0.10).
Fillion 2008	2-arm RCT: group stress management and psychoeducation vs usual care (general information), 4 weeks	Fatigue severity (MFI), Energy Level (POMS-Vigor Scale), Fitness (VO2 Max)	Sig. between group difference for fatigue severity (p=0.03, Cohen's d = 0.49), and energy level (p=0.01, Cohen's d = 0.58) No sig. between group difference in fitness.	3 months post INT	No sig. fatigue effect over time. No sig between-group difference for fatigue severity, energy level and fitness.
Foster 2016	2-arm RCT: web-based intervention vs leaflet, 6 weeks	Fatigue severity (BFI), perceived self- efficacy for fatigue self-management (PSEFSM questionnaire, CS- SES), QOL (FACT-G, PWI, PHQ-9)	No sig. differences between groups noted for all outcomes; near- significant improvement between groups in fatigue self-efficacy (p = 0.09).	3 months post baseline (? post INT)	No sig. differences between groups noted for all outcomes; between-group difference in fatigue self-efficacy decreased becoming negligible.
Freeman 2015	3-arm RCT: live imagery- based behavioural program (LD) vs telemedicine imagery based behavioural program (TD) vs waitlist control, 5 weeks	Fatigue severity (FACIT-F)	Participants in LD and TD reported sig. less severe fatigue than waitlist (p<0.01). No differences between LD and TD groups on any outcome measure.	3 months post treatment completion (? post INT)	No effect on fatigue severity over time.
Galiano-Castillo 2016	2-arm RCT: Internet- based exercise program vs usual care (written exercise) inf, 8 weeks.	Fatigue severity (PFS – revised, Spanish ver.)	Improved total fatigue severity in INT vs CONT (p<0.001; Cohen's d= -0.89, 95CI% - 1.30 to -0.48). Large effect size.	6 months post INT	Improved total fatigue severity in INT vs CONT (p=0.002; Cohen's d=0.74, 95%CI -1.19 to -0.29). Moderate effect size.

Godino 2006	2-arm RCT: patient education program vs written information, 3 weeks	Fatigue severity (FACT-F)	Lower fatigue severity in INT (mean score $121 \pm 25$ ) compared to CONT (mean score $105 \pm 13$ ) clinically sig. however, findings not statistically sig. due to small sample size (p>0.05).	No follow-up assessment completed	N/A
Gokal 2015	2-arm RCT: home-based walking vs usual care (not specified), 12 weeks	Fatigue severity (FACT-F), Physical activity (General Practice Physical Activity Questionnaire & Yamax SW-200 Pedometer)	Sig. reduction in fatigue severity in INT vs CONT over time (F(1,48)=5.77, p=0.02). Sig. improvement in perceived levels of physical activity in INT vs CONT ( $\chi^2(3, N=50) = 17.15, p=0.001$ ).	No follow-up assessment completed	N/A
Golsteijn 2018	2-arm RCT: computer- tailored physical activity program vs waitlist control, 3-4 months	Fatigue severity (CIS- F), Physical activity: moderate-to- vigorous physical activity (MVPA) and days ≥30 min physical activity (SQUASH, ActiGraph GT3X-BT)	Sig. decreased fatigue severity in INT vs CONT (B = $-3.57$ , p = 0.02). Sig. improvement physical activity in terms of both MVPA (B = $133.55$ , p=0.04) and days with at least 30 min of physical activity (B = $0.86$ , p < .001) in INT vs CONT.	2 months post INT	INT had a sig. decrease in fatigue severity in comparison to CONT (B = $-4.16$ , p = 0.009) Sig. improvements in physical activity (MVPA: B = 267.17, p < .001; Days $\geq$ 30 min physical activity: B = 0.98, p < .001). ActiGraph assessed MVPA also increased significantly (MVPA: B = 44.60, p = 0.006), whereas the increase in ActiGraph assessed days $\geq$ 30 min PA was borderline significant (B = 0.38, p = 0.05).
Gregoire 2020	2-arm RCT: Self-care and self-hypnosis vs waitlist control, 8 weeks	Fatigue severity (MFI), Physical Activity (Garmin Vivoactive HR, self- reported number of physical activity hours per week)	All dimensions of fatigue (MFI-20) sig. improved in INT: general fatigue (p < 0.001, Cohen's d = 0.67); physical fatigue (p< 0.001, Cohen's d = 0.60); mental fatigue (p< 0.001, Cohen's d = 0.65); lack of activity (p< 0.001, Cohen's d=0.67); and lack of motivation (p=0.002, Cohen's d = 0.54). No sig. differences for physical activity (steps/day or hrs/wk).	No follow-up assessment completed	N/A

Hoffman 2017	2-arm pilot RCT:	Fatigue severity (BFI),	Sig. improvement in fatigue severity,	No follow-up	N/A
	Perceived self-efficacy -	perceived self-	sig. between-group difference [t (42) =	assessment	
	enhancing (PSE) CRF self-	efficacy (PSE) for	-9.3, p<.001; 95% CI -4.0 to -2.6;	completed	
	management education	fatigue self-	Cohen's d = $1.7$ ].	completeu	
	5	•	-		
	vs usual care (standard	management	PSE for CRF self-management: sig.		
	medical care from health	(PEFSM), CRF Self-	difference between groups [t (41.8) =		
	care providers), 6 weeks	management	7.4, p<.001; 95% CI 2.0 to 3.5; d = 1.3]		
		behaviours -walking			
		& balance – (weekly	PSE Walking: sig. difference between		
		diary that included	groups [t (52.2) = 4.3; p<.001; 95% Cl		
		prescription and	14 to 39; d = 0.8].		
		activities completed),			
		PSE for walking	PSE for Balance: sig. difference		
		duration	between groups at week 6 [t (46.9) =		
		(Survey), Fatigability	4.1; p<.001; 95% CI 6.6 to 19.5; d =		
		<ul> <li>fatigue with activity</li> </ul>	0.7].		
		(6-minute walk test –			
		6MWT)	CRF self-management walking and		
			balance exercise: increase in walking		
			and balance exercise behaviours.		
			Fatigability: sig. difference between		
			the INT and CONT group's fatigue		
			during the 6MWT [t (50.9) = -9.3,		
			p<.001; 95% CI -3.3 to -2.1; d = 1.8)		
Johns 2015	2-arm pilot RCT:	Fatigue severity,	MBSR group demonstrated sig. greater	6 months	MBSR group demonstrated sig. greater
	mindfulness-based stress	interference, &	improvement than CONT in fatigue	post INT	improvement than CONT in fatigue
	reduction (MSBR) vs	frequency (FSI)	interference p<0.001 Cohen's d =-1.43,	P	interference p<0.001 Cohen's d=-1.34,
	waitlist control, 7 weeks		fatigue severity p<0.001 Cohen's d= -		fatigue severity p<0.001 Cohen's d=-1.54,
			1.55, fatigue frequency p<0.001		fatigue frequency p<0.001 Cohen's d=-1.22
			Cohen's d=-1.08		
Johns 2016	2-arm pilot RCT:	Fatigue interference,	No sig. between-group differences for	6 months	No sig. between-group differences for
20110 2020	mindfulness-based stress	fatigue severity,	fatigue interference or fatigue	post INT	fatigue interference, fatigue severity or
	reduction (MSBR) vs	fatigue global	severity.		fatigue global improvement.
	psychoeducation 8	improvement (FSI -	Fatigue global improvement: MBSR		
	weeks		participants were significantly more		
	WEEKS	l	participants were significantly more		

		Fatigue Symptom Inventory scale)	likely than PES participants [58.8 vs. 34.3 %, respectively; $\chi^2$ (1) = 4.176, p = 0.041] to report their CRF as being moderately to completely better		
Johnston 2011	2-arm pilot RCT: Education + acupuncture vs usual care (standard care from physicians), 8 weeks	Fatigue severity (BFI)	INT had clinically sig. 2.38-point greater reduction in fatigue severity (90% CI: 0.586 to 5.014; p=0.08; Cohen's d = 1.85). Not statistically sig. due to small sample size	No follow-up assessment completed	N/A
Lam 2018	2-arm pilot RCT: Integrated physical adventure programme vs various leisure activities, 6 months	Fatigue severity (Fatigue Scale - Chinese ver.), physical activity levels (CUHK-PARCY), physical activity Self- efficacy (PASE)	Sig. lower fatigue severity (p<0.001), and higher physical activity self- efficacy (p<0.001) in INT than CONT. Physical Activity levels: not measured at this timepoint.	3 months post INT	Sig. lower fatigue severity (p=0.010, eta squared= 0.09) and higher levels of physical activity self-efficacy (p=0.004, eta squared =0.11) in INT. Sig. higher physical activity levels in INT (p=0.003, eta squared = 0.12)
Lee 2014	2-arm pilot RCT: diet and exercise web program vs 50-page booklet	Fatigue severity (BFI-K: Korean ver.), exercise & intake of fruit and veg. (exercise & dietary diaries), stage of change (Motivational readiness questionnaire: non- validated)), perceived self-efficacy (Non-validated study questionnaire)	Sig. improvement in fatigue severity (p = 0.032), higher stage of change for exercise (p < 0.0001) and fruit & veg consumption (p = 0.029) in INT. Sig. increase in self-efficacy for exercise management (p = 0.024) and to increase F&V intake (p = 0.023). Sig. increase in moderate intensity aerobic exercise for at least 150 min per week (p < 0.0001) and eating 5 servings of F&V per day (p = 0.001) in INT vs CONT.	No follow-up assessment completed	N/A
Li 2014	2-arm RCT: adventure camp vs attention control (leisure	Fatigue severity (FS-C - Chinese ver.), physical activity	No sig. difference between INT and CONT for fatigue severity (p=0.09).	6 months post INT	Sig. main effects for the INT on all outcomes; INT had lower levels of fatigue severity (p<0.001), higher levels of

	activities), 4 days over six months	levels (CUHK-PARCY), self-efficacy (PA-SE)	Sig. higher physical activity levels (p<0.01) and self-efficacy (p<0.01) in INT vs CONT.		physical activity (p<0.001), and self-efficacy (p<0.001).
Mooney 2016	2-arm RCT: Telephone- based interactive voice symptom reporting system with nurse care vs interactive voice symptom reporting system w/o nurse care, mean 77 days	Fatigue severity (measure not specified; NRS scale 1-10)	Sig. less fatigue severity in intervention (MD: -0.685, p<0.001).	No follow up assessment completed	N/A
Naraphong 2015	2-arm pilot RCT: walking exercise vs information, 12 weeks	Fatigue severity (PFS- R), Physical Fitness 12-MWT (12-min walk test).	Non sig. improvement in fatigue severity in INT vs CONT (no sig. group or time effects differences for fatigue ( $\beta = -0.00$ , p= 0.15). Sig. higher physical fitness over time in INT vs CONT (t = 2.28, p = 0.04).	No follow up assessment completed	N/A
Purcell 2010	4-arm RCT: pre- & post- radiotherapy fatigue education and support (RFES) vs pre-RFES vs post-RES, vs usual care (no RFES), 60 min RFES session	Fatigue severity (MFI), Physical activity (IPAQ- SF), leisure & self-care activity participation (FAI)	No form of RFES achieved sig. reduction in fatigue severity, or the improvement physical activity or leisure and self-care activities.	6 weeks post-RT (? post INT)	No sig. treatment effects on fatigue severity, physical activity, and leisure & self-care activities from the complete RFES programme analysis compared to usual care. Pre-RFES group had sig. increase in vigorous activity (p<0.01) compared to no pre-RFES, post-RFES sig. increase in walking p<0.05) compared to no post-RFES.
Ream 2006	2-arm RCT: fatigue education and management counselling vs usual care (ad hoc fatigue assessment), 3 chemo cycles (3 months).	Fatigue severity, distress caused by fatigue, disruption to work/chores by fatigue, disruption to pastimes/hobbies, (4 study VAS questionnaire – non-	Global mean fatigue score sig. lower in INT (p<0.04, ES=0.18). No stat. sig. difference in fatigue severity (p=0.19) or disruption to work/chores caused by fatigue (p=0.10).	No follow-up assessment completed	N/A

		validated), global mean fatigue score	Sig. lower associated distress (p<0.01), and disruption to hobbies (p=0.02) caused by fatigue.		
Ream 2015	2-arm pilot RCT: motivational interviewing vs usual care, varied depending on chemo cycle (3 cycles)	Fatigue intensity (BFI), fatigue distress (FDS), fatigue self- efficacy (questionnaire)	Improvement in fatigue intensity (ES=0.18; 95% CI: -0.57 to 0.93), fatigue distress (ES=0.62; 95% CI: 0.11 to 1.13), fatigue self-efficacy (ES= - 0.34; 95% CI: -1.10 to 0.71).	No follow-up assessment completed	N/A
Reif 2013	2-arm RCT: fatigue information and management education vs waitlist control, 6 weeks	Fatigue severity (Fatigue assessment questionnaire - FAQ); general self-efficacy (General self-efficacy scale); exercise self- efficacy (Physical exercise self-efficacy scale), physical activity (FFKA); fatigue knowledge (non-validated fatigue knowledge test [F-WT] -created for study)	Comparison with waitlist control not reported.	6 months post INT	Sig. reduction in fatigue severity (F =76.510, p < 0.001, $\eta^2$ = 0.248) in INT over time. Sig. improvements in general self-efficacy (F=27.680, p < 0.001, $\eta^2$ = 0.107), exercise self-efficacy (F= 49.230, p < 0.001, $\eta^2$ = 0.175), physical activity (F= 8.036, p < 0.001, $\eta^2$ = 0.033), and fatigue knowledge (F=55.157, p < 0.001, $\eta^2$ = 0.192).
Ritterband 2012	2-arm pilot RCT: internet CBT program vs waitlist control, 6-9 weeks	Overall fatigue severity (MFSI-SF) and general, mental, emotional, and physical fatigue subscales.	Sig. improvements in overall fatigue severity ( $F_{1,26}$ =13.88; p<0.01, d=1.16), general fatigue ( $F_{1,26}$ =9.46; p<0.01, d=0.91), mental fatigue ( $F_{1,26}$ =8.65; p<0.01, d=0.66). No sig. improvement in emotional ( $F_{1,26}$ =3.44; p=0.08, d=0.77) and	No follow-up assessment completed	N/A

			physical fatigue (F <sub>1,26</sub> =2.83; p=0.11, d=0.47).		
Sandler 2017	2-arm RCT: CBT and exercise program vs education and single exercise session, 12 weeks	Fatigue severity (SOMA), Physical activity (IPAQ)	Sig. fatigue improvement in fatigue severity in INT compared to CONT [(M= 2.55, SD = 3.77; t(36) = -2.56) vs (M= 0.10; SD =2.55)]; p< 0.05 between groups. No sig. difference between groups in physical activity levels (p=0.82)	3 months post INT	Sig. improvements in fatigue within INT (p<0.05); however, no sig. difference was found in change scores between groups (MD = 1.56; 95% CI: -3.77 to 0.48; p= 0.13). No sig. difference between groups in physical activity levels (p=0.96)
Savard 2005	2-arm RCT: CBT vs waitlist control, 8 weeks	Fatigue severity (MFI- Canadian French ver.)	Sig. lower fatigue severity in INT (F <sub>1,158</sub> =11.70; p< 0.001)	3-, 6-, & 12- months post INT	No sig. difference between fatigue post- treatment scores at 3-, 6-, or 12-months post INT.
Schjolberg2014	2-arm RCT: CRF education and coping strategies vs usual care – info on general cancer, 3 weeks	Fatigue Severity [FQ – physical, mental, total, chronic - & LFS (lee fatigue)]	No sig. difference in total fatigue, physical fatigue, mental fatigue, chronic fatigue, and Lee fatigue between groups. No sig. difference within groups for all fatigue measures.	3 months post INT	No sig difference in total fatigue, physical fatigue, mental fatigue, chronic fatigue, and Lee fatigue between groups. No sig. difference within groups for all fatigue measures.
Smith 2019	2-arm RCT: Online CBT vs waitlist control, 10 weeks	Fatigue severity (FACIT-F), self- efficacy (Self-efficacy for Managing Chronic Disease scale)	Not measured.	4.5 months post INT	Sig. reduction in fatigue severity in INT compared with CONT (ES: 0.46; p=0.034). INT had a moderate effect on fatigue (Cohen's d =0.46). No sig. difference in improvements in self- efficacy between the INT and CONT (p >0.05).
Spahn 2013	2-arm RCT: walking sessions & mind body exercises vs home-based walking, 10 weeks	Fatigue severity (MFI), unusual fatigue of the <i>last</i> <i>month</i> & unusual	No sig. group differences for any outcome. Clinically sig. reduction in unusual fatigue in last month (31% reduction in both groups).	3 months post INT	No sig. group differences for any outcome. Clinically sig. reduction in unusual fatigue in last month in INT (37% reduction).

		fatigue of the <i>last week</i> (German Fatigue Questionnaire)			
Stanton 2005	3-arm RCT: information video (VID) vs psychoeducational counselling (CTL) vs info booklet (EDU), varied	Fatigue/energy (SF- 36)	Not measured.	6, 12 months post recruitment	6 months VID intervention produced a sig. greater improvement in fatigue/energy (t <sub>1,405</sub> =2.36; p=0.018), during 6 months than did the CTL. No sig. differences for EDU versus CTL and VID versus EDU. <u>12 months</u> No sig. effects of the intervention on fatigue/energy
Steel 2016	2-arm RCT: web-based collaborative care vs attention control (usual care from medical team), 6 months	Fatigue severity (FACT-F)	Non sig. reduction of fatigue severity with a small effect size (Cohen's d=0.26) was observed between the INT and attention control arm (t(15)=1.80, p= 0.09).	No follow-up assessed	N/A
Vallerand 2017	2-arm RCT: exercise telephone counselling vs self-directed exercise, 12 weeks.	Fatigue severity (FACT-F), Aerobic exercise (GLTEQ)	No sig. between-group difference on fatigue severity. Increased weekly aerobic exercise in INT (adjusted mean between group difference (p < 0.001, Cohen's d= 0.19).	No follow-up assessed	N/A
Van der Lee 2012	2-arm RCT: mindfulness based cognitive therapy vs waitlist control, 9 weeks.	Fatigue Severity (CIS- F)	Sig. lower fatigue severity in INT than waitlist control (p<0.001). Effect size for fatigue =0.74	6 months post INT	Follow-up compared with immediate post INT: no sig. difference (p=0.20). Follow-up compared with baseline: Sig. less fatigue severity (p<0.001).
Van Waart 2015	3-arm RCT: Low intensity home based physical activity (Onco-Move) vs supervised moderate-to- high intensity program (OnTrack) vs Usual care (no prescribed exercise),	Fatigue severity (MFI, Fatigue quality list), Physical fitness (step ramp test, endurance test, physical activity scale for the elderly)	OnTrack reported sig. less general (p=0.041, ES =0.29) and physical fatigue (p<0.001, ES= 0.63), reduced activity (p=0.045, ES= 0.31), reduced motivation (p=0.049, ES= 0.34) than usual care; and sig. less physical fatigue (p=0.021, ES= 0.42) than Onco- Move.	5 months 1 week post INT	No sig. between-group differences for any of the performance-based measures of physical fitness or in self-reported fatigue.

Van Weert 2010	varied – duration of all chemo cycles 3-arm RCT: physical training (PT) vs PT + cognitive behaviour therapy (CBT) vs waitlist control (WLC), 12 weeks	Fatigue severity (MFI – general fatigue, physical fatigue, mental fatigue, reduced motivation, reduced activation)	Physical fitness levels were maintained immediately after completion of chemo in OnTrack but declined in usual care and Onco-Move. Compared to WLC, PT had sig. reduction in 4 domains of fatigue (general mean change: -1.0, p<0.01; physical mean change: -2.4, p<0.001; mental mean change: -2.4, p<0.001; mental mean change: -1.6, p<0.05; reduced activation mean change: -1.3, p<0.05). PT+CBT showed reduction in one fatigue domain only (physical fatigue mean change: -2.7, p<0.01). No sig. differences in reduction in fatigue were found between the PT+CBT and PT groups.	No follow-up assessment completed	N/A
Vargas 2014	2-arm RCT: Cognitive behavioural stress management vs psychoeducation, 10 weeks	Fatigue (FSI – fatigue intensity and fatigue- related daytime interference)	Not measured.	3 months & 9 months post INT	No sig difference over time between INT and CONT for fatigue intensity (p>0.30, Cohen's d=0.37). Sig. lower fatigue-related daytime interference in INT (p<0.05, Cohen's d=0.48)
Wang 2011	2-arm RCT: walking program vs usual care (not specified), 6 weeks	Fatigue severity (FACIT-F), exercise self-efficacy (ESES), exercise behaviour (GLTEQ), exercise capacity (6MWD)	INT had sig. lower fatigue ( $F_{1,60} = 9.74$ , p=0.003), improved exercise self- efficacy (results NR), increased exercise behaviour ( $F_{3,180} = 25.53$ , p<0.001), and exercise capacity ( $F_{1,63} = 12.01$ , p=0.001) compared to CONT over program period.	No follow-up assessment completed	N/A
Wangnum 2013	2-arm RCT: Multidisciplinary education program in self-care fatigue vs nurse	Fatigue severity (PFS), physical fitness (VSAQ), nutritional status (MNA)	Sig. lower fatigue (t=2.15, p=0.036) and improved nutritional status (t=4.01, p=0.002) in INT.	No follow-up assessment completed	N/A

	provided information on exercise strategies, 9 weeks		No sig. difference in physical fitness (t=-0.20, p=0.846) between groups.		
Willems 2017a & 2017b	2-arm RCT: Web-based psychosocial education program vs waitlist control, 6-month access to program	Fatigue severity (CIS- F)	Sig. reduction in fatigue severity (B =- 4.36, p = 0.020, f2 = 0.013, Cohen's d = 0.21).	12 months post baseline	Non-sig. group differences in fatigue severity.
Yates 2005	2-arm RCT: psychoeducational support program vs general cancer education, 3 weeks	Fatigue management behaviours (study questionnaire non- validated), confidence with managing fatigue (11-point NRS - developed for study), Fatigue experiences (fatigue distress, fatigue severity, fatigue impact-PFS, FACT-F), <u>Cancer self- efficacy</u> (24-item questionnaire -non- validated)	Fatigue Experiences: Sig. lower fatigue at worst (p=0.01), fatigue interference (p=0.01), average fatigue (p=0.02), FACT-F score (p=0.05) in INT. No sig. between group differences for best fatigue, fatigue distress and fatigue now.Fatigue management behaviours: INT pts. had sig. greater actions reccommended to (p=0.01) in the number of treatments reccommended. No sig.between group differences for all other fatigue behviours.	3-4 weeks post INT & 6-12 weeks post INT	Fatigue experiences: No. sig between-groupdifferences for all measures of fatigueexperiences. <a href="#">Fatigue management behaviours:</a> No sig.between-group differences forcommunication with health professionalsabout fatigue or number of actions taken tomanage fatigue. INT pts. had sig. greateractions recommended to (p=0.02) in thenumber of treatments recommended.Confidence with managing fatigue: no sig.between group differences in baseline topost INT change scores (all time points).Cancer self-efficacy:no sig. effect of INT (atall time points).
Yun 2012	2-arm RCT: internet- based education vs waitlist control, 12 weeks	Fatigue (BFI, FSS) – global fatigue, fatigue severity, fatigue interference.	INT had sig. greater decrease in global fatigue (p=0.001, Cohen's d=0.29), fatigue severity (p=0.008, Cohen's d=0.21), and fatigue interference scores (p=0.001, Cohen's d=0.29) of BFI, and in the total score of FSS.	No follow-up assessment completed	N/A

			Among patients who had moderate or greater fatigue at baseline, the INT group had clinically more meaningful improvement than CONT in BFI global score, worst fatigue, and FSS score.		
Zhang 2018	2-arm RCT: Nurse-led home-based exercise and cognitive behavioural therapy vs usual care (drug, diet chemo education), 12 weeks	Fatigue severity (PFS- Chinese ver.)	Lower total (t/ $Z$ =-2.558, p=0.011), behavioural (t/ $Z$ =-4.978, p<0.001), sensory (t/ $Z$ =-2.709, p=0.007), and cognitive fatigue (t/ $Z$ =-2.147, p=0.036) in INT vs CONT. No sig. between group difference in affective fatigue (p=0.778).	3 months post INT	Lower total (t/Z= $-3.311$ , p= $0.002$ ), behavioural (t/Z= $-5.855$ , p < $0.001$ ), sensory (t/Z= $-4.033$ , p< $0.001$ ), and cognitive fatigue (t/Z= $-5.388$ , p< $0.001$ ) in INT vs CONT. No significant between group difference in affective fatigue (p= $0.915$ ).

### Appendix 8: Qualitative Examples of Identified Themes (Study 2)

Themes	Theme frequency	Codes	Code frequency	Example quotes for theme	
	· <u> </u>	1		Round 1	_
Tailor information	16	Ensure cancer survivors and their family/caregivers understand the information provided Find out patient's/ cancer survivors information style (e.g., how would they prefer information to be delivered). Place emphasis on delivering information with care no to overload patient.	6 7 3	<ul> <li>Communication levels, jargon, making sure information is not just given but ensure understanding.</li> <li>Important to check how information can be delivered since information can be tiring. Important also to address with how the patient and parents/network can use the information. "</li> <li>Establish an open rapport with patient rather than just providing written documentation. i.e., a video</li> <li>Coping with information overload and having a support person to collate information</li> <li>Provide the patient with" Whenever I see this, I am reminded of how most of this information has just been given to us in a handout with little explanation or exploration. I would like to see a discussion aspect to those practices. The same with "Inform" - while that is the practice there needs to be a check for understanding. I have been informed of many things during medical appointments but never fully understood the implications.</li> <li>It is important that they know their patients and tailor their messages to how the patient process this. Not everyone wants detailed information, so know your client!</li> <li>Avoid jargon as most 'civilians' won't recognize or understand. People generally are embarrassed to ask the meaning.</li> <li>Patients need information on their treatment &amp; how to combat the side effects of both the disease &amp; treatment, but only enough for their needs at that time. There is more than enough going on inside their head.</li> <li>Medical professionals need to show empathy with the patient and family and not see the patient as a statistic and talk over them as if they are just a job.</li> <li>Personalisation of provided information in partnership with the healthcare professional should also be mentioned.</li> </ul>	
Consider patient preferences about involving support network	14	Consider patient preferences when involving support network in self- management support Consider patient preferences when involving other cancer survivors in self- management support Consider patient preferences when collaboratively deciding the coping strategies to be incorporated in the patient's fatigue action plan	10 2 2	<ul> <li>Consultation of provided information in participant with the function of provided the of information of support network should be contingent on an individual's preference</li> <li>Incorporation of support networks, if the patient so chooses / wishes. not every patient aims to exchange experiences with other patients (wording shood reflect this).</li> <li>Must incorporate consent from patient into using the wider support network in action planning, not all patients want their families or friends involved. This should be worded as 'consider incorporating with patient's consent'.</li> <li>Some patients are very shy of having their families/friends involved - they find it difficult to share how they feel.</li> <li>Again, this can depend on the patient and how they feel about sharing</li> <li>"Not all patients have the desire to include their support network into action planning so I think there should be more emphasis on the patient's needs" (Round 2)</li> <li>Integration of evidence-based management strategies for managing fatigue, alongside consideration of patient priorities, cultural context, etc.</li> <li>Patient consent to involvement of support network should be spelled out.</li> </ul>	1
Consider the patient's support network	4	Consider what the patient's support network needs to assist the patient to self-manage. Consider how the patient's cancer- related fatigue affects support network.	2	<ul> <li>Identify caregivers/parents need of support from healthcare team for being able to support the patient</li> <li>To what extent is fatigue degrading the patient's lifestyle and that of his partner/family?</li> <li>Need to consider family.</li> </ul>	
Focus on identifiable risk factors for fatigue	10	Focus on identifiable risk factors for fatigue	10	<ul> <li>Health professionals, specially doctors should also carry the efforts needed to exclude and treat aggravating causes to cancer fatigue, like anemia, hormonal disturbances, insomnia, pain,</li> <li>Conduct testings that identify and eliminate other possibilities related to fatigue such as hormones, thyroid, fibromyalgia etc</li> <li>Identify actionable risk factors for the fatigue itself. I practice in the malignant hematology setting, where we can often adjust or consider breaks of medications like lenalidomide or ibrutinib (being used for long-term disease control for MM or CLL, respectively) that can cause fatigue."</li> <li>Factors could be other disease factors or medical conditions and this should be assessed; and this should be addressed specifically.</li> </ul>	) 2 1 2
Identify who should deliver practices and how	10	Specify constraints to executing practices Specify the multidisciplinary nature of self-management support and the need to engage with other professional support Specify what health profession does what.	4 3 3	<ul> <li>This may need to be split up by [health professional] domain. Physicians like myself aren't equipped to do some of the tasks above, versus a social worker who might be able to provide direct coaching / counselling.</li> <li>This comment pertains to all domains. All of this is great, but who, when and how will this care be delivered</li> <li>all the items are relevant. However, it is very much to do it all</li> <li>very broad and should be focused on key precipitating or predisposing factors specific to fatigue; if we add all these expectations for managing fear of recurrence then fatigue support will not occur. perhaps engage other professional support for managing co-occurring depression, etc. but really focus on the key practice for fatigue.</li> </ul>	

#### **Response to theme**

An additional document was developed, emphasising the importance of underpinning the delivery of practices by the presence of effective, person-centred, health professional communication which involves the ability to establish and develop mutual understanding, rapport, trust, respect, and cooperation with people affected by cancer using clear and plain language.

Wording of practices in Domain 3 - (Improve Patient Knowledge) have been modified to reflect the need to provide information in a diversity of formats to accommodate different learning styles.

Wording changed to Key Practice 3 to acknowledge the need to ensure coping strategies are aligned with patient preferences.

Practices that involve the inclusion of support networks (peer, family/caregiver) have been adjusted to reflect the need to consider individual preference.

A new practice component was added to Domain 1 (Establishing Context and Defining the Problem), Key Practice 1.

Changes made to wording of practice component 1c) to explicitly specify that the purpose of information collection is to identify risk factors for cancer-related fatigue that can inform coping strategies.

Referral and involvement of multidisciplinary healthcare team is already incorporated in Practice Component 8b. A new practice component specifically highlighting referral for managing psychological effects (Practice Component 9b) has been created and put forward for rating and feedback in Round 2. An additional document providing detail on the execution of the key practices and practice components (i.e., who and how) has been created and added to the second-round survey for participants to peruse. Acknowledges that cancer-related fatigue

				<ul> <li>At some point in the recommendations, there should be a reference on how different health care professionals, community services and everybody communicate and work as a care team. This concerns also the acquisition of competencies to work with this population as in the exercise field, psychology, nutrition, social,</li> <li>assume ahead of time that the health professional (clinician or nurse?) will have the capacity to deliver these practice components. Is there not a need to specify such constraints - or an absence of them?</li> <li>It should be an MDT (multidisciplinary team) model of working involving the team's expertise (with referrals if/as necessary)</li> </ul>
Positive attitudes to framework	10	Framework is a useful and comprehensive tool. Incorporation of exchange between cancer survivors and support network is important. Glad this component is included in the framework Examples and strategies provided are useful for framework application. Glad support with communicating fatigue to employers has been included in the framework	3 2 4 1	<ul> <li>Just a comment that this is a very comprehensive set of practices that should, in fact, be in place for every survivor not just those with fatigue. It is so relevant to the many issues we face once treatment has ended and there is no path for us to follow. Most of our family doctors are not aware of the complications we face, and the cancer centres are focussed on active treatment, so we tend to feel abandoned. Thank you for the opportunity to be part of a solution. I would like to think that these practices would become part of our normal follow-up. I look forward to the next stage.</li> <li>[About facilitating exchange between cancer survivors] To me this is by far the most important. Everything I have learnt in cancer management is from other cancer survivors belonging to the same demographics as me, worldwide in 6 different languages, all on social media. And NO it is NOT Facebook nor Dr Google, we are organised in subreddit or twitter discussion groups, all have PhDs, access to research publications, used to do scoping, subscribe to up to date, follow short specialist courses on line all over the world etc we are just ill and do not put Dr in front our name . We then disseminate the knowledge through the closed Facebook groups so patients can drive their care more effectively and co-design, co-create instead of being ""managed "". The idea is to learn how to address our own needs with the support of professionals rather than professionals trying to support our needs, total shift in approach.</li> <li>[About facilitating exchange between cancer survivors] It helps to speak to someone who understands</li> <li>The specific strategies and examples provided are useful</li> <li>These all seem like good ideas!</li> <li>Employers should also be aware of the plan to the extent of documented support. Often an employer feels that because an employer looks' ok that he/she can do the same job as always. They do not see the fatigue after work, the length of preparation it takes to get ready for work, or the</li></ul>
Focus on referrals	6	Engage community services in provision of support. Engage and refer to other health professions.	6	<ul> <li>Holistic practical approach to managing fatigue not just limited to health environment but other pathways to access assistance where financially strapped to access medications gym yoga physio therapies etc</li> <li>Connect patient with community services so people with no one in the world are not left to die alone and receive some help.</li> <li>Helpful to inform individuals about available community-based resources for fatigue management</li> <li>Include appropriate referrals to other agencies and professionals with patient engagement and consent.</li> </ul>
Assist implementation of framework Identify the applicability of	6	Include more examples to aid         implementation of framework practices         Consider using tools to assist with         creating action plans and collecting         information         Consider the relevance of framework to         underserved populations	4 2 2	<ul> <li>Recommend clearer linkage to cancer-related fatigue, as currently it is quite broad. Adding examples of strategies can also aid in implementation.</li> <li>Could automate [the collection of clinical, behavioural, and symptom information] with questionnaires the way that InA (from Savor Health - a text-based oncology nutrition artificial intelligence does): https://www.savorhealth.com/</li> <li>[About provision of fatigue information] A patient brochure would be extremely helpful!</li> <li>Are the action plans co-created with patients in words or all in infographics? multiple choice?</li> <li>Strategies provided are very useful, recommend including for all practices.</li> <li>[Self-management support is] something for the privileged: people who have time to be concerned about their individual wellness rather than the collective, middle class affluent people. Obviously not directed to refugees. We too</li> </ul>
framework				<ul> <li>have cancer.</li> <li>other pathways to access assistance where financially strapped to access medications gym yoga physio therapies etc</li> <li>Round 2</li> </ul>
Provide fatigue specific strategies and examples	7	Contextual elements missing	7	<ul> <li>Contextual element missing? (Home, work, clubs, community organisations)"</li> <li>Incorporate fatigue specific self-management strategies and strategies for coping with fatigue.</li> <li>Coping strategies for management of fatigue is relevant but this should also include fatigue-specific self-management strategies (i.e., adaptive pacing, physical activity, etc). So word as incorporates fatigue specific self-management strategies and strategies and strategies for coping with fatigue.</li> <li>While tailoring is ideal, what is missing is clarity regarding the characteristics on which the provision of information should be tailored (i.e., symptom severity? resources available? intervention preferences?)."</li> <li>Greater specificity would be helpful if these practice components would be used for teaching or evaluating practice. Perhaps "attend to requests for further symptom review, treatment modification, or counselling"</li> </ul>

is multi-factorial and thus will require a multidisciplinary health team to manage. This study presents best practices, implementation of the framework (including how tasks and responsibilities can be shared amongst the healthcare team will be investigated in next research phase. We thank the panel for their feedback. We thank the panel for their feedback.

involvement of community services has already been incorporated (particularly Domain 2 – developing an action plan in practice component 6b.) Referral and involvement of multidisciplinary healthcare team already incorporated in Practice 8b. A new practice component specifically highlighting the need for providing referrals when managing psychological effects (9b) was created.

An accompanying document providing further information on the framework and context was created. All strategies and further examples of items in practice framework were moved to this document. Document will be distributed to panel participants in Round 2 for their perusal.

We agree that further targeted consultation with specific stakeholders is needed to understand relevance and applicability for different groups. Cultural influences on health and disease, and attitudes towards care will need to be considered when adapting the framework to different contexts in the next step.

Strategies and examples of practice scenarios were included in a separate document attached to the survey. Further analysis shows not all panel participants opened document. Panel feedback shows inclusion of these strategies and examples are essential. They have now been presented clearly.

I have a problem with adjective "" general "" i would like it to be targeted and " contextually relevant" otherwise     waste of time. We are already bombarded by these messages constantly so totally disregard it.
---

# Appendix 9: Changes from Round 1 of Modified Delphi Study (Study 2)

Original Practice Statement	Modified Statement based on Panel Feedback
DOMAIN 1: ESTABLISHING CONT	TEXT AND DEFINING THE PROBLEM
Key Practice 1	
1. Collect and use clinical and behavioural information to inform decision making about the patient's self- management of cancer-related fatigue	Reached consensus for inclusion in Round 1 and included in final framework
1a) Conduct a consultation and assessment with the patient, and if available, other key people in the patient's support network (e.g., carer's; family members) to collaboratively define key concerns, problem areas and priorities.	Reached consensus for inclusion in Round 1 and included in final framework
NEW	1b) Identify family members/informal caregivers need to have guidance from healthcare providers to facilitate their ability to support the patient with self-management strategies.
1c) Collect, record, and store key clinical, symptom and behavioural information.	1c) Collect and record key clinical, symptom, behavioural, and psychological information and highlight risk factors that may contribute to the patient's cancer-related fatigue.
NEW	1e) Conduct a fatigue (e.g., Brief Fatigue Inventory Questionnaire, informal questioning) to determine fatigue severity, onset, duration, pattern, associated patient distress, and interference with daily living.
Key Practice 2	
2. Assess the patient's capacity for self-management.	Reached consensus for inclusion in Round 1 and included in final framework
2a) Identify factors within the patient, and the patient's physical and social environment, that they perceive may improve or worsen their cancer-related fatigue (e.g. more severe fatigue in the afternoon).	Reached consensus for inclusion in Round 1 and included in final framework
2b) Identify the patient's beliefs, attitudes, and knowledge about cancer-related fatigue, and identify their current coping strategies.	2b) Identify the patient's <b>and the informal caregivers</b> / <b>support network</b> beliefs, attitudes, and knowledge about cancer-related fatigue, and identify their current coping strategies.
2c) Identify factors (i.e., cultural consideration, language literacy levels, availability of peer support network, pre- existing conditions) that may affect the patient's ability to participate in self-management activities.	Reached consensus for inclusion in Round 1 and included in final framework
	PING AN ACTION PLAN
Key Practice 3	
3. Create a cancer-related fatigue management action plan in collaboration with the patient	Create a cancer-related fatigue management action plan in collaboration with the patient <b>that incorporates evidence- based coping strategies that are aligned with patient</b> <b>preferences.</b>
<b>3a)</b> Reflect on the patient's capacity for self-management (including barriers), and the clinical and behavioural information gathered during pre-assessments, to refine the patient's priorities, needs, and goals concerning their cancer-related fatigue and general lifestyle.	Reached consensus for inclusion in Round 1 and included in final framework
3b) Plan a culturally and contextually relevant fatigue self- management care plan drawing on clinical and psychosocial information gathered during pre-assessments, the patient's capacity for self-management support and the health professional's applied knowledge of cancer-related fatigue management strategies.	3b) Plan a culturally and contextually relevant fatigue self- management care plan drawing on clinical and psychosocial information gathered during pre-assessments, the patient's capacity for self-management support (including addressing identified barriers to self-management or leveraging specific strengths/ capabilities), and the health professional's

	applied knowledge of cancer-related fatigue management strategies.
3c) Incorporate the patient's support networks (family, friends, carers) into action planning.	3c) <b>Consider incorporating</b> the patient's support networks (family, friends, carers) into action planning <b>with the patient's consent.</b>
DOMAIN 3: IMPROVE	PATIENT KNOWLEDGE
Key Practice 4	
4. Provide information (visual, written, verbal) on cancer- related fatigue and common management strategies	Provide <b>tailored evidence-based</b> information on cancer- related fatigue and common management strategies <b>in a</b> <b>diversity of formats to accommodate different learning</b> <b>styles.</b>
4a) Inform the patient and their support network of the differences between cancer-related fatigue and 'normal' fatigue.	Reached consensus for inclusion in Round 1 and included in final framework
4b) Inform the patient and their support network of the causes, key risk factors, presenting characteristics, and the possible effects and interferences of cancer-related fatigue on daily living.	Reached consensus for inclusion in Round 1 and included in final framework
4c) Communicate tailored evidence-based information to the patient and their support network regarding the benefits of exercise and physical activity for managing cancer- related fatigue and have an awareness of the strength of such evidence.	Reached consensus for inclusion in Round 1 and included in final framework
4d) Communicate tailored evidence-based information to the patient and their support network on other management strategies for cancer-related fatigue (along with their risk and benefits) and have an awareness of the strength of such evidence.	4d) Communicate tailored evidence-based information to the patient and their support network on <b>psychological and complementary medicine</b> management strategies for cancerrelated fatigue (along with their risk and benefits) and have an awareness of the strength of such evidence.
Key Practice 5	
5. Provide tailored evidence-based information on managing common psychosocial consequences of cancer and cancer-related fatigue	Provide tailored evidence-based information on managing common psychosocial consequences of cancer and cancer- related fatigue <b>in a variety of formats to accommodate</b> <b>different learning styles.</b>
5a) Provide the patient and their support network with evidence-based information on how psychological and psychosocial factors (e.g., fear of cancer recurrence or progression, anxiety, depression, and stress) contribute to, and are exacerbated by cancer-related fatigue.	Reached consensus for inclusion in Round 1 and included in final framework
5b) Provide the patient with coping strategies for managing psychological and psychosocial factors	This was deleted as panel feedback indicated it was already covered in Domain 4 (Key Practice 9).
5c) Provide the patient with coping strategies for processing and communicating with others about cancer- related fatigue and cancer experiences.	This was deleted as panel feedback indicated it was already covered in Domain 4 (Key Practice 7).
Key Practice 6	
6. Provide information about available social support	<b>Provide tailored evidence-based information</b> about available social support <b>in a variety of formats to</b> <b>accommodate different learning styles and check patient</b> <b>understanding</b>
6a) Facilitate the involvement of the patient's support	6a) <b>Consider</b> facilitating the involvement of the patient's
network (family members, friends, carers, significant	support network (family members, friends, carers, significant
others) in cancer-related fatigue self-management activities	others) in cancer-related fatigue self-management activities with the patient's consent.
6b) Provide the patient and their support network with education and information about how to seek further social support and inform the patient of relevant support services in their community.	Reached consensus for inclusion in Round 1 and included in final framework
6c) Facilitate the exchange of cancer and cancer-related fatigue experiences and/or discussions between the patient and other cancer survivors.	6c) <b>Consider</b> facilitating the exchange of cancer and cancer- related fatigue experiences and/or discussions between the patient and other cancer survivors, <b>if the patient agrees.</b>

<b>DOMAIN 4:</b> TRAINING REHE	EARSAL (STRATEGY BUILDING)
Key Practice 7	
7. Provide the patient with problem solving and evidence- based solution-focused strategies to communicate with their systems of support (includes health professionals, non-health professionals, personal communities; and voluntary and community groups) about cancer-related fatigue.	Reached consensus for inclusion in Round 1 and included in final framework
7a) Provide the patient with the skills to self-advocate and communicate with health professionals, non-health professionals, personal networks, employees, and others about cancer-related fatigue.	Reached consensus for inclusion in Round 1 and included in final framework
7b) Provide coaching and counselling about navigating relationships and social support	Reached consensus for inclusion in Round 1 and included in final framework
Key Practice 8	
8. Provide evidence-based coaching for lifestyle modifications that support living with cancer-related fatigue.	Reached consensus for inclusion in Round 1 and included in final framework
8a) Provide the patient and their support network with evidence-based tailored coaching and practical strategies for exercise (e.g., aerobic, resistance, yoga, balance) and physical activity (e.g., daily walking, morning stretches).	Reached consensus for inclusion in Round 1 and included in final framework
8b) Provide tailored coaching and practical strategies to the patient and their support network that supports the patient's everyday activities	Reached consensus for inclusion in Round 1 and included in final framework
8c) Refer the patient to relevant services and professionals for support when indicated (e.g., if you are unable to provide tailored coaching).	8c) <b>Involve</b> and refer the patient to relevant services and professionals for support when indicated (e.g., if you are unable to provide tailored coaching). <b>and in the planning</b> <b>and decision making of the patient's care.</b>
8d) Involve relevant services and professionals in the planning and decision making of the patient's care	This was combined with practice component 8c) as per panel feedback (due to overlap).
Key Practice 9	
9. Provide the patient with evidence-based problem- solving strategies for coping with the psychological effects or risk factors of cancer-related fatigue.	Reached consensus for inclusion in Round 1 and included in final framework
9a) Provide patients with strategies for coping with anxiety, fear of recurrence or progression, stress, depression, and managing interpersonal relationships.	Reached consensus for inclusion in Round 1 and included in final framework
NEW	9b) Refer the patient to relevant services (e.g., psycho-social oncology) and professionals for support when they are unable to provide tailored coaching. (new)
Key Practice 10	
10. Provide evidence-based general health promotion and education on lifestyle adaptation strategies.	Reached consensus for inclusion in Round 1 and included in final framework
10a) Provide relevant general lifestyle advice and counselling support to the patient and their support network.	Reached consensus for inclusion in Round 1 and included in final framework
DOMAIN 5: CARE CO-ORD	INATION AND MAINTENANCE
Key Practice 11	
11. Provide regular review of self-management goals and action plans in collaboration with the patient (and their social network).	Provide regular review of <b>self-management activities</b> , and self-management goals and action plans in collaboration with the patient, <b>their support network (with the patient's</b> <b>consent)</b> , and their health care team.

11a) Reformulate previously established goals based off	Reached consensus for inclusion in Round 1 and included in
the patient's confidence, needs, and progress (i.e., goal	final framework
attainment).	
11b) Establish long term goals prior to the conclusion of	Reached consensus for inclusion in Round 1 and included in
support to facilitate continual patient self-care beyond the	final framework
self-management support program.	
11c) Provide scheduled reviews to monitor cancer-related	Reached consensus for inclusion in Round 1 and included in
fatigue (e.g., severity and lifestyle interference), associated	final framework
symptoms (e.g., stress, depression), and review of progress	
with self-management behaviours (e.g., confidence,	
physical activity, activity management).	
Key Practice 12	
12. Provide practical support that facilitates ongoing self-	Reached consensus for inclusion in Round 1 and included in
management.	final framework
12a) Provide practical support to assist the self-monitoring	Reached consensus for inclusion in Round 1 and included in
of cancer-related fatigue symptom reporting (e.g., fatigue	final framework
severity, energy levels, and other factors that impacted	
fatigue), and behaviour change (e.g., exercise, mediation	
habits, physical activity, and dietary habits).	
Key Practice 13	
13. Be able available on request to review the symptoms of	Reached consensus for inclusion in Round 1 and included in
cancer-related fatigue.	final framework
13a) Be available for symptom review, treatment	13a) Be available for symptom review, treatment modification
modification or counselling if symptom thresholds are	or counselling if symptom thresholds are exceeded or if
exceeded.	requested by the patient due to a change in support
	requirements.
Key Practice 14	
	This was moved to a practice component of Key practice
Be able available on request to review the symptoms of	13 (due to overlap) as per panel feedback
cancer-related fatigue	
	This was merged with practice component 13a as per
14a) Be available for symptom review, treatment	panel feedback.
modification and/ or counselling if symptom thresholds are	*
exceeded	

Appendix 10: Cancer-Related Fatigue Self-Management Support Practice Framework (Study 2)

# Cancer-Related Fatigue Self-Management Support Practice Framework

## CONTEXT

Cancer-related fatigue is one of the most common and distressing symptoms reported by people affected by cancer [1]. There is need for specific guidance [2, 3] to assist health professionals in facilitating the management of cancerrelated fatigue [4, 5].

This Practice Framework presents the core practices required by health professionals to deliver effective self-management support to cancer survivors (pre- and post-cancer treatment) experiencing cancer-related fatigue, and presents an outline that is pragmatic, contextually adaptable, and clinically relevant. The practices included in this framework revolve around an individual's perceived level of fatigue severity and interference, regardless of cancer diagnosis, treatment type, or phase in the cancer care continuum. This framework presents the core tasks that should be undertaken when providing support for cancer-related fatigue management, rather than explore changes within or between any subgroups.

## FRAMEWORK DEVELOPMENT



#### DOMAINS

The practices included in this framework have been categorised into five domains. The domains of the practice framework provide a broad thematic organization for the self-management support practices and strategies. Each domain addresses a different set of essential processes that should be undertaken to sufficiently support cancer survivors to manage their cancer-related fatigue. Each domain consists of corresponding Key Practices and Practice Components. The content of these Key Practices and Practice Components have been informed by:

- i. a systematic review of self-management support programs for cancer-related fatigue
- ii. The Capabilities for Supporting Prevention and Chronic Condition Self-Management Framework



#### **KEY PRACTICES**

Key practices describe the proposed activities health professionals are required to undertake to provide best practice self-management support for cancer-related fatigue.

#### **PRACTICE COMPONENTS**

Practice components describe the steps needed to complete a key practice. Health professionals are expected to demonstrate all practice components for all the key practices when providing self-management support for cancerrelated fatigue. These practice components may require health professionals to apply, adapt, and integrate new and existing evidence-based knowledge.

## HOW CAN THIS FRAMEWORK BE USED?

Cancer-related fatigue is multifactorial – meaning that there are a diverse range of factors that can contribute to, or cause it (e.g., cancer type, cancer treatment type, anaemia, nutrition factors, psychological factors, etc.); therefore, it is expected that the provision of management support will require a multi-disciplinary approach.

It is <u>not expected</u> that all health professionals be proficient in all practice components, whereby certain practices may fall outside of one's professional expertise or scope (e.g., radiation therapist providing psychological coping strategies). The provision of support for cancerrelated fatigue management requires professional collaboration between differing specialities within a multidisciplinary care approach which includes an informed and supportive health care team.

The execution of the key practices and practice components specified in this document may require health professionals to apply, adapt, and integrate new and existing evidence-based knowledge. This practice framework does not provide, present, or describe the capabilities or competencies required by health professionals. There are already clinical practice guidelines outlining evidence-based management strategies for cancer-related fatigue (e.g., NCCN, CAPO, Macmillan, other (Level I) empirical evidence). Instead, this practice framework outlines the tasks that health professionals and or health care teams should undertake when supporting people affected by cancer to manage their fatigue.

Finally, the execution of practices outlined in the framework should be underpinned by the presence of effective, person-centred, health professional communication which involves the ability to establish and develop mutual understanding, rapport, trust, respect, and cooperation with people affected by cancer using clear and plain language. This includes making appropriate adjustments (e.g., use of appropriate language and detail, use of appropriate verbal and non-verbal cues, confirming that the other person has understood) to meet the communication and information needs of patients and their support network (e.g., caregivers, family, friends) and providing opportunities for the patient and their support network to demonstrate their understanding.

For the individual health professional	<ul> <li>Use the Practice Framework as a tool:</li> <li>to guide the provision of selfmanagement support for CRF</li> <li>for determining your professional development needs</li> <li>for evaluating current practice when providing support to those managing CRF</li> <li>Use the Practice Framework to develop an understanding about:</li> <li>the extent of CRF impact on those affected by cancer and the importance of its management</li> <li>the various roles of different health professionals in the delivery of selfmanagement support for CRF</li> </ul>
For the clinical leader/ educator	<ul> <li>Use the Practice Framework as a tool:</li> <li>to develop clinician awareness and knowledge of evidence-based cancer- related management and assessment strategies</li> <li>to provide training/in-service programs to improve ability to undertake practices (how to use certain questionnaires, practice developing an action plan, etc.)</li> <li>to advocate for system-level changes to provide resources (time, space, and human) to deliver optimal support for CRF management</li> </ul>
For the CRF self-management intervention developer	Use the Practice Framework to aid development of a CRF self-management intervention/program (determining the specific components that are needed).
For the researcher	Use the Practice Framework as a tool:         • to describe self-management support interventions for CRF       • to synthesize evidence on CRF self- management
For the consumer experiencing CRF	<ul> <li>to develop understanding of the various roles of different health professionals in the delivery of self management support for CRF</li> <li>to advocate for improved delivery of self support. to improve your care</li> <li>to advise your health care team(s) about the existence of the practice framework and teaching and learning resources in efforts</li> </ul>

## CANCER-RELATED FATIGUE SELF-MANAGEMENT SUPPORT PRACTICE FRAMEWORK

\*Note 1: The examples 'tools, strategies, and prompts' section provides a guide to help health professionals consider the support expected. These descriptions provide examples but are not intended to be exhaustive. Information for the 'tools, strategies, and prompts' section was derived from the NCCN Cancer-related Fatigue Guidelines [6] and a systematic review of self-management support programs for cancer-related fatigue [7].

**\*\*Note 2:** The 'patient' refers to the individual experiencing cancer-related fatigue (encompasses those who are on active cancer treatment and those who have completed active cancer treatment).

Key Practice 1	Practice Components	*Examples of Tools, Strategies, Prompts, etc.
	<ul> <li>1a) Conduct a consultation and assessment with the **patient, and if available, other key people in the patient's support network (with the patient's consent), to collaboratively define key concerns, problem areas, and priorities.</li> </ul>	<ul> <li>Things for clinicians to consider</li> <li>How is fatigue impacting the patient's daily activities (social life, work life, personal feelings)? Examples of cancer-related fatigue presentation in patients can include lack of energy; lack of concentration; inability to complete daily tasks such as housework; poor concentration [6].</li> <li>Consider involving the patient's support network in information collection [6, 7]. Oftentimes the patient may not be aware that fatigue has impacted their life: however, their support network may be more cognizant of changes and the effect of fatigue [6].</li> </ul>
1. Collect and use clinical and behavioural information to inform decision- making about the	1b) Identify family members/informal caregivers need for guidance from healthcare providers to facilitate their ability to support the patient with self- management strategies.	
patient's self- management of cancer-related fatigue.	clinical, behavioral, and psychological information and highlight clinical, behavioral or psychological risk factors that may contribute to the patient's cancer-related fatigue.	Clinical information can include: cancer treatment type, cancer treatment length, cancer recurrence or progression, referral history, age, medication history, pain levels [6] Clinical risk factors of cancer-related fatigue can include: anaemia, chemoradiation or other systemic treatments, hypothyroidism, hormonal cancer treatment depression, anxiety, stress, fluid/ electrolyte imbalance, pain Behavioural information can include: sleep habits, current level of physical activity, diet, confidence in engaging in self-management behaviours Behavioural risk factors of cancer-related fatigue can include: sleep disturbances, inactivity, poor diet Psychological information can include: psychiatric history, history of depression and anxiety Psychological risk factors of cancer-related fatigue can include: depression, anxiety, stress

#### Domain 1: Establishing Context and Defining the Problem

assessm fatigue duration, p patien	nduct a fatigue nent to determine severity, onset, pattern, associated it distress, and be with daily living.	<ul> <li>Things for clinicians to consider [6]</li> <li>Brief, self-reported, quantitative, and single-item assessments with empirically established cut-off scores are effective.</li> <li>E.g., 0 to 10 numerical rating scale (0 = no fatigue and 10 = worst fatigue), mild fatigue indicated as a score of 1 to 3, moderate fatigue as 4 to 6, and severe fatigue as 7 to 10.</li> <li>Evaluation of fatigue in children can be modified to a scale of 1 to 5 or asked if they are 'tired' or 'not tired'.</li> </ul>
---	--	---

Key Practice 2	Practice Component	Example of Tools, Strategies, Prompts, etc.
	2a) Identify factors within the patient, and the patient's physical and social environment, that they perceive may improve or worsen their cancer-related fatigue.	Things for clinicians to consider [7]. Are there things the patient and/or the patient's support network have noticed that makes their fatigue worse or better? (for example, patients can describe worse fatigue on treatment days; more severe fatigue in different parts of the day; changes during certain weather)
2. Assess the patient's capacity for self- management.	2b) Identify the patient's (and their informal caregiver's/ support network's) beliefs, attitudes, and knowledge about cancer-related fatigue, including differences between patients and informal caregivers and identify their current coping strategies.	<ul> <li>Things for clinicians to consider [7]</li> <li>What does the patient and their support network know about cancer-related fatigue?</li> <li>How has the patient managed (or been managing) their fatigue? (e.g., What strategies have they tried, and not found success with? What do they feel has worked?)</li> </ul>
	2c) Identify factors (i.e., cultural consideration, language literacy levels, availability of peer support network, pre-existing conditions) that may affect the patient's ability to participate in self-management activities.	<ul> <li>Things for clinicians to consider [6, 7]</li> <li>What factors might affect the patient's ability to undertake exercise and other fatigue management activities (e.g., lymphoedema, neuropathy in fingers, lack of transport, financial status, physical status, lack of support to assist with daily tasks.)</li> <li>How does the patient's informal caregivers/ support network feel about providing assistance with support?</li> </ul>

Key Practice 3	Practice Component	Example of Tools, Strategies, Prompts, etc.
	3a) Reflect on the patient's capacity for self- management (including barriers), and the clinical and behavioral information gathered during pre-assessments, to refine the patient's priorities, needs, and goals concerning their cancer- related fatigue and general lifestyle.	<ul> <li>Things for clinicians to consider [6, 7]</li> <li>What does the patient wish to get out of the support provided?</li> <li>How motivated is the patient to address their cancerrelated fatigue?</li> <li>A particular patient may need more support network engagement (health professionals, friends, family, carer) and structural involvement versus another who may be highly self-motivated.</li> <li>Consider using a readiness ruler to guide conversations around priorities, importance, confidence, and personal change.</li> </ul>
3. Create a cancer-related fatigue management action plan* in collaboration with the patient that incorporates evidence-based coping strategies that are aligned with patient preferences.	3b) Plan a <u>culturally and</u> <u>contextually relevant</u> fatigue self-management care plan. The plan should be informed by clinical and psychosocial information gathered during pre- assessments, the patient's capacity for self- management (including addressing identified barriers to self- management or leveraging specific strengths/ capabilities), and the health professional's applied knowledge of cancer- related fatigue management strategies.	<ul> <li>Culturally and contextually relevant takes into consideration sociocultural factors (e.g., cultural and linguistic diversity, age, gender, disability, religion, socioeconomics, geographical locations) and acknowledges and incorporates the importance of culture, the assessment of cross-cultural relations, and the adaptation of services to meet culturally unique needs.</li> <li>Things for clinicians to consider [7]         <ul> <li>Ensure goals are SMART (specific, measurable, actionable, realistic, time-bound), patient-centred, and include specific actions for moderately challenging behaviors.</li> <li>Goal development process may need to include motivational interviewing techniques to address patient priorities and concerns; rolling with resistance when required, and facilitating the patient's self-agency</li> </ul> </li> </ul>
	3c) Consider incorporating the patient's support network (family, friends, carers) into action planning with the patient's consent.	<ul> <li>Things for clinicians to consider</li> <li>The inclusion of a support network in action planning can increase uptake of fatigue self-management strategies [7].</li> <li>How can the patient's support network assist the patient to achieve goals in action plan?</li> <li>For example, in a situation of escalating fatigue at the end of life, family members may wish to designate individuals to assume activities relinquished by the individual with cancer [6].</li> </ul>

#### Domain 2: Developing an Action Plan

\*Definition/ components of a Cancer-related fatigue Action Plan: The fatigue self-management action plan should be developed in collaboration with the patient and (if available) their other supports (including carers). The care plan should facilitate the patient's engagement in their own care (leveraging the patient's strengths and capabilities) and contain patient defined problems and goals, community education programs and resources, community support networks, allocated time for review and follow-up.

Key Practice 4	Practice Component	Example of Tools, Strategies, Prompts, etc.
	4a) Inform the patient and their support network of the differences between cancer- related fatigue and 'normal' fatigue.	<ul> <li>Cancer-related fatigue vs. normal fatigue: Cancer-related fatigue is a physical, emotional and/or cognitive tiredness that is more persistent and severe than normal fatigue and cannot be relieved by adequate sleep and rest [6].</li> <li>Things for clinicians to consider [6]</li> <li>There is a need to relay to the patient that if fatigue does occur often, it is not necessarily an indication that the treatment is not working or that the disease is progressing/recurring.</li> <li>Reassurance is important as fear of progression is a main reason for the under-reporting of fatigue.</li> </ul>
4. Provide tailored evidence-based information on cancer-related fatigue and common management strategies in a diversity of formats to accommodate different learning styles	4b) Inform the patient and their support network of the causes, key risk factors, presenting characteristics, and the possible effects and interferences of cancer-related fatigue on daily living.	<b>Examples of risk factors for cancer-related fatigue</b> [6] Insomnia; depression, anxiety and stress, pain; being overweight, anaemia; sudden weight loss, poor diet, other medical conditions (e.g., heart problems, diabetes). <b>Examples of presenting characteristics</b> <b>and interference on daily living</b> [6] Difficulty making decisions, sleeping, maintaining attention and concentration; poor memory, feeling drained with no energy or strength, inactivity, low mood.
	4c) Communicate tailored evidence-based information to the patient and their support network regarding the benefits of exercise and physical activity for managing cancer-related fatigue and have an awareness of the strength of such evidence.	<ul> <li>Several meta-analyses have been conducted to provide a comprehensive evaluation of the impact of increased activity upon cancer-related fatigue. Studies suggest that exercise and physical activity has a beneficial effect on fatigue in patients during and after cancer treatment [6].</li> <li>It is reasonable to encourage patients to engage in a moderate level of physical activity (3 to 5 hours per week) during and after cancer treatment [6].</li> </ul>
	4d) Communicate tailored evidence-based information to the patient and their support network on psychological and complementary medicine management strategies for cancer-related fatigue (along with their risk and benefits) and have an awareness of the strength of such evidence.	Examples of management strategies (excluding physical activity and exercise) [6] Energy conservation, sleep therapy, activity management, counselling, cognitive behavioural therapy.

## Domain 3: Improving Patient Knowledge

Key Practice 5	Practice Component	Example of Tools, Strategies, Prompts, etc.
5. Provide tailored evidence-based information on managing common psychological consequences of cancer and cancer-related fatigue in a variety of formats to accommodate different learning	5a) Provide the patient and their support network with evidence-based information on how psychological and psychosocial factors (e.g., fear of cancer recurrence or progression, anxiety, depression, and stress) contribute to, and may be exacerbated by cancer- related fatigue.	<ul> <li>Things for clinicians to consider [6]</li> <li>There is a strong correlation between emotional distress (depression, anxiety), stress, and cancer-related fatigue.</li> <li>Psychological and symptom distress also have a negative effect on the patient's confidence to undertake fatigue self-management behaviours.</li> </ul>

stv	00
ວເທ	63

Key Practice 6	Practice Component	Example of Tools, Strategies, Prompts, etc.
<ol> <li>6. Provide tailored evidence-based information about</li> </ol>	6a) Consider facilitating the involvement of the patient's support network (family members, friends, carers, significant others) in cancer-related fatigue self- management activities with the patient's consent.	<ul> <li>Things for clinicians to consider</li> <li>The availability of dependable caregivers can significantly impact the functional, emotional, and financial capacity of a patient and impact their pursuant fatigue [6].</li> <li>Involving the personal support network in self-management solutions (i.e., family members exercising with the patient) may increase compliance/adherence to cancer-related fatigue self-management strategies [7].</li> <li>A personal support network can be particularly key when a patient lacks the economic and supportive resources to obtain tangible support [6].</li> </ul>
available social support in a variety of formats to accommodate different learning styles and check patient	6b) Provide the patient and their support network with education and information about how to seek further social support and inform the patient of relevant support services in their community.	<ul> <li>Things for clinicians to consider</li> <li>What are the available community-based resources for fatigue management?</li> <li>Are there community options for physical activity programs?</li> <li>What are the offerings of local advocacy groups?</li> </ul>
understanding	6c) Consider facilitating the exchange of cancer and cancer-related fatigue experiences and/or discussions between the patient and other cancer survivors (peer support), if the patient wishes to do so.	<ul> <li>Practical support strategies [7]</li> <li>Exchange can occur through in-person discussion groups, patient stories (written), patient videos, online forums.</li> <li>Things for clinicians to consider [6]</li> <li>Use a guided group setting for peer exchange to avoid well-meaning but non-evidence-based suggestions and to allow for consideration of whether someone's experience is truly relevant to another person. Also take into consideration groups being limited to the right stage (in-treatment for primary cancer, living with secondary cancer, etc.).</li> </ul>

### Domain 4: Training Rehearsal (Strategy Building)

	Domain 4: Training Renearsal (Strategy Building)	
Key Practice 7	Practice Component	Example of Tools, Strategies, Prompts, etc.
7. Provide the patient with problem solving and evidence-based solution focused strategies to communicate with their systems of support (includes health professionals, non-health professionals, personal communities;	7a) Provide the patient w the skills to self-advoca and communicate with heal professionals, non-heal professionals, persona networks, employees, ar others about cancer-relat fatigue.	<ul> <li>Providing examples of questions to ask healthcare professionals and examples of goals to prepare for discussions (e.g., "Tomorrow morning I will write a list of questions about fatigue to take with me to my appointment with the nurse next week.")</li> <li>Advising patients to show their fatigue diary to their</li> </ul>
and voluntary and community groups) about cancer-related fatigue.	7b) Provide coaching an counselling about navigating relationships a social support.	
Key Practice 8	Practice Component	Example of Tools, Strategies, Prompts, etc.
8. Provide evidence- based coaching for lifestyle modifications that support living with cancer-related fatigue.	8a) Provide the patient and their support network with evidence-based tailored coaching and practical strategies for exercise (e.g., aerobic, resistance, yoga, balance) and physical activity (e.g., daily walking, morning stretches).	<ul> <li>Things for clinicians to consider [6]</li> <li>Improved satisfaction with physical activity is often reported when family involvement in a program increases.</li> <li>Encourage patients to engage in a moderate level of physical activity (at least 30 minutes a day <u>OR</u> at least 3 to 5 hours per week) during and after cancer treatment.</li> <li>It is critical that: <ul> <li>Patients choose a type of exercise they enjoy (as this increases the likelihood that they engage in physical activity)</li> <li>Providers discuss specific implementation strategies (type of exercise, time of day, days of the week, location of activity) to enable patients to make frequent activity a reality.</li> <li>Activity prescribed is individualized based on age, gender, cancer type, and fitness level.</li> <li>Activity begins with a low level of intensity and duration and is modified as the patient's condition changes.</li> </ul> </li> </ul>
	8b) Provide tailored coaching and practical strategies to the patient and their support network that supports the patient's everyday activities.	<ul> <li>Strategies can include coaching on meditation, relaxation, and breathing exercises; providing nutritional consultation; coaching on changing sleep habits; providing advice on re-structuring daily activities and energy conservation; cognitive behavioural therapy [7].</li> <li>Examples of energy conservation strategies [6] <ul> <li>Maintaining a fatigue diary which will allows the patient to ascertain peak energy periods, and then plan their activities accordingly within a structured routine.</li> <li>Labor-saving techniques (wearing bathrobe instead of drying off, assistive devices such as walker, etc.)</li> </ul> </li> </ul>

Key Practice 8	Practice Component	Example of Tools, Strategies, Prompts, etc.
		<ul> <li>Examples of sleep hygiene strategies [6]</li> <li>Going to bed and waking up at the same time each night, getting out of bed after 20 minutes if unable to fall asleep, avoiding long or late afternoon naps, avoiding caffeine in the afternoon.</li> </ul>
	8c) Refer the patient to relevant services and professionals for support when indicated, and involve these services in the planning and decision making of the patient's care.	<ul> <li>Examples of relevant services or health professions</li> <li>Referral to exercise specialists (e.g., physical therapist, physical medicine, exercise physiologist, rehabilitation specialist) for an assessment and provision of exercise prescription.</li> <li>Referral to cancer specific exercise programs</li> <li>Community outreach programs</li> </ul>
Key Practice 9	Practice Component	Example of Tools, Strategies, Prompts, etc.
9. Provide the patient with evidence-based problems solving strategies for coping with the psychological effects or risk factors of cancer-related fatigue.	9a) Provide patients with strategies for coping with anxiety, fear of recurrence or progression, stress,	<ul> <li>Can occur through [7]: Coaching on interpersonal relationships, distress management, sharing experiences with family and other cancer survivors, and assignments for building mental and social activity.</li> <li>Specific strategies can include [7]:         <ul> <li>Cognitive restructuring and reframing; relaxation exercises; stress-management, breathing exercises, counselling, mindfulness therapy.</li> <li>Activating positive emotions, overcoming depressive periods, breathing exercises, mindfulness therapy.</li> </ul> </li> </ul>
	9b) Refer the patient to relevant services (e.g., psycho-social oncology) and professionals for support when you are unable to provide tailored coaching.	<b>Examples of relevant services or health professions</b> Psychologist, behavioural therapist, counsellor, social worker, support network, community organisations.

Key Practice 10	Practice Component	Example of Tools, Strategies, Prompts, etc.
10. Provide evidence- based health promotion and education on lifestyle adaptation strategies.	10a) Provide relevant lifestyle advice and counselling support to the patient and their support network.	<b>Examples of relevant lifestyle advice</b> Promotion and support for physical activity, exercise, sleep hygiene, diet, activity management and regulation, stress management, and general health and wellbeing (e.g., preventing weight gain, health nutrition, increasing fluid intake, anger management).

Key Practice 11	Practice Component	Example of Tools, Strategies, Prompts, etc.
11. Provide regular review of self- management activities, and self- management goals and action plans in collaboration with	11a) Reformulate previously established goals based off the patient's confidence, needs, and progress (i.e., goal attainment).	<ul> <li>Things for clinicians to consider [7]</li> <li>Review should incorporate constant encouragement to improve fatigue self- management engagement and confidence.</li> </ul>
	11b) Establish long-term goals prior to the conclusion of support to facilitate continual patient self-care beyond the self-management support program.	
the patient, their support network (with the patient's consent), and their health care team	11c) Provide scheduled reviews to monitor cancer-related fatigue (e.g., severity and lifestyle interference), associated symptoms (e.g., stress, depression), and review of progress with self- management behaviours (e.g., confidence, physical activity, activity management).	<ul> <li>Things for clinicians to consider [7]</li> <li>Scheduled reviews can occur during consultations or logbook/diary review.</li> </ul>

Key Practice 12	Practice Component	Example of Tools, Strategies, Prompts, etc.
12. Provide practical support that facilitates ongoing self- management.	12a) Provide practical support to assist the self-monitoring of cancer-related fatigue symptom reporting (e.g., fatigue severity, energy levels, and other factors that impacted fatigue), and behaviour change (e.g., exercise, mediation habits, physical activity, and dietary habits).	<ul> <li>Support Strategies can include [7]</li> <li>Providing diaries, logs, activity trackers, pedometers, heart rate monitors, frequent telephone calls, automated email/text message reminders, home visits.</li> <li>Things for clinicians to consider [7]</li> <li>Consider providing equipment that enables the patient to engage in physical activity or exercise in their homes.</li> <li>Consider recording self-management support coaching/consult sessions to facilitate home practice.</li> </ul>

Key Practice 13	Practice Component	Example of Tools, Strategies, Prompts, etc.
13. Attend to requests to review the symptoms of cancer-related fatigue.	13a) Attend to requests (from the patient) for further symptom review, treatment modification or counselling if symptom thresholds are exceeded or if requested by the patient due to a change in support requirements.	<b>Examples include</b> facilitating additional counselling on request; ensuring patient accessibility to care coordinators, nurse practitioners, and psychologists for additional support.

#### References

1. Weis J, Horneber M. Cancer-Related Fatigue. Tarporley, United Kingdom: Springer Healthcare Iberica; 2014.

2. Pearson E, Morris M, McKinstry C. Cancer related fatigue: Implementing guidelines for optimal management. BMC Health Services Research. 2017;17.

3. Ann MB, Kathi M. Dissemination and Implementation of Guidelines for Cancer-Related Fatigue. Journal of the National Comprehensive Cancer Network. 2016;14(11):1336-8.

4. Hilarius DL, Kloeg PH, van der Wall E, Komen M, Gundy CM, Aaronson NK. Cancerrelated fatigue: clinical practice versus practice guidelines. Supportive Care in Cancer. 2011;19(4):531-8.

5. Jones G, Gollish M, Trudel G, Rutkowski N, Brunet J, Lebel S. A perfect storm and patient provider breakdown in communication: two mechanisms underlying practice gaps in cancer-related fatigue guidelines implementation. Supportive Care in Cancer. 2021;29(4):1873-81.

6. Berger. NCCN clinical practice guidelines in oncology (NCCN Guidelines®) cancerrelated fatigue. 2019.

7. Agbejule OA, Hart NH, Ekberg S, Crichton M, Chan RJ. Self-management support for cancer-related fatigue: A systematic review. International Journal of Nursing Studies. 2022;129:104206.